

Durham E-Theses

Turkish Foreign Policy towards Iran since the Arab Uprisings (December 2010 to July 2018): A Neoclassical Realist Account

ALTUNDAS-AKCAY, CANGUL

How to cite:

ALTUNDAS-AKCAY, CANGUL (2022) Turkish Foreign Policy towards Iran since the Arab Uprisings (December 2010 to July 2018): A Neoclassical Realist Account, Durham theses, Durham University. Available at Durham E-Theses Online: http://etheses.dur.ac.uk/14393/

Use policy

 $The full-text\ may\ be\ used\ and/or\ reproduced,\ and\ given\ to\ third\ parties\ in\ any\ format\ or\ medium,\ without\ prior\ permission\ or\ charge,\ for\ personal\ research\ or\ study,\ educational,\ or\ not-for-profit\ purposes\ provided\ that:$

- a full bibliographic reference is made to the original source
- a link is made to the metadata record in Durham E-Theses
- the full-text is not changed in any way
- The full-text must not be sold in any format or medium without the formal permission of the copyright holders.

Please consult the full Durham E-Theses policy for further details.

Academic Support Office, The Palatine Centre, Durham University, Stockton Road, Durham, DH1 3LE e-mail: e-theses.admin@durham.ac.uk Tel: +44 0191 334 6107 http://etheses.dur.ac.uk

Turkish Foreign Policy towards Iran since the Arab Uprisings (December 2010 to July 2018): A Neoclassical Realist Account

Cangül Altundaş-Akçay

A thesis presented for the degree of Doctor of Philosophy



School of Government and International Affairs Durham University United Kingdom July 2021

ABSTRACT

With the outbreak of the Arab Uprisings across the Middle East and North Africa at the end of 2010, regional and global state actors, including Turkey and Iran, sought to shape and manage these political and social phenomena in alignment with their regional interests and objectives. Although Turkey's foreign policy towards Iran was initially tense, it later became more manageable due to specific unit-level factors. In this context, the Arab Uprisings offer the historical context by which to assess how Turkish Foreign Policy (TFP) towards Iran was shaped and conducted. This thesis explores the rationale behind TFP towards Iran since the Arab Uprisings by asking two central questions:

- 1. To what extent did structural forces shape the formulation of Turkish foreign policy towards Iran following the Arab Uprisings, from December 2010–July 2018?
- 2. How did unit-level (domestic) factors interact with structural imperatives in shaping Turkish foreign policy towards Iran following the Arab Uprisings, from December 2010–July 2018?

Consequently, the thesis uses the perception and domestic security and economic considerations of the AKP elites as intervening variables to clarify the cause-effect relationship between systemic changes and the relative power position of Turkey (as independent variables) to examine the TFP towards Iran (the dependent variable) in the short to medium term. In this regard, NCR provides a useful theoretical framework for the intervening variables to explain the causal link between independent and dependent variables. Schweller's balance of interest theory also helps explain Turkey's alliance choice with Iran, considering Turkey's economic and security considerations. This study contributes to a deeper understanding of TFP and TFP towards Iran in particular by establishing a conceptual framework through the use of NCR while also providing fresh insights concerning the key variables connected to foreign relations and domestic politics in TFP towards Iran. In short, the empirical goal of this work is to provide an innovative contribution to the study of TFP from a theory-driven standpoint. As a result, this thesis provides a solid foundation for scholars to build on and use NCR in future research

that explores the actions of state actors facing seismic regional shifts and new security challenges.

DECLARATION

The work in this thesis based on research carried out at the School of Government and International Affairs, Durham University, United Kingdom. No part of this thesis has been submitted elsewhere for any other degree or qualification and it is all my own work unless referenced to the contrary in the text.

Copyright © 2021.

"The copyright of this thesis rests with the author. No quotations from it should be published without the author's prior written consent and information derived from it should be acknowledged."

ACKNOWLEDGEMENTS

Completing the PhD has been a challenging task, and this compelling process would not have been accomplished without the support of the people who helped me the most. First and foremost, I would like to express my deepest appreciation to my supervisor, Prof Clive A. Jones, for the continues support of my PhD study. His guidance helped me in all the time of research and writing of this thesis. I am also extremely grateful to my sincere friend, Dr Saliha Oner Caliskan, for giving me enough moral support, encouragement, and motivation to accomplish the thesis. I would also like to thank my family for supporting me spiritually throughout writing this thesis and my life in general. Above all, I would like to thank my husband Dr Samet Akcay for his love and constant support.

TABLE OF CONTENTS

CHAPTER 1	INTRODUCTION	1
1.1 Re	SEARCH OBJECTIVE AND SIGNIFICANCE	3
1.2 LI	ferature Review	5
1.2.1	Turkish Foreign Policy	6
1.2.2	Turkish Foreign Policy towards the Middle East	8
1.2.3	Turkish Foreign Policy Towards Iran	15
1.3 Or	RIGINALITY OF THE THESIS	19
1.4 Re	SEARCH METHODOLOGY	20
1.5 St	RUCTURE OF THE THESIS	23
CHAPTER 2	2 THEORETICAL FRAMEWORK	25
2.1 NE	OCLASSICAL REALISM	25
2.1.1	Variables of Neoclassical Realism	28
2.1.1.	I Independent Variable	28
2.1.1.2		
2.1.1.	1	
2.1.2	The Differences between Neoclassical Realism and other International Relations Theories	37
2.1.2.		
2.1.2.2		
2.2 Co	NCLUDING REMARKS	45
CHAPTER 3	3 TURKEY'S RELATIONS WITH IRAN FROM FEBRUARY 1979 TO DECEM	IBER
2010: AN OV	/ERVIEW	47
3.1 BI	LATERAL DETERMINANTS	47
3.1.1	Ideological and Sectarian Division	47
3.1.2	Minority Issues	50
3.1.3	Economic Factors	53
3.2 Sy	STEMIC DETERMINANTS	56
3.2.1	The Iran-Iraq War between 1980-1988	58
3.2.2	The 1991 Gulf War	60

3.2.3 The Collapse of the Soviet Union and the Emergence of the Newly Independent	nt States in the
Caucasus and Central Asia	62
3.2.4 The Palestinian-Israeli Conflict	66
3.2.5 The US-led operations of Afghanistan and Iraq	69
3.3 CONCLUDING REMARKS	72
CHAPTER 4 THE SYSTEM LEVEL — INDEPENDENT VARIABLES IN TURKI	SH FOREIGN
POLICY TOWARDS IRAN SINCE THE ARAB UPRISINGS (DECEMBER 2010-JULY 2	2018)74
4.1 AN ANALYSIS OF THE ARAB UPRISINGS AT A SYSTEMIC LEVEL	77
4.1.1 The Role of Muslim Brotherhood-affiliated Islamic Parties	77
4.1.2 The Sunni-Shia Divide	81
4.1.3 The Role of Global Powers	
4.1.4 Empowerment of Non-State Armed Actors	
4.1.4.1 Islamic State of Iraq and Syria (ISIS)	
4.1.4.2 Kurdish Armed Groups	91
4.2 The Role of the System-Level on Turkish Foreign Policy towards Iran si	INCE THE ARAB
Uprisings, from December 2010–July 2018	
4.3 CONCLUDING REMARKS	102
CHAPTER 5 THE UNIT LEVEL — INTERVENING VARIABLES ON TURKIS	SH FOREIGN
POLICY TOWARDS IRAN SINCE THE ARAB UPRISINGS (DECEMBER 2010-JULY 2	2018)105
5.1 THE AKP ELITE'S PERCEPTION	106
5.1.1 The Role of the AKP Elite's Perception in TFP towards Iran Since the Arab Upris	
2010 to July 2018)	0
5.2 Domestic Considerations	
5.2.1 Turkey's Domestic Considerations in Relation to Iran since the Arab Uprisings (
2010 to July 2018)	, ,
5.2.1.1 Security Consideration	
5.2.1.2 Economic Consideration	
5.3 Concluding Remarks	
CHAPTER 6 TURKISH FOREIGN POLICY TOWARDS IRAN SINCE THE ARAI	
(DECEMBER 2010-JULY 2018): THE CASE OF THE SYRIAN CONFLICT	138
6.1 THE ROLE OF THE SYSTEM-LEVEL ON TURKISH FOREIGN POLICY TOWARDS IRAN SINCE	THE OUTSET OF
THE SYRIAN CONFLICT: MARCH 2011 TO JULY 2018	138
6.1.1 Inter-State Competition	139
6.1.2 The rise of non-state armed groups	144
6.2 THE ROLE OF THE UNIT LEVEL (INTERVENING VARIABLES) ON TURKISH FOREIGN PO	OLICY TOWARDS
IRAN SINCE THE ONSET OF THE SYRIAN CONFLICT: MARCH 2011 TO JULY 2018	148
6.2.1 The AKP Elites' Perception	149
6.2.2 Domestic Considerations	154
6.2.2.1 Security Consideration	155
	vii

	6.2.2.2	Economic Considerations	
6.3	CONC	LUDING REMARKS	
CHAPT	FER 7	TURKISH FOREIGN POLICY TOWARDS IRAN SIN	ICE THE ARAB UPRISINGS:
DECEN	MBER 2	010 TO JULY 2018 — THE CASE OF THE KURDISTAN	REGION OF IRAQ165
7.1	The I	Role of the System-Level on Turkish Foreign Policy	fowards Iran since the Arab
Upri	SINGS (D	ECEMBER 2010 TO JULY 2018): THE CASE OF THE KRI	
7.	1.1 US	policies in Iraq	
7.	1.2 The	e Syrian Conflict	
	7.1.2.1	The rise of ISIS	
	7.1.2.2	The rise of the PYD	
7.2	The 1	ROLE OF THE UNIT-LEVEL ON TURKISH FOREIGN POLICY T	OWARDS IRAN SINCE THE ARAB
Upri	SINGS: D	ECEMBER 2010 TO JULY 2018 — THE CASE OF THE KRI	
7.2	2.1 AK	P Elite Perception	
7.2	2.2 Do	mestic Considerations	
7.3	CONC	LUDING REMARKS	
CHAP	FER 8	CONCLUSION	
8.1	EXEC	UTIVE SUMMARY AND ARGUMENTS	
8.2	CONT	RIBUTIONS OF THE RESEARCH	
8.3	Limit	ATIONS OF THE RESEARCH	
8.4	Pote	NTIAL FURTHER RESEARCH	
BIBLIG	OGRAPI	HY	
APPEN	DIX		I

LIST OF TABLES

Table 2-1: Variables in Neoclassical Realism	37
Table 2-2: Classical Realism, Neorealism and Neoclassical Realism (Lobell et al., 200	9, 20)
	42
Table 3-1 Determinants of TFP towards Iran before the Arab Uprisings	73
Table 4-1The effects of Arab Uprising in each country from 2011 to 2018	76
Table 4-2 The Arab Uprisings and reginal and global state actors from 2011 to 2018	103
Table 5-1: Turkey-Iran Trade Volume (Billion \$) (www.tuik.gov.tr)	134
Table 6-1 TFP towards Iran in the Syrian Conflict from March 2011 to July 2018	164
Table 7-1 TFP towards Iran in the KRI: December 2010 to July 2018	185

LIST OF ABBREVIATIONS

AKP	Justice and Development Party (Adalet ve Kalkinma Partisi)
AQI	Al-Qaeda in Iraq
ASALA	Armenian Secret Army for the Liberation of Armenia
BCM	Billion Cubic Meters
BOTAS	Petroleum Pipeline Corporation
BPD	Barrels Per Day
CENTO	Central Treaty Organisation
CHP	Republican People's Party (Cumhuriyet Halk Partisi)
CR	Classical Realism
ECO	Economic Cooperation Council
EU	European Union
FETO	Fethullahist Terror Organisation (Fethullahci Teror Orgutu)
FSA	Free Syrian Army
GCC	Gulf Cooperation Council
HDP	People's Democratic Party (Halklarin Demokratik Partisi)
IAEA	International Atomic Energy Agency
IRGC	Islamic Revolutionary Guard Corps
ISI	Islamic State in Iraq
ISIS	Islamic State of Iraq and Syria (Daesh, Islamic State)
KDP	Kurdistan Democratic Party
KNC	Kurdish National Council
KRG	Kurdistan Regional Government
KRI	Kurdistan Region of Iraq (Iraqi Kurdistan)
JCPOA	Joint Comprehensive Plan of Action
MB	Muslim Brotherhood
MIT	Turkey's National Intelligence Organisation (Milli Istihbarat Teskilati)
MHP	Nationalist Movement Party (Milliyetci Hareket Partisi)
MOU	Memorandum of Understanding
NATO	North Atlantic Treaty Organisation
NCR	Neoclassical Realism
NPT	Non-Proliferation Treaty
NR	Neorealism

OIC	Organisation of the Islamic Conference
PJAK	Kurdistan Free Life Party (Prtiya Jiyana Azad a Kurdistane)
РКК	Kurdistan Workers' Party (Partiya Karkeren Kurdistane, Kurdistan Isci Partisi)
PLO	Palestine Liberation Organisation
PUK	Patriotic Union of Kurdistan
PYD	Democratic Union Party (Partiya Yekitiya Demokrat)
SDF	Syrian Democratic Forces
SNC	Syrian National Council
TFP	Turkish Foreign Policy
TIKA	Turkish Cooperation and Coordination Agency (Turk Isbirligi ve Koordinasyon Idaresi Baskanligi)
UAE	United Arab Emirates
UK	United Kingdom
UN	United Nations
UNSC	United Nations Security Council
US	United States (United States of America)
USD	United States Dollar
YPG	People's Protection Units (People's Defense Units, Yekineyen Parastina Gel)

Dedicated To my lovely sons,

Kerem Bera Akcay- Erdem Uraz Akcay

"When Turkey sneezes, Iran catches a cold."

Mohammed Reza Pahlavi-the Last Shah of Iran

CHAPTER 1 INTRODUCTION

According to Ray Hinnebusch (2002, 1), the Middle East is a chaotic regional subsystem with a high level of conflict and state rivalry. This region was hit by yet another powerful political and social phenomenon in December 2010, exacerbating tensions within the exiting regional order. The Arab Uprisings (also called the Arab Spring or Arab Awakening) broke out across the Middle East and North Africa, significantly influencing regional politics as autocratic regimes faced a wave of protests demanding political and economic change. These waves began to threaten the existence of the ruling regimes, first in Tunisia and then in other authoritarian regimes, thus altering the political landscape of the region. The Middle East became the centre of international and regional politics and attracted the attention of both regional and international state actors, once again. All these state actors, including Turkey and Iran, sought to shape and manage the developments of this political and social phenomena, in alignment with their regional interests and objectives. As two of the three pivotal non-Arab state actors in the Middle East (the other is Israel) Turkey and Iran viewed the uprisings as an opportunity to fulfil their regional interests and aims by influencing regional politics. The Arab Uprisings, therefore, offer the historical context by which to assess how Turkey conducted foreign policy towards Iran.

Turkey has historically had a roller coaster relationship with Iran. While its economic ties exert influence over its relations with Iran, Turkey has sometimes had a competitive relationship with Iran due to conflicting interests and ideas. Following the collapse of the Soviet Union in 1991, for instance, Turkey competed with Iran to influence the then newly independent Central Asian states. Both countries hoped to establish close ties with these countries to develop their regional interests. Given Turkey's cautious ties with Iran, the Arab Uprisings recast Turkey's policies towards Iran by causing political and security shifts in the regional system.

Middle East politics represents a "primitive anarchic order" with a low level of institutionalization in terms of state norms and rules, with autonomous regional dynamics (Buzan and Wæver 2003, 187; Hinnebusch and Ehteshami 1997, 13). From the outset of the

Arab Uprisings, the international community experienced another systemic change following the 9/11 attacks and the US-led operations in Afghanistan in 2001 and Iraq in 2003. In such an environment, the already anarchic regional politics accelerated and reshaped "political systems, national interests, alliances and rivalries" (Amour 2017, 1). As such, Turkey had to readjust its foreign policies towards the Middle East in general, and Iran in particular.

Following the uprisings and despite initial hesitation, Turkey sought to strengthen its power and enhance its security, economic, and political interests. First, Turkey used its soft power to achieve these goals by presenting itself as a model Islamic, democratic, and economically prosperous government. In other words, its political and economic success in international politics promoted Turkey's self-view as a role model in the post-Arab Uprisings era (Ennis and Momani 2013, 1128), and so it followed an active foreign policy in the wake of the uprisings. Likewise, Iran aimed to utilise the anarchic state of politics to increase its regional position and thus welcomed the uprisings as a movement which would transform the Arab world (Turhan and Yorulmazlar 2018, 110). Consequently, the geopolitical vacuum resulted in renewed competition between Turkey and Iran over regional hegemony and leadership in the Muslim world (Murinson 2012, 22). More precisely, the two countries attempted to influence the regional balance of power due to their similar relative power position in the regional structure following to the uprisings.

Despite their rivalry, Turkey was sometimes forced to establish practical cooperation with Iran due to the significant consequences of the Arab Uprisings, including the empowerment of non-state armed actors. With the rise of non-state armed actors, such as the Islamic State of Iraq and Syria (ISIS) and the Democratic Union Party (PYD), Turkey chose a pragmatic and short-term collaboration with Iran over national security concerns. Economic considerations also forced Ankara to establish close ties with Tehran. While losing its authoritarian economic allies and having financial issues, Ankara hoped to establish closer economic cooperation with Iran, and so followed an interest-driven foreign policy towards it. Bearing in mind Turkey's these ambivalent foreign policies towards Iran from December 2010 to July 2018, this thesis explores and explains how Turkey's relative power position in the systemic shifts and domestic factors played an intervening role between Turkey's relative power position and its foreign policy towards Iran in the short-to-medium term. Neoclassical realism (NCR) is adopted as the theoretical framework since NCR incorporates both systemic and

domestic factors in the explanation of a state's foreign policy. However, first, in this introductory chapter, the research aims, and objectives are defined, and an overview of the research methods and thesis structure is presented.

1.1 Research Objective and Significance

According to Barry Buzan (2003, 141), a region refers to a "subsystem of economic, political, and security relations that exist among a set of states whose fate is that they have been locked into geographical proximity with each other." Al Tamamy (2014, 206) further argues that "the distribution of power among the principal countries in the region makes it difficult for a single country to exercise a hegemonic agenda over the entire region and gives rise to the formation of power clusters among them." Instead of a specific regional hegemon, therefore, rival regional powers play a significant role in the Middle Eastern politics. In this regard, middle/regional state powers (Iran, Saudi Arabia, Turkey, and Israel) "project influence, exert pressure and try and drive agendas which are of direct interest to them. Some work together, sometimes, but they did not have any meaningful strategic partnerships with each other" (Ehteshami 2014, 36). This demonstrates that Turkey and Iran, non-Arab regional powers and two neighbouring states, are among the units which are sensitive to regional power shifts because they want to expand their respective spheres of influence. This situation further provides a critical understanding of their amity and enmity towards each other.

With the onset of the Arab Uprisings in December 2010, the Middle East experienced significant shifts at a sub-systemic level. Many authoritarian regimes were toppled, such as in Tunisia, Egypt, Libya and Yemen, and many-faced domestic revolution, such as Syria and Bahrain, while numerous non-state armed groups used the political chaos to increase their power and influence in the post-uprisings' era. Many external and regional state actors, moreover, were confronted with a decrease in their power and influence in the region. Turkey's and Iran's regional actions drew attention in such an anarchic regional arena since they influenced the uprisings' progress as two prominent regional state powers. Following the uprisings, Ankara and Tehran first followed an interest-maximisation policy, through which Turkey first viewed Iran as a competitor. As a regional middle power, Turkey aimed to increase its relative power position by exploiting the underlying power vacuum; however, taking relative power as the sole consideration fails to explain TFP towards Iran fully. Indeed, domestic factors, including the AKP elites' perception and domestic security and economic

considerations, also played an influential role in shaping TFP towards Iran from December 2010 to July 2018 because these factors hampered the acceleration of Turkey's competitive perspective on Iran. Such a rapid shift in TFP towards Iran while still being governed by the same party is unusual and led to an intriguing enigma. The period of December 2010 to July 2018 has been chosen for this study because the Arab Uprisings started in December 2010, and Turkey officially adopted new political system by switching from a parliamentary system to a Turkish type of presidential system, in which the role of parliament decreased while the role of President increased, in July 2018. In this regard, this thesis accepts the influence of both systemic and domestic levels on TFP towards Iran, and the central questions of this research are as follows:

1.To what extent did structural forces shape the formulation of Turkish foreign policy towards Iran following the Arab Uprisings, from December 2010–July 2018?

2.How did unit-level (domestic) factors interact with structural imperatives in shaping Turkish foreign policy towards Iran following the Arab Uprisings, from December 2010–July 2018?

This thesis utilises NCR and explores the causal relationship between the independent (system level) and intervening (domestic level) variables vis-à-vis the dependent variable (foreign policy behaviour). Through a careful examination and analysis of these different variables, the thesis argues that international power relations and, most importantly, Turkey's relative power played a significant role in shaping TFP towards Iran since the Arab Uprisings, from December 2010 to July 2018. The thesis also explores the intervening role of domestic variables in shaping TFP towards Iran in the same period, based on the literature on TFP and the aims of this study. The thesis first categorises the AKP elites' perception as an intervening variable because the governing AKP party has conducted Turkish foreign policy since 2002 and, therefore, its leaders and founders have political power and influence in Turkey. The next selected intervening variable concerning the analysis of TFP towards Iran is Turkey's domestic security and economic considerations; this aligns with Schweller's balance of interest theory, which argues that states may ally with other states because of their interests and potential profit. As such, domestic considerations (security and economic considerations) and the AKP elites' perceptions were the most significant variables, among other internal factors, in producing a more reliable and credible analysis of TFP towards Iran. Additionally, the Syrian conflict and

the politics of Kurdistan Region of Iraq were chosen as case studies since they mediate fluctuations in TFP towards Iran. In these cases, Turkey pursued short-term pragmatic and interest-driven collaboration with Iran, although their interests and aims conflicted for the longer term. Bearing in mind that TFP towards Iran provides a useful case study for understanding Turkey's wider foreign policy since the Arab Uprisings, this thesis also addresses the following sub-questions as part of its theoretical framework:

- 1. Did Turkey cooperate with its western allies or turn away from them over its relations with Iran from December 2010 to July 2018?
- 2. Did Turkey follow a value- or interest-driven foreign policy from December 2010 to July 2018?
- 3. How did Turkey's relative power position and the analysis of the AKP elite's perceptions and domestic motivations of Turkey shape TFP towards Iran in the Syrian Conflict and the politics of Kurdistan Region of Iraq from December 2010 to July 2018?

Overall, this thesis provides a specific and detailed analysis of TFP towards Iran since the Arab Uprisings, from December 2010 to July 2018. It does not solely attribute TFP decisions to external factors; rather, it adopts a comprehensive approach by exploring the role of different aspects or levels in shaping TFP. More precisely, this thesis is the first attempt to identify links between the independent, intervening, and dependent variables in the formation and implementation of TFP towards Iran since the Arab Uprisings using NCR as a framework. In doing so, the thesis aims to improve the disciplinary knowledge of TFP towards Iran and establish a foundation for future researchers to build on the current literature regarding TFP towards Iran, because previous studies have mainly concentrated on describing their bilateral affairs. Therefore, they have not fully investigated all the possible variables and factors that improve understanding of TFP towards Iran. By assessing and evaluating previous studies and applying a theoretical framework, the thesis aims to undertake a multidimensional research approach to the investigation of TFP towards Iran, and in so doing, develop a more comprehensive study.

1.2 Literature Review

This section broadly reviews the literature on TFP, TFP towards the Middle East, and TFP towards Iran, and aims to assess and evaluate the plausibility of the available competing

paradigms in the literature. In this regard, this section engages with the ongoing debates and extracts information that can help answer and develop the central research question. Therefore, the materials chosen in this review were selected for their relevance and potential contribution to this thesis. Overall, this literature review critiques the wider narrative on TFP towards Iran and demonstrates how the outcome of this thesis will fit into the wider academic discourse.

1.2.1 Turkish Foreign Policy

The literature on Turkish foreign policy (TFP) is vast, and researchers have approached it from diverse angles. According to Philip Robins (2007), interest in TFP increased after the 1990s when Turkey was actively involved in systemic changes, such as the collapse of the Soviet Union, the first Gulf War (1990–1991), NATO and EU enlargement. Hale (2012) coupled a detailed historical examination of TFP with a presentation of how medium-sized power has functioned in the international context. He examined Turkish foreign policy historically from the late Ottoman period (1774) to the emergence of the AKP in power (2002). Bilgin (2009) also explores the role of Turkey's security concerns on its foreign policy behaviour since 1920s. According to her (2009, 121), "Turkey's western-oriented Foreign Policy...was a crucial aspect of its search for security in the face of a European/International Society that had, in the past, refused equal treatment to the Ottoman Empire. At the time, the Foreign Policy of locating Turkey in the West allowed the founding leaders to claim the right to be treated equally and with respect." In other words, at the outset of the republic, Turkey aimed to join the Western alliance. Similarly, Tur and Han (2011, 11) pointed out Turkey's security-oriented foreign policy, stating that "it is hard to draw a clear line between TFP and security policy especially until the late 1990s." Deringil (1989), Aydin (2006) and Sayari (2000) also defines the determinants of Turkish foreign policy till the end of Cold War, from a structural perspective.

Other scholars proposed different arguments regarding TFP after the 1990s. Robins (2003, 159) focuses on the role of international system and the domestic motivations behind TFP mostly in the 1990s, arguing that "ideology is at least as important as geopolitics in the formulation and pursuit of Turkish foreign policy." Therefore, different political and/or religious identities have an influential role in shaping TFP. Bozdaglioglu (2004) further uses a constructivist approach to define TFP, suggesting that "Turkish foreign policy clearly depends on the distribution of power between Secularists and Islamists." When the Islamic-rooted AKP came to power as a single ruling party in 2001, the role of identity became

increasingly influential in the study of TFP. Robins (2007) pointed out that "a cooperative cohabitation" between the "Islamist-rooted AKP" and the "Kemalist" state apparatus was reflected in TFP across several areas, including Turkish-EU relations, the 2003 Iraqi war. Kosebalaban (2011) examined TFP from 1923 to 2011 through the prism of identity politics, highlighting that the ideological rivalry between the secular vs. Islamist and nationalist vs. liberal groups in Turkish domestic politics influenced TFP. Onis (2011, 57) further asserts that "the AKP leadership has been particularly effective in using pro-active foreign policy as a tool for projecting its 'globalist' and progressive image and thereby gaining advantage over its principal 'defensive nationalist rivals', the CHP and the MHP."

However, identity and ideology had limited influence on the first years of TFP under the AKP government because the AKP elites refrained from following identity and ideological foreign policies. Instead, emphasis was placed on Turkey's diplomacyoriented and multi-dimensional realist foreign policy. The beginning of the AKP rule in Turkey coincided with significant structural developments, such as the 9/11 attacks, the US intervention in the Middle East, and the 2001 global financial recession. Therefore, scholars focused more on Turkey's realpolitik and multidimensional foreign policy rather than identity politics. Dikici-Bilgin (2008, 408) noted that "the domestic pro-Islamist rhetoric of political parties might not necessarily transform into policies. Once in power, they are subject to rationalist cost-benefit calculations and moderation." Oguzlu (2007) focused on external and internal factors, such as the Western dimension and desecuritisation, on the rise of Turkey's soft power. He (2010, 658) even read the Europeanization process of TFP as a matter of "realpolitik security considerations." Kanat (2010) claims that new realities of a multi-polar world and Turkey's domestic factors encourage Ankara to follow a multidimensional foreign policy. Turkey thus aimed for a more "autonomous, self-regulating, and self-confident foreign policy." Kirisci (2009) further stated that Ankara's interest in Turkey's foreign policy stemmed from the country's desire to become a trading power. Similarly, Kutlay (2011) highlighted the impact of the 2001 economic crisis on Turkish domestic and foreign policies. Since a state's foreign policy is also dependent on state leaders' perception, some scholars also argued for the role of individuals on Turkey's multidimensional foreign policy to be taken into account. Aras and Gorener (2010, 90) suggested that "AKP foreign policy-makers, led by Foreign Minister Ahmet Davutoglu, have envisioned Turkey as holding multiple roles in world politics, whereas these were previously thought of as incompatible". Accordingly, Turkey under the AKP governments sought to establish cordial affairs with a variety of countries in different regions.

In other words, Ankara actively focused on different regional politics. This perception resulted in the shift of TFP from passive stance to active involvement in international politics.

Nevertheless, the literature on TFP started to involve "axis shift" arguments when Turkey's EU process was interrupted in 2005 due to the suspension of key chapters in membership negotiations and Turkey's subsequent pivot away from the EU and towards the Middle East. The next subsection thus focuses on the literature on TFP under the AKP toward the Middle East.

1.2.2 Turkish Foreign Policy towards the Middle East

The number of studies focusing on TFP towards the Middle East has increased following the AKP shift towards an active and diplomacy-oriented foreign policy towards the Middle East. This stands in contrast to previous Turkish governments. In parallel to Turkey's EU accession process, scholars highlighted the role of Turkey's EU membership process on TFP towards the Middle East. According to Ozcan (2016), Turkey's application for EU membership influenced its Middle Eastern foreign policy and Turkey attempted to work with the EU on regional concerns. Bac and Gursoy (2010) further highlight the role of EU membership process in TFP towards Iraq and Syria by stating that the Europeanization of TFP culminated in the increase of diplomatic and economic tools in shaping TFP towards both Iraq and Syria despite their conflicts being mostly related to Turkey's Kurdish issue.

Following the suspension of key chapters in Turkey's EU membership negotiations, the impact of Turkey's ties with its Western allies on policy towards the Middle East remained significant in the literature. According to Terzi (2016, 2), "It is crucial to notice that the changes of policy towards the Middle Eastern countries are loosely linked with Turkey's relations with the West in general, and with the EU in particular." Onis and Yilmaz (2009) further analysed the loss of Turkey's enthusiasm toward the EU membership due to bilateral issues. Notably, the divergence regarding the Cyprus issue resulted in Turkey's disappointment towards the EU's decision and policies when Turkey's negotiation process was partially suspended after Turkey's refusal to open its ports to vessels from the Republic of Cyprus. As a result, the Turkish elites assumed that the EU was using the Cyprus dispute as a "tool of exclusion" to stymie Turkey's EU accession process (Onis and Yilmaz 2009, 15).

On the other hand, most analysts saw TFP's policy toward the Middle East as a means of improving and deepening relations with its Western allies. According to Öniş (2011) and Altunışık (2009), Turkey's EU accession process still maintained its importance for TFP despite a period of deadlock. Therefore, Turkey's role in the Middle East did not undermine its relationship with the EU. Rather, the AKP sought to demonstrate Turkey's future influential role in the EU as a significant regional power. In his book called *Strategic Depth: Turkey's International Position*, Davutoglu (2001) uses a bow and arrow metaphor to define TFP, representing Turkey as the arrow and foreign policy as the bow. According to him, the closer Turkey gets to the Caucasus and the Middle East, the faster it will be admitted to the EU.

Other analysts pointed out that Turkey's domestic political and economic developments based on its democratisation process led Ankara to strengthen bilateral ties with Middle Eastern countries. Kardas (2010, 116) argued that "Turkey's new geopolitical alignment is a result of a process that predates the JDP [AKP]'s rise to power and has its own unique dynamics". Aras and Toktas (2010, 2) further focused on the significance of domestic politics, suggesting that "Turkey contribut[ing] to peace and stability in the neighbouring regions with the selfconfidence gained through democratization and de-securitization at home." In a similar vein, Aras (2009, 30) noted that "in this environment of pluralism in foreign policy-making, Turkey has acquired a new foreign policy identity, which led to the emergence of a self- confident, multi- dimensional and dynamic foreign policy approach." Tur (2011) argued that the strengthening of Turkey's economic ties with the area resulted from changes in the Turkish economy and the AKP's election to office. Therefore, Turkey's growing role as a mediator, according to these analysts, is related to its new foreign policy stance and domestic changes. Dagi (2008) pointed out that most Middle Eastern states had few concerns regarding Turkey's "new regional activism" due to Turkey's increasing regional soft power policies. Danforth (2008, 83) also noted that "Atatürk's disengagement from the Middle East and the AKP's reengagement with the region were both practical responses to strategic realities."

With the beginning of the uprisings in December 2010, TFP towards Middle East increased its significance among scholars due to Turkey's active roles following the uprisings. At the beginning of the Arab Uprisings, many academics and scholars defined TFP as a value based because Ankara advocated liberal-oriented principles of democracy and human rights and acted as a strategic mediator between the various sides involved in the Arab Uprising countries, including western and Middle Eastern countries (Akgün, Perçinoğlu, and Senyücel

Gündoğar 2012, 19–24; Altunışık 2011, 3). Ahmet Davutoglu (2013, 5–6), then the Turkish Minister of Foreign Affairs, noted that Turkey endorsed peaceful democratic change in the region by saying that "the leaders are temporary and the people are eternal [and] the flow of history is on the side of the masses that have demanded their rights in the Middle East". Similarly, Ayata (2015) highlighted that Turkey followed a value and identity-based domestic-driven foreign policy to consolidate itself as a regional power in the light of the Arab Uprisings. Ipek (2015) and Yesiltas (2013) further posited that Turkey adopted a soft power policy under an identity-based discourse and efforts centred on public diplomacy. Therefore, Ankara's initial foreign policy approach to the uprisings was considered by most observers to be liberal and humanitarian.

Although Turkey initially demonstrated a value-based humanitarian policy towards the uprisings, the primary motivation behind Turkey's regional policies remained self-interest. Murinson (2012, 17) argued that Turkey developed its foreign policy according to its "national self-interest and economic interests, not the promotion of democracy". He used Turkey's support for military intervention in Bahrain to quell Shi'ite opposition protesters as an example to support his observation. Onis (2012) added that Turkey was stuck between ethics and selfinterest given its previous close relations with the authoritarian regimes across the Middle East, including Syria. Kosebalaban (2011, 93-112) similarly argued that Turkey utilised idealistic views to maximise its national interests, which means that its apparent liberal foreign policy really masked calculations of national interest rather than any idealist views. Altunisik (2014) further posited that the concept of "humanitarian diplomacy" was used by Turkey to justify its involvement in the uprisings. As such, Ankara sought to justify its relative power position following the Arab Uprisings, and notably demonstrated more interest-driven policies, although Turkish policymakers were quick to deny the centrality of national-interests as the driver of their actions. In particular, Turkey's approach towards the uprisings in Egypt and Syria highlighted the dominance of interest-driven policies on TFP when Turkey was prepared to embrace sectarian policies to increase its relative power position both during and after the uprisings.

Accordingly, Çuhadar et al. (2020), Aras and Gorener (2010, 81), Gurzel (2014, 98– 99), and Sula (2018, 255–56) analysed the discourse of the Turkish policymakers, stating that Ankara played numerous roles towards the uprisings, including being a 'natural leader' in the region, a historical 'big brother', and the 'protector of Muslim minorities'. Similarly, Ozdamar et al. (2014, 93–102) examined Ankara's foreign policy roles following the uprisings, proposing that Turkey shifted its "soft power instruments to hard roles requiring material capabilities". For this reason, TFP towards the Middle East showed numerous role conceptions, such as 'a defender of regional peace and stability', 'a regional system collaborator', 'a trading state', 'a central/pivotal country', 'an active independent country', 'a developer', 'a protector of the oppressed' and 'a model/example country'. Ozdamar (2016, 94–95) purported that following its opposition to Israel's campaigns in Gaza in 2009 and again in 2014, the AKP sought to be a "regional leader, protector of the oppressed and leader of the Muslim world" to promote its "leadership and ties with populations across the region that was religiously and culturally close to the AKP electorate". Kara and Sozen (2016, 47–62) similarly pointed out that, to evaluate Turkey's position in international politics, the AKP elites defined Turkey as "a regional protector", "a mediator-integrator", "a developer", "a facilitator", "a rising power", and "a global-subsystem collaborator". For this reason, the AKP elite's perception of Turkey's relative power position and international politics has determined TFP towards the Middle East since the Arab Uprisings.

Kesgin (2020) notably analysed the effects of President Erdogan on TFP between March, 2003 and May 2013. According to him (2020, 65-70), Erdogan's religious views influenced his perceptions and pushed him towards following a more assertive foreign policy towards the uprisings. Arkan and Kinacioglu (2016, 383) also focused on Davutoglu's influence on TFP towards the Middle East, stating that "Davutoglu seeks a leadership role in the Muslim world for Turkey, framed by a desire to transform Turkey into a legitimate alternative power centre, as the leader and ultimate representative of a distinct civilisation". Baskan (2018) similarly analysed Davutoglu's discourse, illustrating "how Davutoglu's interpretation heavily borrows from the Islamist interpretive frame of modern Turkish history". Taspinar (2011) and Wastnidge (2019a) defined all role conceptions of TFP under the label of neo-Ottomanism. According to Taspinar (2011, 2), the AKP elites' neo-Ottoman tendencies are aimed at influencing politics across territories that were once part of the Ottoman Empire by using political, economic, and cultural soft power capabilities. Thanks to its secular, modern and Islamic identity, as well as its economic prosperity, the AKP elite aimed to promote Turkey as a role model for uprising countries (Ayata 2015; Cevik 2019). Dalay and Friedman (2013) further focused on Turkey's foreign policy in the Syrian conflict over the history of the AKP's foreign policy, they stated that it "is rooted in a tradition of both continuity and change vis-àvis the AK Party's political Islamist predecessors, the Refah and Fazilet parties". Accordingly,

the values, motivations, failures, and lessons of the AKP's political forebears may help us understand recent TFP towards the Middle East under the AKP government. However, Turkey's soft power policy was exchanged for hard power especially in the Syrian conflict due to a development in its threat perception and the search for assertive and interventionist policies by Turkish foreign policymakers; this is discussed in the following chapters.

Another internal factor shaping TFP towards the Middle East in the light of the uprisings is the 'Kurdish issue.' With the rise of various Kurdish groups representing varied strands of Kurdish nationalism and separatism, the Kurdish issue re-emerged as a burning issue in Turkish domestic politics, being "securitised domestically and exported externally into regional politics" (P. Bilgin 2007; Polat 2008). In particular, due to the re-escalation of the conflict between the AKP government and the Kurdistan Workers' Party (PKK) following the collapse of a two-year ceasefire in 2015, the rise of the Kurdish groups in the region became a domestic issue for Turkey (Robins 2014, 318). In this regard, TFP experienced "a shift from foreign commercial activity to security considerations" (Lesser 2011, 3). Haugom (2019, 215) further evaluated the role of President Erdogan in TFP following the 2016 military coup attempt, but argued that national interests were more significant than ideology in shaping TFP towards the Middle East. As such, securitisation of foreign policy was the dominant aim, notably due to the Kurdish issue and the coup attempt. This thesis examines how TFP evolved towards promoting securitisation as an intervening variable shaping Turkey's policies towards Iran since the Arab Uprisings. This occurred because the AKP elite's threat perception of domestic and non-domestic Kurdish groups elevated the need for nationalist and security-based Turkish foreign policies. Following the rise of non-state Kurdish armed groups in northern Syria post-2013, Turkey's AKP governments have followed more interventionist foreign and domestic policies than were in existence before this date.

Nevertheless, Turkey's domestic factors are insufficient to fully understand TFP towards the Middle East between 2010 and 2018. The anarchic structure of the international system influenced TFP too. In this regard, Mclean (2015) analysed the role of systemic structure in shaping TFP. According to him (2015, 12), "in Turkey, external forces are highly instrumental in shaping private foreign policy responses, and that strategic adjustments are made relatively independent of domestic debates". Oğuzlu (2020, 127–28) also claimed "[TFP] since 2002 can be partially read as Turkey's effort to adapt to external developments at international and regional levels", indicating "how Turkey's responses to the external

developments at regional and systemic levels have largely agreed with structural realist expectations". As such, TFP changed alongside shifts in the systemic structure.

Yorulmazlar and Turhan (2015, 349) argued that the changing geopolitical environment forced Turkey to alter its foreign policy aims and tools. Turkey had to change its "strategic interconnector" position, separating its ties with the West and the Middle East. Notably, the Syrian conflict influenced Turkey's power position and capabilities in the region. Robins (2013, 397) posited that "the longer the Syrian imbroglio has continued, the less plausible has Turkish leadership appeared in the Middle East, let alone further afield". Ozpek and Demirag (2014) noted that the AKP's policy towards the Arab Uprising was based on the decisions of NATO countries in terms of military intervention and the restoration of the countries affected. Indeed, so long as Turkey persuaded NATO countries to intervene, its new foreign policy is viewed as successful, especially in Syria. These authors highlight the role of relations between Turkey and its western allies, especially concerning Syria. In this thesis, it is accepted that the systemic structure and external state powers played a significant role in Turkey's foreign policy behaviour since the Arab Uprisings. However, the thesis also addresses how Turkey failed to respond to the systemic changes arising from the Arab Uprisings.

Other scholars examined TFP towards the Middle East using a more theoretical framework. Parlar Dal (2015) addressed the role of ethics and morality in TFP using a 'normative international relations' approach, especially with regard to the Syrian conflict. She concluded that TFP had normative characteristics at the beginning of the conflict, but its normative aims could not be maintained due to its inability to obtain sufficient international and domestic support. In another study, Ozdamar (2016) studied Turkey's foreign policy towards the Middle East in the post-2009 period, with a 'role theory' approach. By analysing the agent-structure relationship, he proposed that the change in Turkish foreign policymakers' perception regarding Turkey's power and regional and international states led to an ambitious TFP towards the Middle East. Based on these sources, it can be stated that individual perceptions and identities shaped TFP to a certain extent.

However, merely considering the individual level provides insufficient analysis to explain TFP towards the Middle East since the Arab Uprisings. In this regard, Guner and Koc (2018) used a 'structural balance' approach to explore TFP in the Syrian conflict, arguing that the roles of the US and Russia in the trend toward Kurdish independence, and in the Syrian conflict, played a significant part in Turkey's alignment choices. From 'constructivist and realist' approaches, Sadik (2012), Yalvaç (2014) and Kanat (2014) proposed that individual and systemic level factors played roles in TFP towards the Middle East. Kuru (2015) added that the government's policies toward the uprisings can be viewed via three levels of analysis. First, Erdogan attempted to apply the one-person rule at the individual level and used TFP toward the Middle East to manage his own internal constituency. Second, at the state level, Turkey was unable to become the engine of the Arab Uprisings owing to its insufficient military or diplomatic capacity. To explain this further, Kuru argued that Turkish domestic politics demonstrated that Turkish democracy remained frail, and its economy based largely on crony capitalism. Last, Turkish-Western relations deteriorated due to Erdogan's populist discourse and authoritarian policies at the international level, even though Turkey required the support of its NATO allies. Therefore, Kuru highlighted multi-level dimension of Turkey's foreign policy following the uprisings. This thesis is in alignment with these sources, positing that both systemic and domestic variables determined TFP, most notably since the Arab Uprisings. As such, the NCR perspective can be used to examine these variables comprehensively, and it is for this reason that some studies have sought to explain TFP via NCR.

Sönmez (2020), Yesilyurt (2017), and Kardas (2021) respectively explored Turkey's grand strategy making, the causes of Turkey's miscalculations, and Turkey's (re)securitisation policies in the Middle East since the Arab Uprisings. Sonmez evaluated the role of the AKP elite and domestic interest groups in shaping Turkey's grand strategy between 2002–2016. Yesilyurt proposed that the ideological tendencies of the AKP and its quest for consolidation of domestic power were intervening variables which paved the way for miscalculations in TFP. In a similar vein, Kardas took the prevailing security culture and domestic power configurations as the intervening variables that explained Ankara's cross-border operations in Syria and Iraq. Ovali and Özdikmenli (2019) analysed the role of nationalism and Islamism on anti-Western sentiments in TFP under NCR, comparing the Cyprus problem of the 1970s with the Syrian Civil War. Şahin et al. (2020) further addressed Turkey's response to Syrian mass migration and used "Turkey's relative power in the anarchic system" and "Turkey's foreign policy objective [of counteracting] the security threat from armed conflict in Syria" as independent variables, while also applying ideational drivers and domestic politics as intervening variables.

Given the literature regarding TFP towards the Middle East, the next subsection sheds light on the literature concerning TFP towards Iran.

1.2.3 Turkish Foreign Policy Towards Iran

Since the thesis historically covers TFP towards Iran since the Iranian Islamic Revolution in 1979, this subsection covers studies focusing on TFP towards Iran since 1979. The literature concerning the relations between Turkey-Iran relations after the Islamic Revolution has involved the role of 'ideology' as a key determinant in their relations. Due to the Iran's Islamic revolution and subsequent emergence of tensions between a theocratic Iran and secular Turkey, scholars started to analyse the domestic sources of foreign policy behaviours between the two countries. In this sense, an *Innenpolitik* approach gained more importance in the literature of Turkey-Iran relations.

Sinkaya (2005), Ozcan and Ozdamar (2010) examined the ideological tensions between two countries in the 1980s and 1990s due to the alleged role of Iran in the rise of radical Islamists in Turkey. Aras (2001) analyzed Turkish foreign policy toward postrevolutionary Iran by considering the domestic dynamics and examining Turkish politicians' attitudes toward the threat posed by Iran exporting its Islamist revolution to neighbouring countries. Aras and Polat (2008) emphasized the possibility of changing perceptions of Ankara and Tehran toward each other with the desecuritisation process of Kurdish and Islamist questions in Turkish politics. Sinkaya (2012) also noted that the fractious role of ideology in Turkey-Iran relations was replaced by more pragmatic calculations of each side in the 2000s such as trade interactions and security cooperation.

Various scholars further examined different aspects of Turkish-Iranian ties, such as economic collaboration as a significant determinant in their closer ties. According to Bahgat (2015) and Aşkar Karakir and Karasulu (2011), despite the ideological and regional confrontations between Turkey and Iran, economic cooperation between the two states has been pragmatic since the Islamic Revolution. Tür (2011), Ünal and Ersoy (2014) further highlighted economic ties between Ankara and Tehran, pointing out that regional economic conditions improved economic collaboration between Turkey and Iran, and the former in particular became an important trade route for the latter due to Iran's global isolation arising from the sanctions. Similarly, Küntay (2014) noted that, although both countries were regionally competitive, the interdependence between Turkey and Iran prevented the two

countries from clashing. This is because "Turkey does not want a marginalised and isolated Iran that constitutes one of the most significant energy suppliers to Turkey, on its southern border, whereas Iran does not want to lose its significant economic partner and supporter in the international arena" (Küntay 2014, 253). Sinkaya (2012, 139) and Stein and Bleek (2012, 139) also focused on economic collaboration between the two countries. They argued that Turkish-Iranian economic and cultural interactions prevailed over their ideological differences with the onset of the AKP's zero-problems-with-neighbours foreign policy, which emerged as a foreign policy principle of the AKP designed to normalise Turkey's relations with its neighbours. These pragmatic interactions showed a "rationalisation" in their bilateral affairs and "a much more flexible and less ideological approach" in TFP towards Iran. As such, economic interests pushed both countries into closer bilateral ties.

Security considerations emerged as another study area in the literature of Turkish-Iranian affairs. Ayman (2012, 102) reviewed the Turkish-Iranian rapprochement, arguing that mutual security interests resulting from the Arab uprisings pushed Ankara and Tehran closer together as much through need as through any shared normative concerns. In his study, Olson (2004) examined the importance of the blend of external/internal threats of both Kurdish and Azeri separatism in determining Turkey-Iran relations through an omni-balancing approach, which explains the choices of Third-World state leaders facing internal and external threats in establishing alignments. In his other studies, Olson (2000, 2001) highlighted the role of Islamist fundamentalism on Turkish-Iranian affairs. Yucesoy (2020) further assessed their rapprochement following the 2016 coup attempt in Turkey using Steven David's framework of omni-balancing. According to him, regime survival is the primary motivation of the AKP, and shaped its ties with Iran. In this regard, domestic threats to Ankara forced AKP policymakers to cooperate with Iran.

In addition to the *Innenpolitik* approaches on Turkey-Iran relations, more systemic approaches, particularly derived from a neorealist logic, took hold in the literature on Turkey-Iran relations following the Iran's Revolution. One of the most frequently cited work is Elik (2012)'s book that analyzes Turkey-Iran relations according to the concept of Middle Power states (MPs). According to Elik, diplomatic, security and energy relations between the two states from 1979 to 2011 can best be understood by the concept of MPs since they have some unique features that "distinguish them from great power and small power behaviour" and have a limited room for manoeuvre due to the intervention of superpowers and great powers.

Flanagan (2013) also dealt with the Turkey-Iran relationship within the scope of their relations with major powers such as Russia and the US, mainly from 1979 to 2013. Criss and Güner (1999) approached their study within the Turkey-Iran nexus in the region through regional issues and great powers politics. They especially focused on the possible collaborations and conflicts between Russia, Turkey, and Iran in the post-Cold War period. Efegil and Stone (2003) similarly mentioned the possibility of collaboration between Iran and Turkey in the Central Asia following the Cold War. The two sources, however, lack a theoretical framework by only defining determinants of the conflicts and collaboration between Ankara and Iran.

The Arab Uprisings certainly ushered in a new phase in Turkey's foreign policy towards Iran. As such, many scholars have studied the relationship between these two significant actors during the Arab Uprisings. Notably, their competition in this changing geopolitical environment has become a significant subject in their bilateral ties. For instance, Uzun (2013) explained Turkish-Iranian competition during the Arab Uprisings, claiming that "the main competition between Turkey and Iran has been ongoing over-exercising their soft power to attract newly emerging regimes to their models of governance". Ünver (2012, 107) predicated their historical competition on their ideological differences, saying that each state has used "their historical symbolism" to gain an opportunity to influence the Sunni-Shia status quo. Keyman (2016), Monier (2014), Dallmayr (2011) further defined their role in ideological competition in the post-2011 period, by saying that Turkey and Iran have attempted to enhance their regional influence. Wastnidge (2019b, 605) also explored each state's identity in an analysis of their foreign policy behaviour. Such studies indicate that the literature in the field has focused on ideological aspects of Turkish-Iranian affairs in the post-Arab Uprisings term.

Some academics looked at the Turkish-Iranian conflict from a different angle. Sinkaya (2019), Şen (2019) and Keyman and Sazak (2015) studied their periodic estrangement and rapprochement. Notably, Şen (2019, 148) discussed the role of the US in Turkish-Iranian relations since the Arab Uprisings, arguing that the decline of the US presence in the Middle East resulted in competition between the two countries since each tried to increase their geographical influence. However, he also noted that their periodically contentious relations with the US in the post-2016 era drove a rapprochement between Ankara and Tehran. Sinkaya (2019) also analysed the AKP elite as another determinant in their ties, alongside the geopolitical context, and structural factors. According to him, the AKP elite followed "rationalisation, institutionalisation and compartmentalisation strategies" in TFP towards Iran.

Kang and Kim (2016) noted that the cooperation between Ankara and Tehran in the Middle East is derived from Turkey's bandwagoning strategy with the US. Sinkaya (2018) analyzed the role of the Kurdish issue on Iran-Turkey relations from the Iranian perspective and examined how the Kurdish issue has created a security problem for both countries and has been a root of both cooperation and competition for their relations by considering the current developments in the Northern Syria.

Several other scholars have conceptualised Turkish-Iranian relations using a variety of conceptual frameworks. Charountaki (2018) explored the significant determinants of Turkish and Iranian foreign policies and their influence on events in the Middle East since 1979, using a multidimensional interrelations model. Although she adopted a historical review of Turkish-Iranian relations, she also conducted a political analysis of their bilateral affairs during the Arab Uprisings at three levels: the domestic, regional, and international. Charountaki, however, only explained their bilateral relations during this period by analysing the Syrian Crisis and focusing on only energy politics and the Kurdish role in their relations. Further work is therefore required to define TFP towards Iran since the Arab Uprisings by involving different determinants in shaping TFP towards Iran. In another study, MacGillivray (2019) used a historical sociological framework to examine Turkey-Iran relations between 2002 and the signing in 2015 of the Joint Comprehensive Plan of Action (JCPOA) when Iran agreed to strict limits and oversight of its nuclear programme in return for sanctions relief from the United States, Russia, China, Germany, France and the United Kingdom. He concluded that historical, domestic, and regional/international developments influenced their bilateral ties, but his study fell short of covering events surrounding the Arab Uprisings and their impact on bi-lateral ties.

Kumral (2020) further examined Ankara-Tehran ties from an 'affective-normative' and 'psycho-social' perspective. According to him, bilateral emotional entanglements such as brotherly love, trust, and dyadic jealousy, as well as neighbourhood narratives like the Ozsoy legend, the cold shoulders saga, the Caldiran myth and the story of Kasr-I Sirin, shaped Ankara and Tehran's historical ties. Kirchner and Baş (2017, 172) studied Turkish and Iranian policies during the Arab Uprisings using NCR, arguing that the AKP overestimated its local partners, such as the Muslim Brotherhood in Egypt and the Free Syrian Army in Syria, and thus failed to evaluate regional power re-distribution. The authors also pointed out that, while Turkish policymakers wished to export 'the Turkish model,' thus shifting TFP strategies from "non-interference (zero-problems) to interventionism (sectarianism)", Iran followed "a strong sense

of pragmatism and abandoned their local allies when they seemed to prove useful no longer". They thus concluded that Turkey's miscalculations failed those seeking to overthrow authoritarian rule, not least in Syria. However, this study mostly explained Turkish and Iranian perceptions towards the Uprisings individually, rather than specifically focus on TFP towards Iran in the midst of the uprisings.

1.3 Originality of the Thesis

This literature review has explored the debates concerning Turkish foreign policy. While this literature is rich in its empirical and conceptual reach, there remains key gaps that this thesis seeks to address. As shown in the previous section, the literature on Turkish-Iranian ties since the Arab uprisings is limited and most of the studies have focused upon their competition in Syria. Furthermore, few studies have analysed their relations from a theoretical approach. Despite the high number of works on the relations between two countries that incorporate some conceptual approaches, it can be observed that TFP towards Iran between 2011-2018 is still under-theorised because these studies either lack a sufficient period or focus on bilateral affairs rather than emphasising the Turkish side alone. There is support in the literature, and in this thesis, for the co-determination of intervening and independent variables in TFP. The context of the Arab Uprisings is generally not considered, specifically regarding TFP towards Iran, and neither is there an analysis of different unit-level variables in these studies' explanations of TFP.

Based on the literature survey regarding TFP towards the Middle East and TFP towards Iran following the uprisings, the thesis argues that a simple calculation of relative power or domestic factors is insufficient to explain Turkey's policy towards Iran because both are necessary in the analysis of the subject. While Turkey's relative power position determines TFP in the long term, domestic factors have an influential role in shaping TFP towards Iran in the short-to-medium term. This thesis is thus the first comprehensive attempt at exploring Turkish perspective towards Iran since the Arab Uprisings, using neoclassical realism (NCR) which includes system, state, and individual level of analysis. The contribution of this thesis is that it enhances these studies by examining TFP via the lens of NCR while also considering selected intervening variables and Iran. Notably, it employs Schweller's balance of interest theory to define Turkey's domestic considerations, a more appropriate approach for explaining Turkey's alignment choices and rapprochement with Iran since the start of the Arab Uprisings, from December 2010 to July 2018. Therefore, this study is distinguished from other theorydriven studies on TFP towards Iran by its original application of NCR that is more responsive to Turkey's peculiarities in its relations towards Iran.

As the literature survey demonstrated, decisionmakers' responses to the international environment are significant for an understanding of a state's specific foreign policy because their world views can shape their perceptions and personality. Therefore, the AKP elite perception must be taken into consideration to understand how they viewed particular issues regarding the uprisings and Iran. Moreover, as Dueck (2009, 146) posits, a neoclassical realist model "begins by positing that state officials have some conception of the national interest in the face of potential external threats. These conceptions may be misguided, but they are nevertheless genuine." In this regard, Turkey's domestic considerations must be added in examination of TFP towards Iran from December 2010 to July 2018. As such, Schweller's balance of interest theory is useful for exploring Turkey's alignment with Iran. As a result, an explanation of TFP towards Iran must be approached on many levels if it is to capture the complex dynamics of the relationship. The thesis is also the first comprehensive study examining TFP towards Iran in the Kurdistan Region of Iraq under a theoretical framework while providing different theoretical viewpoint on Turkey's amity and enmity affairs with Iran in Syria. The study clearly conceptualises why Turkey choose an alignment with Iran despite its initial competition with Iran in these areas.

1.4 Research Methodology

Based on the research questions, this study uses a qualitative methodological approach to understanding TFP towards Iran from December 2010 to July 2018. To explain the causal chain linking independent variables (systemic changes and the relative power of Turkey) and the dependent variable (TFP towards Iran since the Arab Uprisings), process-tracing is used as the central methodology. This is considered the best tool to evaluate how intervening variables play a role in TFP towards Iran since the Arab Uprisings, because process-tracing allows analysis and evaluation of specific causal mechanisms within the formulation, development, and evolution of a specific foreign policy in a particular period, without the use of controlled comparisons (Lobell, Ripsman, and Taliaferro 2016, 132). As Benneth and George (2005, 205–6) note, it also "attempts to identify the intervening causal process —the causal chain and causal mechanism —between an independent variable (or variables) and the outcome of the dependent

variable". Accordingly, it provides the most appropriate explanation to understand a causal link between variables.

Process-tracing involves the use of "histories, archival documents, interview transcripts, and other sources to see whether the causal process a theory hypothesises or implies in a case is in fact evident in the sequence and values of the intervening variables in that case" (George and Andrew 2005, 6). As such, this thesis utilises two data collection techniques. In terms of existing material, it draws on theoretical and non-theoretical studies, including those using NCR and other IR theories, and examinations of contemporary Turkish policies towards the Middle East in general, and Iran in particular. The survey of the literature examines a multitude of English and Turkish primary and secondary sources such as books, refereed journal articles, online newspaper and magazine articles, think tank reports, documents/notes posted on Turkish governmental websites, and documents published by research centres. To reflect the AKP elite's perception, especially, pro-AKP newspapers and state-sponsored think thank reports were used as well as unbiased secondary sources.

Interviews were used as the second research technique, to resolve the paucity of substantial secondary data on this topic; these interviews were used to substantiate findings from a wide range of sources, most of which are non-refereed. Additionally, interviews can fill potential gaps between the literature and new information provided by interviewees. They are complementary to the analytical research and used to support or challenge the insights derived from secondary data. As such, this study drew data from twelve semi-structured interviews with open-ended questions which were designed to unpack several previously unconsidered topics regarding TFP towards Iran since the Arab Uprisings, the Syrian conflict, and the politics of Kurdistan Region of Iraq. The interviewees were selected using two criteria: expert knowledge on the topic, and ease of access. Access to interviewees was gained from a combination of strategies, including personal contacts, introductions made by third parties, cold calling, and credible affiliations with previous interviewees.

Moreover, to maintain an accurate record, the interviews were audio-recorded with the permission of the interviewees, who were notified that they could withdraw at any time. The interviews were conducted in English to minimise errors in communication where possible since all interviewees were proficient in English. Following the interviews, the gathered data were assessed using content analysis to identify certain words, themes, and concepts, selected

based on their relevance and contribution to this study. The extracted items were then categorised and assessed for both their quality and value to the research and applied as appropriate to chapters in the thesis.

However, it should be noted that just like any study, this thesis is not without drawbacks. There are methodological limitations concerning the limited access to AKP elite interviewees due both to the turmoil being experienced in Turkey at the time of the research, and possibly the reluctance of politicians to fully reveal their views and ideas. This resulted in a shortfall in interviews with political personalities, specifically AKP politicians. To mitigate this issue and increase reliability and objectivity, expert/specialist interviewees were identified according to their different ideological standpoints and professions regarding their support for the AKP or detailed knowledge of its members during the period of the study. These interviews are complementary to the analytical research and used to endorse or challenge the insights derived from the secondary data. All the secondary sources were corroborated by expert interviews to help explore the research questions. The identities of the interviewees remain anonymous to prevent any possible pressure on them in future. Therefore, they are identified only by their job title, such as academic, journalist, and expert. Due to the Covid-19 pandemic, the interviews were conducted between July and August 2020 via Skype.

As case studies, the thesis also explores the impact and influence of the Syrian conflict and the politics of Kurdistan Region of Iraq on TFP towards Iran since the Arab Uprisings, from December 2010 to July 2018. According to Gerring (2008, 645), a "case-study analysis focuses on one or several cases that are expected to provide insight into a larger population", and as such they permit explanations of research through an in-depth analysis of a single unit. The cases of the Syrian conflict and the Kurdistan Region of Iraq were selected as the appropriate examples to represent the causal mechanism of TFP towards Iran from December 2010 to July 2018 since they can provide sufficient analysis to understand how independent variables (systemic structure following the uprisings and Turkey's relative power position) and unit-level variables (AKP elite perception and domestic security and economic considerations) influenced TFP towards Iran.
1.5 Structure of the Thesis

The thesis is composed of eight chapters. The aim of this introductory chapter was to present the objectives, significance, and methodology of the study. Chapter 2 introduces NCR as the theoretical framework and provides an overview of the alternative theoretical approaches by comparing them with NCR. It investigates how and where NCR positions itself within both the realist school and the broader international relations literature. It also explains how NCR can be used to understand TFP in Turkey.

Chapter 3 provides a historical review of Turkey's relations with Iran by looking at systemic and bilateral factors. In terms of systemic factors, it analyses the Iran-Iraq war, the 1991 Gulf war, the collapse of the Soviet Union and the emergence of the newly independent states in the Caucasus and Central Asia, the Palestinian-Israeli Conflict, and the US invasion of Afghanistan and Iraq. As bilateral factors, it examines ideological and sectarian divisions, minority issues, the Iranian nuclear programme, and economic relations.

Chapter 4 examines the independent variables of TFP Iran since the Arab Uprisings, analysing the uprisings on a systemic level and evaluating the systemic changes and Turkey's relative power position. This chapter mainly focuses on independent variables of the analysis and examines the role of these variables on TFP towards Iran. Chapter 5 explores the intervening (unit-level) variables in TFP towards Iran since the Arab Uprisings, highlighting the role of the AKP elites' perceptions and domestic considerations of Turkey. The two chapters help to define TFP towards Iran from December 2010 to July 2018 as the dependent variable of the thesis.

The remainder of the thesis contextualises the research by using detailed thematic case studies to shed light on TFP towards Iran since the uprisings, with the aim of demonstrating that NCR allows a greater understanding of the complexity of Turkey's affairs with Iran. In this light, the Syrian conflict and the politics in Kurdistan Region of Iraq are analysed to understand TFP towards Iran through an NCR understanding in Chapters 6–7, respectively. In the case study chapters, TFP towards Iran is explored in detail by comparing the influence of systemic and domestic level variables. Chapter 8 reviews the original questions posed in the introduction chapter and discusses the findings of the thesis.

This thesis utilises the theoretical framework of neoclassical realism (NCR) to examine Turkish foreign policy (TFP) towards Iran since the Arab Uprisings, from December 2010 to July 2018. This chapter begins by analysing NCR, which integrates system level and unit level variables to define the foreign policy behaviour of states, and then engages with the rival theories and critics of NCR in much more detail to understand the reasons for the rejection of possible alternative theories. It then explains how NCR is used to analyse TFP towards Iran in the post-Arab Uprisings era and concludes with an explanation of how the conceptual framework will be employed in subsequent chapters, and in the context of this thesis as a whole.

2.1 Neoclassical Realism

Neoclassical Realism was first outlined and developed by Gideon Rose in his 1998 article *Neoclassical Realism and Theories of Foreign Policy* as a new foreign policy approach, including systemic and unit-level variables of states. In this regard, NCR first seeks to find a logical answer to the following questions:

Under what conditions do nations expand their political interests abroad? What is the relationship between a nation's external behaviour and its domestic mobilisation? How do political elites perceive and think about world politics? How do states assess and adopt to changes in their relative power? How do states respond to threats and opportunities in their external environment, and do different kinds of states respond in different ways? What explains variation in state alliance strategies, whether they choose to balance, buck-pass, bandwagon, chain gang, or avoid alliances altogether? (Elman and Elman 2003, 317–18)

To understand these questions, NCR notably accentuates certain significant premises, many of which are also seen in other realist approaches. To begin with, 'the primacy of the conflict group' is the premise of NCR. Like other variants of the Realist approach towards international relations theory, NCR deems that states are the essential units and assumes that competition between states shapes international politics. As Kenneth Waltz (2010, 94) argues, "[S]tates ... set the terms of the intercourse When the crunch comes, states remake the rules

by which other actors operate". But while Waltz sees states as unitary actors shaped by the anarchic structure of the international system, NCR therefore assumes that each state is unique due to its domestic factors. As such NCR argues that while states clearly operate within an global order defined by anarchy, domestic factors can and do influence how states define their interests and adopt particular policies in meeting wider systemic challenges (Schweller 1998, 60–61).

Second, NCR views international politics as having 'an anarchic nature' due to the lack of central authority either in implementing agreements or protecting states under threat, and this anarchic environment culminates in a continual struggle between self-interested groups to dominate conditions of general scarcity and uncertainty (Juneau 2015; Lobell, Ripsman, and Taliaferro 2009). As a result, competition for power and security in international politics emerges as a fundamental aspect in states' external affairs, although anarchy may be mitigated through institution-building (Kitchen 2010b, 140).

While trying to control this anarchic sphere, however, states determine their policies by considering the probability of conflict. In the event of low probability, states tend to concentrate on other matters; otherwise, security may be their main priority (Juneau 2015, 30). As such, NCR perceives anarchy as a "permissive condition, rather than an independent causal force" (Taliaferro, Lobell, and Ripsman 2009, 7). Instead of assuming that states seek security, it assumes that they respond to the uncertainties of international anarchy by seeking to control and shape their external environment. In contrast, security is only one element that states value when defining their interests and, as such, they are likely to seek more external influence (Rose 1998, 152). States' foreign policy may also lead to different results even if the structural condition of anarchy forces certain directions to be taken, because a state's foreign policy is ultimately conducted by those who evaluate the power and intentions of other states (Lobell, Ripsman, and Taliaferro 2009, 7).

'The primacy of power' is the third premise of NCR, as states require the power either to achieve international domination or, more reasonably, to secure themselves (Lobell, Ripsman, and Taliaferro 2009, 14–15). The capabilities of states and their place in the international system shape their intentions and behaviours (Juneau 2015, 30). As such, the central empirical prediction of NCR is that an increase in the relative material power of a state will eventually lead to a corresponding expansion in the ambition and scope of its foreign policy activity, and that a decrease in such power eventually leads to a corresponding contraction. The relative amount of material power resources possessed by countries likely shapes the magnitude and ambition of their foreign policies in the long term; that is, as the relative power of a state rises, it will seek more influence abroad, and as it falls, the actions and ambitions of the state will be scaled back accordingly (Rose 1998, 152,167).

However, NCR adopts a flexible approach to the operationalisation and measurement of power due to the unit-level variable of states (Juneau 2015, 30). Although the relative power of a state in international politics is a predominant factor shaping a state's foreign policy behaviour, systemic pressure should be translated through intervening unit-level variables, such as policymakers, perceptions, and state structure (Lobell, Ripsman, and Taliaferro 2009, 5). Notably, foreign policymakers' perception of capabilities has a more significant role than actual and physical resources and capabilities (Rose 1998, 147). Based on unit-level variables, states may target more than one foreign policy outcome, rather than merely power maximisation or security (Juneau 2010, 6; Rose 1998, 152). NCR thus emphasises the importance of the unit-level factors of states, as well as their power and capabilities.

'Confined rationality' is another assumption of NCR. In essence, rationality is a contested feature in realist canon. For instance, Waltz (2010, 118) points out that "[The balance-of-power] theory requires no assumptions of rationality or constancy of will on the part of all of the actors" since the systemic structure shapes their policies, while Keohane (1986, 165) posits that states, as rational actors, calculate the costs and benefits of their choices. In this regard, NCR adopts confined rationality, finding a middle ground in the debate on the theme. According to NCR, systemic pressures force "states to act within a set of parameters", already creating opportunities and constraints for them. Therefore, states are not assumed to be rational, but, at the same time, states assess the advantages and disadvantages of achieving their goals (Juneau 2015, 31).

Given the above assumptions inherent in NCR, it can be stated that NCR adopts a multilevel analysis in examining the foreign policy of states. As such, as Taliaferro (2012, 79) argues, "neoclassical realism['s] comparative advantage lies in its willingness to integrate unitlevel and systemic-level, as well as ideational and material, variables into a coherent explanatory framework". Foulon (2015) further summarises NCR, arguing that the theory bridges gaps in international relations literature by involving space (domestic versus international), cognition (do ideas matter, or is there nothing more than objective facts) and time (the present versus the future).

2.1.1 Variables of Neoclassical Realism

This subsection examines the variables of NCR to understand how they define a state's foreign policy behaviour.

2.1.1.1 Independent Variable

NCR assumes that the international system is significant in the shaping of a state's foreign policy. The international system is, however, defined differently by various NCR proponents. According to Schweller (2003, 332) "[t]he inclusion of interaction in the definition of the system allows process variables (such as institutions, norms, or rules) as well as structural variables to define the nature of world politics and to have an effect on their operation and dynamics". Ripsman et all (2016, 40–56), however, emphasise certain material factors, which can be structural modifiers such as geography, rates of technological diffusion, and the offence-defence balance in military technologies, as well as the variables of systemic clarity and the nature of the strategic environment. Despite the different explanations, all neoclassical realists agree that the international system can be influenced by the foreign policies and grand strategies of great powers. That is why a state's relative power position is the independent variable. As such, "the scope and ambition of a country's foreign policy are driven first and 0opforemost by its place in the international system and specifically by its relative material power capabilities" (Rose 1998, 146).

States prioritize survival and power because there is no higher authority above them. Therefore, the international system is susceptible to conflicts and wars. NCR argues that the anarchic international system forces states to determine their foreign policy by providing them with the stimulus "to emulate the successful political, military, and technological practices of the system's leading states or to counter such practices through innovation" (Taliaferro 2006, 467). This means that each state aims to be an economic and military power in international politics since "powerful states have a wider field of action and more available options" than less powerful ones (Edwards 2013, 55). Power is thus a tool by which states can enhance their external influence. By seeking more control and impact in their external sphere, rather than security to respond to the murkiness of international anarchy, states seek to increase their

relative material power resources, and desire more external influence (Rose 1998, 146, 152). In doing so, their position in the international system becomes significant enough for them to achieve their aims and interests, and the greater the relative power in the international system, the more external influence they have. In this regard, "the relative amount of their [states] capabilities will influence and shape the ambition and size of their foreign policy" (Rose 1998, 152). However, the anarchic international system forces states to determine their behaviours in line with the system by rewarding or punishing them. In the long term, a state's foreign policy has to be within the boundaries of the limits and opportunities determined by the international environment (Rose 1998, 167).

Moreover, influential NCR supporters define 'power' as a state's ability to mobilise internal resources to respond to a threat. Since states have limited ability to extract national power and mobilise resources, they thus separate state power from national power because foreign policy is not made by the nation, but by the government (Taliaferro 2006, 296). As such, state power means "the portion of national power which the government can extract for its purposes and reflects the ease with which central decisionmakers can achieve their end" (Zakaria 1999, 9). Wohlforth (1993, 303) adds that " 'power' is determined in part by how given material distributions are interpreted. Since many interpretations are always possible, state leaderships will tend to interpret particular changes opportunistically". That is to say, foreign policymakers of states seek ambitious foreign policy goals in line with national interests when they have gained enough authority; however, they may still be constrained by other factors that influence foreign policy behaviour. Therefore, NCR proponents have doubts about the measurability of power (Lindemann 2014), indicating that intervening variables are another significant factor determining a state's foreign policy outcomes.

2.1.1.2 Intervening Variable

The concept of the intervening variable is among the most significant contributions of NCR, which asserts that the 'transmission belt' between the relative distribution of power and states' foreign policy behaviours is not perfect (Rose 1998, 157; Taliaferro 2006, 465). Owing to ambiguous and unclear signals from the anarchic international system, states need to interpret this anarchic environment "according to subjective rules of thumb" (Rose 1998, 152). However, states may not respond rationally and/or directly to the systemic pressures and outcomes of these imperatives. In this regard, a state's domestic variables can provide a clear

explanation of its foreign policy shaping (Zakaria 1999, 12). Intervening variables should thus be included in an analysis of a state's foreign policy behaviour.

Intervening variables can also indicate that a state may adopt different foreign policies from other states despite identical systemic challenges. Stated differently, "[w]hen states do not respond ideally to their structural situations, neoclassical realism tells us we should find evidence of domestic politics and ideas distorting the decision-making process" (Rathbun 2008, 206). This demonstrates that states are not unitary actors because of distinctive domestic factors, and that an objective assessment of relative power is insufficient to examine a state's foreign policy choices in the short- and medium-term (Lobell, Ripsman, and Taliaferro 2009, 28; Taliaferro 2006, 485). These domestic factors may thus result in states deviating from systemic incentives.

Neoclassical realists classify various intervening variables in their analyses to enhance the theoretical validity of NCR, meaning that there is no specific intervening variable in determining a state's foreign policy outcome. Rather, different intervening variables can be selected based on the research questions of a study. Decisionmakers' perception is one common intervening variable because studying the attributes of individuals is the primary concern of neoclassical realists (Lobell, Ripsman, and Taliaferro 2009). Accordingly, they believe that the way leaders perceive the world determines their states' foreign policy behaviours, since they interpret the particular systemic structure from their own perspective, and therefore their perception of the international environment and relative power of their state within the international system plays a significant role in shaping foreign policy choices.

Zakaria (1999, 42), for instance, asserts that "statesmen, not states, are the primary actors in international affairs, and their perceptions of shifts in power, rather than objective measures, are critical". Lobell et all (2009, 42) also emphasises the leaders' role in determining foreign policy, arguing that leaders operate on two levels. When responding to the external sphere, they must extract and mobilise resources from domestic society, while also collaborating with domestic institutions and preserving the support of key stakeholders. Therefore, how the way decisionmakers think about the state's capability may prevail over the state's actual power capacity at specific times, even if the international system provides clear information about the threats and constraints that states face.

Taliaferro (2012, 78) adds that foreign policymakers must fulfil several tasks: making subjective probability assessments; prioritising various threats and opportunities; and discerning future intentions and shifts in the distribution of power. When fulfilling these tasks, they may misconceive the state's real power capacity. Accordingly, they "are also exposed to misperceptions and time lags between real changes and their (mis)perception of these changes" (Jervis 2017). Since the internal and external considerations of decision-makers play a significant role in a state's response to threats and opportunities (Schweller 2004, 5), as Brawley (2009, 97) posits, "a strictly top-down approach is unable to dictate the precise foreign policy of states and/or explain variances that might occur. This is most apparent when units with similar attributes choose different paths, or when units with different attributes follow the same path".

Political domestic constraint on national power and state-society relations is another common intervening variable that influences a decisionmaker's perception. This is because foreign policymakers may face difficulty in quickly attaining and extracting a state's overall material power resources from their societies when shaping foreign policy and describing national interests (Lobell, Ripsman, and Taliaferro 2009, 26; Rose 1998, 161). As such, state power is believed to be "the portion of national power the government can extract for its purposes and reflects the ease with which central decisionmakers can achieve their ends" (Zakaria 1999, 9). This indicates that foreign policy is not an unmediated product of the decisionmakers.

In another study, Schweller (2004) lists four other domestic variables constraining decisionmakers: elite consensus, elite cohesion, social cohesion, and regime vulnerability. According to him, elite consensus is the degree of shared intra-elite perception regarding certain global facts of a particular nature and necessary policy remedies. In contrast, elite cohesion is the degree to which constant internal divisions fragment the central leadership. Social cohesion further explains the reflection of the society's support for existing institutions by responding to whether they are legitimate and appropriate (Schweller 2004). Regime vulnerability also seeks to respond to whether the governing elite groups confront serious non-democratic challenges from the military, opposing parties, or other powerful groups in the society (Schweller 2004, 173). He (2004, 169) further discusses that "elite consensus and cohesion primarily affect the state's willingness to balance (interests), whereas government/regime vulnerability and social cohesion influence the state's ability to extract

resources for this task. The combination of these four variables determines the degree of state coherence".

When comparing the US and China's foreign policies, Schweller (2018) further unpacks nationalism as another intervening variable. He (2018, 23) proposes that "The interaction between nationalism and power trajectory produces entirely different foreign policy orientations in rising and declining powers—the former embraces an outward-looking, extroverted foreign policy of expansion, while the latter adopts an inward-looking, introverted foreign policy of restraint and retrenchment". He concludes that China seeks more global influence in contrast to the US due to nationalist factors.

Taliaferro (2006, 486) also lists state institutions, state-sponsored nationalism, and statist or anti-statist ideology as domestic variables affecting the ability of leaders or decisionmakers to extract or mobilise resources from domestic society. According to him (2006, 491), state-sponsored nationalism has a tendency to escalate social cohesion and the desire of individuals to identify with the state, thereby allowing leaders to extract and mobilise resources efficiently from society for national security aims; in contrast, ideology is able to ease or restrict leaders' attempts to extract and mobilise resources, relying on "the content of ideology and the extent to which elites and the public hold common ideas about the proper role of the state vis-à-vis society and the economy".

Lobell et all (2009, 58–79) add four intervening variables: the images and perceptions of state leaders, strategic culture, state-society relations, and domestic institutional arrangements. They conclude:

These variables include psychological, bureaucratic/organisational, societal, and institutional models, which reflect alternative approaches to foreign policy analysis. They reflect the various constraints on the central actors, the interactions within and between decision makers and society as a whole, and the processes and mechanism by which foreign policy is formulated, each of which can affect the manner in which states respond to? the external stimuli (2009, 59).

Kitchen (2010a) examines the role of strategic ideas in order to demonstrate the ideational factor of intervening variables. Of the different intervening variables, Lindemann (2014) suggests that the main ones identified by neoclassical realist studies can be placed into

four groups: "the state's structure and domestic competition (measuring the ability of a state to pursue a certain policy) as well as perception and identity (measuring the willingness of a state to pursue a certain policy)".

Bearing in mind that intervening variables play a secondary role in shaping a state's foreign policy, the international environment prevails over domestic factors in the long-term (Rose 1998, 151). This is because anarchy and the distribution of power result in systemic incentives and constraints on states and compel them to adopt similar strategies long-term (Rose 1998, 146–47). Therefore, in Dueck (2009, 139)'s words, "domestic politics 'matter,' not as a primary cause of intervention, but rather as a powerful influence on its exact form". This indicates that NCR uses a multi-level analysis involving both the state unit and its embedded system. Moreover, a state's foreign policy outcome is likely to be differentiated owing to various intervening domestic variables, and thus states may behave neither rationally nor coherently due to complex decision-making processes (Schweller 2004). As a result, intervening variables play a co-determinant role in the analysis of a state's foreign policy in connection to independent variables. Notably, in the short-to-medium term, the analysis of intervening variables can facilitate a clear and proper explanation for a state's foreign policy shaping by employing what might be termed their 'transmission belt role' between independent and dependent variables.

2.1.1.3 Dependent Variable

In the causal chain of NCR, the state's foreign policy is the dependent variable. In particular, "why different states, or the same state at different historical moments, have different intentions, goals, and preferences toward the outside world" is a particular interest of neoclassical realists (Zakaria 1999, 14). In relations to systemic structures, NCR addresses that when the relative material power of a state increases, the ambition, and scope of its foreign policy behaviour will expand; conversely, the objective and scope of its foreign policy behaviour will contract (Rose 1998, 167). However, the domestic factors of states may result in different foreign policy responses in the short-to-medium term (Lobell, Ripsman, and Taliaferro 2009). Therefore, the foreign policy of a state is affected by two different levels — the systemic and the unit —over different terms: short-to-medium, and long-term.

State behaviour is defined differently by the view that a state follows different aims when shaping its foreign policy. Mearsheimer (2001, 30–36), as a supporter of offensive

realism, assumes that, in the pursuit of survival in the international system, state behaviour demonstrates three patterns: fear, self-help, and power maximisation. However, he adds that power maximisation is a more dominant pattern of state behaviour since states, notably great powers, desire to alter international power relations on their own behalf. As such, states tend to follow revisionist policies in the international system. In contrast to Mearsheimer, Waltz (2010, 126) claims that security is the chief goal of most states as they wish to maintain their systemic position. By supporting the status quo in the international system, states thus balance against any aggressive, powerful state as a potential threat (Waltz 2010, 102,128). Based on these definitions of the systemic level, the aims of states regarding the international structure may determine whether the tendencies of a state are revisionist or status quo.

For NCR, however, there is no single assumption (security or power maximisation) shaping the strategy of either revisionism or the status quo, and this challenges Waltz's claim that "security seekers are highly war-prone", while also accepting the difficulty of recognising the aims of other states (Davidson 2016, 9). Security seekers may have a tendency to revise the status quo if there is a strategic benefit for them, yet, at the same time, power maximisers may not naturally alter the distribution of goods —involving territory, status, markets, the expansion of ideology, and the creation or change of international law and institutions —in the event that "further aggrandisement is a liability rather than an asset to their power position" (Davidson 2016, 13). On these bases, NCR proposes several ways of understanding state behaviour.

Schweller (1994, 88) marks that "Satisfied powers will join the status quo coalition, even when it is the stronger side; dissatisfied powers, motivated by profit more than security, will bandwagon with an ascending revisionist state". Schweller divides the revisionist states into two categories, "jackals and wolves" (1994, 99–100), whereby "Jackals will pay high costs to defend their possessions but even greater costs to extend their values," and are "risk-averse and opportunistic". Wolves are predatory states that "value what they covet far more than what they possess", and are thus willing to take significant risks, often becoming reckless. However, revisionist states may not exhibit aggressive policies, and can thus be classified as limited-aims revisionists and unlimited-aims revisionists. Schweller (2015, 7) points out that,

The key question is whether the rising power views the protection and promotion of its essential values as dependent on fundamental changes in the existing international order; or whether it is merely dissatisfied with its prestige and portions of the status quo, e.g., its sphere of influence, certain international norms. If the former, then it is a revolutionary state that cannot be satisfied without destroying essential elements of the international order. If the latter, then its grievances can be satisfied, while preserving, and in some cases actually strengthening, the established order.

It seems that the more dangerous states are of the unlimited-aim type. As such, regional powers can be defined as limited-aim revisionist when they "seek either compensatory territorial adjustments to reflect their increased power, recognition as an equal among the great powers, or changes in the rules and decision-making procedures within, but not the basic norms and principles of, existing regimes" (Schweller 2015, 7–8). All revisionist states are prone to cooperate with more powerful ones to benefit from the cooperation (Davidson 2016, 9). In this regard, the pursuit of profit may result in revisionist state foreign policy tendencies, whereby states jump on the bandwagon for profit to ride "free on the offensive efforts of others to gain unearned spoils" (Schweller 1994, 74–75).

NCR further explains the foreign policy of status quo states. For instance, Schweller (1994) proposes that satisfied states will ally with other satisfied states to maintain their position in the system and can this be divided as "lions and lambs". He goes on to add that "lions are states that will pay high costs to protect what they possess but only a small price to increase what they value... As delighted states, they are likely to be status quo powers of the first rank", while "lambs are weak countries that will pay only low costs to defend or extend their values" (Schweller 1994, 101–2). As such, a status quo state will tend to cooperate with the most powerful status quo state in the international system because it poses no inherent danger. Schweller further explains that such "states can signal their intentions by spending on defence regardless of whether the offence or defence is dominant" (Davidson 2016, 9). In doing so, they tend to maintain the status quo and distribution of power.

Three other kinds of signals can show whether a state has revisionist or status quo tendencies. The first is "shared normative principles about the legitimacy of ruling elites, the sanctity of borders and treaties, the limits on the use of force, and respect for another state's sovereignty, territorial integrity, and traditional sphere of influence" (Chan, Hu, and He 2019, 6). The second is "pertinent to discerning a state's foreign policy orientation", including its "record of joining international accords and conventions which communicate its predisposition to abide by the international consensus as represented by these multilateral treaties" (Chan, Hu, and He 2019, 6). The third is "a state's voting record in the UN that shows the state's

willingness to align with the rest of the global community" (Chan, Hu, and He 2019, 6). When defining states according to their aims, moreover, Cooley (2019) also proposes a "two-dimensional property space" to generate four state types. Accordingly, "status quo actors" are satisfied with the international order and distribution of power, but "revolutionary ones" seek to change both situations. "Reformist states" seek to alter only the international order while "positionalist ones" seek to shift the distribution of power.

Intervening domestic factors are another influential determinant of a state's foreign policy behaviour. In this regard, Davidson (2016) focuses on the preferences of powerful domestic groups. The dominance of "externally oriented" groups, such as nationalists, may result in revisionist policies for rising states while causing status quo strategies for declining ones. The dominance of "internally oriented" groups, such as welfare beneficiaries, is more likely to cause the opposite case: "rising states are less likely to become revisionist, and declining states are less likely to become status quo seekers" (Davidson 2016, 2).

Ripsman et all (2016, 83) further divide dependent variables into time frames. In this regard, "in the shorter term, defined in terms of days, weeks, and months, states navigate rather fixed international circumstances" because "relative power is unlikely to shift within such a short time frame". In "the short-to-medium term —defined in terms of months and years, but not decades —policymaking is more forward-looking and less responsive to fixed conditions and imperatives. Instead, policymakers engage in strategic planning, or an attempt to construct grand strategy" (2016, 85). In another time frame, "the medium-to-longer term, defined in termed of years and decades, allows the strategic choices of the different great powers to interact and to have an impact on international systemic outcomes", which means "observable political phenomena resulting from the coaction and interaction of the strategies pursued by two or more actors in the international arena" (2016, 86). Finally, in "the longer term, defined in terms of decades, international outcomes and the policies and grand strategies of the principal units themselves can help reshape international structure" (2016, 87).

To sum up, depending merely on the influence of external variables is short-sighted and unreliable in determining the foreign policies of states, since foreign policy does not inevitably follow systemic pressures. The reliance on internal variables alone, however, is a limited approach, because systemic imperatives influence the context of foreign policy. Rather, the connection between external and internal variables is necessary to capture the complexity of foreign policy, thus making NCR more efficient and robust than purely structural theories or traditional multilevel theories for the analysis of the relationship between international affairs and foreign policy. In the light of these assumptions, Table 2.1 demonstrates the variables of neoclassical realism.

<i>Table 2-1:</i>	Variables	in Neocl	lassical	Realism
10010 2 1.	<i>v ui uvi us</i>	mncoci	assicai	ncuism

Independent Variable	Intervening Variable	Dependent Variable
Relative power in the international system	Domestic Factors	Foreign Policy

2.1.2 The Differences between Neoclassical Realism and other International Relations Theories

This section begins by examining why NCR is different from classical realism (CR) and neorealism (NR), and then compares NCR and Innenpolitik in this theoretical debate. Accordingly, this section reveals why NCR is a suitable international relations theory in explaining TFP towards Iran since the Arab Uprisings, from December 2010 to July 2018.

2.1.2.1 Neoclassical Realism and other Realist Accounts

In essence, NCR shares some significant aspects with CR and NR as a realist account. To begin with, power is the main subject of these three realist accounts. Proponents of CR deem that power is part of human nature due to the innate human desire to dominate. As such, power in international affairs is defined as an aspiration which states seek to obtain for their own interests (Morgenthau 1947, 193). For NR, power is a security issue since states aim to have greater power to protect themselves in an anarchic environment. That is to say, self-protection is the primary motivation for states to increase their capabilities and military resources (Waltz 2002). Therefore, the CR perspective assumes that power-seeking is essential for states, while the NR perspective perceives power as a tool for self-protection.

These definitions of power establish the basis for NCR, which agrees with NR's evaluation regarding the correlation between the nature of the system and power. Both view

power relations between states as a determining factor in driving the foreign policy of those states, and argue that "the distribution of power is the only element of the international system that varies and that a country's position in the hierarchy is a major constraint on that country's behaviour" (Taliaferro 2006, 464). NR is, however, lacking an explanation of exactly how power can be understood and measured by states due to its absolute focus on anarchy and the distribution of state capabilities (Schweller 2006, 330).

In contrast, NCR emphasises how states evaluate and perceive that distribution of power. It argues that international anarchy is so ambiguous that states have difficulty determining whether security is abundant or scarce due to the absence of "immediate or perfect transmission belt linking to foreign policy behaviour". As such, states seek to control and shape their external sphere through a system of "filtering" (Rose 1998, 152). In contrast to NR, NCR further deems the system a tolerant structure rather than hostile, thus assuming power is primarily for influence not just security; ensuring its own existence is not the only goal for a state. Rather, states can use power to influence the international system since they aim to control their environment and anticipate other states' strategies due to the uncertainties produced by the anarchic-permissive international system (Carpes 2014, 33). In this regard, NCR assumes much of the tradition of CR, in that states respond to the uncertainties of international anarchy by seeking to control and shape their external environment, while noting that security is an element in defining state interests.

The nature of the international system is another common theme covered by the three realist accounts. NR defines the system as a whole and classifies political structures into three features: ordering principles (the decentralised structure of anarchy between states); the character of the units (the functions showed by differentiated states); and the distribution of capabilities (states in the anarchic system are separable in terms of their greater or lesser capacity towards similar tasks). In this regard, NR perceives the international system as an essential structure in which the characteristics of states, their behaviours, and their interactions with other states are unimportant (Waltz 2010, 79). The anarchic international political system is a more significant determinant of state behaviour since it forces states, particularly the great powers, either to follow identical policies or to risk elimination as independent entities to ensure their security (Jervis 1978, 169; Rathbun 2008, 301–4). States accordingly compete for security by perceiving other states as potential threats and try to maintain their status by increasing their power (Grieco 1990, 29). This indicates that the distribution of relative power

and material capabilities in the international system are influential in explaining international outcomes in which states are unitary actors. States track and mind their interests due to the unreliability of other states and zero-sum game security between them, either boosting their capabilities or balancing them through alliances (Waltz 2010, 118).

In this regard, Waltz (2010, 71–72)'s definition of international politics:

[international politics] can tell us what pressures are exerted and what possibilities are posed by systems of different structure, but it cannot tell us just how, and how effectively, the units of a system will respond to those pressures and possibilities. (...) We cannot predict how they will react to the pressures without knowledge of their internal dispositions. (...) An international-political theory does not imply or require a theory of foreign policy any more than a market theory implies or requires a theory of the firm.

Thus, it appears that a change in the international system will lead to a change in state behaviour. If states react differently and even irrationally to systemic requirements, they are most likely punished by the international system. Therefore, the pressures of international competition are more important than ideological preferences or the internal political constraints of states, and result in similarly situated states behaving alike (Lobell, Ripsman, and Taliaferro 2009, 20; Waltz 1986, 329).

NCR is aligned with NR in that the international system provides incentives to states on how to act, and yet it also states that the signals sent by the international system are not always clear or can be misunderstood even if they are clear. In essence, anarchy is what statesmen and women make of it, because "nations do not formulate and implement foreign policy and extract resources to those ends; governments do" (Zakaria 1999, 187). As such, NCR emphasises the role of domestic variables in analysing state responses towards the signals sent by the international system, perceiving the interactions between states as a significant part of the (Buzan, Jones, and Little 1993; Lobell, Ripsman, and Taliaferro 2016).

NR is thus criticised both for lacking an explanation of "why and how states choose among different types of 'internal' balancing strategies, such as emulation, innovation, or the continuation of existing strategies", and for ignorance of specific domestic events, such as the human motivations of individual leaders or foreign policies (Foulon 2015, 648; Taliaferro 2006, 66). In this regard, a state's domestic factors come to prominence, since "international pressures from the structure are indirect and translate downward through state-specific domestic intervening variables at state level" (Foulon 2015, 648). Put differently, intervening domestic variables affect a state's foreign policy manner by filtering systemic pressures; this is in contrast to the view within NR that states have absolute competency in extracting and mobilising resources from domestic society (Schweller 2006, 164; Taliaferro 2006, 478).

This significantly different viewpoint between NCR and NR is its inclusion of "domestic politics, internal extraction capacity and processes, state power and intentions, and leaders' perceptions of capabilities and relative power" in examining the foreign policy behaviours of states (Schweller 2003, 317). NCR assumes that the international system does not dictate foreign policy decisions, and instead confines them while the intervening variables of states play this role; thus, it "provides a fuller conception of the state by specifying how systemic imperatives will likely translate, through the medium of state power, into actual foreign and security policies" (Taliaferro 2006, 468). While NR focuses on international political outcomes such as the polarity, balance of power, and alliance behaviour in the international system, NCR is a foreign policy theory focusing on individual states and their policies by utilising multilevel analysis (Lobell, Ripsman, and Taliaferro 2009, 19). As such, NCR conceptualises the international system more efficiently than NR by sharing the same idea with CR in terms of recognising the importance of domestic factors in state behaviour (Rose 1998).

While NCR shares distinct similarities with CR, Rose (1998, 153) argues that CR is not a coherent research programme for developing a generalisable theory of foreign policy because it "covers a host of authors who differ greatly from one another in assumptions, objectives, and methodologies, and thus is not helpful for current purposes". In contrast to CR, NCR is viewed as an attempt at improving a generalisable and definite foreign policy with a specific methodology which posits the role of intervening variables that requires the most examination (Lobell, Ripsman, and Taliaferro 2009, 19).

Moreover, the two realist accounts differentiate the determinants of state behaviour even though they share mutual ideas regarding states' domestic factors. CR deals with "the sources and uses of national power in international politics and problems that leaders encounter in conducting foreign policy", and concentrates on "the relative distribution of capabilities between specific pairs of states or coalitions of states" since behaviour is shaped by power and the interests of states (Lobell, Ripsman, and Taliaferro 2009, 16; Schweller 2006, 331). NCR, however, argues that "a state's relative position in the international system determines its national behaviour, so nations expand their interests abroad when their relative power increases", thus aligning with the principal insight of NR, which is that the international system is the dominant factor in international politics (Lobell, Ripsman, and Taliaferro 2009, 19; Zakaria 1999, 482).

NCR also has a different view on power vis-à-vis CR, while both share similar assumptions on state-domestic society relations. As Barkin (2003, 336) posits, CR focuses on the role of moral ideals to understand international politics, and adds that the human motivations of individual state leaders influence the foreign policy behaviour of states. In contrast to CR, NCR argues that state authority has to consider or bargain with other domestic forces which exist at the unit level in order to extract or mobilise resources to support their chosen policies (Lobell, Ripsman, and Taliaferro 2016; Rathbun 2008; Rose 1998).

In summary, NCR specifically focuses on the foreign policies of states and argues that unit-level variables play a significant role in explaining state foreign policy, despite its realist premise being that the relative international power position of a state defines its foreign policy behaviour (Rose 1998, 150). In this way, it differs from its realist relatives despite having similarities with both CR and NR. Kitchen (2010a, 118) therefore argues that "[w]hat makes NCR 'new' is its ongoing attempt to systematise the wide and varied insights of classical realists within parsimonious theory, or to put it another way, to identify the appropriate intervening variables that can imbue realism's structural variant with a greater explanatory richness". Similarly, Lobell et al. (2009, 13) elucidate that "NCR builds upon the complex relationship between the state and society found in CR without sacrificing the central insight of NR about the constraints of the international system". NCR is, *per se*, a 'new' theory of foreign policy consisting of independent, dependent, and intervening variables by which to explain specific state behaviour. Table 2.2 thus demonstrates each theory's assumptions to illustrate their similarities and differences.

Theory	View of	View of the	Dependent	Causal Logic
	the IR	Units	Variable	
	System			
Classical	Somewhat	Differentiated	Foreign	Power
Realism	Important		Policies of	Distribution→Foreign
			States	Policy
Neorealism	Very	Undifferentiated	International	Relative Power
	Important		Political	Distributions→International
			Outcomes	Outcomes
Neoclassical	Important	Differentiated	Foreign	Relative Power
Realism			Policies of	Distributions→Internal
			States	Factors \rightarrow Foreign Policy

2.1.2.2 Neoclassical Realism and Innenpolitik Theories

Several criticisms exist of NCR due to its analysis of states' domestic factors. According to Legro and Moravcsik (1999), NCR is problematic since it supports the assumptions which realism traditionally rejects. They term this new realist approach as minimal realism and claim that secondary factors (state preferences, beliefs, and international institutions) are often treated as more important than power in minimal realism. The realist theory thus becomes indistinguishable from liberalism. Narizny (2017) similarly notes that NCR incorporates domestic factors, such as public opinion or regime stability, into realist interpretations focusing on the structure of the international system. He adds that realism can involve key domestic factors, while liberalism explicitly includes international factors and thus, he claims that NCR becomes incoherent, unnecessary, and prone to methodological errors. The critics of NCR, therefore, emphasise that "post hoc efforts [...] explain away the anomalies of neorealism, making use of whatever tools are necessary to plug the holes of a sinking ship" (Rathbun 2008, 295). As such, these scholars see NCR as theoretically weaker than other realist theories, and

similar to alternative international relations approaches such as liberalism and constructivism (Vasquez 1997). In this regard, this section will now underline the differences between *Innenpolitik* theories and NCR.

Innenpolitik, a German term, refers to a state's domestic policy, and thus examines the internal factors, including politics, economics, ideology, national character, partisan politics, or socio-economic structure. In other words, Innenpolitik theories focus on domestic determinants. For instance, liberalism highlights "individual", "peace", and "economics" (Dunne 2008) while constructivism highlights how perceptions remain bound by "socially constructed ideas" and "norms" (Barnett 2005). According to Innenpolitik approaches, the domestic dynamics of states shape their foreign policy behaviour, and so examining the choices and structures of the main domestic actors can help interpret a state's foreign policy outcomes (Rose 1998, 145). To illustrate, domestic peace theory, one such Innenpolitik approach, states that the foreign policy behaviours of democratic states are different from non-democratic ones (Doyle 1986). Liberalism further focuses on state-society relations in explaining state preferences and state behaviour in world politics (Moravcsik 1997). This suggests that Innenpolitik theories view individual and national level factors as the determinant variables shaping a state's foreign policy behaviour (Waltz 2001). For example, as agent-oriented, foreign policy analysis (FPA) highlights the role of human factors in the process of foreign policy decision-making (V. M. Hudson 2005). Therefore, Innenpolitik theories propose that states' internal variables are sufficient to examine their foreign policy behaviours, and leave little room for the structural impact of the international system (Hadfield-Amkhan 2010, 29). Put differently, such theories highlight internal aspects more than systemic ones.

To be sure, NCR is in alignment with *Innenpolitik* theories in arguing that domestic variables shape a state's foreign policy behaviour. Domestic factors are especially important "in the process of self-help inherent to an anarchic system", therefore, they are "fair game for realism and neoclassical realists have taken up this mantle" (Rathbun 2008, 301). As such, decisionmakers' perceptions, ideas, and desires have an influential role in foreign policy outcomes, but their policies are not completely free from other domestic factors. In contrast to FPA, for example, NCR asserts that foreign policy makers can be influenced by other internal factors as well as external pressures. Moreover, state leaders define "national interests" and direct foreign policy based on other countries' intentions and relative power shifts, but they can be inhibited by other domestic constraints (Taliaferro, Lobell, and Ripsman 2009).

Another example is the view of constructivism regarding anarchy. According to Wendt (1992), states' perceptions of other states determine the definition of anarchy. A state's perception of the other state(s) as a friend or an enemy influences the international system, either through conflict or peace. Therefore, the supremacy of structure is underrated, which is in opposition to the realist approach. According to NCR, structure is the primary constraint on states' behaviour, but there are several ways in which ideas can assist states in making decisions. It establishes a clear hierarchical relationship between main factors (anarchic system and domestic factors) influencing a state's foreign behaviour and deems state behaviour a topdown manner in contrast to Innenpolitik theories (Coetzee and Hudson 2012). As a result, in the shorter term, especially, in the event of unambiguous international pressures, Innenpolitik theories may be more explanatory than realist ones since they analyse a state only on domestic variables involving domestic political coalitions or leader preferences. They may also be more useful in explaining the behaviour of non-state actors, be they corporations, societal interest groups, national non-governmental organisations, or international non-governmental organisations, since realist theories are state-centric. This is because NCR "could help explain the conditions under which non-state actors are more likely to influence the foreign policy behaviour of states when they attempt to do so" (Lobell, Ripsman, and Taliaferro 2016, 96-97).

Nevertheless, NCR notes that states and the international system are dependent on one another since the latter is composed of states motivated by different aims (Rose 1998, 148). Thus, the influence of domestic factors is analysed when a state's behaviour deviates from systemic structures (Rathbun 2008, 295–96), because when domestic political factors have more overwhelming influence in the decision-making process than international variables, there will be consequences such as punishment and condemnation by the international system in the long-term (Rathbun 2008, 317). Moreover, states with similar domestic variables may follow different foreign policies, but states with different domestic variables may conduct a similar foreign policy approach (Rose 1998, 148). However, *Innenpolitik* approaches are unable to explain why states or governments with similar functions behave in different ways (Zakaria, 1998). As such, explanatory power of *Innenpolitik* theories is limited in contrast to NCR since involving both systemic and domestic variables provides a richer and more complete understanding of international politics and the details of states' foreign policy behaviours (Turpin 2019, 5–6). In Wivel's (2005, 374) words:

Realist foreign policy analysis is not an either/or choice between rigorous theoretical parsimony and empirically rich case studies. Instead, realist foreign policy analysis should ideally proceed in stages from the parsimonious and highly general starting point in structural realism to the rich case studies allowing us to explain the specific foreign policy of individual countries.

Therefore, NCR finds a middle way between systemic theories and *Innenpolitik* approaches to increase its explanatory power (Rose 1998, 152). While *Innenpolitik* theories, such as constructivism and liberalism, claim that foreign policy is the product of domestic factors of states, NCR perceives domestic factors as intervening. Notably, in the longer term, NCR "has greater explanatory power than structural realism or any of its *Innenpolitik* competitors" (Lobell, Ripsman, and Taliaferro 2016, 98). Onea (2021, 277) summarises the importance of NCR, purporting that "if NCR adds value compared to its competitors, this comes from its willingness to accept that one gains more insight by combining diverse variables rather than considering them separately and exclusively".

Given the flaws and strengths of the approaches discussed above, we can see that an ideal solution would be an approach that consistently seeks a balance between, on the one hand, taking account of both the nature of the structure of the international and regional system and on the other the ideational and unit-level factors that motivate and influence foreign policy choices at the domestic and individual level. That is, NCR combines material and non-material explanations, thereby widening its explanatory scope.

2.2 Concluding Remarks

Neoclassical Realism (NCR) is one of a number of the theories that look to explain state behaviour in the international system. Its most significant feature is to analyse state foreign policy using both systemic and domestic drivers. NCR contends that focusing solely on systemic or domestic issues is insufficient to understand a country's foreign policy. As a result, it varies from both systemic and Innenpolitik theories in that it provides a balanced perspective of foreign and domestic actors.

While acknowledging the criticisms and limitations identified above, the thesis nonetheless uses NCR to analyse Turkish Foreign Policy (TFP) towards Iran since the Arab Uprisings, from December 2010 to July 2018. This is precisely because NCR incorporates the

assumptions of *Innenpolitik*, NR, and CR, and it can be argued that, as individual approaches, they have low explanatory value with regard to a state's foreign policy. Using domestic factors as an independent variable, which *Innenpolitik* approaches propose, does not explain TFP towards Iran alone because Turkey's relative power is a significant determinant in conducting foreign policy towards Iran. Equally, system-level approaches cannot accurately explain TFP towards Iran even though NCR uses it as an independent variable. This is because the AKP elites' perceptions and Turkey's domestic considerations have played a significant role in shaping TFP, through which the elites have mediated between the imperatives of the international system and TFP towards Iran. The thesis thus uses NCR since it offers an empirically detailed and theoretically informed analyses of TFP towards Iran since the Arab Uprisings. To date, no other study of Turkish Iranian relations has used NCR to fully understand and critique how this relationship has reshaped much of the contemporary Middle East. The next section explores why NCR is better for analysing TFP by evaluating significant studies on TFP towards Iran since the Arab Uprisings.

Given the theoretical background of the thesis, the next chapter examines Turkey's historical relations with Iran from 1979 to 2010 to provide a clearer insight into TFP towards Iran. In so doing, the thesis aims to demonstrate how systemic and bilateral determinants historically shaped TFP towards Iran before the Arab Uprisings term.

CHAPTER 3 TURKEY'S RELATIONS WITH IRAN FROM FEBRUARY 1979 TO DECEMBER 2010: AN OVERVIEW

This chapter examines Turkey's historical ties with Iran up to the start Arab Uprisings. Instead of a broad chronological examination, the chapter focuses on which factors determined TFP towards Iran before the uprisings. As such, bilateral determinants and systemic determinants of Turkish historical ties with Iran are analysed. Since the thesis explains recent TFP towards Iran, this chapter's primary purpose is to present Turkish affairs with Iran after the Islamic Revolution. It thus focuses on the period beginning with the Islamic Revolution in Iran in February 1979 through to December 2010.

3.1 Bilateral Determinants

Turkey has shared with Iran a 560 km long border since the 1639 Treaty of Qasr-i Shirin was signed by the Safavid and Ottoman Empires; it has also shared mutual geopolitics with Iran by virtue of being situated near significant and volatile regions, including the Middle East and Central Asian and Caucasus. As such, cultural, political, social, and economic aspects are such common ties that "in many ways, Turkey and Iran are mirror images of each other" (Barkey 2010, 163; Wang 2011, 2). These shared features naturally bring close ties between the two. Nevertheless, these similarities have at times led to competition rather than collaboration. Due to their historical heritage and similar power position, and different regime type and identities, both countries compete for power in the Middle East. As a result, Turkey's perception of cooperation and collaboration towards Iran have an influential role in determining not only how its shapes its foreign policy towards Iran, but also how it chooses its policy behaviours towards regional and international politics. In the following section, the bilateral determinants between Turkey and Iran are explained to understand TFP towards Iran.

3.1.1 Ideological and Sectarian Division

Ideology and religious sects were undeniably key in the bilateral relations between Turkey and Iran. At the beginning of the sixteenth century, the Sunni Ottoman and Shia Safavid empires each promoted themselves as major sects of Islam and thus entered into sectarian competition. Following World War I (1914-1918), the newly founded Turkish Republic also endorsed western principles by defining itself as a western state and seeking to develop standards in alignment with this, while Iran established a monarchy under the Reza Shah Pahlavi rule (Keskin 2009, 46; N. A. Özcan and Özdamar 2010, 103). Prior to 1979, Turkey was thus at odds with Iran for ideological and sectarian reasons, despite their periodic collaboration.

Turkey's concern about Iranian ideological policies was reinforced when Iran, following the declaration of the 1979 Islamic revolution, embraced a theocratic Shia-based regime and cut its ties with the US and its allies. Given its secular government and status as a US ally, Turkey was concerned about the possibility of the Iranian revolutionary movement being exported to the entire Middle East (Pahlavan 1996, 73). In contrast to Iran, therefore, Turkey promoted itself as a "model of secularization and modernization in the Muslim world" (Wang 2011, 4–5). In particular, the Turkish military adopted an obdurate stance towards Iran owing to Iran's Islamic identity (Fuller 2007, 111). In the light of this, Huseyin Kivrikoglu, the former Chief of the Turkish General Staff (1998–2002), claimed that "Iran's intentions have never changed. From 1639 there has not been a war between us, but Iran never wanted a strong Turkey. It seems clear that Iran's intention is to show Turkey as an aggressive country" (Olson 2000, 877–78). Accordingly, Turkey was more concerned about the ideological differences between the two countries following the 1979 Islamic revolution, which brought an ideological conflict.

Their ideological conflict intensified with the rise of Turkic nationalism in Central Asian and Caucasus following the collapse of the Soviet Union in 1991. While Turkey, with its secular and Turkish identity, aspired to establish closer ties with these countries, Iran sought to expand political Islam and support radical Islamic groups in its proximity. Therefore, Turkey feared becoming a target for these radical Islamic groups. It was claimed that Iranian intelligence cooperated with the jihadist groups in the assassinations of Turkish secularists, including the journalists Cetin Emec and Ugur Mumcu and the academics Muammer Aksoy and Bahriye Ucok (Ayman 2014, 9). Turkey's then Minister of the Interior even claimed that Iranian security forces assisted radical jihadist groups with their military and ideological training to conduct terrorist attacks against Turkish citizens and Iranian opposition in Turkey (Barrans 2016, 31). Various Turkish newspapers also alleged that religious organisations in Turkey resulted from movements in Iran (Pahlavan 1996, 73). As a result, ideological competition in Central Asia and the Caucasus placed Ankara and Tehran in opposition, with

Ankara continuing to blame Tehran for supporting terrorist activities in Turkey during the early years of the 1990s.

Nevertheless, in June 1996, the Turkish government changed and the Islamist-based Welfare Party (in Turkish, Refah Partisi) formed a coalition government with the centre-right True Path Party (in Turkish, Dogruyol Partisi). With the election of the Welfare Party, Turkish foreign policy started to focus more on the Middle East due to the Islamic identity of the Welfare Party. The leader of the Welfare Party, Necmettin Erbakan, espoused ideological views which paved the way for cordial bilateral ties with Iran. In 1997, Turkey and Iran established the Developing-8 (D-8) Organisation for Economic Cooperation, involving eight developing Muslim countries: Turkey, Iran, Bangladesh, Egypt, Indonesia, Malaysia, Nigeria, and Pakistan. This policy shift from a western- to a Middle Eastern-based foreign policy caused debate among Turkish secularists and military, who claimed that the Welfare Party had Iranian ties (Larrabee and Lesser 2003, 147). On 28th of February 1997, after just one year, the Turkish army issued a military memorandum which forced the coalition government to resign by claiming that the Welfare Party was following an Islamic plan in opposition to secularism, thus interrupting Erbakan's quest for close ties with Tehran. This military memorandum was then called as a post-modern coup against the Turkish government.

In 1999, the Merve Kavakci affair, related to the headscarf issue in Turkey, further broke down the bilateral affairs between Turkey and Iran. Merve Kavakci, who was elected deputy of the Welfare Party, was dismissed from the Turkish Parliament for wearing a türban (headscarf) as deputy in the swearing-in ceremony. In this regard, many Iranian students, who may have been backed by Iranian officials, conducted demonstrations to support Turkish women wearing a headscarf and other Islamic clothing. Turkey perceived this as an intervention in its domestic issues and blamed Iran for attempting to export revolution (Keskin 2009, 49; Olson 2000, 875–76). As a result, it seems that ideological difference was a strong determinant in TFP towards Iran during the 1990s. Even though some Turkish politicians, such as Necmettin Erbakan, aspired to follow closer ties with Tehran, both the secularists and the Turkish army strongly opposed Iran's Islamic Shia identity and opted for restrictions on Turkey's affairs with Iran. Since these groups played a more dominant role in TFP during the 1990s than those who sought closer ties with Iran, Turkey's foreign policy towards Iran exhibited mostly ambivalent and limited characteristics.

Having come to power in 2002, the AKP, with its Islamic-based identity, redefined TFP towards the Middle East in general and Iran in particular by promoting the Muslim and Ottoman heritage of Turkey (Dismorr 2008, 215; Ehteshami and Elik 2011, 646). Accordingly, the Islamic-dominated government's focus on the Middle East brought warmer ties between Turkey and Iran under the principles of "zero-problems-with-neighbours" and "economic interdependence" (Fuller 2007, 109; McCurdy 2008, 88). Turkey under the AKP government opted for close ties with Tehran to a certain extent, especially following the Ergenekon trials. These trials started in 2008 and prompted the dismissal of suspected members of an alleged criminal network, including secularists, ultra-nationalists, and the Turkish military, all of whom allegedly have sought to oust the AKP. As a result of these trials, both the political pressure of these groups over the AKP government terminated and the AKP consolidated its legitimacy with consecutive electoral successes. Therefore, the AKP government started to establish closer ties with Iran than before. During its tenure as a non-permanent council member of the UN in 2010, for instance, AKP-led Turkey and Brazil sought to mediate between the UN and Iran and signed a joint declaration with Tehran on Iran's nuclear programme. In this regard, it seems that the less secular and nationalist Turkish political ideology became, the more Turkish-Iranian affairs warmed, until the Arab Uprisings.

However, the Arab Uprisings ushered in a period of ideological and sectarian competition between Turkey and Iran. The AKP started to follow Sunni sectarian policies towards the Middle East, and its Sunni-based policies resulted in a challenge to Shia-based Iran. Despite this rivalry, TFP towards Iran under the AKP governments demonstrated interest-based approaches, examined in the following chapters.

3.1.2 Minority Issues

Due to their multi-ethnicity, Turkey and Iran faced minority issues for many years before the uprisings. While dealing with this internal issue, they sometimes experienced bilateral disagreement due to this problem. Before the Iranian Islamic revolution, for instance, Iran had been concerned about Pan-Turkism movements of the Young Turks —which had aimed for the cultural and political unification of all Turkic peoples —in the early 20th century due to its the Azerbaijani population (Wang 2011, 3). On the other hand, following the revolution, the Azeris of Iran were prone to be less mindful of their 'Turkishness' by integrating into Iran both culturally and economically (Fuller 2007, 110). Therefore, the disagreement over the Iranian

Azeris had little impact on their bilateral affairs. Following to Islamic Revolution in 1979, however, their main shared minority problem became Kurdish groups. Since the revolution, the Kurdish issue has remained a mutual and ongoing minority concern for both countries due to their significant respective Kurdish populations. Since 1984, Turkey's Kurdish issue has pertained to the Kurdistan Workers Party (PKK), which has been conducting an armed struggle against the Turkish government. For this reason, Turkey has deemed the PKK to be a potential danger to the country's territorial integrity.

Prior to the Arab Uprisings, Turkey's Kurdish issue shaped TFP towards Iran because Turkey occasionally accused Iran of provoking its Kurdish minorities against Turkey. In particular, Turkey slammed Iran for its apparently close ties with the PKK. For instance, in 1991, Ankara arrested an Iranian-flagged ship on suspicion of transporting weapons to the PKK (Fuller 2007, 110). According to Wang (2011, 3), "for years, the Iranian government has had acquiescence or even support for the PKK militants launching attacks against Turkey from its territory, which became an important focus of conflict between the two countries". Similarly, Imset (1992, 168) points out that Osman Ocalan, brother of the PKK leader Abdullah Ocalan, had a personal relationship with the Iranian Revolutionary Guards in the early 1990s, and the PKK was allowed to use Iranian territory to carry out attacks against Turkey. The main reasons behind Iran's arguable support for the PKK were "the regime differences between Iran and Turkey, presence of Iranian asylum seekers in Turkey, Turkish policy towards Azerbaijan, even its policy towards Iraq, and its membership in the NATO" (Ozcan 2000, 52). More precisely, Iran allegedly endorsed the PKK to prevent Turkish influence in international politics, while Ankara considered that Iran's policy was an intervention in its domestic politics. Ankara was thus concerned about any possible close ties between the PKK and Tehran. Ankara's accusations, on the other hand, were expressed verbally as public declarations rather than pushing for more serious diplomatic action, such as the recall of diplomats or the cancellation of contracts.

In essence, Iran dealt with its Kurdish minorities and suffered from the attacks of the PKK-linked Kurdistan Free Life Party (PJAK) in its territory. Therefore, Iran sought to establish security ties with Turkey due to their mutual concern; for instance, both opposed the creation of a semi-autonomous state in northern Iraq, assuming that it would motivate their Kurdish minorities to press for greater independence. Turkey was also concerned about the control of the oil and gas fields of Northern Iraq by the Kurds (Olson 2000, 165). As such,

Turkey aimed for collaboration with Iran. They further collaborated to prevent the PKK and PJAK launching terrorist attacks from their territories, and Iran suspended its ties with the PKK following Ocalan's expulsion from Syria and his arrest by Turkey in 1999 (Yesilyurt and Akdevelioglu 2009, 45). In a joint security meeting in October 2001, having halted its support for the PKK, Iran gave assurances that it would not let the PKK use its territory against Turkey. In return, Turkey agreed to end political support for the South Azerbaijan National Liberation Movement in Azerbaijan (Wang 2011, 13). They also signed a Memorandum of Understanding on security cooperation against PKK/PJAK terrorism, respectively, on July 24, 2004, and in April 2008, to promote this collaboration (Jenkins 2012, 33; Karacasulu and Aşkar Karakir 2014, 115).

Additionally, both countries entered active bilateral cooperation in the fight against terrorism by organising mutual attacks on these groups to protect their borders. In doing so, Turkey aimed to improve its relationship with Iran, adopting a "maximum cooperation" approach to relations with Iran regarding security. Turkey also sought to restrain the possible Iranian hostile policies towards Turkey when Iran strengthened its influence in the region after the 2003 Iraq War (Yetim and Kalayci 2011, 92). Their mutual Kurdish issue, therefore, paved the way for both cooperation and accusation between them. As long as they aimed mutual benefit, they opted to cooperate against any Kurdish rebellion movement. Otherwise, especially Ankara accused Iran of supporting the PKK in its violent operations against Turkey.

The Kurdish minority group was not only conflict area for TFP towards Iran, however, as the Armenian ethnic minorities in Turkey also emerged as another point of conflict. From the 1980s to the early 1990s, Turkey accused Iran of supporting the Armenian Secret Army for the Liberation of Armenia (ASALA), which wanted to force Turkey to publicly acknowledge its responsibility for the Armenian Genocide in 1915, pay reparations and abandon the Eastern Anatolia region for an Armenian homeland. Apart from the Armenian and Kurdish minorities, Turkey was also very concerned about Iran's influence on its Shiite, Alevi, and Qizilibash populations (Wang 2011, 3), but these minority groups had a limited role in TFP towards Iran in contrast to the Kurdish minorities. In essence, the Kurdish issue demonstrated uneasy affairs between Turkey and Iran.

3.1.3 Economic Factors

Economic factors have always influenced on Turkish-Iranian relations. Their quest for close economic ties has brought various economic alliances either on bilateral or multinational levels, such as the Saadabad Pact of 1937 and the Economic Cooperation Organisation (ECO) in 1985 (Elik 2012, 161). They also played leading roles in the establishment of the D-8 in 1997 for economic collaboration. Economic interests thus played an influential role in TFP towards Iran before the Arab Uprisings. Even in periods of conflict, Turkey opted to boost economic ties with Iran (Sinkaya 2004, 110).

Given its pragmatic economic ties with Tehran, Ankara refused to impose an embargo against Tehran during the 1980s when the US froze Iranian assets and imposed a trade embargo due to the hostage crisis, in which the Americans in the US Embassy in Tehran were held hostage by the Iranian protesters after the declaration of the Islamic Revolution. In return, Turkish-Iranian economic ties improved significantly in the first half of the 1980s. For instance, Turkish exports to Iran increased from USD 45 million in 1978 to USD 1.088 billion in 1983, while Turkish imports from Iran increased collaterally from USD 189 million to USD 1.548 billion in 1984 (Elik 2012, 163). Following the collapse of the Soviet Union in 1991, Ankara maintained economic ties with Iran despite its competition with Tehran over the newly independent countries in Central Asian and Caucasus (M. Aydin and Aras 2005, 31). Both countries signed two essential agreements promising that Iran would supply gas and a gas pipeline construction project to Turkey, and both would increase their bilateral trade in merchandise to USD 2.5 billion a year (Kibaroglu 2009).

Sinkaya (2004, 111) added that the degree of economic ties between Ankara and Tehran depended on their bilateral political relations. For instance, their economic affairs declined during the 1990s when Turkey blamed Iran for supporting terrorist activities in Turkish territory, and so their trade volume fell below that of the 1980s (Inat 2015). In contrast to their weak economic ties in the 1990s, Ankara sought to strengthen its economic relations with Tehran during the 2000s, based on its warming and political relations with Tehran (Kunu and Hopoglu 2016, 112). With the principles of "zero-problems-with-neighbours" and "economic interdependency", Ankara aspired to fortify economic ties with Tehran. In this regard, Turkish imports and exports to Iran in 2002 were USD 921 million and USD 334 million, respectively. They also grew to USD 11,965 million and USD 9,923 million in 2012 (Sariaslan 2013, 75–

76). Therefore, closer political ties culminated in closer economic ties between Ankara and Tehran while economic gains always hold its priority in TFP towards Iran. More precisely, Turkey's quest for pragmatic and interest-based political ties with Tehran brought economic collaboration between the two countries.

Despite their regional political competition, the Arab Uprisings demonstrated that Ankara preferred economic cooperation with Iran. Turkey now looked for closer economic ties with Iran, a result of its increasing regional isolation and a political confrontation with erstwhile Western allies. In this thesis, therefore, Turkey's quest for economic ties with Iran is used as the context in which to examine TFP towards Iran. Chapter 5 contains an analysis of Turkey's interest-based economic policies, to explain why Turkey opted for a rapprochement and alliance with Tehran, despite their competitiveness.

3.1.4 The Iranian Nuclear Programme

The Iranian nuclear programme became another bilateral determinant in TFP towards Iran. Since the beginning of the Islamic Revolution in 1979, Iran has occasionally been accused of advancing its nuclear programme in order to produce a nuclear bomb in the future. In particular, the US, despite its initial assistance to the Shah's regime, has always been concerned about Iran's nuclear programme and assumed that Iran had a nuclear capability, as it did with Saddam Hussein's Iraq. Even though the US President Barak Obama aimed at a diplomatic solution through an "engagement policy" leading to the signing of the Joint Comprehensive Plan of Action-JCPOA (and in contrast to the administration of George W. Bush), the United Nations Security Council (UNSC) approved and adopted numerous sanctions against Iran in May 2010 (N. A. Özcan and Özdamar 2010, 114).

Despite global disapproval of the Iranian nuclear programme, Turkey highlighted Iran's legitimate right to develop peaceful nuclear technology as a member of the nuclear non-proliferation treaty (NPT) signed on 1st July 1968 to limit the spread of nuclear weapons. As such, Turkey voted against UNSC Resolution 1929 relating to Iranian sanctions on June 9, 2010, as a non-permanent member of the UNSC. Ankara believed that western countries took sides while ignoring Israel's possession of nuclear weapons (Pieper 2013, 85). Mercan (2010, 18–19), then Chairman of the Foreign affairs Committee of the Turkish Grand National Assembly, defined Turkey's policy towards Iran's nuclear programme as:

We advocate diplomatic and economic engagement of Iran rather than isolationist policies as a more effective way to address the challenges that we are facing in the region. We will continue to encourage all our counterparts to take a conciliatory approach in order to better tackle the problems in the Middle East. [...] Any interference from the outside world will have a boomerang effect and will be counterproductive. Therefore, the international community should refrain from any attempt to interfere in Iran to the detriment of the social and political fibre of Iran.

Given its support for Iran's nuclear programme, Turkey offered to mediate between Iran and the West and hosted various P5+1 (US, Russia, China, France, the UK and Germany) negotiations with Iran at the political directors-level in Istanbul. Additionally, Turkey, along with Brazil, signed the Tehran Declaration in May 2010, to propose a diplomatic initiative. According to this agreement, Iran was required to send 1,200 kg of low-enriched uranium to Turkey for safekeeping in exchange for receiving 120 kg of fuel rods for the Tehran Research Reactor, which produces isotopes for medical use. In this context, Pieper (2013, 86) pointed out that:

Turkey's motivation for the 2010 Tehran declaration arguably was a mixture of its insistence on a resolution of the nuclear conflict through dialogue and engagement, geostrategic manoeuvring in line with a newly discovered regional policy and a pragmatic attempt to engage Iran in order to reduce tensions. The latter reading is thus very much in line with the perception of Turkey's role in the Iranian nuclear dossier as being that of a 'facilitator'. A further motivation and one that is underlying Turkey's Iran policy in general was the need not to disturb regional trade relations through heightened tensions surrounding the Iran case.

Turkey's other motivations in the 2010 Tehran declaration were regarding regional security and its economy. Turkey assumed that the existence of nuclear weapons in Iran would cause a nuclear arms race in the Middle East by prompting other countries to follow suit. Ankara thus argued that a race regarding nuclear weapons would reduce regional security and cause significant economic loss. Moreover, with the successful declaration, Iran would terminate its backing of non-state armed groups in the Middle East and be integrated into the international structure. As a result, Iran would not be a geopolitical threat for Turkish economic and political interests (Ozdamar 2015, 2–4). More precisely, Turkey prioritised its future possible gains and followed interest-based policies towards the Iranian nuclear programme.

Nevertheless, Turkey's attempt regarding the declaration failed due to the P5+1's resistance. Since Ankara did not aim uncouple from the West over the Iranian nuclear programme, it adopted a pragmatic policy which would not come at the cost of its Western orientation (McCurdy 2008, 89). As such, Kibaroglu (2009) argues:

Bearing in mind the rivalry between the Turks and the Iranians throughout history, despite the fact that some common concerns exist as regards their national interests, the scope and the content of Turkish-Iranian relations may not go far beyond the present levels unless Turkey makes a radical turn in its relations with the West in general, and with the United States in particular, even if they may not be at satisfactory levels either.

Therefore, Ankara was forced to consider its ties with its Western allies for long-term interests. Turkish policymakers then realised that Iran would increase its regional significance and influence by reintegrating into the world system. Despite the economic and security advantages for Turkey in an optimistic scenario, the rise of Iranian influence in the international system would threaten Turkey's relative power position and policies in the region in the long term. In this regard, the possibility of Iran's close ties with the Western powers pushed Ankara to revise its policies towards the Iranian nuclear programme (Ozdamar 2015, 4). In essence, the Arab Uprisings emerged after nearly seven months from the 2010 Tehran declaration. Ankara entered into a competition with Tehran to influence the future of the uprisings, as explained in Chapter 4 of the thesis. Ultimately, Turkey made cost-benefit calculations in its policies towards Iran's nuclear programme; despite its initial support for the programme, the Arab Uprisings forced Ankara to reconsider the regional balance of power.

3.2 Systemic Determinants

Since Turkey and Iran are two middle powers in international politics, the systemic and subsystemic level changes determined their bilateral affairs. During the Cold War, the pressure of the communist threat of the Soviet Union resulted in a close relationship between Turkey and Iran, which sought alliances with the US and Western Europe to counter this threat. Along with Iran, Turkey followed a pro-Western foreign policy. The US also encouraged Turkey to establish bilateral or multilateral cooperation with Iran to counter the influence of the Soviet Union in the Middle East. Under the aegis of the US, Turkey entered a regional alliance with Iran in the Bagdad Pact (later called the Central Treaty Organisation) in 1955 and Regional Cooperation for Development (later called the Economic Cooperation Organisation) in 1964. Turkey mostly viewed Iran "as a friendly and status quo power within the Western camp, not willing to jeopardise Turkey's position in the region or its security" (Kibaroglu and Nun 2008, 158–59) despite its occasional concern about the closeness between the Soviet Union and Iran's leftist groups (Kumral 2020, 120). Hence, regional great power politics mostly drove Turkey's rapprochement policies towards Iran during this period.

Following the 1979 Islamic Revolution, Iran's political structure altered radically, and foreign policy was now shaped by a new radical interpretation of Shia Islam, most visibly seen in the breaking of diplomatic ties with the US and the taking hostage of 400 American diplomats. Iran sought to export its revolution to neighbouring states, including Iraq with its own Shiite majority and across the wider Middle East, most notably Lebanon (Sayigh and Shlaim 1997), causing tensions to re-emerge between Ankara and Tehran. Ankara was clearly concerned about Iranian willingness to export its revolution, as well the Iranian separation from the Western alliance (Kibaroglu and Nun 2008, 159). Despite its concerns regarding the Islamic regime, however, Ankara almost instantly recognised the new Iranian regime. From this perspective Kibaroglu and Nur (2008, 152) proposed that:

The Islamic Republic of Iran's militant Islamist statements and foreign policy fuelled tension and mutual distrust. However, both countries sought to prevent conflict or a rupture in relations. This reluctance to escalate the tensions stemmed largely from their desire to protect their economic interests, given that Turkey was an exporter of goods to Iran and Iran was a major energy supplier for Turkey.

There were further reasons behind Turkey's recognition of the new Iranian regime: Turkey had already criticised the Shah and CENTO alliance; the country was concerned about a possible civil war in Iran on the grounds of the potential risk of a Kurdish separatist movement and Soviet control in Iran; Turkey believed that Iranian regional prestige and capabilities would diminish in favour of Turkey; and Ankara aimed to establish a trade relationship with a weakened and isolated Iran for its bankrupt economy (Cetinsaya 2002). Moreover, Turkey aimed to prevent Iran from establishing close relationships with the Soviet Union (Turel 1999, 86). For this reason, Turkey approached the Iranian Islamic Revolution on the grounds of the principles of self-determination and non-intervention in the internal affairs of a state (Bleda 2000, 107). In this regard, Turkey also warned the US and the Soviet Union against interfering in Iran's internal affairs and refused a US demand for the use of Incirlik airbases in southeast Turkey against Iran following the US embassy takeover and seizure of diplomats as hostages in Tehran (Ayman 2014, 10). As a result, Turkey's interests and cost-benefit calculations culminated in a detente between Ankara and Tehran. Alongside the declaration of the Iranian Islamic Revolution, other regional factors also shaped Turkey's foreign policy towards Iran until the Arab Uprisings.

3.2.1 The Iran-Iraq War between 1980-1988

The outbreak of the Iran-Iraq War in September 1980 influenced Turkey's policies towards Iran at the regional level. With the Islamic revolution in Iran, the countries neighbouring Iran became concerned about the former exporting revolutionary ideas across the region. With its secular and socialist ideology, the Iraqi Ba'ath regime notably increased its criticism of the Islamic revolution following the Al-Sadr Uprising in Iraq, which had been allegedly supported by the Iranian Islamic regime (Aziz 1993, 207). Iran further aspired to establish sectarian links with Iraqi Shi'ites, support Kurdish groups against the Iraqi regime, and resolve the dispute with Iraq over the Shatt-ul Arab waterway (U. Aydin 1994, 89). In this way, the war seemed to be an indispensable gain for the Iranian regime, only a year after the Islamic Revolution.

During the war, Turkey adopted a neutral foreign policy towards Iran and Iraq, a policy dictated by its own internal politics. Following the putsch of 12 September 1980, the Turkish military took over the political system. Turkish policymakers opted to concentrate domestic policy even though they were concerned about Iran's support of Iraqi Kurdish groups. Moreover, Turkey had to deal with an economic crisis as many Western states, including many in Europe opposed the new military regime and reduced bilateral trade ties with Turkey. Ankara looked to liberalise its economy during this economic difficulty through greater privatisation and moves towards a more export-oriented economy. Therefore, Turkey deemed the Iran-Iraq war an opportunity to help its ailing economy and enhance economic ties with Iran and Iraq, its two leading energy suppliers (Olson 1998, 29). A neutral foreign policy towards the war emerged as a more appropriate choice for Turkish cost-benefit calculations, in that intervening in the war would have been more costly for Turkey's fragile economy since it would have lost its significant economic partners.

Given Turkey's aims and policies, Iran appreciated Turkish neutrality, since it hoped to consolidate the legitimacy of its regime and gain an edge over Iraq. Iran also believed that Turkey's neutral policy would diminish Iranian regional and international isolation (Olson
1998, 30). Accordingly, Iran too followed pragmatic policies and refrained from using its Islamist ideology against Turkey. It also hoped to increase economic ties with Turkey since it needed Turkish ports in the Black and Mediterranean seas to import key materials. Turkey therefore had strategic and economic leverage over Iran and remained the only viable trade and transport route during the war (Gundogan 2003, 4). Due to their mutual interest, Turkey and Iran signed barter agreements for the former to provide manufactured products and foodstuffs in exchange for oil and gas. By 1983, Turkey had become Iran's largest trading partner (Jenkins 2012, 17). Interest-driven aims thus paved the way for close Turkish-Iranian ties during the war.

In the same period, Iraq too viewed Turkey as a viable route since "its economic lifeline went through Turkey with the Kirkuk-Yumurtalik pipeline to the Mediterranean as the only secure route for its oil exports" (Hentov 2013, 125). Turkey thus benefited from an oil pipeline from Iraq, allowing Baghdad to lay a significant pipeline on its territory and providing an alternative outlet to its oil at a time when tankers were being increasingly targeted by Iran in the Gulf. As a result, Turkey's neutrality provided economic gain by obtaining valuable energy supplies and foreign income, supporting Turkish consumption, and generating transit revenues through the export of oil (Hentov 2013, 125). With its neutral and pragmatist policy, Ankara achieved economic gains from the dispute between Baghdad and Tehran.

As well as economic gains, Turkey obtained political gains. During the war, it maintained the unique position of holding both credible impartiality and handling the complexities of the war. Turkey became a crucial player in the mediation efforts of the Organisation of the Islamic Conference (OIC) (Hentov 2013, 125). The Iran-Iraq war allowed Turkey to obtain regional economic and political supremacy, and Turkish policies towards Iran were dependent on economic and political interests. Given Turkey's gains, maintaining the regional power balance became Turkey's initial aim. Turkish policymakers believed that, if either Iran and Iraq saw their regional power diminished, the Soviet Union would seek to benefit from the power vacuum and a long-term war could emerge in the Middle East. Therefore, Turkey hoped the war would end quickly to prevent any Soviet threats from its north (U. Aydin 1994, 89; Gundogan 2003, 2).

Nevertheless, tensions continued to mark relations between Turkey and Iran as Ankara remained concerned by the long-term regional ambitions of Tehran. One Turkish worry concerned impaired safety for maritime passage in the Gulf, and the security of the transportation of oil over land through Turkish territory. Great concern was expressed about the fate of "Turkish tankers plying the Gulf routes and noted with concern the Iranian's public threat to hit the Kirkuk-Dortyol pipeline, which they declared to be a legitimate strategic target" (Liel 2001, 178). Turkish diplomatic relations with Iran also became tense due to the number of Iranian refugees who had been settled in camps in eastern Turkey, and some Iranian regime opposition groups had launched attacks from Turkish territory across the border. By 1985, Turkey was hosting one million Iranian refugees (Liel 2001, 172).

In 1983, Turkey felt itself provoked when the war spread to Iraq's northern Kurdish region. This impacted on Turkish-Iraqi trade ties, which deteriorated. The oil flowing through the Kirkuk-Iskenderun pipeline was under direct threat from the war because the pipeline provided Turkey with significant energy revenue, almost 40 per cent of oil volume pumped (Hentov 2013, 32–33). Equally, Turkey became increasingly concerned that any Kurdish success in one region would promote Kurdish rebellion in other , and so Ankara opposed Iran's support for a Kurdish insurgency against Iraq believing this would radicalise its own Kurdish population (Liel 2001, 173). Ankara thought that the Patriotic Union of Kurdistan (PUK) and the Kurdistan Democratic Party (KDP) would collaborate with the Kurdistan Workers' Party (PKK) in Turkey, which had moved its base of operations to northern Iraq and northern Iran in the early years of the war (Olson 1998, 30). As such, although the Iraq-Iran war created some gains for Turkey, there were also security concerns that influenced policy towards Iran. However, in total, Ankara followed an interest-driven policy towards the war and Iran because of its regional position and domestic issues. As long as Turkey gained benefits, Turkey pursued to establish cordial ties with Iran.

3.2.2 The 1991 Gulf War

By invading the oil-rich regions of Kuwait in 1990, Iraq hoped to control the world's secondlargest oil reserves. The invasion led to the formation of a US-led Coalition deployed to Saudi Arabia (Kibaroglu and Nun 2008, 152; N. A. Özcan and Özdamar 2010, 108). Turkey joined the allied coalition to blockade Iraq despite their developed economic affairs. Hoping to use the war to improve its regional influence, Ankara had several aims, which included demonstrating its strategic significance to Europe and the US, reasserting its strategic position in the post-Cold War era, and assuming an essential role in any future Middle East settlement (Mango 1994, 112; N. A. Özcan and Özdamar 2010, 108). In particular, Ankara viewed the crisis as a chance for Turkey to reiterate its interest in EU membership, and to develop a strategic relationship with the US (Larrabee and Lesser 2003, 134). Given these aims, Ankara cooperated with the US-led forces to liberate Kuwait from Iraq under the UNSC Resolution 678. Ankara also allowed the US to use its Incirlik Airbase during the crisis, and in doing so, Turkey abandoned its previous policy of neutrality and non-intervention in the regional disputes (Narli 1993, 279). Therefore, the 1991 Gulf War was a systemic variable which determined Turkey's regional policies.

In contrast to Turkey, Tehran adopted a neutral policy towards the war, believing it would lessen Iraq's relative power (N. A. Özcan and Özdamar 2010, 108). In essence, this war strengthened Iran's regional position. Having fought a bloody eight year war with Iran that it had in initiated, Iraq now had to accept Iranian territorial conditions which included the recognition of Iranian and Iraqi joint sovereignty over the contested region of Shat-al-Arab (Kibaroglu and Nun 2008, 152). As such, Tehran proved not only its regime legitimacy but also its durability against a Baghdad whose own regional influence was much diminished by the outcome of the war. While this outcome reduced Iranian concerns about Iraq's regional designs, Tehran remained wary of Ankara's increasingly pro-western policies. In response, Iran cooperated with Russia and Armenia and used religious and Kurdish political cards against Turkey (Ehteshami and Elik 2011, 652). The way in which Iran's policies undermined Turkey's security exasperated Ankara and led to hostility from Turkey towards Iran.

The main criticism levied by Turkey at Iran was the Iranian support for the PKK, which had already carried out the most prominent attacks against Turkey, by utilising the vacuum created by the US and its allies north of the 36th parallel in Iraq in April 1991. Even though Turkey called on Iran to terminate its support for the PKK, its diplomatic attempts were ineffective. Moreover, Turkey entered into a competition with Iran in the northern part of Iraq during this period. While Turkey supported Massoud Barzani's Kurdistan Democratic Party (KDP), Iran supported Jelal Talabani's Patriotic Union of Kurdistan (PUK) and the PKK. As such, Turkey blamed Iran for assisting opposition groups and violating international law in Northern Iraq while competing with Iran (Narli 1993, 280; N. A. Özcan and Özdamar 2010, 108). In this regard, Turkish-Iranian competition spread to northern Iraq, with both countries vying to increase their position of power and endorsing different Kurdish groups in the area. Therefore, closer ties with the Kurdish groups in northern Iraq emerged as a crucial aim for both countries.

Despite this rivalry over the north of Iraq, Turkey signed an agreement with Iran regarding control of terrorist groups, fearful that Kurdish irredentism could eventually challenge the territorial cohesion as well as security of the Turkish state. According to the agreement, neither country would allow terrorist activities within their borders, accepting that collaboration was essential against political violence and separatism (Kibaroglu and Nun 2008, 163). Both also agreed on the territorial integrity of Iraq by opposing the foundation of an independent Kurdistan, which threatened their internal security (Narli 1993, 280). When facing a mutual security threat from Kurdish armed groups again, Ankara opted for bilateral cooperation with Tehran because this seemed less costly than opposing Tehran. If Tehran had not provided financial and military support to Kurdish armed groups acting against Turkey, these groups would in turn have not received support from other states, thus enabling Turkey to manage the groups more easily. Based on these cost-benefit calculations, Turkey sought an occasional détente policy towards Tehran over mutual security concerns.

Following the war, Turkey further engaged in more cooperative ties with Iran, especially on energy, owing to the economic sanctions against Iraq, which had once been a significant trading partner with Turkey (Calabrese 1998, 85). Since petroleum imports from Iraq had ceased, Turkey had lost billions of dollars in pipeline fees and trade revenue. The international isolation of Iraq also "cost the Turkish treasury an estimated \$35 billion and damaged the local economy of the region bordering Iraq" (Cook and Sherwood-Randall 2006, 10; Larrabee and Lesser 2003, 135). Turkey therefore saw its access to Iranian oil as providing some amelioration for its economy (Eralp 1996, 109). Ultimately, the 1991 Gulf war culminated in both cooperation and competition areas in TFP towards Iran. Considering its interest-driven aims, Turkey opted for rapprochement with Iran but, at the same time, aiming to increase its relative power position created competition between them.

3.2.3 The Collapse of the Soviet Union and the Emergence of the Newly Independent States in the Caucasus and Central Asia

With the collapse of the Soviet Union in late 1991, the bipolar power balance which had influenced global politics ended and significant change followed on both international and regional levels. The fall of the Soviet Union caused the birth of newly independent states, including six republics (namely Azerbaijan, Kazakhstan, Kyrgyzstan, Uzbekistan, Turkmenistan, and Tajikistan) in the Caucasus and Central Asia. As such, various regional and global state actors sought to establish closer ties with these new states. Turkey was no different.

Not only was geographical proximity a driver but it also shared the same ethnic origins, religion, language, and culture. Suleyman Demirel, the ninth President of Turkey from 1993–2000, emphasised these affinities when he declared:

It is quite natural that Turkey acts by taking into account its national interests in terms of its security, its economic and social relations. However, Turkey, which considers its national interests in this issue, has some moral responsibilities. These moral responsibilities come from Turkey's history...It is impossible to isolate yourself from your history...History offers opportunities as well as responsibilities and difficulties (Bal 2000, 44).

From the economic aspect, Ankara aimed to establish free and efficient international telecommunications and transportation connections, the creation of international banking links, and the finance through exporting consumer goods and entrepreneurial know-how (Arik 1992, 34; Eralp 1996, 104). Turkey also tried to cooperate with these states in terms of its oil and gas needs, to meet its increasing energy demands, while reducing dependency on the Middle East oil and gas, and obtain royalties (Dikkaya 1999, 197; Kramer and Cimen 2001, 156). It thus strengthened its ties through the establishment of an investment bank, the extension of Turkish TV broadcasts into Central Asia, and the promotion of the Latin alphabet (Narli 1993, 282; Winrow 1997, 174). Turkey also founded a development agency, Turkish International Cooperation and Development Agency (TIKA), opened Turkish cultural centres and created state-owned television TRT-Avrasya with customised broadcasting for the Turkic populations of the former Soviet Union (Sinkaya 2004, 86). Therefore, closer cultural and social ties with these newly emerged countries served Turkey's economic and political aims.

From a political perspective, moreover, Central Asia and the Caucasus intersected with Turkish interests relating to both security concerns and Turkey's standing in international politics (Bal 2000, 45). Based on its national interests, Turkey sought to export its experience of democracy and a market economy to these countries as a model state, "emphasizing ethnic Turkic ties, secularism, integration into Western economic and political institutions, and increased trade and cultural ties" (N. A. Özcan and Özdamar 2010, 107). Turkey, therefore, viewed the collapse of the Soviet Union and the birth of these countries as an opportunity to assert a dominant role in the region. In this context, Turgut Ozal, the eight President of Turkey (1989–1993), claimed that "Turkey should have not missed such an opportunity that appeared for the first time after 400 years" (Sinkaya 2004, 83). This shows that Ankara aspired to

promote its soft power in the Caucasus and Central Asia, and the newly founded states provided Ankara an opportunity to enhance its power position.

Western powers also endorsed Turkey's attempt in the region due to Turkey's westernbased foreign policy. In particular, the US propped up the model of Turkey as secular and democratic state with a free market-economy, believing this to be a model for the post-Soviet states in the Caucasus and Central Asia. This is because Washington was concerned that in any power vacuum, political Islam in its varied forms, as well as the influence of Iran would spread throughout the region. As such, it was hoped that Turkey, a democratic western ally, could restrict theocratic Iranian influence in the region (Eralp 1996, 104; Kibaroglu and Nun 2008, 163). Given its concerns regarding Iran, the US also "vehemently opposed the transport of Caspian oil and gas through Iran, instead of supporting projects that would go through Turkey, and prevented countries like Georgia and Armenia from developing stronger economic and political ties with Iran" (N. A. Özcan and Özdamar 2010, 107). Accordingly, Washington's support for Turkey encouraged Ankara to pursue assertive policies in the Caucasus and Central Asia, and as a result Ankara increased its diplomatic overtures towards these new states.

As well as encouragement from the US, Turkey aimed to gain the European Community's (EC) support. Due to their complicated relations over Turkey's membership, Ankara assumed that the EC would reconsider its decision over rejection of Turkey's application to become a full member of the EC if Turkey maintained its assertive policies in the region. Clearly, Turkey was trying to forge new alliances in an effort to stakeout a new and influential position in the emerging regional order (Oran 2001, 158–60). Therefore, the emergence of these new states following the fall of the Soviet Union provided economic and political opportunities for Turkish interests. Ankara sought to use these opportunities to improve its relative power position on the international stage.

Like Turkey, Iran also aimed at establishing close ties with these new states, many of which are adjacent to its northern border. Iran mainly tried to utilise its Islamic ideology, supplemented by territorial and cultural commonalities to legitimise and enlarge its presence and influence. Based on its interests, Iran supported Islamist movements and increased some economic ties and energy trade with these states. In doing so, Iran, like Turkey, tried to fill the power vacuum created by the demise of the Soviet Union and escape regional isolation by "using these states as a corridor to open to the world political and economic markets" (Ertosun

2012, 10; Menashri 1998). As a result of Iran's interest-based policies towards the new states, Turkish-Iranian relations cooled considerably again, and an intensive rivalry emerged for economic and political influence in the Caucasus and Central Asia (Akkoyunlu 2012, 263; N. A. Özcan and Özdamar 2010, 107). For example, Turkey competed with Iran over Azerbaijan, in which both countries had close ties. While Turkey shared affinities of language, ethnicity and secular government, Iran did so in terms of religion, kinship, and history through its own Azeri population. Therefore, Turkey entered into a rivalry with Iran to establish a form of political and cultural hegemony in Azerbaijan and take advantage of power imbalances (Hentov 2012, 145). Despite its shared concern regarding the extension of ethnic conflicts following the dissolution of the Soviet Union, Turkey's regional difference from Iran was most visibly demonstrated when it supported Azerbaijan, in contrast to Iran's support for Armenia in the Nagorno-Karabakh dispute in 1992 (Gurkas 2011, 111). As such, enhancing the relative power position was a more dominant aim for Turkey than tackling the ethnic conflict in its proximity. The Nagorno-Karabakh dispute indicated Ankara's long-term rivalry with Tehran over regional influence.

In turn, Iran had concerns about Ankara's "Turkism" policy, which aimed to establish closer ties between Turkey and Turkic-origin countries due to its ethnic Turkish minorities, such as the Azeris. Tehran believed that Ankara's policy would trigger its Turkish minorities to attempt to establish cordial ties with Ankara. For this reason, Turkey's closer ties with the newly founded Turkic countries placed Ankara and Tehran in opposition, and Tehran hoped to limit Turkish policies in its proximity (Keskin 2009, 49). Ultimately, their competition to export their respective models and economic rivalry for the transportation routes of the Caspian hydrocarbon resources determined Turkey's perceptions towards Iran (Sinkaya 2004, 95). Turkey deemed Iran a rival in Central Asian and Caucasus. As Elik (2012, 94) reveals, "both Turkey and Iran have been trying to gain leverage against great power influence and also negating each other's hegemonic ambition in this critical geography". In this regard, Turkey followed revisionist policies towards the newly founded Turkic states and aspired to increase its relative power position, primarily using soft power policies to solidify cordial ties with these countries. Ankara's these policies brought another rivalry with Tehran.

3.2.4 The Palestinian-Israeli Conflict

With the establishment of Israel as a Jewish state in Palestine on May 14, 1948, the long-term Palestinian-Israeli conflict (or the Arab-Israeli conflict) officially engulfed international and regional states and resulted in a more turbulent regional environment. Since then, Middle Eastern politics has mainly been framed by the conflict and wars between Israel and Palestine and between Israel and Arab countries, as well as other regional issues. International state powers and other non-Arab regional state powers, such as Turkey and Iran, either directly or indirectly participated in these conflicts for different reasons. In this regard, the Palestinian-Israeli conflict and more particularly Turkish-Israeli relations shaped Turkish-Iranian affairs and their policies towards each other. Since the beginning of the Palestinian-Israeli conflict, Turkey has sought to play an active role in the dispute; however, due to the time scope of this thesis, this subsection examines Turkish and Iranian policies towards the conflict since 1979.

Turkey has followed fluctuant policies towards the conflict due to its changing interests. In the early 1980s, Ankara first opted to maintain closer ties with Arab countries rather than Israel due to "Turkey's highly dependent [status] on oil imports from its Arab neighbours" (Gruen 1995, 48). In 1980, Ankara's diplomatic representation in Israel was reduced from the level of consulate to second secretary after Israel declared a united Jerusalem as the capital of Israel. On the other hand, Turkish policymakers then realised that Turkey required a balanced policy toward Arab countries and Israel in order to be able to increase ties with western countries and fix its fragile economy (Gruen 1995, 49). Even in 1984, the Turkish prime minister, Turgut Ozal, pointed out,

Turkey's relations with Israel will be maintained. They will neither improve nor deteriorate...As I told our Arab friends, we are members of NATO and have close ties with the US. The strength of the Israeli lobby in the US is well known...Furthermore, it is a good thing for the Islamic world that an Islamic country should have an open window [to the West] (Gruen 1995, 50).

Since the late of the 1980s, Turkey sought such a close economic tie with Israel that Turkey's trade amount with Israel demonstrated a high rise from \$54 million in 1987 to \$100 million in 1991 (Çakmak and Ustaoğlu 2017, 308–9). While following close ties with Israel, Turkey also maintained its quest for active involvement in the dispute. Ankara recognised the

declaration of Palestinian state in November 1988. In the 1990s, Turkey enjoyed close ties with Israel, with US support, and Turkish-Israeli affairs were upgraded to the ambassadorial level in 1991. Having established an 'entente' with Israel, however, Ankara strengthened its ties with Israel rather than Arab countries. This was because Ankara had been facing problems with the latter over Cyprus, in which Turkey had no support from the Arab countries, and because of Syria's support for the PKK. With the 1991 Arab-Israeli peace process in the Madrid Peace Conference between the representatives of Israel, Syria, Jordan, and the Palestinians and the 1993 Oslo Accords between Israel and the Palestinian Liberation Movement, Ankara's ties with Israel showed a high level of diplomatic and economic ties. Although the Islamist-based Welfare Party in Turkey followed a pro-Muslim policy in the region and criticised Israel use to political pressure of the Turkish army on the Turkish government (Gökçe 2016, 59–61). As such, Turkey mostly followed interest-based policies with Israel and these interests determined its perception of the Palestinian-Israeli conflict.

Because of Turkey's regional interests, Turkish foreign policymakers aimed to preserve a balance between Turkey's relations with Israel and Arab states at a certain level (Ayman 2012, 10; Sayari 1997, 45). More precisely, they opted for a balanced foreign policy towards the Palestinian-Israeli conflict by refraining from breaking ties with both sides in the conflict and standing with an international consensus that called for a two state solution (Ertosun 2012, 14). As such, they followed a cautious and prudent foreign policy towards the conflict. Due to economic and political concerns, notably, Ankara opted to maintain ties with both Israel and Arab countries when Turkey's relations with western countries became problematic following the military coup against the Turkish government on September 12, 1980 (Aras 1997, 61). Therefore, Ankara perceived Israel and Arab countries as alternative political and economic partners and followed pragmatist and status quo driven policies to the conflict, showing that cost-benefit calculations determined Turkey's perception of the dispute.

In contrast to Turkey, with the 1979 Islamic Revolution, Tehran followed hostile policies to Israel, believing that Israel sought to weaken Muslim states, and urged other Muslim countries to collaborate against Israel (Aras 1999, 201). Therefore, Tehran wholeheartedly adopted anti-Israel policies in the dispute and supported the Palestinian groups against Israel. Given Tehran's anti-Israel policies, Tehran welcomed the cooling ties between Israel and Ankara. According to Iran, Israel was using Turkey to achieve its long-term regional interests

and Turkey served Israeli policies in the region (Ozturk 1999, 254–55). Tehran further claimed that Israel provided intelligence about Iran across the border between Turkey and Iran, thus perceiving the Turkish-Israeli rapprochement as a threat to its national security (Aras 1996; Parsi 2007, 171). Iran pressurised Turkey to abrogate its alignment with Israel, claiming that cooperation with the "Zionist regime" would result in civil unrest across Turkey. Having blamed Turkey for serving Israeli and US interests, Iran allegedly attempted to sway Turkish society against Israel following the 1997 February 'Sincan affair'. At this event, "the mayor of Sincan, a Justice and Welfare Party member, held special activities in a big tent resembling the al-Aqsa mosque and decorated with posters of members of Hamas and the Lebanese Hezbollah"; Iranian Ambassador Muhammad Riza Bagheri delivered a speech to the people by calling on them to punish those who signed agreements with the US and Israel (Bengio 2004, 160-61). This case thus resulted in a diplomatic crisis between Turkey and Iran and the reciprocal recalling of their ambassadors. Turkey saw this as a crass example of interference in its internal affairs and vehemently criticised Iran for exporting the Islamic revolution to Turkey, and destabilising its secular regime by supporting radical Islamist groups (Akkoyunlu 2012, 264).

With the AKP's accession to power in 2002, Turkey maintained friendly ties with Israel under the foreign policy principles of 'multidimensional' and 'zero problems with neighbour'. However, these ties became increasingly tense following President Erdogan's acerbic words to Israeli President Shimon Peres on 29 January 2009 at the Davos Summit due to a series of Israeli attacks on Gaza called Operation Cast Lead between 27 December 2008 and 18 January 2009. According to Đidić and Kösebalaban (2019, 7), "Both the AKP's conservative and liberal democratic orientations clashed with Israel's treatment of Palestinians". With the deaths of nine Turkish activists during the Israeli raid on the Mavi Marmara vessel taking aid to Gaza on 30 May 2010, relations nose-dived politically and militarily. At the same time, Turkey started to follow an active pro-Palestinian policy by supporting Palestine's observer status at the UN and urging the removal of the Israeli blockade against the Gaze strip. Then Hamas leader, Khaled Mashal, and President Erdogan even met many times in Turkey to talk about recent developments on the Palestinian issue. With Turkey's more interventionist policy towards Palestine, Ankara was said to be pursuing a wider leadership role, not just in the Palestine-Israeli conflict but more broadly across the Middle East (Balci 2013, 202; Gumus 2013, 84). Therefore, the AKP used its ideology and ideas for interest-based policies towards the conflict in the pre-uprisings term.

Iran, however, maintained its hostile view of Turkey, claiming that Ankara's hostile policies towards Israel were just for show, believing that NATO's missile defence system targeting Iran and Turkey served the US and Israeli goals in the region (Gumus 2013, 91). Tehran further opposed Turkish involvement in the Palestinian-Israeli conflict, assuming that Turkey's greater engagement would diminish Iranian influence (Boz 2013).

In return, Turkey hoped for a deterioration in Hamas-Iranian relations to reduce the Iranian effect on the Palestinian-Israeli conflict. As such, Turkey welcomed the breakdown of Hamas-Iranian relations in 2012 when Hamas and Iran endorsed different groups in the Syrian conflict. Hamas refused to support the Assad regime as an ally of Iran in the Middle East and disagreed with the Iranian presence in the conflict. The disagreement then deepened when Hamas endorsed the legitimacy of Yemeni President Abed Rabbo Mansour Hadi's regime, in contrast to Iranian support for the Houthis (Tuysuzoglu 2014, 595).

In conclusion, the Palestinian-Israeli conflict occasionally resulted in competition between Turkey and Iran. Turkey adopted mostly interest-based policies towards the conflict to increase its relative power position and/or forge close ties either with Arab countries or Israel. While the Palestinian-Israeli conflict had limited influence on TFP towards Iran during the 1980s and 1990s, Turkey's active involvement in the dispute since 2009 paved the way for a rivalry between Ankara and Tehran due to AKP's pragmatist and utilitarian policies in the conflict.

3.2.5 The US-led operations of Afghanistan and Iraq

The 9/11 terrorist attacks on the Pentagon and the World Trade Centre became a turning point generally in international politics, but particularly in Middle Eastern politics. The respective US-led operations of Afghanistan in 2001 and Iraq in 2003 following the 9/11 attacks resulted in significant regional changes, including a shift in the power balance. In this regard, Turkey and Iran, as two regional significant state actors, were influenced by developments arising from the US-led operations. In the US-led Afghanistan war, Turkey gave the US full support by condemning the terrorist attacks as a country that had been fighting against terrorism for more than thirty years, and by sending troops to Afghanistan. After the overthrow of the Taliban regime in the late of 2001, Turkey stationed troops in the northern region of Afghanistan for military educations and intelligence aid and twice led the NATO international security

assistance force between June 2002 and February 2003 (Cook and Sherwood-Randall 2006, 8). As such, Turkey entirely supported the US-led Afghanistan operation and played a significant role in the developments of the war as a NATO country.

In the US-led operation of Iraq in 2003, however, Turkey hesitated to play an active role due to security concerns. In particular, Turkey highlighted the possible emergence of an independent Kurdish state in northern Iraq, and the future of oil-rich city Kirkuk, as red lines for its national security (A. Karakoc 2009, 28). Ayman (2014, 11–12), therefore, notes that "Turkey's anxieties were much more focused on the possibility of the spill over of the Kurdish demands for independence and their potential to undermine Turkey's security and stability than on the deployment of American troops in Iraq". This statement shows that Ankara was concerned about the emergence of US-Kurdish alliance and the possibility of the empowerment of the Kurdish groups in its proximity.

In consequence, despite being a US and NATO ally, Turkey refused to participate in the US-led Iraq war when the Turkish Parliament voted against the US decision to open a second front in March 2003, causing a severe decline Turkey-US relation (Ozdamar and Taydas 2008). In this context, Ayman (2012, 10) addresses that "Sectarian violence in Iraq and growing Kurdish separatism that could spill over, urged Turkey to focus more on the Middle East in an effort to prevent the establishment of a Kurdish state and contribute to peace and stability in Turkey's neighbourhood". Uslu (2000, 200) further points out that:

The general Turkish view on Iraq is that Iraq's independence, sovereignty and territorial integrity should be preserved, and Iraq should comply fully with the relevant UN Security Council resolutions. While the U.S. remains focused on removing Saddam from power and challenges Iraq's territorial integrity with its policies, Turkey does not want to see Iraq destroyed and divided, fearing that this would destabilise the balance of power in the region and contribute to the expansion of the Iranian influence.

Therefore, the main reasons for Turkey's hesitation about joining the US-led Iraq operations were the twin fears of having a Kurdish state in its proximity, and the rise of Iranian influence in the region despite its active support to the US-led operations in Afghanistan. Accordingly, Ankara deemed the US-led operation in Iraq more costly and profitless for its security interests.

Iran had concerns about the US-led operations, even though its key protagonists, the Taliban in Afghanistan and Saddam Hussein in Iraq, had been toppled by the end of the operations. Its primary concern was the blockade and threat from the US and its regional allies to its survival. As such, Iran sought the elimination of US military pressure over the invasion of Iraq because it had backed US-led efforts in Afghanistan. In particular, US President George Bush's State of the Union address on 29 January 2002, regarding a possible US-led attack against Iran, increased Tehran's fears since Bush labelled Iran, Iraq and North Korea as the "axis of evil". Following this statement, Iran was thought to have become a potential military target for the US after Iraq (Celikkanat 2007, 128).

Additionally, Tehran feared any increase in Turkish influence in the region, assuming that Turkey would enhance its control over Northern Iraq by cooperating with the US and controlling the area's oil fields. For Iran, this meant reducing its own influence over regional politics. Moreover, the growing military cooperation between Turkey and Israel angered Iran, and Tehran perceived this alliance as a threat to its national security (N. A. Özcan and Özdamar 2010, 110). Turkey and Iran therefore deemed one another rivals following especially the US-led operations in Iraq, and TFP towards Iran was based partly on this perception.

Nevertheless, Turkey and Iran shared a concern regarding the possible disintegration of Iraq because a divided Iraq would, they reasoned, only encourage Kurdish separatists. Therefore, the Kurdish issue re-emerged as a mutual policy following the US-led operations (Wang 2011, 9). Based on their mutual fear, Turkey changed its position towards Iran. Indeed, as Cetinsaya (2003) highlights, "following the events leading up to the US invasion of Iraq in 2003, the relationship between Turkey and Iran seemingly [entered] a new phase. Similar concerns about the probable consequences of developments in Iraq [caused] the two countries' positions with respect to regional political issues to converge". Given their mutual concern, Turkey required Iran's cooperation since both faced a loss of influence in northern Iraq due to the US presence in Iraq (Olson 2004, 217). With the US now the dominant power in the region, both had little chance to influence the developments of regional politics. Turkey, along with Iran, "called for Muslim countries [to act] together to solve the Iraq crisis peacefully and [to come] together under the Turkish initiative to hold regional meetings with the participation of foreign ministers of regional countries to find a solution to Iraq crisis before and after the war" (Ayman 2012, 10). When considering cost-benefit calculations, therefore, Turkey opted to establish a short-term pragmatist collaboration with Iran following the US-led operation in Iraq.

Nevertheless, this short-term rapprochement was interrupted when Iran sought closer ties with the Shia-dominated Iraqi government in the aftermath of the 2003 invasion. Ankara was concerned about Iran's policies in Iraq, including cooperating with Shia cleric Moqtada al-Sadr's Mahdi Army and other insurgent groups in Iraq. Ankara believed that Tehran sought to increase its influence and power in the post-Iraq war. Iran's quest for a weakened Iraq and its sectarian policies vindicated Turkey's concerns regarding Iran (Kane 2011, 1). Given its fears regarding Iran's Iraqi policies, Turkey preferred there to be no dominant group in the Iraqi political process; rather, it aimed for a united Iraq for its security and economic interests and regional stability. Therefore, Ankara propped up the secular Iraqiya alliance of Allawi, emphasising the "Iraqi identity rather than any religious or ethnic identity" (ibid). In this regard, Turkey strongly opposed friendly ties between the Shia-dominated Maliki government and Tehran. At the same time, President Erdogan expressed his annoyance over Iran's rising influence in Iraq after the decrease of the US presence in Iraq in late December 2011, complaining that the US abandoned Iraq to the hands of Iran (Ayman 2014, 15).

Ultimately, Turkey maintained its ambivalent policies, including conflict and cooperation, towards Iran following the US-led operations of Afghanistan and Iraq. While the Kurdish issue was a mutual concern, the warming ties between Iran and the Shia-dominated Maliki government and the rise of Iran's influence on Iraqi politics increased Turkey's fears regarding Iran. While aiming to increase its regional power position, Ankara perceived Tehran as a rival and sought to hamper Iranian influence, especially in Iraq.

3.3 Concluding Remarks

From 1979 to 2011, bilateral and systemic factors shaped TFP towards Iran. Before the Arab Uprisings, Turkey and Iran occasionally became rivals due to conflicts of interest. For instance, following the collapse of the Soviet Union, Ankara increased its power position in the region, forging cordial economic and political ties with the newly founded Turkic states. After the US-led operations in Afghanistan in 2001 and Iraq in 2003, Turkey occasionally became concerned about the rise of Iranian influence in its proximity, assuming that increased Iranian influence would lead to a decrease in Turkey's. Therefore, Ankara did not welcome the closer ties between Baghdad and Tehran. Notably, Iran's Shia-based regional policies evolved contrary to Turkey's regional aims and voided Ankara's calculations regarding regional developments. Ankara thus perceived Tehran's policies as a threat to its interests and benefits.

Ankara also formed a temporary alliance with Iran due to its Kurdish issue and economic interests. Despite at times slamming Iran's pro-PKK policies, Turkey required Tehran's collaboration to be able to hamper the rise of the PKK in its proximity. Given Turkey's ambivalent policies towards Iran, it can be concluded that Turkey's interest-based aims and cost-benefit calculations primarily determined its foreign policy approach towards Tehran. In cases where cooperation with Tehran was more appropriate for Turkey's interests than their competition, Ankara opted to work closely with Tehran, but otherwise Ankara deemed Tehran a rival in cases where cooperation with Tehran was costly and provided less benefit for its interests. Based on the arguments made in this chapter, TFP towards Iran exhibited a realistbased approach. This chapter further evaluated determinants of TFP towards Iran before the Arab Uprisings. While some determinants, such as minorities and economic interests, seem unchanged, other systemic determinants stemming from the Arab Uprisings influenced TFP towards Iran since the outset of the uprisings. Table 3.1 thus demonstrates the constant and changeable before the Arab Uprisings. In the following chapters, there will be a further examination of the realist paradigms in TFP towards Iran. Chapter 4 contains an examination of the role of recent systemic factors and Turkey's relative power position in shaping TFP towards Iran since the Arab Uprisings, as the independent variables explored in this thesis.

Constant Determinants	Changeable Determinants
Minorities	Systemic Factors
Economic factors	Ideology (from secularism to political Islam)

Table 3-1 Determinants of TFP towards Iran before the Arab Uprisings

CHAPTER 4 THE SYSTEM LEVEL — INDEPENDENT VARIABLES IN TURKISH FOREIGN POLICY TOWARDS IRAN SINCE THE ARAB UPRISINGS (DECEMBER 2010–JULY 2018)

According to Samuel Huntington, "the third wave of democratisation that swept the world in the 1970s and 1980s might have become a dominant feature of Middle Eastern and North African politics in the 1990s" (Abushouk 2016, 52). Before the Arab uprisings, in essence, the Middle East experienced some mass demonstrations, including the Kefaya Movement, which was against the fifth presidential term of the Mubarak regime in Egypt in 2005, and Yemen's Southern Movement, which were anti-government protests in 2008 and 2009 (Dincer and Hecan 2020, 6). With the start of the Arab uprisings in late December 2010, the Middle East was faced with chaos, insurgency, civil war, or revolution once more.

Democratisation and social justice emerged as two significant goals among the wave of protests, as most people in these countries lacked basic services and lived in poverty. Furthermore, according to World Bank data, "in the decade preceding the uprisings, the share of income that went to the top 20 percent of the populations of Syria, Yemen, Morocco, and Egypt ranged from 41 to 48 percent, while the bottom 20 percent controlled between 7 and 9 percent" (Al-AIi 2019, 2). In this regard, the Middle East has suffered from anti-democratic regimes because of "a combination of oil wealth and conflict." Notably, a large and vibrant youth population was dissatisfied with rising inequality and declining levels of secure employment, prompting them to seek a political and social change (Elbadawi and Maksidi 2017). As a result, the protesters used 'aish, horreya, adala ijtema'eya' [bread, freedom, social justice] to express their demands during the uprisings.

With the suicide of a Tunisian hawker called Muhammed Bouazizi in December 2010, a wave of rebellion emerged first in Tunisia and then in Egypt, Libya, Yemen, Bahrain, and Syria. According to Dalacoura (2012, 66), Tunisia, Egypt, Libya, Bahrain, Yemen, and Syria were isolated "from the rest of the region which experienced comparatively minor fallout from those events," therefore not all monarchies were impacted equally by the Arab Uprisings. In this regard, the initial demonstrations in Tunisia had a domino effect and quickly affected much of the Middle East and North Africa more or less. However, the effects of the uprisings were different for each individual country. Some countries experienced regimes change; others enacted hasty reforms. When losing their authoritarian allies, for instance, the oil-rich dynastic Gulf monarchies, such as Saudi Arabia and the United Arab Emirates (UAE), maintained their existing family rule and established strategies by increasing public spending. They also supported other poorer monarchies, such as Morocco, Jordan, and Oman, supplying diplomatic and financial assistance, and in one case even gave direct military support to Bahrain (Bank, Richter, and Sunik 2014, 177). As such, the above table demonstrates where the uprisings happened and what the results of the uprisings for each country were.

Tunisia	 The first protests, also known as the Jasmine Revolution, occurred in response to the self-immolation of a street vendor, Mohamed Bouazizi, who was protesting his treatment by local officials. President Zine al-Abidine Ben Ali, an authoritarian leader, was deposed on January 14, 2011, and a free election was held in October 2011 to select a President and Prime Minister. A new Constitution was promulgated in January 2014.
Egypt	 Protests began on January 25, 2011, and another authoritarian leader, Hosni Mubarak, was overthrown on February 11, 2011. Mohammed Morsi's Muslim-Brotherhood affiliated government was elected in June 2012. The democratically elected leader Morsi was deposed by Egyptian military, and General Abdel Fattah al-Sisi took over as president in July 2013.
Libya	 Protests against Muammar al-Qaddafi erupted in mid-February 2011. NATO launched a campaign of air strikes against Qaddafi's forces in March. Following Qaddafi's assassination in October 2011, a Transitional National Council was formed and recognised internationally. However, a bloody civil war erupted in 2014.The country was divided into rival eastern and western-based administrations.
Yemen	 Demonstrations erupted in late 2011, and Saleh signed an internationally mediated agreement calling for a phased transfer of power to the Vice President, Abd Rabbuh Mansur Hadi, in November 2011. However, a civil war erupted in Yemen in 2014, when Hadi's government was confronted with armed conflict and rebellion by the Houthis.

Syria	 Protests erupted in southern Syria against Bashar al-Assad in mid-March 2011. However, Assad retains power as a result of Russian, Chinese, and Iranian support, and a civil war is ongoing.
Bahrain	 Demonstrations took place in mid-February 2011. Saudi Arabia and the United Arab Emirates aided the Bahraini security forces in March 2011.
Other countries (Algeria, Jordan, Morocco, Oman)	• Minor pro-democracy protests occurred, and the rulers made a variety of concessions ranging from the dismissal of unpopular officials to constitutional changes.

Table 4-1The effects of Arab Uprising in each country from 2011 to 2018

Such circumstances affected the regional structure and caused power vacuums which pulled in other states in the Middle East. "Structure" is defined by the relative distribution of capabilities and the ordering principle of the international system-anarchy (Waltz, 1979 pp. 91–99). Accordingly, the political upheaval invariably impacted upon the regional balance of power. Turkey was too affected by this change in the regional environment and tried to adapt foreign policy in parallel with its new surroundings. Following this thesis' methodology and theoretical framework, this chapter is concerned with the role of the Arab Uprisings in Turkish foreign policy (TFP) towards Iran and Turkey's relative power position as the independent variable. In the first section, distinct but interrelated trends are presented which were transforming Middle Eastern politics since the Arab Uprisings, from December 2010 to July 2018, at the systemic level. These trends are the role of Islamic parties, the Sunni-Shia divide, the role of the global powers in the Middle East, and the empowerment of non-state armed actors. The second section sheds light on how the Arab Uprisings affected Turkey's regional power position and its bilateral ties with Iran from December 2010 to July 2018.

Overall, this chapter aims to answer the primary and overarching question: *To what extent did structural forces shape the formulation of Turkish policy towards Iran following the Arab Uprisings, from December 2010–July 2018?* In this process, other sub-questions will be answered on the role of Turkey's relative power position in its foreign policy behaviour and how Turkey recalibrated its foreign policy in the wake of a geopolitical shift.

4.1 An Analysis of the Arab Uprisings at a Systemic Level

The uprisings increased the uncertainty in the region and had the potential to shift the regional balance of power in which all regional and international states were attempting to manage, either by establishing a new Middle Eastern order or maintaining the status quo in line with their interests. Lynch (2013, 17) notes that "the Arab uprisings erupted within the context of a decade long cold war that polarised the region into a resistance axis led by Iran, which included Syria, Hezbollah, and Hamas as well as a broad swath of public opinion, and a moderate axis aligning the US with Israel and most of the other Arab states". Accordingly, new regional alliances were established following the overthrow of the authoritarian regimes, because as the Arab Uprisings continued to transform the balance of power, the former authoritarian regimes became weak and fragile.

The Arab Uprisings further paved the way to an increased role for Islamic parties and non-state armed groups in regional politics. As such, new power centres emerged to fill the power vacuum, and there were significant threats and opportunities for regional and global state powers to make gains according to their interests. In this tumultuous environment, the role of Muslim Brotherhood (MB)-affiliated Islamic parties in Tunisia, Egypt and Libya, the growing sectarian divide between Sunni and Shi'a, the presence of global state powers across the region, and the empowerment of non-state armed groups were all factors in the re-casting of regional state competition, the balance of power, and the relative power position of the regional state actors in the wake of the initial uprisings.

4.1.1 The Role of Muslim Brotherhood-affiliated Islamic Parties

The role of Islamic parties proved significant following the Arab Uprisings. Most Islamic parties won electoral victories in various countries, including Egypt, Libya, Morocco, and Tunisia. They even played crucial roles in the developments in Syria and Yemen (Lynch 2016a). For instance, the MB-affiliated Freedom and Justice Party (FJP) won 37.5 per cent of the vote in the 2011–12 parliamentary elections in Egypt, and the MB-affiliated leader Mohamed Morsi became the first democratically elected President of Egypt in 2012. The similarly affiliated Ennahda in Tunisia won 37 per cent of the vote in the constituent assembly elections of October 2011, while the Justice and Development Party (PJD) achieved the largest parliamentary representation in November 2011. The advent of these parties to power

demonstrated that Islamic sentiment was a more dominant motivation than pan-Arabism among the public (Totten, Schenker, and Abdul-Hussain 2012). As such, following the uprisings, a dense Islamic political network re-emerged, and Islamist movements re-engaged with mainstream politics (Brown 2011).

According to Gerges (2013, 391), the reason why the Islamic parties gained power is that they were "slowly moving away from their traditional agendas of establishing an authoritarian Islamic state towards a new focus on creating civil Islam that permeates society and accepts political pluralism". In this respect, the shift in their perception and focus aroused sympathy from the public, who wanted to see an end to an authoritarian state order. Ayoob (2014, 10) posits that "the impressive electoral performance of the Islamist parties is likely to have regressive implications for Arab societies and highly negative consequences for the Arab world's relations with the West", especially with those having closer ties with authoritarian regimes and being concerned about Islamist movements. Therefore, the rise of Islamic parties had the potential to influence the regional balance of power.

The electoral victory of Islamic parties presented regional actors with an apparent dichotomy. Initially, Israel watched the transformations in regional politics with great concern because Arab-Israeli relations had been calculable and predictable since the signing of the Camp David Accords with Egypt in 1979 and the Peace Treaty with Jordan in 1994. Israel had viewed both Egypt and Jordan as largely passive strategic partners who would maintain the peace accords and the balance of power on behalf of Israeli interests (Amour 2017, 3–4). While the foreign policies of the previous authoritarian regimes were largely predictive, Israel had difficulty forecasting the trajectory of the new political dispensations' trajectory that now emerged. The fall of the Mubarak regime in Egypt, a reliable ally of Israel for over 30 years, became a worrying situation for Israel. In this regard, Israel particularly feared that its Arab allies would lose power and anti-Israel governments would rise in the region (Ayoob 2012, 86). The revival of Islamism and the empowerment of Islamic extremists would threaten Israel's security. As such, the electoral victory of MB-affiliated parties was viewed with deep suspicion in Israel, especially when the MB-affiliated government under Mohammad Morsi sought to cooperate with Turkey and Qatar and establish an "Islamist-based axis", even though "Morsi reiterated his party's commitment to uphold the peace treaty with Israel and preserve the strategic partnership with the US" (Amour 2017, 2–7; Shlaim 2014, 397). Based on a shared understanding of regional security due to the rise of MB-affiliated parties and Iranian influence throughout the region, Israel sought to cooperate with Arab Gulf states, in particular the United Arab Emirates (UAE) and Saudi Arabia, under a relationship called a "tacit security regime" (Jones and Guzansky 2017).

Alongside Israel, the Gulf Cooperation Council (GCC) —a regional intergovernmental political and economic union established in 1981 by Saudi Arabia, the UAE, Qatar, Oman, Kuwait, and Bahrain—was at the centre of shaping the regional order and maintaining both internal security and the regional status quo. In this regard, the members of the GCC, except for Qatar, were concerned about challenges to their Islamic legitimacy and wished to maintain their unique Islamic model in the region (Al-Rasheed 2011; Duran and Yilmaz 2013, 148). In this era, therefore, the GCC, except for Doha, defended the regional status quo and spread the wealth and political support to conservative regimes across the region (Ryan 2009, 27). They also conducted anti-MB policies, especially in Egypt, and assisted the military government after the fall of the Morsi regime in July 2013 even though they (notably Riyadh) endorsed other Islamic opposition groups in Syria.

Their ambivalent policies towards the Islamic opposition groups pertained to their perceptions towards these groups. While supporting other Islamic groups in the region for their interests, Saudi-led Gulf countries perceived the MB-affiliated groups as a threat against their security due to their religious ideology. Notably, the regional rise of the MB, with its Sunni Islam identity, posed a threat to Riyad's Wahhabi identity; as another version of Sunni Islam, this identity was used by Saudi Arabia to legitimise its regime and counteract Iran's Shia identity (Darwich 2017, 1290). As well as an identity challenge, these states assumed that the rise of MB-affiliated parties could increase Iranian influence in the region. In essence, Iran's MB-oriented policies vindicated these countries' fears regarding Iran. Tehran envisaged that "if Islamists come to power in Egypt, Iraq, and elsewhere they may be more inclined toward good relations with Iran". Iran thus welcomed the rise of Islamist movements, especially the MB-affiliated ones, in the region (Byman and Pollack 2011, 247), and strove to establish warmer ties with the newly-emerged governments to increase its regional position. The advent of the Morsi government in Egypt, for instance, encouraged Tehran to enhance its relationship with Cairo further and, in doing so, the Saudi-led Gulf countries were opposed to the MBaffiliated government and its state allies, including Turkey, Qatar and Iran.

Given their support for the MB-affiliated groups, Doha, along with Ankara, welcomed the rise of the MB-affiliated governments following the uprisings, seeing these groups as significant in improving their geopolitical status and regional influence. As such, Ankara and Doha aimed to establish a new axis of power in the Middle East in the midst of the power vacuum created by the Arab Uprisings. Notably, Doha became a financial supporter of the MBaffiliated governments (Ozkan and Korkut 2013, 171). Following the election of the MBaffiliated government in Egypt, Doha offered Cairo loans and investments estimated at around \$10 billion from the total Gulf support of \$20 billion in total economic aid promised to Egypt, as well as financial and tourism agreements (Khatib 2013, 428).

Qatar further endorsed other MB offsets in the region, such as moderate Islamist Syrian MB against the Assad regime. According to Roberts (2014, 54), the reason for Qatar's backing of the MB "originated as the result of a structural necessity to staff positions without inculcating any systems that would automatically defer authority to Saudi Arabia [...] It also continued to make Qatar an important spoke of the wider Muslim Brotherhood wheel, expanding its importance regionally. These networks played the central role in Qatar seeking to augment its influence" since the Arab Uprisings. Therefore, Doha followed a pragmatist and ideologicalbased policies in supporting the MB (Booysen 2020). However, the pro-MB policies of Qatar soon encountered difficulties. Doha was accused of supporting a de facto movement whose ideology remained anathema to the idea of hereditary rule, an ironic position for Doha to adopt given the absolute fiat exercised by the ruling family, the Al-Thani. As a result, the 2017 Qatar crisis emerged when Saudi Arabia, along with Egypt, the UAE and Bahrain, severed diplomatic relations with Qatar, accusing it of supporting extremism and destabilising the region (Trager 2017). With this crisis, the Saudi-led these Arab countries applied an embargo against Qatar who Turkey and Iran soon assisted. Ultimately, new alliances emerged with the inclusion of Ankara, Doha and Tehran against the Riyadh-led GCC countries.

Nevertheless, the success of the MB-affiliated parties started to be questioned when the Morsi government in Egypt was unable to meet the expectations of the uprising. Al-Anani (2015, 533) highlights that three main factors caused the fall of the MB-affiliated Morsi government: "the burden of conservatism and lack of a revolutionary agenda, organisational inertia and stagnation of the MB, and the lack of governance experience among Brotherhood leaders, particularly President Morsi". As a result, the democratically elected Morsi regime was ousted by the Egyptian military on July 3, 2013. Following the Egyptian coup, the Ennahda

Party in Tunisia resigned in January 2014, while other Islamist parties demonstrated poor performance in the 2018 elections (Lynch 2016a). Again, the regional balance shifted towards anti-MB states, and, with the failure of these MB-governments, influence of pro-MB states started to decrease throughout the region (Ryan 2009, 27). By contrast, the influence of Saudi-led Gulf countries increased, notably in Egypt, and they tried to rebalance the regional forces that had so disrupted the balance of power across the region.

To sum up, the MB-affiliated Islamist parties were an essential determinant top understanding the Arab uprisings, prompting both global and regional powers to jockey for position amid a changing regional balance of power in an attempt to either protect or enhance their regional interests.

4.1.2 The Sunni-Shia Divide

The Sunni-Shia divide was another significant determinant of the regional balance of power that emerged with the Arab Uprisings. This division originated from the US-led invasion of Iraq in 2003 when the Sunni-dominated Ba'athist regime of Saddam Hussein was deposed, and a power vacuum emerged. In the ensuing chaos and despite the efforts of Washington to hold free and fair elections, Iran saw an opportunity to enhance its influence, playing an essential role in Iraqi politics by exercising more power over different Iraqi Shia actors, political parties whose sectarian, as opposed to the ideological character was all too easy to discern (Salloukh 2013, 34–35).

With the outbreak of the Arab Uprisings at the end of 2010, Sunni-Shia competition escalated and a range of actors, many militia-based, sought to exercise influence and power over what were, in effect, captive constituencies. This rise in the power of non-state armed groups and their affiliate parties attracted Saudi Arabian-led Sunni states and Shia Iran, with these countries offering increased support to non-state armed actors mainly along sectarian lines. As Malmvig (2014, 146) suggests, "no longer just Sunni Arab leaders have allied with the West, who advance a sectarian narrative or invoke a Shia Crescent to legitimise Iran and its allies, but also civil society groups and popular religious leaders, even some who have been actively seeking to mend Sunni-Shia relations". More precisely, compared to the pre-Arab Uprisings order, the Sunni-Shia division appeared to have both deepened and broadened in scope.

Sectarian rivalry in the region involved three power centres: Iran and the Shia militias that supported Iran in Syria, Lebanon, and Iraq; Saudi Arabia and the GCC (plus Morocco and Jordan) and ISIS, the latest iteration of religious extremism in the region (Khoury 2013). However, the escalation of the sectarian divide was mainly accompanied by Iran's increasing influence in the region, which posed a threat for Saudi Arabia-led Sunni states. Iran maintained to enhance its regional power within and among Shia communities across the Middle East by supporting Shia uprisings in Bahrain and Yemen while endorsing the Alawaite regime in Syria (Khoury 2016b, 14). Tehran's pro-Assad stance put it on a collision course with the Sunni states regarding Syria. In Syria, Sunni-led pro-western states such as Saudi Arabia and Jordan supported the opposition, perceiving themselves as balancing the rise of Iranian and other Shia influences in the region. Since they viewed Iraq as a loss to Iran and the Shia camp, they thought that "they would gain an important advantage in the regional Shia-Sunni balance by turning Syria's Shia-affiliated Alawi regime into a Sunni one" (Aras and Akarçeşme 2012, 22). In this respect, Khoury (2016b, 14-15) addresses that as long as the uprisings continued to be unresolved, "the balance of power between Iran and Saudi Arabia will continue to fluctuate, along with the fortunes of the forces and factions they champion in the various proxy wars around them."

Thus, Riyadh viewed "Iran and Shia populations within or near its borders as explicit threats to continued stability" (Ennis and Momani 2013, 1128). Despite its domestic challenges, Saudi Arabia, hoped to expand its influence in the region because Riyadh was concerned that a decrease in its strategic regional influence would impact its affairs with the other Gulf countries. As such, the survival of the monarchical regimes emerged as a security interest, and so Riyadh developed an active foreign policy to maintain the regional status quo and prevent the increase of Iranian influence (Ennis and Momani 2013, 1128–30). In particular, the Saudi-Iranian rivalry extended military activism in the area. Saudi Arabia, for instance, supported NATO military interventions in Libya and Syria and Bahrain to maintain dynastic rule and impede the rise of Iranian influence across the region. Given its hostile policies against Tehran, Riyadh continued rally regional for an anti-Iranian axis, placing itself at the apex of such an alliance. In this regard, other Arab regimes, except for Doha, endorsed the recent actions of Saudi Arabia in Syria and Yemen against the rise of Iranian influence (Yorulmazlar and Turhan 2015).

These states with sectarian identities also sought to advance their regional interests by supporting non-state ideology-based armed groups, leading to a proxy war between Riyadh and Tehran. To illustrate, Riyadh propped up "proxy forces involved in armed struggle and, at least until late 2013, turned a blind eye to the recruitment of Saudis to join Al Qaeda- affiliated groups outside Saudi borders" (Norton 2015, 141); this was in addition to becoming a sponsor of Sunni armed groups in Syria. In Yemen, Iran collaborated with the Houthis, who fought against the ousted President Ali Abdullah Saleh, an ally of Saudi Arabia. Accordingly, the civil wars in Syria and Yemen aggravated the Riyadh-Tehran sectarian competition. Overall, as Gause (2014, 8) notes, sectarianism emerged as a weapon which changed the balance of power. Indeed, the Sunni-Shia competition between Riyadh and Tehran shaped the foreign policies of other regional and external actors following the uprisings.

4.1.3 The Role of Global Powers

The US and the EU established friendly relations with authoritarian regimes in the Middle East, ignoring undemocratic practises in these countries to protect Israel's security, maintain the flow of oil and energy supplies to the West, and limit the rise of Iranian influence. As a result, authoritarian regimes served the interests of Western powers in the region (Metawe 2013, 141). When the authoritarian regimes faced strong opposition in the region and began to fall, Western influence in the region began to wane. In this regard, other global powers (Russia and China) and regional powers (Turkey and Iran) attempted to fill the power gap. However, the US and the EU tried to maintain their regional influence despite their initial hesitation. By accelerating the anarchic environment in the region and upending the structure of the regional system, therefore, the Arab Uprisings carried the risk that regime shifts could transform an allied state into a neutral one or even an enemy of global state powers. Therefore, global powers hoped to manage the future of the uprisings to transform the developments of the uprisings based on their regional interests. In this regard, the US, the European Union (EU), Russia, and China were global powers who were critical of events and sought to shape the future of the uprisings. Following the 2008–2009 global economic recession and the failure of the prolonged conflicts in Afghanistan in 2001 and Iraq in 2003, the distribution of global power experienced a significant shift in which the US dominance in the international order declined, while a more multipolar world emerged. As such, the US lost military, economic and moral prestige and was in the cross hairs, as most young Arabs questioned the US presence in the region and saw the

US-led wars as but an extensions of age old western imperialism (Held and Ulrichsen 2014; Shabbir 2013, 32).

At the outset of the Arab uprisings, the US support for most Arab autocrats was the focus of much ire of Arabs protesting across the Middle East. The public in Egypt, Jordan, the Palestinian areas, and in Turkey criticised the US policies, and identified Washington as responsible for aiding and abetting autocratic regimes (Shabbir 2013, 33). In this regard, Kitchen (2012, 55) posits that "the pro-democracy movements have not themselves improved perceptions of the US, with views remaining profoundly negative, as they have been for a decade". Among these global actors, therefore, the US presence in the region was questionable in the changing regional balance of power in the wake of the uprisings. Shabbir (2013, 32) notes that the US presence in the Middle East was especially diminished and its influence across the region reduced since it had "less familiarity with the new actors in the region and [an] even less powerful position from which to influence those actors on the ground".

On the other hand, it was believed that the US maintained its dominant position in the region due to "the strong positions of its regional partners and the stagnation or decline of its rivals" (Juneau 2014, 40). In the wake of the Arab Uprisings, the US allies, including Saudi Arabia, Egypt, Turkey, and Israel, preserved their US-oriented policies since they needed the US as an external power to strengthen their policies in the wake of the uprisings. In essence, Washington pursued its own regional interest after the uprisings, putting aside its values regarding "national self-determination" since its application to Middle Eastern countries would lead to the rise of anti-US governments following the uprisings (Byman and Pollack 2011, 8). In an interview with the Atlantic, then the US President Barack Hussein Obama highlighted "American credibility" in the region and believed that "a president should place American soldiers at great risk in order to prevent humanitarian disasters, unless those disasters pose a direct security threat to the US" (Goldberg 2016). As such, cost-benefit calculations determined the policies of the Obama administration following the uprisings, and so long as any intervention was beneficial to the US interests, Washington opted to conduct an operation in the region.

Washington pursued developments which would create a more conducive environment for its strategic interests, and dealt with each uprising on a "case-by-case and ad hoc basis" (Ciplak 2014, 112; Yorulmazlar and Turhan 2015, 342). The US seemed to have a common understanding with its Sunni state allies about the rise of Iranian influence and at first opposed the Assad regime, supporting Sunni state policies. However, the empowerment of Sunni radical extremists like ISIS revived US tensions regarding the jihadist groups due to their anti-US and anti-western policies (Bazzi 2017; C. Phillips 2016, 466). Based on interest-driven policies, Washington was also concerned about the threat to energy supplies from the rise of international terrorism and jihadist groups (Lesch and Haas 2012, 7). To prevent this future aggravation to the US, Washington aimed to share the burden of, and responsibility for, any intervention during the uprisings with the EU and regional countries, as occurred with the Libyan intervention in 2011, under the 'responsibility doctrine'. Washington also then stayed passive on the Sunni-Shia division while its Sunni state allies sought a greater US role against Iran, changing its focus from Iran to the jihadist groups (Hachigian and Shorr 2013).

Despite its initial hesitation to intervening in the uprisings, Washington then withdrew their support from the oppressive regimes in Egypt and Libya, but ignored violent suppression by the authoritarian regimes in Bahrain and other Gulf States (Held and Ulrichsen 2014). Washington was also concerned for security reasons, on its own behalf and that of its allies, about the empowerment of Islamic parties. For this reason, it attempted to influence the newly elected parties in Egypt and Tunisia, hoping to encourage political reforms that would allow secular political systems, similar to Turkey's to emerge. Even so (?), Washington was accused of endorsing the military coup, despite Obama highlighting his concern over the military coup led by Sisi and the future of Egypt (J. Hudson 2013). As a result, the US role became a key determinant following the uprisings, affecting the policies of regional states and the developments of the uprisings.

The EU shared a common understanding with the US in terms of awaiting the outcomes of the uprisings. Having close economic relations with the authoritarian regimes, the EU aimed to maintain close trading ties in the region. The EU's policies towards the uprisings were further subjugated to security concerns regarding terrorism and illegal migration from the southern Mediterranean coast and the Arab world (Santini 2011, 285–86). In this regard, the EU's response to the Arab Uprisings was "limited and inspired by a conservative, rather than progressive, attitude to Euro-Mediterranean relations" (Bicchi 2014, 443). On this issue, Dias (2014, 59) unpacked the EU's policies at the beginning of the uprisings:

The EU's renewed approach southwards does not seem to differ significantly from the previous ones [...] When read carefully, the discourses and texts framing EU relations with the Southern Mediterranean since the Arab Uprisings, one realises that the EU is reproducing the same jargon and perpetrating the weaknesses of previous frameworks. Security concerns, including migration issues, remain at the top of the EU's agenda towards its southern neighbourhood, therefore reflecting the EU's interests and its vision on how the region should evolve.

As such, the EU had difficulty in articulating a coherent approach towards the Arab uprisings, opting instead to ameliorate challenges to their regional interests by adopting short term measures rather than supporting wholesale normative change. Following the overthrow of the MB-affiliated party in Egypt in 2013, for instance, the EU largely played a passive role due to its concern about the success of Islamic parties in the region, even though they seemed to support democratic movements in the Middle East, as did the US (Academic 3, personal communication July 9, 2020).

Since the previous authoritarian regimes, notably the Mubarak regime in Egypt, had mostly cooperated with Washington (T. Oğuzlu 2011, 5), China and Russia deemed the Arab Uprisings to be a diminishing of western influence in the region. They thus aspired to sway regional politics in favour of their own strategic interests, with both countries adopting policies aimed at helping to diminish Western, and particularly American influence still further (Dannreuther 2015, 80–93; J. D. Pollack 2011, 300). In particular, Moscow endeavoured to increase its influence in Syria after 2015 as a non-western global state power. As such, Bagdonas (2012, 55) argues that "for Russia, [the Syrian conflict] was primarily about the world order and Russia's place in it". Similarly, Trenin (2016, 27) states that Russia's intervention in Syria indicated the significance of military force in reaching its historic idea which was to "keep Russia in one piece and return it to its rightful place among the world's powers". Since 2015, therefore, Russia managed to reassert its influence in Syria which had waned following the end of the Cold War.

Based on its long-term pragmatic and overriding interests, Moscow aspired to maintain its alliance with the Assad regime in Syria and had concerns about the possibility of fall of the Assad regime due to its considerable interests, including a military base and economic and military relations with the regime. Notably, Moscow feared that "if a pro-Saudi regime replaced the Assad regime, Saudi influence would further spread and reach Muslims in the North Caucasus, as it happened in Chechnya" (Aras and Akarçeşme 2012, 22). In this regard, Moscow perceived Ankara and Tehran as short-term allies against the western camp, including the US. The three states temporarily cooperated to determine the future of the Syrian conflict through the Astana process in January 2017 to achieve a ceasefire between the Assad regime and Syrian opposition groups following the failure of the 2012 UN-backed Geneva talks to prevent escalation of the conflict.

Moreover, Russia started to play an active role to supersede the US, undercutting longstanding US relationships in the Middle East and restructuring the regional order on its own behalf. In this regard, Moscow attempted to establish economic and security ties with US allies such as Egypt. Russia and Egypt formed cordial relations and signed arms agreements, keeping a regular schedule of meetings throughout the process. In the wake of the 2013 military coup in Egypt, Russia also maintained ties with the new Egyptian government, while by contrast, the US compelled General Abdel Fattah al Sisi, the new Egyptian President, to make reforms and suspended aid (Goldenberg and Smith 2017).

Like Moscow, Beijing continued to adopt pragmatic policies towards political transformation, maintaining its economic and financial interests even though it had lost its long-standing relationship with many of the authoritarian governments in Tunisia, Egypt and Libya (Chang 2014, 177). "China's growing dependence on energy imports from the Middle East, its central role in the financing and development of major oil fields in the Persian Gulf, and the heightened investment of Chinese multinationals across the Middle East and North Africa" resulted in the continued reflection of China's strategic calculus toward the region (J. D. Pollack 2011, 298). In contrast to western efforts regarding human rights, democracy, and economic reform, therefore, Beijing aspired to forge close ties with undemocratic states like the Assad regime in Syria (Salman, Pieper, and Geeraerts 2015, 590). From the latter phase of the uprisings, therefore, the role of Moscow and Beijing in the region increased the questionability of the US presence in the Middle East. The US regional allies, including Ankara, looked for an alternative external power to achieve their regional interest following the uprisings. As a result, the role of global powers determined the sub-systemic policies from December 2010 to July 2018.

4.1.4 Empowerment of Non-State Armed Actors

Before the Arab Uprisings, non-state armed actors in the Middle East started to become empowered due to "the successive failures of states in the Arab-Israeli conflict, 9/11 attacks and lack of adequate participation in the decision-making process" (Shabbir 2013, 31). In fact, in some cases, they undertook more of a role than the states in the region. For instance, Hezbollah gained legendary status through its resistance to the Israeli invasion of Lebanon in July 2006, and which saw widespread approbation for the movement on the Arab street, much to the chagrin of some Arab Gulf states who saw Hezbollah as merely an extension of Iran (Kamal 2009, 98). The Arab Uprisings thus increased the already upended power structure and provided non-state armed actors to strengthen their regional aims and policies.

The transformation of states from being a security provider to a source of insecurity pushed sub-national ethnic and religious groups to pursue their security mechanisms. To illustrate, the Kurds, Turkmen, and Sunnis in Iraq sought to maintain their interests due to the weakness of the Iraqi state and the government's transformation into a source of insecurity (Yeşiltaş 2014, 31). As well as these groups, non-state armed actors conducted their own political, religious and territorial aims and improved their status through pragmatic ties and alignments with states in the absence of a central authority (Elhusseini 2015). More precisely, those failed states whose structure had collapsed or weakened experienced the empowerment of non-state armed groups.

Previous regional non-state armed groups like Hamas and Hezbollah arguably lagged behind recent political transformations, and their power decreased in the region. Although Hamas enjoyed warming relations with Turkey and Qatar and was able to consolidate its regional position during the first period of the uprisings, the Saudi Arabia-led Gulf countries considered Hamas to be part of the MB and thus came out against its policies. The overthrown of the Morsi regime in Egypt further scuttled Hamas's regional aims since it lost a possible state ally. Additionally, Hezbollah's support for the Assad regime in Syria and Shia administration in Iraq decreased its prestige (Aras 2014a, 40–41). On the other hand, ISIS and Kurdish armed groups like the PYD/YPG increased their influence and, in so doing, threatened the institutions, ideologies, and economic structures of sovereign states more than the former groups. These groups directly challenged the region's modern formal borders by threatening territorial boundaries established during the first quarter of the twentieth century (Lesch and Haas 2016, 184). More precisely, the unpredictable policies of these newly emerged non-state armed groups made external and regional states more concerned, while these groups benefited from the interstate competition, sectarian division, and political chaos (Yorulmazlar and Turhan 2015).

4.1.4.1 Islamic State of Iraq and Syria (ISIS)

A detrimental outcome of the Arab Uprisings was the emergence of ISIS (also called ISIL and DAESH), a jihadist group which was attempting to garner power in Syria and Iraq through the use of violence. Despite being widely known as an opponent of the Assad regime in Syria, the armed group's roots can be traced back to Al-Qaeda in Iraq (AQI), established by Abu Musab al-Zarqawi. Following Zarqawi's death in 2006, AQI established the Islamic State in Iraq (ISI) as an umbrella organisation. Under the leadership of Abu Bakr al-Baghdadi, the armed group sought to increase its power and influence. Initially, the group started to carry out attacks against non-Sunni targets in Iraq and, with the advent of the uprisings in Syria, the group then supported the Al-Nusra Front, the Syrian branch of Al-Qaeda, against the Assad regime. Al-Baghdadi also unified his forces in Syria and Iraq under the name ISIS and found support among those who had left Al-Qaeda. As a result, ISIS used extreme violence as an instrument in its efforts to implement its rigid Wahhabi interpretation of Islam and became an influential non-state armed group that increased its control rapidly in the region, with its anti-western discourse and intentions (J. Karakoc 2016, 1-5). A failed central state and sectarian strife in Iraq thus led to the empowerment of ISIS. After the toppling of the largely Sunni Ba'athist regime by the US-led operation, many Sunnis were marginalised and excluded from state service by the Shia government, an occurrence which ISIS exploited as evidence of the victimisation of Sunnis at the hands of Shias and the US (Işiksal 2017b, 92–94). In other words, the deteriorating sectarian balance between Sunni-Shia in Iraq provided ISIS with the chance to gain control. As such, the empowerment of ISIS was not a coincidence; rather, various regional developments gave rise to the empowerment of the armed group.

The indirect and latent support of Turkey, Qatar and Saudi Arabia for the Syrian opposition groups further resulted in the unintentional empowerment of ISIS. They initially provided financial and military support to opposition groups, without distinguishing between them, even though this included radical Islamist organisations like Al-Nusra and ISIS. Remarkably, Tehran blamed Riyadh for providing direct support to ISIS due to the higher participation rate of Saudi citizens in ISIS from Riyadh and the financial backing of wealthy

Saudi citizens (Işiksal 2017b, 98). This is because ISIS offered a virulently anti-Shia interpretation of Islam derived from Wahhabism. Therefore, Iran was the state most concerned about the rise of ISIS in its proximity. ISIS's military victories in Syria and Iraq threatened Iran's two closest regional allies —the Assad regime and the Shia Kurdish-dominated government in Iraq. The jihadist group also threatened the Iraqi central government, with its annexation, and caused the collapse or disintegration of Iraq. For Iran, the prospect of sectarian chaos and an armed and organised extremist Sunni entity on its western border was unacceptable (Lesch and Haas 2016, 185).

In essence, the empowerment of ISIS challenged both Sunni and western states' security and interests, as well as those of Tehran. With its radical Salafist ideology aimed at the establishment of an Islamic Sharia state under the rule of the Caliph, ISIS threatened the legitimacy of other Sunni states. For instance, an ISIS victory would have challenged the House of Saud's legitimacy, which is based on Wahhabism (Steinberg 2017, 35). Al-Baghdadi even proclaimed himself caliph, the spiritual leader of Islam, to legitimate ISIS's operations in the eyes of the Islamic world (Cockburn 2015, 28). The attacks of ISIS against the Kurdish region in northern Iraq, a reliable US ally in the region, were further perceived as a threat by the US, bringing the situation to "a point where it [was] no longer insisting on the overthrow of the Assad regime in Syria, much to the satisfaction of Iran" (Işiksal 2017b, 98; J. Karakoc 2016, 1–19). As such, ISIS emerged as a mutual security concern for regional and global state actors. Even, the rise of ISIS became another shared concern for Washington and Tehran following the fall of the Taliban in Afghanistan and the regime of Saddam Hussein in Iraq (Lesch and Haas 2016, 185–86). Therefore, with its radical ideology and violent policies, the empowerment of ISIS posed a security threat to most countries following the uprisings.

The violent attacks of ISIS changed regional state borders. ISIS had captured a third of Iraq's territory by August 2014 and half of Syria by May 2015 following the declaration of Raqqa as a capital of the caliphate in June 2014 (Shaheen 2015). The armed group thus gained more territory than any jihadist organisation in history, and became the wealthiest and deadliest terrorist organisation in history, possessing substantial financial reserves derived from oil smuggling, bank robberies, ransom money, and taxes, all of which enabled it to hire recruits (Işiksal 2017b, 92). Taking all of this into account, ISIS posed a significant challenge to the regional status quo. Its empowerment especially posed a threat to Iraq and Syria's neighbours

because it aimed to enhance its territory and financial capability, bringing sectarian, territorial, and humanitarian issues.

To hamper the empowerment of ISIS, regional and global states have conducted international military operations against this armed group since August 2014. However, the states fighting against ISIS exhibited different geopolitical ambitions. To illustrate, in September 2015, Russia asserted that its military intervention in Syria aimed to prevent the spread of ISIS and other radical Islamist groups. By this intervention, Russia aimed to become an alternative determinant (dominant?) power in the Middle East, supporting its close allies and challenging US power in Syria (BBC News 2015; Hubbard et al. 2019). Moreover, the US endorsed its regional partners in the fighting against ISIS, both to defend its interests and to prevent a possible jihadist threat against itself and its allies (Esfandiary and Tabatabai 2017, 455). To this end, from August 2014 the US and its allies conducted thousands of military strikes to weaken ISIS. As Perra (2016, 371–72) notes, "the global fight against the ISIS has indeed opened the Pandora box providing opportunities for everyone to advance larger strategic plans aimed at either maintaining the status quo or at prompting regimes to change". Ultimately, the empowerment of ISIS affected the regional balance and states' policies in the wake of the Arab Uprisings by July 2018.

4.1.4.2 Kurdish Armed Groups

Like ISIS, Kurdish armed groups have reinforced their position since the Arab uprisings. With the outbreak of the Syrian conflict in March 2011 and the empowerment of ISIS in 2014, armed Kurds attracted unprecedented attention from international and regional state actors and gained opportunities to advance their goals. Particularly in Syria and Iraq, these groups took advantage of weakened central governments, and there were two main political effects in this regard. The first was their "determined resistance against the expansion of extremist Islamism across the region", and the second their "pro-modern outlook", which allowed them to gain sympathy from the international community in their struggle against ISIS (Sabio 2015, 332). Accordingly, the armed Kurds engaged in combat with ISIS in the Siege of Kobane (September 2014 to March 2015) and the Shingal battles (2014–2015), where the empowerment of the armed Kurds became more visible. Karakoc (2016, 18–19) argues that "Whereas the decades-old Kurdish struggle in different countries in the Middle East previously attracted little attention around the world, after ISIS attacked Kobane, the resistance and fight put up by the Kurds made them prominent actors for Western countries almost overnight".

Bearing in mind that most Kurdish armed groups looked to enhance their claims to national self-determination, they aimed to change the regional status quo (Aras 2014a, 40). In particular, the Syrian conflict created a knock-on effect in those states with large Kurdish populations. As such, the anarchic regional environment pushed these groups to seek "various degrees of independence, autonomy, or self-rule that they would not have otherwise" (Ünver 2016, 65). In this regard, Iraqi Kurdish groups sought to consolidate their semi-state status through the Kurdistan Regional Government's (KRG) de facto independence in 2017. In the north of Syria, in 2016, the Syrian Kurds also declared an autonomous region called Rojava that emerged as a potential game-changer in the area. According to Gunter (2015, 102), "the precipitous rise of the Kurds in Syria could become a factor in changing the artificial borders of the Middle East established after World War I by the notorious Sykes-Picot Agreement".

Utilising the conflicts, these armed groups further increased their external support, with the US and Israel notably seeming to endorse them against jihadist groups. While the US backed them against ISIS, Israel liaised with these groups against the empowerment of Al Qaeda-backed Sunni extremists, and Iranian-backed Shia forces (Kavak 2017, 203). Even the Israeli Prime Minister, Benjamin Netanyahu highlighted Israel's support for the establishment of a Kurdish state, claiming that "creation of Kurdish state would aid in formation of alliance of moderate powers in Middle East" (The Guardian 2014). Due to its hostile policies against Iran and the Assad regime, Riyadh also began to establish ties with the PYD through financial backing. In May 2018, it was claimed that Saudi Arabia, along with the UAE and Jordan, were aiming to cooperate with the PYD to establish Arab military units led by Al-Sanadid forces, which are part of the PYD-dominated Syrian Democratic forces (Middle East Monitor 2018; Yeni Şafak 2018). As well as Iran and the Assad regime, this collaboration would also pose a threat for Turkey, even though in 2016 Ankara and Riyadh had previously cooperated against ISIS and the PYD (Sharma 2016; The Guardian 2016). Therefore, the regional aims of states contributed the empowerment of Kurdish armed groups.

Nevertheless, the empowerment of these groups posed a security threat to certain regional states, including Turkey and Iran, due to their long-term Kurdish minority issues. for this reason, these states sought to unite against the rise of armed Kurdish groups in the region, while also changing their choice of alliance (discussed in Chapter 5). Ultimately, the empowerment of armed Kurdish groups was an influential factor in changing the regional balance following the Arab Uprisings. In particular, the revisionist policies of the armed Kurdish

generally influenced states' alliance choices as these states pursued their own interests. Given the significant factors shaping the developments following the Arab Uprisings, the next section sheds light on the role of these determinants regarding Turkey's relative power position and TFP towards Iran.

4.2 The Role of the System-Level on Turkish Foreign Policy towards Iran since the Arab Uprisings, from December 2010–July 2018

According to NCR, a state's relative power position in the international system determines its national behaviour in the long term (Zakaria 1999, 482). NCR is further concerned with the impact of the international sphere upon the state's foreign policy, arguing that "over the long run a state's foreign policy cannot transcend the limits and opportunities thrown up by the international environment" (Rose 1998, 151). In this regard, Turkey's power position determines its foreign policies. Having explained the sub-systemic changes after the Arab Uprisings, this section contains an analysis of how Turkey fitted into the international system as a regional middle power and followed a foreign policy approach following the uprisings. This section thus includes an examination of the role of the system-level in Turkish foreign policy towards Iran since the uprisings, from December 2010–July 2018.

The concept of a "middle power" is so debatable that there is no entirely satisfactory definition of it.¹ Nevertheless, the traditional definition of a middle power has included those states occupying "a middle-level position in the international power spectrum, just below superpowers or great powers", with "military capabilities, economic strength, and geostrategic position, leadership capacity, and impact and legitimacy in the international arena" (Yilmaz 2017). Although they are mainly affected by the affairs of the great powers in international politics, middle powers may declare themselves to be "the local great power", with or without

¹ See Details: Holbraad, C. (1984). *Middle powers in international politics*. Springer.; Cooper, A. F., Higgott, R. A., & Nossal, K. R. (1993). *Relocating middle powers: Australia and Canada in a changing world order* (Vol. 6). UBC Press.; Challenging Contemporary Notions of Middle Power Influence: Implications of the Proliferation Security Initiative for "Middle Power Theory"; Chapnick, A. (1999). The middle power. *Canadian Foreign Policy Journal*, *7*(2), 73-82.; Jordaan, E. (2003). The concept of a middle power in international relations: distinguishing between emerging and traditional middle powers. *Politikon*, *30*(1), 165-181.; Shabbir, M. (2013). Emerging Middle East: Interplay of the New Power Centers. *ISSRA PAPERS*, 25.; Patience, A. (2014). Imagining middle powers. *Australian Journal of International Affairs*, *68*(2), 210-224.; Jordaan, E. (2017). The emerging middle power concept: Time to say goodbye? *South African Journal of International Affairs*, *24*(3), 395-412.; Robertson, J. (2017). Middle-power definitions: confusion reigns supreme. *Australian Journal of International Affairs*, *71*(4), 355-370.

the support of great powers or small states in their proximity. In the regions where they are located, middle powers may follow different policies to the great powers, based on their own power potential and capabilities. Therefore, they can be defined as a regional power, which "has the potential to balance other forces, maintain codes of conduct, stabilise spheres of influences and police unruly states" (Müftüler and Yüksel 1997, 186–88).

Since the end of the Cold War, Turkey's power capacity, due to its population, military and economic strengths, and geostrategic position, has been conceptualised as a middle power.² Since coming to power in 2002, the AKP government has aimed to make Turkey a 'strategic country' by increasing Turkey's middle power capability. Between 2002 and 2011, Turkey demonstrated significant economic growth following the 2001 global financial crisis and increased its trade volume nearly fourfold in ten years (Bagdonas 2015, 315; Öniş 2014). In 2008, Turkey's status in the International Monetary Fund (IMF) was also upgraded from a 'debtor' to a 'creditor' and Turkey also became an active member of Group of 20 (G20) involving the world's major economic states. As well as economic growth, Turkey enjoyed political and social development. Notably, its desire for EU membership prompted several reforms, including "improvements in its human rights regime, freedom of expression, freedom of assembly, gender equality and minority rights" (Öniş and Kutlay 2017, 171–73). As such, the EU membership process accelerated democratic reforms in Turkey by the uprisings-term. In terms of military capacity, Turkey had the second-largest army in NATO at 495,000

² See Details: Hale, W. (1992). Turkey, the Middle East and the Gulf crisis. *International Affairs*, 68(4), 679-692.; Müftüler, M., & Yüksel, M. (1997). Turkey: a middle power in the new order. In *Niche Diplomacy* (pp. 184-196). Palgrave Macmillan, London.; Hale, W. (2012). *Turkish foreign policy since 1774*. Routledge.; Kardaş, Ş. (2013). Turkey: A regional power facing a changing international system. *Turkish studies*, *14*(4), 637-660.; Cagaptay, S. (2013). Defining Turkish power: Turkey as a rising power embedded in the Western international system. *Turkish Studies*, *14*(4), 797-811.; Sandal, N. A. (2014). Middle powerhood as a legitimation strategy in the developing world: The cases of Brazil and Turkey. *International Politics*, *51*(6), 693-708.; Bagdonas, Ą. (2015). Turkey as a great power? Back to reality. *Turkish studies*, *16*(3), 310-331.; Ongur, H. Ö., & Zengin, H. (2016). Transforming habitus of the foreign policy: A Bourdieusian analysis of Turkey as an emerging middle power. *Rising Powers Quarterly*, *1*(2), 117-133; Parlar Dal, E. (2016). Conceptualising and testing the 'emerging regional power' of Turkey in the shifting international order. *Third World Quarterly*, *37*(8), 1425-1453.; Öniş, Z., & Kutlay, M. (2017). The dynamics of emerging middle-power influence in regional and global governance: the paradoxical case of Turkey. *Australian Journal of International Affairs*, *71*(2), 164-183.; Süsler, B. (2019). Turkey: An Emerging Middle Power in a Changing World? *LSE Ideas Strategic Update*, *May*.
members of the Turkish Armed Forces, and significant developments were made by joining international NATO military operations (Daily Sabah 2012).

Similarly, Turkey enjoyed high-level bilateral ties with various states, including western and Middle Eastern states. Based on its geostrategic location, historical roots, and religious and cultural bonds, Turkey under the AKP government continued to seek a more proactive role to improve relations with other Middle Eastern countries by signing numerous political and cultural agreements as part of the policies of 'zero-problems-with-neighbours', and 'economic interdependency' (Ennis and Momani 2013, 1128; Işiksal 2017a, 211). As a result, Turkey's regional policies toward several states, such as Iran and Syria, entered into a period of détente, although they had been intended to resolve areas of tension in bilateral ties (Ennis and Momani 2013, 1129).

According to Sandal (2014, 694) "middle powers are seen as multilateralists that actively participate in institution building, peacekeeping operations and humanitarian mission". In this regard, Turkey further sought to mediate various disputes in the Middle East, including between Israel and Syria in 2008–09, Israel and Hamas in 2009, and Iran and the P5+1 (the US, UK, France, China, Russia, and Germany) in 2010. Ankara also hoped to play a significant role in various international and regional issues through its temporary role as a UNSC member in 2009–10, and its observer status in the Arab League in 2006. Turkey further pursued humanitarian diplomacy to "increase Turkey's assistance to developing nations and [respond] to major humanitarian crises around the world" (Susler 2019, 10). As a result of its humanitarian diplomacy, Ankara's overall amount of humanitarian assistance reached \$7.5 billion from 2004 to 2012 (Haşimi 2014, 134). Accordingly, until the Arab Uprisings, Turkey was perceived as an influential regional power due to its increasing power capabilities, and Turkey's relative power position increased between 2002–2011.

Given the rise of its power capability and regional influence in the pre-uprisings period, Turkey initially hoped to maintain its power status in the region, and thus faced "a dilemma between ethics and interests" (Öniş 2012). In particular, Turkey's previous pragmatic economic and political gains with the authoritarian regimes prevented Ankara from responding immediately to the uprisings. In the case of Libya, Turkey first objected to external intervention, stating that military intervention would increase chaos by damaging politics in the region, even though it ultimately recognised the NATO intervention by offering active support. In doing so, Turkey first attempted to maintain its previous interests in Libya, with whom Turkey had close trade relations. Before the Arab Uprisings, Libya had been a significant economic partner for Turkey due to its "source of crude oil as well as for the construction contracts of Turkish businesses, which amounted to approximately \$20 billion" (Altunişik and Martin 2011, 575). In the case of Syria, Turkey also demonstrated a similar attitude during the first few months of the conflict by opposing international intervention "on behalf of the protesters" and recommending that "any reforms be initiated and controlled by the Bashar al-Assad regime" (Lesch and Haas 2012, 153). Therefore, Ankara opted to maintain the status quo in the region to a certain extent, based on its economic and political gains with the authoritarian regimes.

Because maintaining ties with the regimes started to be costly for Turkish interests, Ankara abandoned its previous ties and declared support for pro-democracy movements. If Turkey had maintained its close ties with the regimes, Ankara's prestige would have been damaged, and Turkey's regional influence might have suffered both at the time and in the future context of the uprisings. In essence, Turkey's leading role was in jeopardy since other regional and international actors had, by now, begun to interfere in the regional conflicts. While Turkey was following a wait-and-see policy, for instance, France and Britain were adopting an interventionist policy in Libya (Yeşilyurt 2017, 69). Oguzlu (2012, 5) adds that "appearing too supportive of Assad might have cost Turkey a critical role to play in a post-Assad era", and so Turkey endorsed the opposition groups. Based on interest-driven calculations, Turkey opted to promote democracy following the uprisings to protect its regional goals in its immediate environment and reinforce its status as a regional power.

Turkey thus recalibrated its policies towards the uprisings and followed revisionist policies by supporting the uprisings. These volte-face policies reflected its cost-benefit calculations and demonstrated its interest-based approach towards the uprisings. As a corollary of Turkey's volte-face policy, Ankara's zero-problems-with- neighbours policy lost its validity in Turkey's ties with its authoritarian neighbours, including Syria, Libya, and Egypt. In Dalacoura's (2012, 76) words, Turkey's zero-problems-with-neighbours policy "was thrown [up] in the air as Arab populations challenged their governments, leaving Turkey searching for a coherent response".

According to Onis and Kutlay (2013, 1412), "economic interdependence" and "prodemocracy rhetoric" became two significant pillars of TFP at the beginning of the uprisings. Aras and Yorulmazlar (2014, 114) added "humanitarian protection, security, and regional diplomacy" as three pillars determining Turkey's involvement in the Arab uprisings. However, these soft power tools served Turkey's regional interests in that Ankara sought to use liberal values to improve its relative power position. In the case of Bahrain in February 2011, for instance, Turkey remained silent and avoided criticising the repression of protests by the Gulf monarchies (led by Saudi Arabia), to avoid jeopardising its strong economic ties in particular Saudi Arabia, Qatar, and the UAE (Elik 2018, 107). Turkey further opposed the autonomy of the Syrian Kurds in the north of Syria due to its national security concerns and conducted several military operations against the PYD/YPG to prevent their empowerment in its proximity since 2013. Therefore, Turkey's policies towards Bahrain and the Kurds in the north of Syria fell contrary to its principle of promoting democracy. As a result, as Stein (2014, 36-57) describes, Turkey's changing policies towards the uprisings in each country "cannot be described as an effort to promote democracy. Instead, it has been far more nuanced, based on assumptions made about a changing regional order and how the upending of the Arab world's political status quo would benefit Turkey". This shows that Turkey conducted interest-based foreign policies based on cost-benefit calculations.

Given the positive image it held before the uprisings, Ankara pursued a leadership role and promoted itself as a model in the region. Turkey's efficacious economic success and regional political prestige allowed it to propagate a positive image during the uprisings (Ennis and Momani 2013, 1128). It mostly attempted to become a 'central country' by adopting a strong and significant position in regional systems, and also intended to become a world power in the long term (Aras 2010). As such, Turkey perceived itself as an influential model, presenting an excellent example of a democratic Muslim country in the Middle East. Although Turkey seemed to balance idealism and realism (Karacasulu 2015, 34), TFP exhibited more realist aspects based on national interests.

At the beginning of the uprisings, Ankara further sought to consolidate its power position by cooperating with global powers and bringing its foreign policy in line with those of the EU and the US. In doing so, Ankara hoped to empower its internal and external roles and eliminate potential opposition from Brussels and Washington towards its political activism in the Middle East (Ciplak 2014, 93). The Obama administration encouraged Turkey to assert

itself as an ideological model throughout the Islamic world, with a view to protecting US interests through the spread of Turkey's "liberal Islamic system" in the political realities of the era. Washington reasoned that "If Turkey [did] not provide the ideological inspiration in domestically vulnerable regimes, more radical, and more anti-American, Islamist groups might emerge" (Lesch and Haas 2012, 166). As well as the US, the EU supported the Turkish model, putting aside its concerns regarding the growing Islamisation of TFP (Tocci et al. 2011, 9). Despite its Islamic aspects, the Turkish model overlapped with EU principles, and this support encouraged Turkey to follow a more active foreign policy in the region. As a result, potential mutual benefit paved the way for cooperation between Ankara and its Western allies, the EU and the US. Turkey initially appeared to be more willing to seek a common understanding with western states and with the US in particular due to its lack of ability to affect regional developments and undertake a leadership role (Murinson 2012, 12). This is because, as Demir (2017, 3) explains, "the efficacy of Turkey's foreign policy as a middle power depends on the quality of both relationships between the great powers and Turkey's own relationships with them". After a brief hesitation, Turkey cooperated with NATO to topple the Libyan leader Qaddafi and had common aims with the West in attempting to topple the Assad regime following the uprisings.

On the other hand, Zakaria (2011, 38) argues "that rising powers should no longer be seen as having to choose between integrating into or rejecting the western order. On the contrary, rising powers today seem to be entering the western order but doing so on their own terms —thus reshaping the system itself". From the beginning of the Arab Uprisings until mid-2012, both systemic pressures and the regional environment were permissive. To respond to these pressures, Turkey adopted two policies: supporting the transformation by seeking a mediator role, and direct intervention in the uprisings when transformation seemed difficult (Expert 3, personal communication, July 21, 2020). The power vacuum, in which the regional roles of the US and the EU were lacking, Russia did not yet intervene in the uprisings, and some Arab countries were dealing with domestic issues, provided Turkey with an opportunity to enhance its power position and regional interests, by constituting a new regional order and cementing its international status (Karacasulu 2015, 34). Therefore, TFP demonstrated revisionist aspects by following assertive and active policies in the region.

Turkey viewed the Arab Uprisings as an historic opportunity to further increase economic and political cooperation with new governments in the region, especially those with which Turkey had more limited relations (Işiksal 2017a, 213). After the removal of the Ben Ali regime, for instance, Turkey attempted to develop political and economic ties with Tunisia (Altunışık 2014, 345). Notably, pro-MB policies emerged as being of "strategic importance" for Turkey's future policies in the regional politics because MB-affiliated governments would be allies supporting Turkey's influence in the region (Süsler 2020, 166), thereby increasing Turkey's relative power position. Ankara thus started to support the opposition groups, including MB-affiliated parties/groups, by welcoming the overthrow of the Mubarak regime in Egypt and attempting to establish close economic and political ties with the new MB-affiliated parties government (Lesch and Haas 2012, 153). Turkey's pro-democracy policies thus served Ankara's interest in Egypt, Libya, and Syria in where Turkey supported MB-affiliated parties and groups. According to Expert 3 (personal communication, July 21, 2020), Turkey became successful and earned economic and political benefits following the uprisings. In Libya, for instance, Turkey gained economic benefits and in Egypt political ones. Egypt and Turkey put aside their competition and formed a political alliance in the region, and Turkey's economic gains doubled in contrast to pre-uprisings term.

Bearing in mind that the influence of the great powers had decreased to a certain extent, the regional middle powers were attempting to fill the resulting power gap. Turkey thus entered into competition with Iran and Saudi Arabia to promote its political tendencies in the future of the uprisings. Riyadh deemed Turkey's growing role as a threat to its position as the leader of the Sunnis because of Turkey's backing of the MB-affiliated parties (Roy 2012, 17). Tunisia's Al-Nahda leader Rachid Ghannouchi, and Saad el-Katatni, the head of the Egyptian MB's new party Freedom and Justice, viewed the Turkish AKP government as a useful Islamic model (Dalacoura 2012, 77). Similarly, Iran was concerned about the Turkish model in the region since, as Murinson (2012) notes, "Turkey's emerging regional leadership role offered the Arab Islamists a Sunni alternative to Iran". There are two main reasons why the Turkish model was more attractive to the new governments. First, compared to the religious Saudi and Iranian models, the Turkish model offered Islamic liberalism involving "a marriage of formal democracy, free-market capitalism and (a toned-down) conservative Islam" (Tugal 2016, 4). Second, while the Saudi and Iranian models combined "natural resources and rentier politics to achieve economic growth", the Turkish model raised its "economic productivity through the combination of and educated Turkish society and strong manufacturing" (Salem 2011, 8). Therefore, the Turkish model was viewed as more applicable in the public opinion of the Arab uprising countries (Ennis and Momani 2013, 1129).

Turkey and Iran vindicated neoclassical realist expectations which argue that states, as influence maximisers, aim to expand their power over their external environment and utilise this power to achieve better opportunities to better affect the regional order (Juneau 2015, 104). The two non-Arab actors attempted to play a leading role in the fate of the Arab Uprisings, and hoped to become a model country in the post-Arab Uprisings era (Legrenzi 2015). Like Turkey, Tehran followed pragmatist and interest-based policies following the Arab Uprisings and perceived the uprisings as an opportunity to enhance its regional influence to a certain extent, assuming that Iran might be a good model since its 1979 Islamic revolution had toppled the west-supported authoritarian Shah regime (Academic 3, personal communication, July 9, 2020). This competition over being the best 'model' is reflected in the two states' perceptions of the uprisings. Turkey viewed the uprisings as an internal dynamic that resulted from undemocratic and corrupt regimes. For instance, then Foreign Minister Ahmet Davutoglu pointed out that,

At that time, the question of what we attempted to find out an answer was: What it the ordinary course of history? Who stands on the right side of history? Are the protesters in Tunisia or Ben Ali? Is Mubarak or Youngs in Tahrir Square? These kinds of questions came out. We decided that this is a natural order and there is no conspiracy. The Arab Spring was started by the young Arab generation which should be respected (Uzun 2013, 149).

Under this self-perception, Besir Atalay, the former Turkish Deputy Prime Minister, further pointed out that Turkey was ready to share its experience with the affected countries, to become part of the democratic transformation in the region (Uzun 2013, 149). However, Iran saw foreign interference behind the uprisings. Iran's former president Mahmoud Ahmadinejad, claimed, that:

We are going to make greater efforts to encourage both the government of Syria and the other side, all parties to reach an understanding. But I think and we believe that there should be no interference from outside. The positions of the United States are not going to help. They have never helped. They could do things better in Libya, for example. From the beginning, we said there should be an international team to mediate in order to encourage all parties to reach an understanding. But NATO had ambitions in Libya. They wanted the oil resources in Libya. There was no need to kill so many people. This is the situation in Syria, too (Uzun 2013, 151).

In this tumultuous environment, Iran also emerged as another regional non-Arab power by using the emerging power vacuum in Iraq, as well as its long-term support for President Bashar al-Assad, to maintain and expand its presence across what some referred to as the Shia crescent (Khoury 2016b, 14). Iran now gained more opportunity than Turkey to raise its level of influence following the decreased US presence in Iraq and the 2013 draft nuclear deal, since the balance between Sunni and Shia states changed in favour of Iran (Academic 4, personal communication, July 6, 2020). In other words, the declining of the US influence in the region fell completely in line with Tehran's wishes. Turkey and Iran thus competed to fill the regional power vacuum arising from Arab fragmentation and turbulence, while most Arab actors were dealing with domestic issues, and the roles of the great powers were limited. The two states emerged as rival ideological models for those countries which experienced the Arab Uprisings, and their ideological differences caused sectarian rivalry (Academic 2, personal communication, August 6, 2020). According to Expert 2 (personal communication, August 19, 2020), Turkey was indeed disturbed by the rise of Shia Iranian influence in Syria and Iraq. The sectarian rivalry between Ankara and Tehran thus led to collaboration between Riyadh and Ankara over Bahrain and Syria at the beginning of the uprisings. To prevent the rise of Iranian influence, Turkey supported Saudi Arabia against Iran in the proxy Sunni-Shia war between Tehran and Riyadh and agreed with the Saudi objective of restricting Iranian Shia influence in the Persian Gulf. In the Yemeni uprising of 2011, Turkey supported the Saudi campaign and sought to make economic and military agreements with pro-Saudi groups, as well as Bahrain and Syria (Elik 2018, 107).

The divergent policies of Iran and Turkey expanded to Syria, where the states supported opposite ideological sides in the war. Turkey deemed the Syrian conflict to be about the democratic demands of Syrians and was therefore more critical of the Alawaite Assad regime. Turkey was at loggerheads with Iran due to its support for the Syrian Sunni opposition groups. On this issue, Nuray Mert, a Turkish journalist and political scientist, discussed the competition between Turkey and Iran in Syria:

If there is going to be a regime change in Syria, the whole power balance will change. If Iran loses Syria, they will lose a significant base of power in the Middle East. So, it will be a significant defeat for Iran, and within this framework, Turkey sides with the dissidents and supports some sort of regime change. Iran will take it directly against itself (Uzun 2013, 158). As a result, the Arab Uprisings disrupted the balance between Ankara and Tehran, bringing their conflicting interests, visions, and projects to the fore (Lesch and Haas 2012, 8; Uzun 2013, 158). Turkey perceived the rise of Iranian influence as a threat, since it might lead to a decrease in Turkish influence (Academic 3, personal communication, July 9, 2020). As a regional middle power, Turkey thus aspired to enhance its relative power position and regional interests in the wake of the uprisings to cope with the rise of Iranian influence. TFP towards Iran demonstrated competitive and revisionist characteristics when their interests overlapped. Expert 2 (personal communication, August 19, 2020) stated that if Turkey fails to balance Iran in the region, Iran may become Turkey's greatest enemy because its relative power would increase if Iran cooperated with US and Gulf-countries. This demonstrates that systemic variables shaped TFP towards Iran, and TFP towards Iran was influenced by Turkey's relations with international actors, including the US. Given its interest-driven calculations, Ankara pursued collaboration with other state actors, such as Saudi Arabia and the US, to hamper the Iranian influence to a certain extent.

4.3 Concluding Remarks

The Arab Uprisings not only changed the domestic politics of the affected countries, but they also led to regional reassessments of political systems, national interests, alliances, and rivalries. The uprisings accelerated the already anarchic environment of inter-state competition, and the rise of non-state armed groups. The above table indicates the precise link between the AUs as a change of executive powers in several countries and spill-over effects in the shape of changes of direction in the foreign policies of other countries.

Saudi Arabia and the United Arab Emirates	 Both states saw minor protests but suppressed them. They aided Bahrain's Sunni regime in putting down demonstrations by the Shiite Muslim majority. In Egypt, they perceived the MB-related government as a threat and supported the military coup.
	 Saudi-led coalition supported the Yemeni regime.
	• However, they supported Sunni rebellion groups against the Allawi Assad regime in Syria.

	• In Libya, they endorsed the eastern- based government.
Iran	• Supported the rebellions in Bahrain and Yemen but aided the Assad regime in Syria.
Turkey	• Supported the MB-affiliated parties, Syrian opposition, and the western- based government in Libya.
Global State Powers	 NATO conducted air strikes against the Qaddafi regime in Libya. In Egypt, the US and the EU played a passive role during the military coup attempt. The US and the EU supported the Syrian opposition before shifting their focus to combating the ISIS. Russia has backed the Assad regime in Syria since 2015 and supported the eastern-based government in Libya.

 Table 4-2 The Arab Uprisings and reginal and global state actors from 2011 to 2018

In such an environment, the role of Islamic parties, the Sunni-Shia division and the competition in the intra-Sunni group, the role of great powers and the empowerment of nonstate armed groups determined Turkey's affairs with Iran. As two regional middle powers, Turkey and Iran followed interventionist and revisionist policies after the Arab Uprisings, hoping to utilise the political vagueness to enhance their interests; they thus welcomed the uprisings as movements that would transform the Arab world under the principles and ideologies they respectively embraced (Işiksal 2017a, 211; Yorulmazlar and Turhan 2015, 343). As such, Turkey perceived Iranian policies as a threat to its power position and was concerned about the rise of Iranian influence in the region. In this regard, Ankara collaborated with Saudi Arabia to prevent the rise of Iranian influence, notably in Syria, Yemen, and Bahrain, and further sought US support to for its regional policies. In Syria, Turkey initially opted for collaboration with NATO while supporting NATO operations in Libya. Although the regional dynamics that stemmed from the Arab Uprisings and its groundbreaking effects in the region were influential in understanding the evolution of TFP towards Iran, however, these dynamics alone cannot provide a comprehensive understanding unless Turkey's domestic variables are examined. The argument made throughout this thesis is that external structural changes primarily induced a change in TFP towards Iran in the long term. However, the thesis maintains that unit-level dynamics determined TFP towards Iran from December 2010 to July 2018, which is a short-to-medium term.

Based on unit-level variables in short-term foreign policy, Turkey opted for pragmatist and interest-driven collaboration with Iran. Therefore, the next chapter will shed light on the AKP elite's perception and Turkey's domestic considerations, including economic and security ones, to examine TFP towards Iran since the Arab Uprisings. The next chapter contains an examination of how the AKP failed to grasp the systemic changes and conducted foreign policies above Turkey's middle power capacity, and how Schweller's balance of interest theory explains Turkey's short-term alliance choice with Iran between December 2010 and July 2018.

CHAPTER 5 THE UNIT LEVEL — INTERVENING VARIABLES ON TURKISH FOREIGN POLICY TOWARDS IRAN SINCE THE ARAB UPRISINGS (DECEMBER 2010-JULY 2018)

According to NCR, even though foreign policy outcomes reflect the distribution of power in the long term, they may not follow the imperatives of power distribution renders in the short to medium term. More precisely, in short to medium term, state foreign policy may not harmonise with state material capabilities since domestic unit-level factors play a determinant role in shaping state foreign policy (Rose 1998, 146; Schweller 2004, 6). These unit-level variables provide a tool to examine the domestic processes of states, which "arrive at policies and decide on actions" in response to systemic forces (Juneau 2010, 2).

In this chapter, further analysis is made of the unit-level variables which filtered the systemic pressures of the Arab Uprisings and Turkey's relative power position and shaped the outcome of TFP towards Iran. Following the period of the uprisings, Turkey specifically shaped its policy responses towards Iran on the basis of certain domestic factors and the changing regional structure and balance of power. This chapter sheds light on how the intervening variables prevented Turkey from entering a conflict with Iran and resulted in an alliance between the two neighbours. Thus, the chapter presents the response to the second research question concerning "*how unit-level factors interacted with structural imperatives in shaping TFP towards Iran from December 2010 to July 2018*".

The chapter is divided into three sections: an identification of the AKP elite's perception in TFP towards Iran; an examination of Turkish domestic security and economic considerations on TFP choices towards Iran; and conclusion remarks of the findings. In particular, the chapter highlights the role of the AKP elite's (mis)perception of the systemic changes and Turkey's middle power capabilities, and how Schweller's balance of interest approach can be used to view Turkey's choice of alliance with Iran.

5.1 The AKP Elite's Perception

In NCR, there is an emphasis on politicians as the pivotal actors in international affairs, since their perception of shifts in power influences a state's foreign policy behaviour (Wohlforth 1993, 6; Zakaria 1999, 42). Notably, a change in the perceptions of the state's leaders or other policymakers is more rapid than shifts in the state's material capabilities. As such, state foreign policy behaviour can be examined through the perceptions of its foreign policymakers (Schweller 2003, 39). This suggests that over the short to the medium term, state foreign policy is influenced by foreign policymakers' perceptions more than the state's objective material power (Rose 1998, 147).

A state's foreign policy is a process of three stages, including "strategic assessment", "strategy formulation" and "implementation of the strategy". The strategic assessment phase comprises the perception of the state's decision-makers dominant role (Tang 2009). To understand a state's foreign policy behaviour, it is essential to examine in detail foreign policymakers' understanding of a state's situation in international affairs (Rose 1998, 158). Schweller (2004, 5) argues that "states respond (or not) to threats and opportunities in ways determined by both internal and external considerations of policy elites, who must reach a consensus within an often decentralised and competitive political process".

According to Barkin (2009, 245), "the need for prescription underlines the possibility that prediction might fail. This is the case in as much as it involves having to tell decision makers to do what they have been predicted to do or having to warn them not to do what it has been predicted they will not do anyway. To the extent therefore that we expect that prescription might work, we must accept that prediction might fail". Therefore, foreign policy makers have difficulty in predicting international developments or foreign policies of other states. This also implies that a variety of factors, such as beliefs or perceptions, domestic interest groups, may influence their forecasting of international developments or foreign policies of other states (Kaarbo 2015, 16). Furthermore, decision makers' choices may be influenced by the external behaviour of other states owing to the absence of an international system mechanism to control all foreign policy choices. As a result, the logical goal of national security decisions can sometimes result in irrational outcomes (Schmidt 2008, 190).

Bearing in mind that "psychological, societal, ideational, political, institutional and material" factors influenced state decisionmakers' responses to domestic and international environment (Darwich and Kaarbo 2020, 227), identity shapes the system because states tend to collaborate when they share common ideas on principles of governance. Since political elites shape state foreign policy, they usually seek to mobilise or construct identities to promote specific interests (Haas 2014). More precisely, identity becomes an instrument for the pursuit of interest, and so the ideology of the ruling elite may inform both the perceptions of relative power and the adaptability of potential policy, and influence the mobilisation of state power (Lobell 2009, 21).

Turkey's national security executive was comprised of the head of government, ministers, and officials, and together they were responsible for state foreign policy, from December 2010 to July 2018 (until the official declaration of Presidency in Turkey). Accordingly, the significant key players in shaping Turkish foreign policy have been Recep Tayyip Erdogan (as Prime Minister between 2003 and 2014, and President between 2014 to 2018), Ahmet Davutoglu (as Prime Minister between 2014 and 2016, and Foreign Affairs Minister between 2009 and 2014), Abdullah Gul (President between 2007 and 2014), Mevlut Cavusoglu (Foreign Affairs Minister between 2015 and 2018), Binali Yildirim (Prime Minister between 2016 and 2018), as well as other AKP officials. Given the importance of the perception of state elites, this section examines the perception of the AKP elites, including the president, prime minister, key cabinet members, ministers, and advisors, and how this perception played a significant role in shaping TFP towards Iran, from December 2010 to July 2018. To a certain extent, the AKP elites' identity and perceptions defined Turkish national interest from December 2010 to July 2018; for this reason, TFP was examined using an ideological framework to a certain extent (Cornell 2012, 18; Yavuz 2009, 208).

In August 2001, a core group established the AKP, splitting from the Islamist Welfare Party (in Turkish *Refah Partisi*). Even though AKP cadre described the party as centre-right when it was founded, their religious background became increasingly dominant in shaping foreign policy. Many of its founding elites graduated from Imam Hatip High schools, which focus on the interpretation of religious-legal matters and served various Islamist background parties for years. Notably, their perception reflected a "National View" (*Milli Gorus*), based on the orthodox Naqshbandiya order (Cornell 2012, 18). In this regard, the AKP cadre followed an "ethnic-national identity over a religious and civilisational one (umma)" and viewed the

Turkish nation as a "Sunni/ Halafi/ ethnic-Turk homogeneous entity" (Yavuz 2009, 209). Their viewpoint under "a conservative Islamic identity" thus shaped their internal and external policies (Han 2016, 59).

During their first term in government, however, AKP elites often refrained from promoting avowedly Islamist policies, and focused on a multi-dimensional foreign policy emphasising the party's centre-right direction. Ahmet Davutoglu, who was then the Foreign Minister of Turkey, elaborated on this:

Turkey enjoys multiple regional identities and thus has the capacity as well as the responsibility to follow an integrated and multidimensional foreign policy. The unique combination of our history and geography brings with it a sense of responsibility. To contribute actively towards conflict resolution and international peace and security in all these areas is a call of duty arising from the depths of a multidimensional history for Turkey (Davutoğlu 2009, 12).

As such, the AKP elites aspired to present Turkey not only as a "Middle Eastern actor but also as an active member of the Islamic world and a European and development-oriented Muslim country through taking the initiative in regional and international relations" (Kiani 2015, 175). This aspiration laid the foundation for multidimensional policies, including ties with both the Middle East and western Europe. Notably, Ahmet Davutoglu and Abdullah Gul, another founding member of the AKP and then President of Turkey from 2007 to 2014, emphasised the combination of Turkey's "Eastern and Islamic identity" and the AKP's determination to follow western values and join the EU (Yavuz 2009, 202). Thus, their understanding was that the more active Turkey was in the Middle East, the more important it would become to the West by connecting the West and Middle East politically and economically (Expert 3, personal communication, July 21, 2020; Journalist 1, personal communication, July 1, 2020).

In doing so, the AKP elites also aimed to protect the party from Kemalists and the Turkish military, who traditionally supported pro-western policies, a secular order and the existence of the Republic, and who suspected that the AKP would transform the country into an authoritarian, religious state (Dinc and Yetim 2012, 72). The AKP elites thus refrained from demonstrating Islamic identity-based policies to avoid experiencing the same fate as former Islamist parties, who were banned either by military intervention or constitutional court rulings

(Yavuz 2009, 3). In essence, the AKP cadre twice faced trials which could have led to its closure in October 2002 and March 2008. The 2002 trial concerned whether the AKP leader, Recep Tayyip Erdogan, could become a deputy and the official leader of the party, since he had been sentenced for inciting religious hatred in 1998 (New York Times 1998). The 2008 trial aimed to close the AKP party and ban from politics for five years seventy-one of its leading members, including then President Abdullah Gul and Prime Minister Recep Tayyip Erdogan, because of the AKP's anti-secular policies. Although the Constitutional Court of Turkey rejected the closure request, ten of the eleven judges declared that the party had violated secular policies and agreed on a partial withdrawal of state funding for the party (Milliyet 2008). To avoid repeating this experience, the AKP deemed that a multi-dimensional policy involving both Middle Eastern and western values would be more appropriate as a basis for TFP.

When consolidating its power in Turkish domestic politics between 2007 and 2011, with the consecutive election success and the decrease of the Turkish military and the seculars on Turkish politics, the AKP sought to form a cohesive elite political group with strong political and economic support from the Muslim bourgeoisie (Altunişık and Martin 2011; Öniş 2014). In this regard, most AKP elites increased the expression of Islamic identity in the public sphere to influence "the rise of a new middle-class stratum of pious Muslims that may lead to the formation of credible, moderate political Islam compatible with regional stability" (Rabasa and Larrabee 2008, 68,75; Roussos 2017, 99). This political transformation reverberated in TFP, in which the definition of national security and threat perceptions were re-evaluated. Unlike the Kemalists' policies which had aimed to closer ties with the Western states, the AKP's policy move was to form a deeper relationship with the Middle Eastern countries.

Issues in the Turkey's EU membership process during the post-2005 period accelerated this policy shift from western- to Middle Eastern-based policies. Although Turkish-EU affairs had entered a new phase through Turkey's accession negotiation on October 3, 2005, Turkey's desire for EU membership diminished when the European Council (EC) suspended eight of thirty-five negotiation chapters on December 11, 2006. The Council claimed that Ankara had not fulfilled its responsibility to open its ports to the Republic of Cyprus (Southern Cyprus), which had been an EU member since 2004. The French President Nicholas Sarkozy further had shown reluctance about Turkey's EU membership, claiming that "Turkey was geographically not part of Europe" and "had no place in an already overstretched union" (Bilefsky 2007); this reluctance paved the way for Turkey to search for alternative policies. As a result, the group in

the AKP cadre supporting the closer ties with the Middle Eastern countries increased their influence in foreign policy and focused more on the Middle East. This resulted in the impression that TFP was undergoing a fundamental pivot away from a western-based foreign policy into a Middle East-focused agenda (Altunişik and Martin 2011, 571,578; Yavuz 2009, 211).

Then, with the 'strategic depth doctrine', the AKP elites highlighted the importance of Turkey's Ottoman past and its historical and cultural ties to the Middle East, the Balkans, and Central Asia. They advocated that Turkey's Ottoman inheritance is a significant asset for Turkey's quest for becoming a regional power (Rabasa and Larrabee 2008, 76). Foreign Minister Davutoglu's 'zero-problems with-neighbours' policy further aimed to establish cordial economic and political ties with Middle East countries, assuming that, as well as historical and cultural relations, strong economic interdependence would increase Turkey's relative power position in the Middle East (Küntay 2014, 235). Based on these aspirations, then Prime Minister Erdogan enjoyed cordial personal ties with numerous authoritarian leaders, including the Libyan leader Muammar Qaddafi and the Syrian leader Bashar al-Assad. In November 2010, Qaddafi even awarded Turkish Prime Minister Erdogan with the Qaddafi Prize for Human Rights.

On the other hand, these policies increased the discussions related to 'Neo-Ottomanism', which originated from then President Turgut Ozal's term in the 1990s; indeed, the Turkish foreign policy literature also referred to the term in regard to Turkey's close political engagement with the regions formerly under the Ottoman Empire.³ However, the AKP

³ See Details: Cagaptay, S. (2009). The AKP's Foreign Policy: The Misnomer of Neo-Ottomanism. *Turkey Analyst*, 2(8), 1-3.; Rüma, İ. (2010). Turkish foreign policy towards the Balkans: New activism, neo-ottomanism or/so what. *Turkish Policy Quarterly*, 9(4), 133-140.; Somun, H. (2011). Turkish Foreign Policy in the Balkans and" Neo-Ottomanism": A Personal Account. *Insight Turkey*, 13(3).; Taşpınar, Ö. (2012). Turkey's strategic vision and Syria. *The Washington Quarterly*, 35(3), 127-140.; Tüysüzoğlu, G. (2014). Strategic depth: A neo-Ottomanism versus Neo-Eurasianism. *Mediterranean Quarterly*, 25(2), 85-104.; Torbakov, I. (2017). Neo-Ottomanism versus Neo-Eurasianism?: Nationalism and Symbolic Geography in Postimperial Turkey and Russia. *Mediterranean Quarterly*, 28(2), 125-145.; Ergin, M., & Karakaya, Y. (2017). Between neo-Ottomanism and Ottomania: navigating state-led and popular cultural representations of the past. *New perspectives on Turkey*, 56, 33-59.; Akca, A. (2019). Neo-Ottomanism: Turkey's foreign policy approach to Africa. *New Perspectives in Foreign Policy*, (17), 03-08.; Wastnidge, E. (2019). Imperial grandeur and selective memory: reassessing neo-Ottomanism in Turkish foreign and domestic politics. *Middle East Critique*, 28(1), 7-28.; Hoffmann, C. (2019). Neo-Ottomanism, Eurasianism or securing the region? A longer view on Turkey's interventionism. *Conflict, Security & Development*, 19(3), 301-307.; Gontijo, L. C., & Barbosa, R. S. (2020).

elite, notably then Foreign Minister Ahmet Davutoglu, disputed that the discourse of Neo-Ottomanism was a malicious attack on the AKP's foreign policies. At the same time, the AKP elite highlighted that Turkey had a responsibility to maintain relations with all regions in international politics (Sabah 2009). However, with the Arab Uprisings having upended the regional power structure and politics since December 2010, the Middle East became more prominent in the AKP elite's foreign policy approach, and so it aspired to play an active role in this sphere. Thus, the following section examines the role of the AKP elites' perception in TFP towards the Arab Uprisings and how their perception shaped TFP towards Iran from December 2010 to July 2018.

5.1.1 The Role of the AKP Elite's Perception in TFP towards Iran Since the Arab Uprisings (December 2010 to July 2018)

When the AKP rose to power in 2002, Ankara followed de-securitisation policies; this contrasted with previous Turkish governments, which had been mainly under the control of the Turkish military in the 1990s. Therefore, the AKP opted to establish closer ties with Tehran, especially under the foreign principle of zero-problems-with-neighbours (Expert 1, personal communication, August 17, 2020). In this regard, then Prime Minister Erdogan called Iranian President Mahmoud Ahmadinejad a "brother" when the Iranian leader visited Turkey in 2008 and welcomed his re-election as Iranian president in 2009, in the hope of closer ties between Turkey and Iran (Williams 2019, 379). Given Turkey's desire for cordial ties with Tehran, Ankara sought to play a mediator role between Tehran and the P5+1 countries over the uranium exchange deal signed between Turkey, Iran and Brazil in May 2010. Accordingly, TFP towards Iran became more positive until Ankara, under the AKP, aimed to take a lead in regional developments that was based on its religious perception of the Arab Uprisings.

A new period of political transformation stemming from the uprisings solidified the AKP elites' pursuit of involvement in this unique political system. Accordingly, then Foreign Minister Davutoglu asserted that, in this grand transformation, Turkey would play an active role by leading regional politics and added that Turkey would "manage the wave of change in

Erdoğan's pragmatism and the ascension of AKP in Turkey: Islam and neo-Ottomanism. *Digest of Middle East Studies*, 29(1), 76-91.; Yavuz, M. H. (2020). *Nostalgia for the Empire: The Politics of Neo-Ottomanism*. Oxford University Press.

the Middle East. Just as the ideal we have in our minds about Turkey, we have an ideal of a new Middle East. We will be the leader and the spokesperson of a new peaceful order, no matter what they say" (Küntay 2014, 243; Tür 2015, 75). At the same time, then Prime Minister Erdogan emphasised the importance of Turkey during the uprisings and began the so-called Arab Uprisings tour to promote Turkey's standing as a potential role model for Arab Uprising countries. According to Erdogan, Turkey's duty was to help Arab countries build Islamist democracies (Rezaei 2019, 197). As such, the AKP elites believed that Turkey would become a model because of its democratic system and economic growth and assumed that a smooth transition would emerge in the aftermath of Arab Uprisings (Journalist 1, personal communication, July 1, 2020; Academic 5, personal communication, July 22, 2020). They thus promoted Turkey as the "central country", "regional leader", "leader of Islamic world", and "protector of the oppressed", and followed an assertive policy and hoped to enhance Turkey's regional influence (Özdamar and Devlen 2019, 187). This aspiration demonstrates that the elite sought interest-driven policies towards the uprisings because they hoped to increase Turkey's middle power capacity by playing an active future role in the context of the uprisings.

Notably, Turkey's participation in the 2011 NATO Libyan operation paved the way for Turkey's active intervention across the region. Lesch (2012, 169) argues that "Turkish leaders in Middle Eastern politics have become increasingly ideological as the Arab uprisings have progressed, with a growing focus on democracy promotion in the region as a central strategy for promoting Turkey's interests". Similarly, Tahiroglu and Taleblu (2015, 127) note that "the liberal West's love affair with a mildly Islamist party that upheld Turkey's modern tradition further encouraged Ankara to support Islamist parties in Egypt, Libya, Tunisia, and Syria". From a pro-democracy position, the AKP elite particularly propped up Islamist groups across the region. This policy demonstrated the abandonment of its previous policy of neither revealing the elites' Islamic identity nor viewing the Middle East from a religious identity perspective (Journalist 1, personal communication, July 1, 2020). The main reason for this volte-face policy was that the AKP elites believed the rise of moderate Islamist movements and political parties would enhance AKP's popularity and influence due to their ideological similarities with the AKP. They lauded the fact that MB-affiliated governments were pursuing a similar ideology as the AKP following the uprisings (Ayata 2015; Cornell 2012). Their Islamist-based ideology was pragmatic, and pro-Islamic groups/parties seemed more beneficial to their interests.

Notably, Erdogan pursued cordial political and economic ties with the MB-affiliated government of Mohammed Morsi in Cairo, presenting Turkey as a prosperous democracy which was politically Islamic. The AKP elites further supported the Sunni opposition in Syria, presenting themselves as their protector from Assad's Alawite regime. In August, 2011, the AKP government hosted the anti-Assad Syrian National Council (SCN) in Istanbul and privileged the position of the Syrian MB (Kirchner and Baş 2017, 169). The Turkish policymakers defined this support as "part of expectations derived from the country's protector of the oppressed role" (Özdamar and Devlen 2019, 186). In this regard, the MB-affiliated groups and parties emerged as a close ally for the AKP cadre. When Erdogan delivered a speech at Cairo University in November 2012, for example, he underlined the ever-deepening bilateral affairs between Egypt and Turkey by using strong Islamic references. Like Erdogan, Davutoglu stressed "the emergence of a new axis between Turkey and Egypt, since Morsi's election and deemed this axis extremely important to the emergence of a new order in the Middle East" (Yeşilyurt 2017, 72). As well as Egypt, the AKP cadre maintained their religious identity policies in Tunisia, Libya, and Syria through active political engagement with MB-affiliated parties and groups; however, in Syria, the elites also followed more interventionist and militarist policies by supporting Sunni armed groups, as well as endorsing the MB-affiliated groups/parties. The AKP elites thus viewed the Middle East from an identity perspective and generally supported Sunni Muslim protest and movements.

Given the AKP elites' search for regional leadership, Turkey initially viewed Iran as a competitor, primarily due to the different viewpoints towards the uprisings. According to Journalist 1 (personal communication, July 1, 2020), TFP demonstrated both Islamic and sectarian features at the same time, but Iranian foreign policy changed based on the situation. Iran used either its Islamic identity, as in its support for Palestinians or the MB in Egypt, or its Shia identity, as in its support for the Alawaite Assad regime in Syria. When Iran followed its Shia identity to increase its influence, Turkey viewed Iran as a competitor, as in the Syrian conflict. The AKP officials, therefore, blamed Iran for promoting unrest to expand its influence in the Persian Gulf. Accordingly, in a TV channel interview in 2015, President Erdogan said,

Iran is trying to dominate the region. Could this be allowed? This has begun annoying us, Saudi Arabia, and the Gulf countries. This is really not tolerable, and Iran has to change its view. It has to withdraw any forces, whatever it has in Yemen, as well as Syria and Iraq and respect their territorial integrity (Pamuk 2015). As such, Turkish-Iranian ties entered a new phase of competition in the region. Ankara's unwavering and pragmatic support for Sunni groups cooled relations between the two regional middle powers. Given their ideological rivalry, Ankara and Tehran were in conflict in Syria because they starkly endorsed different sectarian groups. While Ankara and other Sunni states endorsed Sunni armed groups trying to topple the Assad regime and prevented the rise of Iranian influence, Tehran maintained its support for the Assad regime. Turkey further opposed the Iranian influence over the Shia-dominated Maliki government in Iraq and sought to ally with the KDP under the Sunni Barzanis, against the partnership between Iran and the Maliki government following the decreased US presence in Iraq after 2011.

In the case of Bahrain, the AKP elites followed a sectarian policy and confronted Iran's policies. When the members of Bahrain's Shia majority protested the repressive policies of the ruling Sunni elite under the al-Khalifa dynasty, Turkish policymakers remained silent about the ensuring repression of the protestors. As such, the AKP's passive stance towards the Shiaoriented protests led to accusations against the AKP. Since then, the accusations towards the AKP elites' policies increased, and the AKP was blamed for following Sunni-based policies and aiming for regional leadership via a Sunni Islamist identity. Although the AKP elites denied this, their policies indicated that they perceived that likeminded Islamist groups would come to power across the Middle East and establish a "powerful and promising proxy" of the AKP in the region (Yeşilyurt 2017, 70). This sectarian-based perception of the elites thus resulted in the rise of competition in TFP towards Iran at a certain extent.

While the systemic stimulus revealed clear information regarding the developments of the Arab Uprisings, however, the AKP elites failed to respond to this stimulus due to their biased understanding of the internal and systemic forces shaping the developments of the uprisings. The MB- affiliated parties were unable to meet security and economic demands, and thus lost influence in their countries (Yesilyurt 2017, 71). For instance, in 2014, the Ennahdaled government lost the ensuing parliamentary and presidential elections to its secularist rival, Nidaa Tounis. Likewise, the Egyptian MB-affiliated rule was toppled by the Egyptian military in June 2013. As such, TFP saw a deadlock in the region, indicating that the AKP elites failed to predict the future of MB-affiliated parties, assuming that they could facilitate a rise in Turkish influence in the region due to their similar religious background (Özdamar and Devlen 2019, 182–86). Moreover, as Murinson (2012, 24) highlights, the Turkish model of democracy, "based upon the AKP's embrace of moderate Islam, does not match the social and economic

trajectories which the Arab countries have followed since gaining independence after World War II [...] Westernisation, a process that began long ago during the late Ottoman period, is distinguishing Turkey's path from most other Muslim countries in the Middle East". As a result, the AKP's desire to act as a role model to the new governments failed, and the AKP cadres' Islamic sectarian-based policies resulted in Turkey's isolation in the region.

According to NR theory, Turkey should have followed a rationalist foreign policy to respond to the systemic shifts of the Arab Uprisings, because the systemic stimulus revealed clear information regarding the developments of the Arab Uprisings. It was also apparent that many regional and international state actors opposed the rise of the MB-affiliated parties in the region. The AKP should have predicted this possibility and recalculated its foreign policy approach, but instead they maintained a pro-MB regional policy. Turkey's supportive policies towards the MB worsened Turkey's relations with several countries in the region, including Saudi-led Gulf countries. Even though Turkey and the Saudi -led Gulf countries aimed to limit Iranian influence in the light of Arab Uprisings, Turkey's support for the MB in Egypt deteriorated Turkey's relations with these Gulf countries. While Turkey, along with Qatar, criticised the Egyptian military coup against the Morsi government in July 2013, the Saudi-led Gulf countries supported the Egyptian military because the empowerment of the MB threatened their regimes. Turkey also failed to understand whether the global state powers would allow possible anti-western and/or anti-Russian governments to intervene in the region. In Egypt, these powers, most notably the US, were concerned about the MB-linked Morsi regime and, ultimately, the passive roles of the great powers meant that a military coup toppled the regime. Elsewhere, there was no sufficient democratic development in other countries which experienced the Arab Uprisings, and so the uprisings transformed from "Arab Spring" to "Arab Winter" (Academic 4, personal communication, July 6, 2020).

As a result of these misperception, Turkey lost not only its greatest Sunni allies but also its regional reputation as "an honest broker and future regional powerhouse" (Kirchner and Baş 2017, 167). Academic 6 (personal communication, August 25, 2020) added that with the outset of the Arab Uprisings, the Saudi-led states were disturbed at the AKP elites' self-confidence and discourse regarding neo-Ottomanism and the legacy of the Ottoman Empire, which they eschewed and deemed a positive event. After the overthrow of the Morsi regime in Egypt in 2013, moreover, Turkey's excessive support for the MB was seen by the new Egyptian military government as an outside actor pushing a particular agenda, and no longer a neutral partner. Similarly, Turkey's close ties with the MB-affiliated Justice and Construction Party in Libya put Ankara in a difficult position. The Libyan forces under the command of General Khalifa Haftar, who launched and led Operation Dignity in May 2014, was supported by the UAE, Egypt, and Saudi Arabia, who blamed Turkey for assisting armed groups in Libya (Stein 2015b, 55). As such, Turkey's zero-problems-with- neighbours foreign policy turned into "precious loneliness" following the fall of the MB-affiliated parties in the region and the overthrow of the MB-dominated government in Egypt in 2013 (Özdamar and Devlen 2019, 187). While the Saudi-led states, which aimed regional status quo, triumphed the Arab Uprisings, Turkey lost due to their revisionist policies. In this regard, Expert 2 (personal communication, August 19, 2020) claimed that Ahmadinejad, Erdogan, and Morsi became unpopular leaders in the eyes of the US, Israel, Saudi Arabia, and the EU. These three leaders thus constituted a crescent (Shia/Sunni and Arab/non-Arab) against other regional actors.

In Syria, the AKP elites also followed a more ambitious and interventionist foreign policy than the systemic shifts revealed, in that they supported Sunni-armed groups and believed that the Assad regime would be toppled. In contrast to Turkey's expectations, the Assad regime maintained its grip on power with Russian and Iranian support, and the US changed its priority from toppling the Assad regime to fighting jihadi armed groups. Ultimately, the AKP elites exaggerated Turkey's actual power distribution and position in the international system, indicating that Turkey was conducting foreign policy beyond its middle power. NCR helps to explain Turkey's isolation, in that states are generally punished when they do not follow structural forces. In this context, Turkish policymakers were expected to return to the realities of power distribution and reformulate the policy response to adapt to the new status quo, voiding Ankara's previous calculations (Yesilyurt 2017, 66). Therefore, Turkey had limited ability to counterbalance Iran's influence in the region (Kirchner and Baş 2017, 165), and needed to revise its foreign policy behaviour towards Iran.

Since 2013, moreover, the AKP elites' perception has undergone revision due to domestic issues. Following the Gezi protests (the demonstrations protesting the urban development plan for Taksim Gezi Park in Istanbul, in May 2013), Turkey, under the AKP government, started to reveal authoritarian tendencies because the Turkish government permitted the use of disproportionate force by police against the protestors. After the Gezi protests, the AKP faced another dispute with its erstwhile domestic ally. Although they shared a similar religious ideology, the AKP and the Gulen movement came into dispute in December

2013 due to a corruption scandal involving several significant AKP members. After the coup attempt against the AKP government in July 2016, the AKP declared a state of emergency and enacted a purge against thousands of alleged sympathisers of Fethullah Gulen, the cleric leader of the Gulen movement and member of a 'parallel structure (state)'. Alongside these domestic issues, ties between Turkey under the AKP and its western allies started to deteriorate due to a lack of trust. While its western allies were concerned about the rise of autocratic tendencies in Turkish politics, Turkey under the AKP government accused the US and EU of protecting "FETO members" and of sponsoring the coup attempt in July 2016 and other protests. US support for the PYD since 2014 accelerated criticism from the AKP elites against the US, since the former deemed the PYD to be a terrorist group given its affiliation with the PKK. Therefore, the AKP started to use anti-western discourse and believed that the US-led western powers had tried to topple the AKP (Göksel 2019, 21). According to Expert 2 (personal communication, August 19, 2020), following the coup attempt in Turkey, no other countries except for Qatar, Iran and Russia supported Erdogan, while other countries arguably welcomed the coup attempt. For this reason, there was a change in the ideology-based perception of the AKP elites regarding TFP towards Iran, and instead they began to pursue a pragmatic and interest-based foreign policy with Tehran, viewing it as less of a threat than the US, the EU, and the Saudiled Sunni bloc. They even claimed that the Gulen movement "is worse than the Shiites", implying Tehran (Norton 2015, 130).

Another domestic trend in Turkish politics was that the People's Democratic Party (HDP), a PKK-affiliated party, crossed the ten per cent threshold after the national election in June 2015. While the HDP gained almost thirteen per cent of the national vote in parliament, the AKP lost its majority and required a coalition partner to form a government. In this regard, the election result was an unwelcome surprise for the AKP, and Turkey began a period of internal disorder until the next national election was held in November 2015, when the AKP regained its majority with almost forty-nine per cent of the vote. During this period of disorder, nationalist discourse in Turkish politics increased when the AKP established internal cooperation with a far-right political party, the Nationalist Movement Party (MHP), against the rise of the HDP. As a result, the AKP elites' perspective started to exhibit nationalist and conservative features. Following the constitutional referendum supported by the AKP and MHP in April 2017, President Erdogan highlighted the theme of "one nation, one flag, one homeland, one state" (Williams 2019, 385). These developments paved the way for the emergence of re-securitisation in Turkish politics (Kösebalaban 2020; Journalist 1, personal

communication, July 1, 2020). According to Taliaferro (Taliaferro 2006), states may use nationalism as "a viable propaganda tool" to persuade the public to increase military expenditure and power politics. In this regard, the AKP elites used nationalist discourse with the Turkish public to consolidate Turkey's domestic and external military operations against the Kurdish armed groups, especially against the PKK-affiliated PYD/YPG in the north of Syria.

Given the perception shift of the AKP elites from sectarian to nationalist and antiwestern, the AKP elites sought to engage in a constructive dialogue with Tehran. In 2015, Erdogan highlighted that "I don't look at the sect. It does not concern me whether those killed are Shiite or Sunni; what concerns me is [that they are] Muslims. Iran and Turkey should mediate between the battling parties and hope to obtain a result. We have to put an end to this bloodshed, this death" (Hurriyet Daily News 2015). In 2017, Erdogan further noted that "the Turkey-Iran border was established much earlier than the [political] formation of America", implying that Turkish-Iranian affairs had been longer in its term (Kumral 2020, 286). Similarly, Davutoglu stated during his trip to Iran in March 2016 that "we may have different views, but we cannot change our history or our geography. Turkey and Iran need to develop some common perspectives to end our region's fight among brothers, to stop the ethnic and sectarian conflicts" (Yackley and Sharafedin 2016). Other AKP elites further emphasised the importance of Turkish-Iranian relations. For instance, the current Minister of Foreign Affairs of Turkey, Mevlut Cavusoglu, stressed that "despite differences between Turkey and Iran, the two countries have a brotherly relationship and the strong ties between Turkey and Iran cannot even be questioned" (Noi 2015). Such discourse from the AKP elites indicates that they aspired to strengthen the Turkish-Iranian relationship and perceived the ties with Tehran as more beneficial than those with Turkey's western allies. More precisely, Turkey followed an interestbased policy towards Iran, mainly because the AKP elites aimed for a status quo in Turkish politics.

As well as the elite's (mis)perception, domestic considerations played a role in TFP towards Iran as Turkey sought to resolve economic and security issues. In this respect, the AKP elites sought to facilitate the expansion of bilateral relations between Turkey and Iran, even though they were dissatisfied with Iran's ideological foreign policy towards the Arab uprising countries (Uygur 2012, 26). Notably, the AKP elites highlighted trade and economic ties with Tehran and refrained from intervening in Iran's domestic affairs for bilateral economic and

political benefit. For instance, Erdogan emphasised energy security, especially concerning Eurasian resources, and thus sought collaboration with Iran (Ebrahimi, Yusoff, and Seyed Jalili 2017, 76–77). In the following section, there is a focus on the role of domestic considerations in TFP towards Iran since the Arab Uprisings, from December 2010 to July 2018.

5.2 Domestic Considerations

TFP towards Iran from December 2010 to July 2018 cannot be understood primarily by only analysing the perceptions held by AKP elites because alliances are not formed in response to power alone but are instead shaped by other factors (Walt 2009, 103). Randall Schweller underlines the analysis of interests and motivations as unit-level intervening variables in the causal chain of neoclassical realist foreign policy examination. According to Schweller (1994, 79), states may cooperate with each other to gain an opportunity. Therefore, similarly situated states may respond differently to similar external constraints. Schweller proposes this assumption as an approach thus views status quo and revisionist interests as intervening variables and calculates balance of interest theory.

In essence, the implications of this theory are twofold, at the unit and systemic levels. "At the unit level, it [the theory] refers to the costs a state is willing to pay to defend its values (status quo) relative to the costs it is willing to pay to extend its values (revisionist). At the systemic level, it refers to the relative strengths of the status quo and revisionist states" (Schweller 1994, 99). The state interest at the unit level as:

State Interest $(n) = Value \ of \ Revision - Value \ of \ Status \ Quo$

Considering that states, powerful or weak, will cooperate with other countries sharing the same aims of protecting or changing the system, satisfied states will ally with other satisfied ones to maintain their position in the system, and vice versa dissatisfied countries will cooperate with other dissatisfied ones to change the status quo. Schweller's balance of interest theory has heuristic value in defining the 'zoology of states', whereby the "lions are states that will pay high costs to protect what they possess but only a small price to increase what they value... As delighted states, they are likely to be status quo powers of the first rank"; in contrast, "lambs are weak countries that will pay only low costs to defend or extend their values" (Schweller 1994, 101–2). The theory goes on to state that, "jackals are states that will pay high

costs to defend their possessions but even greater costs to extend their values", while wolves, as predatory states, "value what they covet far more than what they possess" (Schweller 1994, 103).

In line with Schweller's interest theory, TFP demonstrated profit-oriented behaviours in both regional and international politics. Before the uprisings, Turkey had not only experienced intense economic and security affairs with Middle East countries, but it had also been perceived as a powerful economic and political actor in the region. At the outset of the uprisings, Turkey lost several economic allies in the region and faced growing security threats in its surrounding area. These economic and security shifts challenged Turkey's regional interests. Moreover, the AKP elites' foreign policy calculations resulted in a loss of prestige in both regional and international arenas. Collaterally, the AKP's legitimacy was questioned at home. Turkey under the AKP government, therefore, sought to cooperate with those states which shared similar interests, to preserve its value both at home and internationally.

As noted by Eder (2014, 81), "changing capabilities do lead to changes in the perception of threats, interests and opportunities in the neoclassical realist system". Before the Arab Uprisings, regional developments in the Middle East, such as the US-led operation in Iraq in 2003, not only deteriorated Turkey-US affairs but also increased Turkey's anxiety about its vital regional interests. In this era, Turkey enhanced its diplomatic and economic relations with Iran (Küntay 2014, 236). Following the Arab Uprisings, moreover, Turkey chose to forge ties with Iran in terms of political, security and economic aspects even though Iran demonstrated a revisionist tendency and different approach from Turkey. Therefore, it could be argued that the AKP government followed interest-driven and profit-oriented policies towards Iran, and Turkish domestic considerations were another determinant of TFP towards Iran between December 2010 and July 2018.

5.2.1 Turkey's Domestic Considerations in Relation to Iran since the Arab Uprisings (from December 2010 to July 2018)

Turkish foreign policy faced formidable challenges, including the loss of economic and political benefits with authoritarian regimes and the rise of non-state armed actors in its proximity. Although Ankara claimed to follow pro-democratic movements, financial and political motivations regarding cost-benefit calculations and interest-driven policies mostly shaped Turkey's policies towards the Arab Uprisings. After losing its previous authoritarian allies, Turkey's regional vision was claimed to be promoting full economic and political affairs by signing security and trade agreements with the new governments under "economic interdependence" (Davutoglu 2013). However, the AKP elites miscalculated the regional developments due to biased sectarian-based policies and, worse still, Ankara lost its significant regional and external partners, including Saudi Arabia and the US. Thus, protecting economic and political interests became the top priority for Turkey.

Since coming to power in 2002, in essence, the AKP governments have signed several agreements and memoranda of understanding, bringing closer economic, political, and security relationships with Iran under various foreign policy principles, such as 'zero-problems-with-neighbours' and 'economic dependency'. Bilateral trade between the two countries saw a ten-fold increase compared to before the AKP term. Iran became Turkey's largest supplier of crude oil and second largest supplier of natural gas, and in return became Turkey's third-largest goods export market by 2012 (Gürzel 2014, 103). Their political ties also increased, and Turkey hoped to ameliorate international sanctions against Iran through the 2010 nuclear exchange deal. Between 2004–2008, Turkey also signed various security deals with Iran to combat terrorism, drug smuggling, and organised crime (Sinkaya 2019). As such, Turkey under the AKP governments followed a policy of détente towards Iran, a contrast to the less than friendly relations that had marked bilateral ties between Ankara and Tehran before 2002. Although the Arab Uprisings and period following them became a test for TFP towards Iran and brought a competition, Turkey aimed to maintain its ties with Iran at a reasonable level, because the economic, security and political developments from the uprisings confined Turkey's ability.

Regarding security considerations, the rise of ISIS forced Turkey to seek an ally in its fight with the jihadist armed group, especially since ISIS threatened Turkey's security interest as a Wahhabist militant group. Turkey was initially not troubled by the rise of ISIS since the group seemed to oppose the Assad regime, but over time Turkey realised that ISIS was a threat to its national and regional security, as well as economic interests. Notably, ISIS occupied Iraqi territory, in which the main oil pipeline from Kirkuk runs, thus alarming Turkey by influencing its financial affairs with Iraqi Kurdistan. Turkey therefore joined the Iraqi Kurds and Iraqi central government in pushing ISIS back from northern Iraq (Hale 2015, 198). The empowerment of Syrian Kurdish armed groups, the PYD, and its military wing YPG, also became a security issue for Turkey. Being a PKK-affiliated non-state armed group, the

PYD/YPG exacerbated the security challenge of Turkey's own Kurdish problem. Since 2015, when the peace process between Turkey and the PKK terminated, the empowerment of PYD/YPG sabotaged Ankara's domestic interests. Ankara sensed that the rise of the PYD might create a Kurdish ethnic problem at home, assuming that a possible Kurdish state in its proximity would trigger Kurdish groups in Turkey to attempt to gain independence or autonomy. As such, the rise of the HDP (the political party of the PKK) in Turkish domestic politics increased the concerns of the AKP government and resulted in a perception shift among the AKP elites.

As a result, domestic considerations determined TFP after the Arab Uprisings as another intervening variable, because systemic developments, including the rise of non-state armed groups, the extension of the Syrian conflict, and the fall of MB-affiliated parties diminished Turkey's regional position. Despite its unilateral military operations, Turkey realised that it was difficult to resolve such security issues alone, and the groups it was backing could not curb the growth of these non-state armed groups (Expert 3, personal communication, July 21, 2020). In this regard, Ankara required the support of Iran via the formation of a regional axis to protect its security and financial interests in the Middle East. As a result, the AKP elites did not maintain their sectarian-based policies towards Iran. Rather, domestic interests were more dominant than the sectarian-based identity in TFP towards Iran. Ankara pursued collaboration with Iran over security and economic considerations.

As Aydin (2015, 128) argues, "Turkey is a historical partner for Iran and their relations depend on the mutual interests in regional politics; besides, their destiny in international politics is the same". However, As Interviews A (personal communication, August 4, 2020) and C (personal communication, August 6, 2020) noted, we cannot say that there has been consistent cooperation or competition between Turkey and Iran; rather, they are eternal competitors and collaborators in the region. For instance, Turkey perceived Iran to be the instigator of security threats against Turkey due to its previous partnership with the PKK, but also saw it as a potential ally to resolve these self-same challenges. Furthermore, as Yorulmazlar and Turhan (Yorulmazlar and Turhan 2015) further argue, "against growing regional instability, Turkey would preferably prioritise domestic concerns, that is, security, economic development and Kurdish reconciliation". Based on cost-benefit calculations and profit-oriented policies, as Schweller's balance of interest theory, Ankara opted to ally with Tehran for a short-to medium term even though their similar relative power position in the

international system may bring a competition to enhance their regional influence in the longer term. Therefore, the following subsections examine Turkey's security and economic considerations in seeking short-to-medium term collaboration with Iran.

5.2.1.1 Security Consideration

The Arab Uprisings not only created significant security concerns for Turkey, but they also triggered a new phase of relations with Iran. The toppling of authoritarian regimes, instability, and failed state structures gave non-state armed actors, such as ISIS and the PYD a space to exploit. Throughout the region, these non-state armed actors challenged both external and regional powers, enhancing their power and control via repressive operations. The growth of these groups' influences particularly threatened Turkey's southern periphery, leaving it concerned about the empowerment of ISIS and the PYD, which it viewed as significant threats to domestic security.

5.2.1.1.1 The Rise of ISIS

Despite the international community's concerns about the rise of ISIS as a jihadi armed group, Turkey initially remained unresponsive, perceiving it as essential to its regional interests. This is because Ankara hoped for the overthrow of the Assad regime and the prevention of Iranian influence in its proximity. When ISIS emerged as a challenge to the Assad regime as an opponent, and to Iran as a sectarian hostile, Ankara assumed that ISIS would serve Turkey's aims regarding the Assad regime and Tehran. More importantly, Ankara was pleased with the conflict between ISIS and the PYD in the siege of Kobane because it perceived the PYD to be more of a threat than ISIS to its security interests (Cağaptay 2020, 123–25; Işiksal 2017b, 96).

However, Ankara miscalculated the rise of ISIS in its proximity. A corollary of Turkey's initial reluctance to fight against the jihadi group was that Turkey was accused of supporting ISIS by allowing the armed group to receive free treatment in hospitals in south-eastern Turkey, and to cross over its border into Syria. Turkey's western allies also became suspicious of Turkey's alleged collusion with the group, and the US officials, including those in the Pentagon, perceived Ankara as collaborating with the US enemies in Syria. As such, Turkey's initial hesitation in combatting ISIS was a portent of the difficult times to come for Turkey, whose indifference was increasingly challenged by the international community. The jihadi group also posed a security threat to Turkey's external and internal security by

conducting terrorist attacks. As such, Turkey faced more security threats compared to its rivals, including Iran (Cağaptay 2020, 123–25; Işiksal 2017b, 96).

Bearing in mind that Ankara initially faced a "multifaceted dilemma" regarding whether to fight against ISIS or prevent the rise of Iran's regional influence and the possibility of Kurdish independence on its southern periphery (Aras and Yorulmazlar 2016b, 5), Ankara then opted to counter the threat of ISIS to its national security. This volte-face policy resulted from ISIS attacks in Turkey's immediate vicinity. While enhancing its territory and causing instability in Iraq and Syria, ISIS persecuted certain groups, including Iraqi Kurds, Turkmens, and Yazidis, and damaged the cultural and economic resources of the Kurdistan Region of Iraq (KRI). In this region, Turkey sought to protect its economic and political affairs as well as the security of Turkmens (Aras 2014b, 12). More importantly, Turkey was forced to move the tomb of Suleyman Shah in northern Syria towards the Turkish border in March 2014, due to ISIS attacks. ISIS also abducted many Turkish citizens, including a mission head and truck drivers, during its seizure of significant cities in Iraq, such as Mosul.

As well as its proximity, ISIS started to threaten Turkey's domestic security, as ISIS' aggressive policies increased the refugee influx into Turkey. According to the Ministry of Interior, in 2018, Turkey hosted approximately 3.9 million refugees fleeing terrorism and civil war in Iraq and Syria (Hurriyet Daily News 2018). With the empowerment of ISIS, the scale of its activities and of extremist ISIS sympathisers also increased in Turkey, posing a threat to Turkey's domestic security. In 2015, ISIS organised a terrorist attack killing 34 people in Suruc, a southern Turkish town near the Syrian border, while the deadliest terror attack killed 103 civilians in the central train station bombing in Ankara. ISIS conducted several attacks and caused the death of numerous Turkish citizens in various Turkish cities, including 13 people in Istanbul in January 2016, at least 41 people at Istanbul Ataturk Airport, 54 people in a wedding ceremony in Gaziantep in August 2016, and another 39 in a club shooting on New Year's Eve, 2017. These devastating cases culminated in more Turkish operations against the armed group. On 24 July 2015, Turkey hit two ISIS headquarters near Havar, a village in Syria under an operation called Martyr Yalcin. With the military operation called Euphrates Shield, in August 2016, Ankara also aimed to purge ISIS from Jarablus, as well as the PYD from northern Syria. These military operations were followed by the arrest of numerous ISIS members and the closure of ISIS-related media and organisations in Turkey. More than 1,200 ISIS members were arrested in 2015 alone (Işiksal 2017b, 97).

Further to the fight against ISIS, in September 2014, Turkey collaborated with the USled coalition by increasing intelligence sharing and allowed the US Air Force to use Incirlik and Diyarbakir air bases in the struggle against ISIS in July 2015. Turkey also liaised with the EU regarding irregular and illegal refugee flow, including of the jihadist group members, from Syria to Europe (Özer and Kaçar 2018, 187). However, Turkey claimed that its western allies provided insufficient support during its fighting against the two non-state armed groups, ISIS and the PYD/YPG, because while the US and EU shared Turkey's concerns regarding ISIS but had different viewpoints about the PYD. On 28 July 2015, for instance, Turkey felt that NATO's response to Turkey's call under Article 4 of the NATO Charter was insufficient, since NATO members did not see the PYD/PKK as a security issue but agreed on ISIS (Bicer 2016, 93). As such, the AKP and pro-AKP newspapers, such as Daily Sabah, blamed European countries for ignoring attacks on their ally (Daily Sabah 2016b). In this regard, the lack of a shared threat perception between Turkey and its western allies against ISIS and the PYD culminated in insufficient cooperation between them.

In February 2016, Ankara also collaborated with Saudi Arabia against ISIS, announcing that Riyadh "was sending troops and fighter jets to Turkey's Incirlik airbase and the two countries were ready to launch an operation from land if the West comes up with a comprehensive strategy to counter ISIS" (Tol 2016). However, this cooperation was short-lived due to disagreements between Turkey and the West over the PYD, and Turkey and Saudi Arabia over the 2017 Qatar crisis and Syrian opposition groups. On this issue, Erdogan's advisor, Ibrahim Kalin, warned that "in the post-ISIS Syria, a new power game is brewing. The US in eastern Syria was attempting to be a counterweight to Russia and Iran. In this case, Turkey's cooperation with Iran and Russia is essential" (Frantzman 2017). Therefore, despite their different perceptions, Turkey opted to establish closer ties with Iran against the rise of ISIS, since the empowerment of ISIS threatened both countries due to its antagonistic ideological beliefs against Shi'ism and Sunni political Islam. ISIS' sectarian policy forced Ankara to cooperate with Tehran to maintain the balance of Sunni-Shia order in their proximity (Aras and Yorulmazlar 2014, 115–20).

Given Turkish-Iranian security collaboration against ISIS, the officials of the two countries united to resolve security threats arising from ISIS. For instance, in August 2017 General Mohammad Hossein Bagheri, the chief of the Iranian Armed Forces, visited Turkey to discuss security issues, and was the first Iranian military chief to do so since the 1979 Islamic

revolution in Iran (Majidyar 2017). In essence, ISIS has targeted Iran since Iran's Shia Islam clashes with ISIS's religious beliefs. Further, Iran's Sunni communities, such as Kurds and Baluchis, were a potential target for ISIS in their mission to provoke riots against Iran. Since these communities suffered from poverty, repression, and black-market economies, for which they blamed Iran, they became a viable target for ISIS. Hence, in contrast with Sunni countries, Iran needed an ally to fight ISIS (Tabatabai and Esfandiary 2017, 132). As such, mutual security considerations over the rise of ISIS brought collaboration between Ankara and Tehran. Turkey found cooperation with Iran more beneficial than that with western powers and Riyadh in terms of calculating the cost-benefit of a short-term alliance with Tehran. The two states' mutual concern thus increased the degree of their cooperation in the fight against ISIS.

5.2.1.1.2 The Rise of Kurdish Armed Groups

The Syrian conflict culminated in the empowerment of Kurdish armed groups. They benefited from the instability and power vacuum created throughout the region. Disenfranchised and repressed by successive Syrian regimes, the Syrian armed Kurds mainly took advantage of the country's instability to claim territories in which they constituted a majority (Dalay 2017a). They also received support from numerous regional and external actors. The US, for instance, supported the YPG's fight against ISIS. When jihadi ISIS advanced on the Kurdish-held town of Kobane in October 2014, the US Air Force attacked the jihadi group, launching an extraordinary and successful relationship that proved to be the most successful effort at dislodging ISIS from the territory it occupied (Işiksal 2017b, 90). In May 2017, the US President Trump approved the overt supply of weapons to Kurdish factions within the Syrian Democratic Forces to assist the liberation of Raqqa from ISIS (BBC News 2017a). As well as the US, Saudi Arabia, the UAE, and Egypt also sought to establish close ties with the YPG. In October 2017, for instance, the Saudi Minister of Gulf Affairs, Thamer al-Sabhan, and then US Special Envoy to the coalition against ISIS, Brett McGurk, went to Raqqa under YPG control, and Riyadh promised to contribute economic and military aid for the non-state armed group (Başkan and Pala 2020a, 66).

By contrast, the Kurdish issue constituted a critical security concern for Turkey due to its ongoing PKK issue since 1984, despite several ceasefires. With the failure of peace negotiations between Turkey and the PKK in July 2015, this conflict accelerated hostility between the Turkish government and the PKK. When the PKK-affiliated political party, HDP, caused the AKP to lose its majority in parliament in the 2015 national election, Turkey under

the AKP became further concerned about the empowerment of Kurdish groups in its proximity. After 2015, domestic security considerations regarding the Kurdish issue predominated in TFP (Journalist 1, personal communication, July 1, 2020). Given Turkey's concerns regarding the rise of Kurdish groups, Turkey was concerned about Washington and Riyadh's aid for the YPG/PYD, assuming that this support would cause the transfer of weapons to the PKK and the possibility of a PKK attack on Turkish soil (Ergun 2018, 165).

The growth of Kurdish influence in Rojava in north-western Syria, therefore, became a critical factor in the regional and domestic policies of Turkey. Following the victory of the YPG/PYD in Kobane in 2014, Turkey's concerns increased since the PKK's substantial participation in the international campaign against ISIS offered the armed group significant opportunities to strengthen its position vis-à-vis Turkey (Romano 2015, 53–54). During this time, the PKK enjoyed support from the Assad regime due to the legacy of often tense relations that had marked ties between Ankara and Damascus (J. Karakoc 2016, 11). Accordingly, Ankara assumed that the victory in Kobane would trigger the PKK's quest for separatism, despite the Peace Process signed in 2013 between Ankara and the armed group. In essence, the PKK had imitated the Arab Uprisings and attempted to affect a 'Kurdish Spring' among the Kurds in southern Turkey in the wake of the uprisings. On February 14, 2011, Abdullah Ocalan, the imprisoned leader of the PKK, made a statement to his followers to the effect that "the Kurds could only be free if they pour on to the streets and call for their rights in the Kurdish cities, like Diyarbakir" (Noi 2012, 23-26). For these reasons, Ankara's concerns re-emerged after the victory of the Kurdish armed groups in Kobane. At the time, Erdogan was harshly critical of the US intervention in Kobane as he and his party viewed the YPG as a "greater scourge" than ISIS. Erdogan rejected any attempt by the Syrian armed Kurds to establish an autonomous area on Turkey's southern border, similar to the existing ones in Iraq since Such a state would mean that the PKK would pose an increasing security threat under the PYD's control (Barkey 2016; Oktav 2015, 87).

Following the empowerment of the PYD/YPG, Ankara adopted offensive policies in Syria and Iraq and attempted to find an alternative entity for its national Kurdish population living on its southern borders. In this regard, Turkey encouraged the Kurdistan Democratic Party (KDP) under the Barzanis in the KRI to lead a Kurdish entity forming in northern Syria. In doing so, Turkey hoped to prevent the rise of the PYD as a significant and powerful Kurdish group in its proximity, and to control the rise of Kurdish armed groups through its collaboration with the KDP. However, Turkish plans backfired since the PYD was unwilling to partner with the KDP controlled by Turkey (Journalist 1, personal communication, July 1, 2020). Since 2016, Turkey, along with Turkey-aligned Syrian opposition armed groups, conducted cross-border military operations against the PYD/YPG, including operations Euphrates Shield (August 2016–March 2017) and Olive Branch (January-March 2018). Since Kurds in Rojava had military troops and gained external support, however, Turkey realised that its direct involvement in northern Syria and support for Syrian opposition groups were limited in curbing the rise of the Kurdish threat. Ankara therefore needed a state ally to thwart regional Kurdish aims.

In this regard, Tehran emerged as an alternative partner for Ankara due to their mutual Kurdish problem. Likewise, Iran has suffered from several Kurdish insurgent movements. For instance, in mid-2016, Iran faced an apparent, perhaps short-lived, revival of the mostly dormant insurgency in its Kurdish region despite a cease-fire agreement in 2011 (Dalay 2017a). In essence, Iran initially established a pragmatist partnership with the YPG to limit the rise of Turkish influence in Syria and Iraq, because Iran was concerned about Turkey's so-called ties with Jabhat Fatah al-Sham [formerly Jabhat al-Nusra] and its support for Ahrar al-Sham, another militant Salafi group. However, Iran was concerned about the success of the YPG/PYD in establishing an autonomous state through western support, since it would trigger protests among Iran's Kurdish communities and threaten Tehran's increasing role in the region(Oktav 2018, 204–6).

In March 2016, moreover, Iran's YPG/PYD policies shifted from considering the group a tactical ally to a potential strategic threat, due to the YPG/PYD's ties with the US and Russia and declaration of a federal system in the territory under its control. In this respect, an Iranian national security official expressed that, "an autonomous Kurdish region [in Syria] will trigger the fragmentation of Iran, Iraq, Syria and Turkey, replacing major regional states with an archipelago of weak states" (International Crisis Group 2016). Lastly, in July 2016, the coup attempt in Turkey became another turning point in Iran's quest for cooperation with Turkey. Iran immediately declared support for the AKP government due to its national security concerns and assumed that if the coup had been to overthrow the AKP government, Turkish Kurds would have taken the opportunity to rebel against Turkey, influencing Iranian Kurdish communities to secede likewise (Expert 3, personal communication, July 21, 2020).

Based on interest-driven and profit-oriented policies, security collaboration with Tehran became a rational approach for Ankara as it found security ties with Tehran to be more beneficial than inter-state competition over Kurdish groups. Given their mutual security concerns, in October 2011, Ankara and Tehran first reiterated their determination to work together on the Kurdish issue. Following a meeting at the Turkish Foreign Ministry in Ankara between their respective foreign ministers, Davutoglu declared that "[the] joint determination to struggle against the PKK and the PJAK will continue in the strongest way. From now on, we will work together in a joint action plan until this terrorist threat is totally eliminated" (BBC News 2011). He also claimed that there had been an essential cooperation between the intelligence and security institutions of the two countries and significant strides had been made in this joint fight. In this regard, the two countries managed to set aside their differences to tackle the issue, and pledged to coordinate efforts (Kaya 2012, 16). Turkey further attempted to conduct joint military actions with Iran against the Kurdish separatists. On this issue during a trip to Iran in 2017, President Erdogan emphasised that "joint action against terrorist groups" that became a threat is always on the agenda. This issue was discussed between the two military chiefs, and I discussed more broadly how this should be carried out" (Reuters 2017). Ultimately, Turkey's PKK issue and the support from Turkey's state allies to the PYD/YPG forced Ankara to cooperate with Iran in the short-to-medium term.

Mutual concerns also forced Ankara to cooperate with Tehran following the independence referendum held by the Kurdistan Regional Government (KRG). When the KRG pursued independence from the Iraqi government on 25 September 2017, Turkey attempted to establish regional opposition in cooperation with Iran to prevent the KRG from reaching its aim, and thus both Turkey and Iran (along with Iraq's central government) opposed the referendum. Both countries imposed sanctions on the KRG to dissuade it from declaring an independent state. With this objection, they further aimed to consolidate their opposition towards the creation of a possible Kurdish state by the PKK and PJAK (Mohammed 2011, 74). Accordingly, the Kurdish issue in Turkey is intricately tied to its relations with Iran and international allies, such as the US. While Turkey has mutual security concerns with Iran, the two countries are likely to conduct joint policies. From the perspective of Schweller's balance of interest theory, Ankara's desire for Tehran's support against the rise of Kurdish groups reflected the former's interest-driven and profit-oriented policies towards Tehran.

5.2.1.2 Economic Consideration

States' economic interests play an influential role in shaping foreign policy since the success of foreign policies is often determined by their economic impact (Aloudah 2016, 280). Following an eight-month financial recession from March to November 2001, the AKP thus targeted becoming an economic power as an indicator of its success in government (Patton 2006, 513). 'Zero-problems-with neighbours' and 'economic interdependency' emerged as the foreign policy principles by which the AKP hoped to establish economic as well as political ties with Middle Eastern countries, regardless of their governmental form (Tür 2011). As such, TFP under the AKP government hoped to forge closer economic and trade affairs with then authoritarian regimes in the Middle East.

With the onset of the Arab Uprisings, however, Turkey's quest for greater economic ties with authoritarian governments across the Middle East was no longer viable and Turkey lost economic allies in the region even after exerting itself to build sustainable political and economic relations and cooperation after 2002. The Arab Uprisings thus resulted in more detrimental economic loss to Turkish exports than in imports from the uprisings countries. For instance, at the outset of the uprisings, the Turkish export and import demands decreased by nearly four per cent and six per cent on average, respectively (Sorhun 2012, 12). Notably, the Libyan and the Syrian conflicts became a test for Turkey's economic interests since Ankara lost its two close economic allies in the region. Due to its territorial closeness, the Syrian conflict also represented an economic burden on the Turkish economy because of the resulting refugee crisis. According to Cavusoglu (2017, 12), in 2017, Turkey provided numerous services, including health and education, to approximately 2.7 million Syrian refugees.

Moreover, Turkey's pro-MB policies in the region caused a downgrade in its political and economic relations with the Saudi-led Gulf countries following anti-MB policies. While blaming Turkey and Qatar for endorsing terrorist organisations in the region, the oil-rich Gulf countries reduced their economic ties with Turkey. For instance, the values of Ankara-Riyadh and Ankara-Abu Dhabi trade respectively fell by thirty-eight per cent and twenty-three per cent in 2016 (Kerr 2017). After the 2017 Qatar crisis, Turkey also faced significant economic loss due to its pro-Qatar policies. While Saudi Arabia, the UAE and Egypt emerged as critical to financial resources and markets, Turkey prioritised working with Qatar due to its mutual ideological policies rather than following a neutral stance; in so doing, Turkey jeopardised its economic interests with these countries (Başkan and Pala 2020b, 67).
Turkey had further economic issues due to political disagreements with its western allies. Its ties with the US soured due to US support for the PYD/YPG, the US' reluctance to extradite Fethullah Gulen following the 2016 military coup attempt, and Turkey's detention of US pastor Andrew Brunson in October 2016. Turkey hoped to establish security ties with Russia by signing a deal on the delivery of the S-400 missile system in September 2017, but Turkey's NATO allies were concerned about this deal, assuming to be in contravention of NATO principles. In this regard, US officials, notably in the Pentagon, voiced the need for economic sanctions to be levied against Turkey (Expert 2, personal communication, August 19, 2020). As such, Turkey-US political issues reflected on their bilateral economic affairs, and the Turkish lira started to lose value against the US dollar.

There were some issues between Turkey and the European countries regarding the Syrian immigrants, as well as the increasingly authoritarian nature of Turkish politics following the coup attempt in Turkey in 2016. The issue of Syrian refugees fleeing through Turkey to reach Europe exacerbated further tensions with the EU. In this regard, Mevlut Cavusoglu, Turkey's Minister of Foreign Affairs, blamed the EU for not fulfilling its duty towards Turkey, claiming that:

Turkey was already dealing with millions of Syrians and others fleeing civil wars with little international assistance. When the number of migrants overflowing into the EU became unbearable, the EU reached out to Turkey [...] Furthermore, financial assistance from the EU for the Syrian refugees in Turkey has been slow in disbursement. Turkey has spent around 30 billion USD for the Syrians, whereas total contributions from the international community amounted to only 526 million USD. Moreover, once the irregular crossings were reduced, the voluntary humanitarian admission scheme should have begun, but it did not. The 18 March Agreement will maintain its success as long as both sides fulfil their pledges, and the EU has several to fulfil (Republic of Turkey Ministry of Foreign Affairs 2017).

Therefore, as Dursun-Ozkanca (2019, 83) highlights "The EU- Turkey relationship has become even more controversial due to the refugee deal brokered between the two parties on March 18, 2016".

Ultimately, Turkey's problems with the West and the other Sunni states forced Ankara to find alternative allies for its economic issues. According to Expert 2 (personal communication, August 19, 2020), economic and trade dependency prevented Turkey from

abrogating ties with Iran, and so Turkey separated its political issues with Iran from its economic relations. Since the beginning of the AKP's rule in Turkey, the government pursued better trade and commercial ties with Iran under the "maximal mutual interest-based" policy (Cheema 2015, 102). Iran's substantial potential and economic capacity has become attractive for Turkey's export-led growth policy. Moreover, with rising energy demands, Turkey has pursued better relations with Iran (Küntay 2014, 235–36). Each year, Turkey imported large amounts of oil and gas from Iran. While close cooperation with Iran over energy began in 1996 under the Erbakan government, it intensified under Erdogan's regime. In July 2007, Turkey and Iran signed a Memorandum of Understanding (MOU) to transport 30 billion cubic metres of Iranian and Turkmen natural gas to Europe. The deal envisaged the construction of two separate pipelines to ship natural gas from Iran and Turkmenistan to Europe (Rabasa and Larrabee 2008, 87). Turkish dependence on Iranian natural gas and Iranian dependence on Turkey as a transit route for energy revenues (either in the form of money or gold) helped nurture an interdependent energy relationship. Each year, almost one-fifth of Turkey's natural gas imports came from Iran. Turkey's general dependency on imported energy as a growing economy-and Iran's significance in this context-was highlighted by Davutoglu when he described Turkey's need for Iranian energy as a "natural extension of Turkey's national interests" (Grunstein 2008).

Considering Turkish trade relations with Iran, Turkey was eager to access to Iranian gas on easy terms and aimed to promote the transfer of Iranian gas to western markets. It also hoped to overcome the negative effects of international sanctions imposed by the US on Iran, because the Turkish economy in general and Turkish business groups in particular were negatively impacted by the sanctions, leading to its forfeiting of Iranian trade (Aras and Yorulmazlar 2016b; Gürzel 2014, 103). Bearing in mind that Turkey favoured a diplomatic solution to the nuclear issue and criticised sanctions and the use of military force against Iran, Ankara aimed to revive negotiations between Iran and the P5+1 over Iran's nuclear enrichment programme by hosting talks in 2012 (Kaya 2012, 10).

Given strained ties with the Western powers and the Saudi-led Gulf countries, Turkey needed Iran more than ever, and aimed for pragmatic economic collaboration to compensate for other economic losses. Despite the US-led sanctions, for example, in February 2013, Ankara declared that it would continue to purchase oil and natural gas from Tehran, and indeed the Turkish state-owned Halkbank allegedly took the financial risk of exchanging gold for

Iranian oil and natural gas from 2012 to 2016 (Kumral 2020, 270). Turkey also supported the Obama administration with the 2015 nuclear deal and opposed the US withdrawal from it, and indeed the ongoing US-Iran conflict also meant a decrease in the supply of Iranian oil for Turkey (Expert 1, personal communication, August 17, 2020). In this regard, in April 2015, President Erdogan highlighted Turkey's determination to increase annual trade with Iran to \$30 billion when Iran supplied 21% of Turkey's imports of oil and 16% of its imports of gas (Millivet 2015). Ankara further agreed on the use of national currencies in its bilateral trade with Tehran in 2017. From these developments, Iran became Turkey's first and second trading partner in the Gulf in terms of balance of trade between 2015 and 2017 (Martin 2019, 265). As a result, Ankara endeavoured to integrate Iran's economy into the international system and opposed the declaration by former US President Donal Trump of withdrawal from the Joint Comprehensive Plan of Action (JCPOA) in May 2018. If the sanctions had been lifted, Turkey would have established greater economic cooperation with Iran. As such, Turkish policymakers opposed the US sanctions against Tehran, and even President Erdogan said that "Iran is our neighbour and our strategic partner", and "severing ties with Tehran on America's whim goes against Turkey's sovereignty" (Dursun-Özkanca 2019, 109).

In addition, Turkey aimed to consolidate economic ties with Iran to establish a close relationship with the Iraqi central government, as an ally of Iran. The reason for this was because, if Turkey were to achieve strong economic and political ties with the Iraqi central government, it could gain further diplomatic, energy, and trade benefits in the region (Cağaptay 2016). Therefore, despite its rivalry with Tehran over Iraq, Ankara realised that better ties with Tehran would promote a closer relationship between Ankara and Baghdad.

Overall, as noted by Ehteshami and Elik (2011, 654), "in economic terms, Iran was the most important partner in terms of Turkey's 'return to the arc of crisis'". In line with the political reconciliation, the Turkish-Iranian trade volume rose for the first time since 2012, standing at \$10.7 billion at the end of 2017, and representing a 1.1 billion USD increase from 2016 (Cetingules 2018). Turkey did not want a marginalised and isolated Iran, as one of its most significant energy suppliers, whereas Iran did not want to lose its significant economic partner and supporter in the international arena (Küntay 2014, 253). In the same vein, Turkey did not support the imposition of severe sanctions against Iran and supported the Iranian nuclear agreement. Turkey supported Iran's right to use nuclear energy for peaceful purposes and criticised the double standards in US non-proliferation policies towards Iran and Israel. In

essence, Turkey's attempts to promote confidence between Iran and the international community, its emphasis on diplomatic solutions and rejection of military attacks against Iran, and its opposition to further economic sanctions—especially the unilateral sanctions led by the US—are significant elements of the rapprochement process (Uzun 2013, 148). Iran is a significant energy supplier, and so it was not easy to decrease Turkey's dependence on Iran for economic, technological and geographical reasons (Küntay 2014, 250). Table 5.1 illustrates that Turkey maintained close economic and trade affairs with Iran, despite the conflicting issues between them. However, Turkey had to follow the requirements of the US economic sanctions against Iran, because it did not want to be at loggerheads with the US due to their long-term partnership (Expert 4, personal communication, July 24, 2020). As such, Turkey's economic interest-driven approach towards Iran is in line with Schweller's balance of interest theory, since Ankara perceived economic ties more beneficial than competition; therefore, Ankara opted to ally with Tehran in the short-to-medium term.

Year	Turkey-Iran	Turkey-Iran	Turkey-Iran Total
	Exports	Imports	Trade
2011	3.590	12.460	16.050
2012	9.920	11.970	21.890
2013	4.190	10.380	14.570
2014	3.880	9.830	13.710
2015	3.670	6.100	9.770
2016	4.500	4.700	9.200
2017	3.300	7.500	10.800

Table 5-1: Turkey-Iran Trade Volume (Billion \$) (www.tuik.gov.tr)

5.3 Concluding Remarks

From the beginning of the Arab Uprisings, Turkey maintained an up-and-down relationship with Iran. There were times when Turkey joined in a coalition against Iran and times when it collaborated closely with it. Although Turkey competed with Iran to influence the trajectory of the Arab Uprisings, Turkish domestic factors brought a detente which shaped TFP towards Iran. As such, NCR explains this cooperation by arguing that unit-level variables play an influential role in shaping a state's foreign policy in the short to medium term.

NCR emphasises the pivotal role of policymakers' perceptions with regard to their state's position in the international balance of power. The AKP elite's perception thus culminated in a shift in TFP. In particular, their Sunni identity played a central role in shaping TFP in the first years of the Arab Uprisings. Sunni rooted AKP elites attempted to establish close affairs with the MB affiliated-parties due to their ideological similarities, thus overtly supporting Morsi's rule in Egypt, the Ennahda-led government in Tunisia and Sunni opposition groups in Syria. Given the elites' sectarian perception, they prioritised their sectarian identity in shaping foreign policy towards Iran and were disturbed by the rise of Shia Iranian influence in the region at a certain extent.

However, irrational policies and the rise of authoritarian tendencies among the AKP elites culminated in Turkey's isolation in regional and international politics. The decrease in the AKP's domestic popularity, and the rise of the Kurdish HDP party, increased the nationalist perspective among the AKP elites, who began cooperating with the ultra-nationalist MHP party. Accordingly, Turkey put aside its Sunni-based perception of Shia Iran in favour of maintaining close ties. Iran thus emerged as an alternative ally for the AKP elites in a tumultuous period for Turkey. In parallel, Turkey re-evaluated relations with its western allies following the failed military coup in Turkey in July 2016, when Russia and Iran immediately expressed full support for Erdogan unlike the US and the EU. As such, the rise of anti-westernism in Turkish politics also pushed Turkey under the AKP to greater cooperation with Iran.

Domestic considerations emerged as another unit-level factor in TFP towards Iran. While Turkey lost its significant economic partners, such as Libya and Syria, Turkey's overambitious and pro-MB policies accelerated its financial losses from the outset of the uprisings. Turkey experienced strained economic affairs with other regional states, including Saudi Arabia and Egypt, and western allies such as the US. In this regard, Iran maintained its economic importance for Turkey to compensate for the latter's financial losses. Moreover, the growing anarchic environment rapidly spreading in the region due to the advent of ISIS and the PYD/YPG, the regionally isolated Turkey saw Iran as a new ally that could cooperate regarding regional stability. Notably, the rise of the YPG/PYD in the Syrian conflict forced Turkey to change its policy on tackling Iran due to the shift in its security priority. The reason for Turkey's desire for close ties with Iran lies in Turkey's interests and gains: by assuming that warmer ties with Iran would benefit Turkish interests, Turkey opted to ally with Iran. This indicates that the balance of interest theory can account for the increased degree of cooperation in TFP towards Iran; put simply, Ankara needed Tehran.

Overall, this chapter points out that structural imperatives alone are insufficient to reach a thorough understanding of TFP towards Iran. Indeed, according to structural imperatives, Turkey would have been at odds with Iran since the uprisings because their interests clearly overlapped as the two significant middle powers in the region. Accordingly, they perceived each other as rivals based on their similar relative power position in the region. The importance of the AKP elite's perception and domestic motivations as intervening variables demonstrate that systemic structures are not the only major determinants of a state's foreign policy, especially in the short-to-medium term. Through the intervening variables of TFP towards Iran, Turkey adopted profit-oriented and interest-driven policies to manage its relationship with Iran and maintain amity rather than enmity whenever closer ties were more beneficial than rivalry. However, as Academic 4 (personal communication, July 6, 2020) noted, Turkey's collaboration with Iran was for a ceasefire, not a peace. When they have mutual concerns, they collaborate, but this is not consistent. Turkey's fluctuant accusation against Iran over Tehran's alleged support to PKK and Ankara's long-term requirement for US support due to its middle power capacity resulted in short-term rapprochement with Iran. Therefore, unit-level variables played a role in the short-to-medium term in TFP towards Iran. Otherwise, Turkey would have maintained its competition with Iran in the longer term due to both states' middle power activism in the region. This indicates that the main determinant of TFP towards Iran is systemic variables, including their relative power capacity, even though Turkey followed rapprochement with Iran in the short term.

This chapter has elaborated on the main theoretical research questions by explaining how systemic determinants and Turkey's relative power have impacted Turkey's policies towards Iran, while also considering unit-level variables (the AKP elite's perception and domestic considerations) from December 2010 to July 2018. The next two chapters shed light on two significant case studies as empirical discussions to further understand the causal chain between systemic and unit-level variables. In particular, the Syrian conflict and the Kurdistan Region of Iraq (KRI) were selected as case studies for two reasons. First, they help explain the anarchic and competitive international and regional dynamics resulting from the Arab Uprisings and their effects on TFP behaviour towards Iran. Second, they facilitate an understanding of the influence of Turkey's domestic politics and society on the implementation of its revised policy choices towards Iran. In this regard, the Syrian conflict is the first case study of the thesis to corroborate the role of independent and intervening variables on TFP towards Iran since the Arab Uprisings.

CHAPTER 6 TURKISH FOREIGN POLICY TOWARDS IRAN SINCE THE ARAB UPRISINGS (DECEMBER 2010-JULY 2018): THE CASE OF THE SYRIAN CONFLICT

This chapter discusses Turkey's role in the Syrian conflict in the context of interactions with Iran, by asking what the motivation behind Turkey's stance towards Iran was from the outset of the Syrian conflict to July 2018. Therefore, the focus in this chapter concerns how Turkey's relative power position, the AKP elite's perceptions, and the domestic considerations of Turkey shaped TFP towards Iran in the Syrian conflict from March 2011 to July 2018. To understand TFP towards Iran, the chapter is divided into three areas: first, the role of systemic (independent) variables, including certain regional political shifts deriving from the Syrian conflict and Turkey's relative power position; second, the role of Turkish domestic factors regarding the Syrian conflict as unit-level (intervening) variables ---the significance of the AKP elite's perception and domestic security and economic considerations for Turkey's relationship with Iran in the Syrian conflict; and, third, how such variables played filtering roles between systemic factors and TFP choices towards Iran. The influence of unit-level variables is important because they forced Turkey to conduct a short-term collaborative foreign policy towards Iran, even though systemic factors were pushing for the exact opposite. As such, Schweller's balance of interest theory can sufficiently explain Turkey's alliance with Iran, especially for the analysis of domestic considerations. The chapter concludes that NCR convincingly explains Turkey's changing attitudes towards Iran from the outset of the Syrian conflict to July 2018.

6.1 The Role of the System-Level on Turkish Foreign Policy towards Iran since the outset of the Syrian Conflict: March 2011 to July 2018

NCR argues that the balance of power and relative power position of a country shape a state's foreign policy as systemic determinants. Understanding Turkey's policy attitude towards Iran since the outset of the Syrian conflict in terms of the effects of systemic determinants, therefore, requires an assessment of state and non-state armed actors' respective interests and their position towards the conflict, as well as Turkey's relative power position in the conflict.

Accordingly, this section analyses the internationalisation of the Syrian conflict and Turkey's relative power position within it as systemic (independent) variables. These determinants had significant impacts on TFP towards Iran because Turkey has deemed Iran a competitor in the Syrian conflict, particularly for its long-term interests, due to the major regional systemic shifts and their similar position in the international power structure.

The Syrian conflict became an anarchic environment in which there were severe regional competition from the outbreak of the conflict to July 2018, thus accelerating the multipolarity of the context and numerous regional aspirations (C. Phillips 2017b, 37). Numerous external and regional states sought to shape the conflict in terms of their interests and goals in the Middle East by either directly intervening or indirectly controlling the future of the conflict and by backing non-state armed groups. Their seek of enhancing their interests thus ignited inter-state competition in Syria. As well as states, non-state armed actors participated in this competition to influence their regional position by means of state support. These developments internationalised the conflict and shifted regional power relations. To understand the role of systemic (independent variables) on TFP towards Iran, this section is thus divided two subsections: inter-state competition and the rise of non-state armed groups.

6.1.1 Inter-State Competition

In the tumultuous environment of the Syrian conflict, states wondered who would gain most from the newly emerged political sphere and win a relative advantage over others. To do so, numerous international and regional states explicitly formed groups in two camps: either supporting or opposing the Assad regime. The pro-Assad group of Russia, Iran, and China aimed to keep the regime in power, while other states, such as the US, Turkey, and Saudi Arabia, aimed for regime change. Since Turkey and Iran had different goals in Syria, interstate competition determined Turkey's attitude towards Iran.

The pro-Assad powers liaised with the regime on the basis of their regional interests. Russia sought to increase its international relative power position over the US through the Syrian conflict. In Charountaki (2018, 251) words, "the Syrian crisis granted Russia a means of reviving its determinant and influential role in the Middle Eastern region and helped it make a comeback on the regional political scene". Indeed, the Assad regime has always been an indispensable Russian ally, with the Russia's -only Mediterranean naval base being located at the Syrian port of Tartus. Moscow emphasised the protection of the Assad regime to maintain its military base, which provided Russian forces with unconditional access to the Mediterranean (Carpenter 2013, 9; Pierini 2016, 7). Moscow criticised the interventionist policies of western actors and their regional allies, who opposed the Assad regime, and was particularly concerned about a possible NATO military campaign to attack Syria under the UN Security Council Resolution 1973, that looked to replicate the Libyan intervention in March 2011. As such, in February 2012, Moscow used its veto against the UN Security Council resolution, which would have demanded an end to all violence and reprisals against the regime in Damascus for its use of chemical weapons against its own citizens. In doing so, Moscow assumed that a western-led intervention in Syria would result in a similar outcome to the Libyan intervention; this outcome would impact Russia adversely because of the potential for increasing US influence across the region at Moscow's expense (Crosston 2014, 95; C. Phillips 2017b, 41–96). Russia could therefore prevent further western-led regime changes in the Middle East by supporting the status quo and shaping the conflict in terms of its regional intervests.

China was another supporter of the status quo in the Syrian conflict and aimed to maintain its Middle Eastern interests despite its limited ties with the Assad regime. At the outset of the Syrian conflict, China was one of Syria's main trading partners and a crucial participant in Syria's oil industry. Like Moscow, Beijing considered that the US and its allies would try to subvert regional order by demanding a regime change in Syria, and thus used its veto as a permanent member in the UNSC against any condemnation of the Assad regime in February 2012. In this regard, Wang Keihan, then Deputy Director at the Chinese Foreign Ministry, blamed some western countries in early August 2012, for hindering and sabotaging the diplomatic process of the Syrian conflict, and for pushing for regime change in Syria (Carpenter 2013, 9). This statement indicates that the removal of the Assad regime would be against Chinese interests in the Middle East, and Beijing maintained the support of the Assad regime as long as protecting it would increase China's regional power.

As well as Russia and China, Iran depicted the Assad regime as a valuable ally in the Middle East due to its strategic and historical ties with Damascus. Ever since the Islamic Revolution in Iran, Syria has been an Iranian ally alongside Hezbollah, the Shia Islamist political party and militant group based in Lebanon. In particularly, Syria's proximity to Israel served as a convenient conduit to Hezbollah in Iran's "proxy confrontation" with Israel

(Ospina, Gray, and Creek 2014; C. Phillips 2017b, 40). In this regard, the Iranian diplomat for Middle East Affairs, Hossein Amir Abdollahian, stated that "we don't [want] permanent collaboration with the Assad regime but there is no better option. Syria is seen as a door between Iran and Hezbollah and an only Arab ally in the region" (Academic 2, personal communication, August 6, 2020). Like Russia, Iran was also concerned about the US presence in the region, because Tehran felt US pressure and assumed that a US intervention in Syria would promote Israeli interests and threaten Iran's presence and interests. As such, Tehran initially perceived the Syrian conflict as the outcome of "a pre-planned agenda by western countries aimed at weakening and destroying Islamic resistance opposed to the Zionist regime [Israel]" (Akbarzadeh and Barry 2017, 5; Academic 2, personal communication, August 6 2020). Since Iran viewed Syria as a buffer zone between the US and Israel, Tehran immediately sent military and financial aid to Assad following Washington's declaration in late summer, 2011, of seeking to remove the Assad regime (Crosston 2014, 104; Journalist 2, personal communication, August 14, 2020). As well as the US, Sunni Wahhabi Saudi Arabia was a prominent rival for Shia Iran in a sectarian proxy war that increasingly defined the Syrian civil war, as a Sunni-based regime in Syria could undermine Iranian regional interests. Accordingly, Tehran viewed the survival of the Assad regime as vital to Iran's regional interests.

On the other hand, the anti-Assad group sought to topple the regime, and aimed to block pro-Assad policies. The US was notably concerned about the Chinese and Russian vetoes regarding the Assad regime at the UNSC. For instance, in February 2012, Susan Rice, former US ambassador to the UN, pointed out that China and Russia had taken shameful and unforgivable action by vetoing policies against the Assad regime, and the US was tired of such acts. Likewise, Hillary Clinton, then US Secretary of State, proclaimed that the use of a veto by China and Russia was degrading when people were being murdered in Syria (Carpenter 2013, 8). In addition to her criticism of Russia and China, she further claimed that Iran would be a significant lifeline for the Assad regime to maintain its "proxy" and "crony" in Syria (Crosston 2014, 102). In this regard, Washington was concerned about the policies of the pro-Assad states in the conflict.

Other actors too supported the removal of Assad. While the EU played a limited role due to policy differences among its members (Ayman 2014, 22), Saudi Arabia, Qatar and Turkey, as the leading opposition states, shared a vision of actively and directly supporting the Syrian opposition groups. To achieve their goal, each state implemented diplomatic, military

and financial policies, such as diplomatic and economic sanctions, supplying arms to opposition groups, and pressuring member states of the UNSC—particularly the US —to intervene in Syria (C. Phillips 2017b, 36). These three states even contributed to the establishment of the Syrian National Council (SNC), and then participated in the International Contact Group known as the Friends of Syria (Hokayem 2014, 65). Importantly, sectarian policies determined their Syrian policies because each state aimed to prevent the rise of Iranian influence across the Middle East, and to change the balance of power in their favour (C. Phillips 2016, 39). In so doing, they sought US assistance, and tried to persuade the US to intervene in the Syrian conflict. As a result, the anti- and pro-Assad groups competed for regional supremacy and to alter the balance of power in their favour in the first years of the conflict.

Moreover, the sectarian competition between Saudi- and Iranian-led groups was pronounced in the Syrian civil war: Saudi Arabia supported the Sunni group including Turkey, Qatar, and anti-Assad opposition groups, such as the Free Syrian Army and the Islamic Front, whereas Iran sided openly with the Shia groups, including Hezbollah as well as the Assad regime at a certain extent (Aras and Yorulmazlar 2016a, 2268). As a result, the Syrian conflict became the battlefield for a Sunni-Shia proxy war in the Middle East allowing sectarianism to become a dominant characteristic of shifting alliances that now recast much of the regional order.

This inter-state competition led to a clear deterioration of TFP towards Iran since Ankara first competed with Tehran and supported anti-Assad groups in Syria. Turkey attempted to increase its regional influence through a reconfiguration of the balance of power, and at the same time to prevent the rise of Iranian influence; thus, Turkey initially pursued largely US-led policies in the Syrian conflict. To consolidate its regional influence, Ankara followed a Saudi-led Sunni policy against the Assad regime and indirectly against Tehran (Ayman 2014, 20–21; Crosston 2014, 104). In this regard, Iran's close ties with the Assad regime disturbed Turkey, since Iran could gain leverage over Turkey in the shifting regional balance of power (Ataman and Özdemir 2018, 23). To change this possibility, Turkey played an instrumental role in facilitating the Free Syrian Army as the main opposition to the Assad regime, by allowing Turkish territory to be used for the transfer of funds, weapons, and recruits. This exacerbated tensions with Iran. Although Ankara insisted that its Syrian policy was in favour of democracy, reforms and stability, and was not sectarian, it propped up Islamist armed opposition groups, such as Al-Nusra, against the Assad regime, and even initially ignored ISIS's aggressive policies in Syria since the jihadist group was fighting against the Assad regime (Aras and Yorulmazlar 2014, 115–16). Therefore, an indirect Turkish-Iranian proxy war also emerged with the beginning of the Syrian conflict in 2011 and TFP towards Iran shifted from "polite competition to delicate brinkmanship" (Oktav 2013, 201). Similarly, Academic 2 (personal communication, August 6, 2020) noted that there was hard balancing and competition between Turkey and Iran. Between 2012–2016 especially, there was high tension even though they tried to mitigate their tension, and yet, in the beginning of the Syrian conflict, their attempt at detente was seen as difficult because hard balancing played a determinant role. As such, Turkey's unfriendly behaviour towards Iran demonstrated that Turkey perceived Iranian policies in Syria as a threat against its long-term regional interests.

At the outbreak of the Syrian conflict, Turkey, despite its initial hesitance, arguably adopted a more militarist and security-oriented stance against the Assad regime by propping up Syrian opposition groups. According to Expert 3 (personal communication, July 21, 2020), the Syrian conflict shifted from political armed conflict to military conflict due to the involvement of external powers. In such an environment, Turkey opted to support the opposition and followed a western-based policy which meant close ties with the US and EU, and sanctions were announced against the Assad regime akin to US and EU policies (Erdogan 2016, 116). In doing this, Ankara tried to play a significant role in shaping the balance of power and enhancing its regional power situation in the Syrian conflict.

As the conflict intensified, Turkey opted for a more interventionist policy against the Assad regime than its western allies, particularly the US, and increased military aid to the Syrian opposition, notably the Free Syrian Army (FSA), by sending arms and Arab Islamist fighters into Syria (Gupta 2016, 33). Turkey became a significant state actor by hosting the major Syrian opposition institutions of the SNC and the leadership of the FSA, and by allowing the transfer of arms, funds, and humanitarian aid to reach rebel-controlled areas in northern Syria (Lynch 2016b, 141). As well as financial and military aid, Ankara endeavoured to establish both no-fly and buffer zones for the Syrian opposition groups to enhance its regional power and influence in Syria. In particular, the Turkmen-Sunnis inhabiting the mountains in Latakia Province and the Sunni Islamist groups, such as the MB-affiliated opposition, became pivotal allies for Turkey in its attempts to achieve its regional interests in the conflict. In return, these groups promoted Turkey as their regional ally. For instance, Mohammed Riad Shaqfa, the

leader of Syria's MB, said in November 2011, that "he would accept military intervention to stop the Assad regime's deadly crackdown on pro-democracy protesters if it was led by Turkey instead of western powers" (Oktav 2013, 199). In this way, Ankara liaised with various opposition groups to achieve its regional goals.

Nevertheless, sectarianism did indeed play a limited role in TFP towards Iran, owing to the different viewpoints between Turkey and Saudi Arabia on supporting different Sunni groups. While Riyadh preferred backing mostly moderate Sunni groups, Ankara provided military and financial aid to more radical Islamist groups and MB-affiliated groups (Ayman 2014, 20–21). In this regard, a Saudi-Turkish alliance could become ineffective against the rise of Iranian influence, and Turkey then played a balancing role between Iran and Saudi-Arabia (Ayoob 2011, 64).

Furthermore, Ankara sought to establish warming ties with Tehran owing to its complicated affairs with other certain countries, such as the US. While pursuing a US policy in the Syrian conflict, Ankara did indeed adopt different goals to the US. For instance, Turkey insisted on toppling the Assad regime in the first years of the Syrian conflict, while the US changed its priorities from deposing the regime to protecting US national security interests in the region; this clearly formed a breaking point between the two allies and put severe tension on their bilateral ties. Even though inter-state competition determined TFP towards Iran at the first years of the Syrian conflict, Turkey tried to avoid any high-level tension with Iran. It seems that other factors were also key in shaping TFP towards Iran in the Syrian conflict.

6.1.2 The rise of non-state armed groups

The rise of non-state armed groups was another determining factor in the changing regional balance of power from the outset of the Syrian conflict to July 2018. The anarchic environment stemming from the conflict emboldened numerous non-state armed actors to strengthen their position and compete in a period of rapid change in the regional system. Owing to their antagonistic and separatist policies, they aggravated the anarchic environment of the conflict, with religious or nationalist (C. Phillips 2016, 24).

Religious-based policies were arguably more visible in Shia-dominated Hezbollah and Wahhabi-based ISIS' activism in the Syrian conflict when they attempted to cement their regional influence and position. For instance, Hezbollah sent almost 2,000 dedicated fighters to defend the Assad regime and played a significant role in capturing al-Qusayr in June 2013, on behalf of the Assad regime (Byman 2014, 92–93). As well as Hezbollah, various Iraqi Shia armed groups, such as Asa'ib Ahl al-Haq, became prominent in supporting the Assad regime (Academic 1, personal communication, August 4, 2020). ISIS, with its Salafi doctrine, also emerged as a jihadist armed group in Syria, especially in the second half of 2014, and established a strict religious order in the primarily Sunni areas of Syria (Unver 2018, 40). Its small but relatively well-armed and equipped military force and its insurgent tactics made ISIS the major opposition force in Syria. Strong financial reserves derived from oil smuggling, bank robberies, ransom money, and the forceful collection of taxes also enabled ISIS to hire recruits while its military victories in 2014 emboldened other extremist jihadist movements to increase their terrorist activities in Syria (Işiksal 2018, 26).

Various international and regional states also encouraged these armed groups to implement sectarian policies in the Syrian conflict, seeking to use them for their regional interests and goals. Saudi Arabia and Iran arguably demonstrated the aforementioned solidarity with numerous fundamentalist groups to enhance their regional power position and caused a "proxy war" in the conflict. Riyadh particularly endorsed the small Sunni groups, such as the Farouq Brigade, within the SNC both to topple the Assad regime and prevent the rise of Shia influence (Baylouny and Mullins 2018, 1003). In contrast to Riyadh, Tehran sent financial and/or military aid to Shia groups in the Syrian conflict. According to Juneau (2018, 32), "Iran has backed Hezbollah's growing role in Syria. Though Hezbollah has significantly diversified its sources of support over the years, much of its advanced weaponry still comes from Iran, through Syria". Therefore, state support to the non-state armed groups increased the already anarchic environment in the conflict.

Despite cordial ties between non-state armed groups and states over sectarianism, however, individual states found themselves in a dilemma over the empowerment of jihadist groups, as their victory would encourage home-grown militant groups. To illustrate, Moscow was frightened that the victory of jihadist groups in the Syrian conflict would influence Russian jihadists, and thus it boosted its political and military aid to the Assad regime by sending an air force and up to 2,000 military personnel in September 2015 (C. Phillips 2017b, 41). This allowed Russia to launch intensive air campaigns, mostly against the rebels. The rise of Sunni groups also did not fall entirely in line with Iran's regional expectations, which were to increase

Shia influence in the Middle East and keep the Alawite Assad regime in power as Iranian ally. For Iran, in mid-2012, the conflict became worse when Sunni groups and tribes started to revolt against the regime following the receipt of financial and military support from major Sunni-dominated states (Saudi Arabia, Qatar and Turkey), and the US and its western allies (Gupta 2016, 31). In this tumultuous environment, states used the particular non-state armed groups, which had similar ideology or aim, to enhance their interests and prevent the growing influence of other state actors.

According to Journalist 2 (personal communication, August 14, 2020), the rivalry between Ankara and Tehran in the 1990s re-emerged in the Syrian conflict despite Turkey's quest for cooperation with Iran in the 2000s. Given that Turkey sought to improve its regional power position by playing a more active role and increasing its support to the Sunni opposition groups, TFP towards Iran had the potential for intense competition with no short-term possibility of improvement at the beginning of the conflict. In this environment, Turkey directly propped up Sunni armed groups against the Alawite Assad regime but did so indirectly against Iran. For instance, Ahrar al-Sham, which has much in common with al-Nusra and adopts Salafi jihadist ideology and extreme sectarianism, became a significant partner for Turkey, Qatar, and the Gulf countries. Turkey, along with Qatar and Saudi Arabia, further pioneered the foundation of Jaysh al-Fateh, another al-Nusra affiliated jihadist group. With aid from these three states, Jaysh al-Fateh captured Idlib and advanced in Latakia and other significant regime areas (Lynch 2016b, 143, 154, 169). To achieve the overthrow of the Assad regime, moreover, Turkey overlooked the flow of fighters, money and weapons into ISIS-controlled areas, which facilitated the rise of jihadist groups (Zaman 2014). Accordingly, Turkey's Sunni-based policies increased its confrontation with Iran.

According to Academic 1 (personal communication, August 4, 2020), Turkey did not even hesitate to target pro-Iranian militias in Syria. Ankara envisaged more relative advantage against Iran, as the former believed it had the capability to shape the Syrian conflict by means of close ties with the opposition groups. In doing so, Ankara endeavoured to obtain the most benefit from the new political environment and a relative advantage over Iran; Turkey foresaw that its support for Sunni groups would strengthen its international position and allow it to establish much closer economic relations with Syria in the post-Assad period. As such, Turkey, although hoping to "play an order-instituting role in the region, with western support" (Oktav 2013, 199), implemented a more interventionist policy, and this was a sign of the discord between Turkey and Iran. Therefore, Turkey increased competition towards Iran, claiming its position as the leading regional power.

Nevertheless, the divisions within Sunni-dominated state groups prevented the growth of indirect competition between Turkey and Iran. Whereas Riyadh propped up the Salafi rebels, Turkey, along with Qatar, supported MB-affiliated groups. Saudi and other Gulf states were thus at odds over their support for these groups and, more importantly, viewed them as a threat to domestic security (Lynch 2016b, 94,142). This caused another regional political struggle, which damaged the Ankara-Riyadh alliance against the rise of Shia influence. Turkey was also at loggerheads with the US over the latter's supply of military and financial aid to some jihadist groups. For instance, former Vice President Joseph Biden stated in October 2014,

Our allies in the region were our largest problem... [They] were so determined to take down (Syrian President) Assad and have a proxy Sunni-Shia war, they poured hundreds of millions of dollars and thousands of tons weapons into anyone who would fight against Assad. Except that the people who were being supplied were (Jabbat) al Nusra and al Qaida, and the extremist elements of jihadis coming from other parts of the world (Parasiliti, Reedy, and Wasser 2017, 2).

In particular, Turkey, along with Qatar, was willing to establish close ties with the al-Qaeda affiliate Jabhat al-Nusra, which was seen as a terrorist group by the US. As such, Washington had Ankara in its crosshairs owing to reasonable concerns regarding the extremist groups which might ultimately use the weapons to attack the US and its regional allies (Stein 2015a). As a result, Ankara's policies in the conflict started to be questioned by its Western and Sunni-dominated state allies.

As well as sectarianism, nationalist factors became visible in the policies of non-state armed actors. Notably, the PYD pursued its nationalist aims in the Syrian conflict, and thus obtained a de facto autonomy in 2012 when the Assad regime forces left control of three mainly Kurdish areas (Afrin, Jazira and Kobane) to local militias. Therefore, PYD arguably became a key player by taking over large swathes of land in northern Syria, which attracted the attention of various international and regional actors. Notably, the US established cordial ties with the PYD in the fighting against ISIS. However, the PYD posed a national security threat to some states, including Turkey. With the empowerment of the PYD, Ankara shifted its priorities from preventing the rise of Iranian influence to protecting its national security. Turkey's several western allies occasionally criticised its military operations against the PYD (S. Aydin and Ucar 2018), and thus Turkey fuelled the need for Iran's partnership in combating the PYD for a short-term period.

From the beginning of the Syrian conflict to July 2018, Ankara aspired to influence developments in the conflict in favour of its long-term interests, including enhancing its power position. According to Expert 4 (personal communication, July 24, 2020), Turkey's affairs with Iran depended on a global state actor, including the US and EU due to its limited power capabilities. At the beginning of the Syrian conflict, Ankara thus sought US/EU collaboration to topple the Assad regime and prevent the rise of Iranian influence. However, it was disappointed with the EU's limited role and US reluctance to be drawn further into the conflict. Turkey thus sought to collaborate with Russia, as another global state power, while also increasing support for the disorganised opposition groups. What this indicates is that Ankara was maintaining its rivalry with Iran not only to gain a more advantageous position but also a higher profile than Iran across the region in the longer term. However, domestic factors compelled Ankara to ally with Tehran in the short-to-medium term. In the next section, there is a focus on how unit-level factors operated regarding Turkey's relative power position in the anarchic environment and TFP towards Iran from the beginning of the Syrian conflict to July 2018.

6.2 The Role of the Unit Level (Intervening Variables) on Turkish Foreign Policy towards Iran since the onset of the Syrian Conflict: March 2011 to July 2018

While seeing Iran as a rival for its long-term regional interests, Ankara sought a short-term alliance with Tehran for domestic reasons. Therefore, without the context of the intervening (unit-level) variables, TFP towards Iran in the Syrian conflict cannot be understood, as domestic factors greatly determined the character, type, and adjustment of Turkish attitudes towards Iran. In this section, the AKP elite's perception of the Syrian conflict is examined as a domestic factor determining TFP towards Iran, and there is also an analysis of how profit-oriented security and economic considerations prevailed over Turkey's competition with Iran. Overall, in this section it is argued that the AKP elites' perception determined TFP towards Iran, and that Turkey followed profit-oriented and interest-based policies towards Iran, in line with Schweller's balance of interest theory.

6.2.1 The AKP Elites' Perception

In the Syrian conflict, the AKP elite's perception determined Turkey's policy behaviour, since, as Hinnebusch (2015, 15) says, "Turkey's response to the Syrian conflict was a function of the way the change in Turkey's identity under the AKP reshaped the government's conceptions of Turkey's interests and of the threats to them". That is to say, the AKP elites' identity, policies and perceptions regarding the conflict played a significant role in shaping TFP towards Iran.

Before the Arab Uprisings, the AKP elites had shared a common vision with the Assad regime and Erdogan and Basher Assad had become close friends. Other remarks from the AKP elites regarding Bashar Assad and the Syrians had also been friendly, calling them "friends", "family", and "brothers". In return, after becoming the first Syrian leader to visit Ankara in 2004, Bashar al-Assad compromised on the territorial conflict over Hatay⁴ in 2005 and the Euphrates water issue⁵ in 2008. This cordiality culminated in a reduction in the tension of the 1990s derived from Syria's support for the PKK and Turkish construction on the Euphrates, but also the rise of economic and political interaction between the two countries. Accordingly, the AKP elites did not envisaged that similar outcomes to the Egyptian or Tunisian Uprisings would occur in Syria, and few even believed that Turkey would lead an anti-Assad group in the conflict (C. Phillips 2016, 36, 59).

Following the onset of the conflict in March 2011, Turkish policymakers expressed their support for Assad, and even Erdogan portrayed Bashar Hafez al-Assad as a good friend

⁴ After one-year independence, Hatay (the Sanjak of Alexandra), which had been previously under French occupation, unified with Turkey in 1939. Therefore, Hatay became a territorial issue between Turkey and Syria which never approved this unification by believing that this unification was illegal and Western imperial designed. See details: Sanjian, A. K. (1956). The Sanjak of Alexandretta (Hatay): Its Impact on Turkish-Syrian Relations (1939-1956). *Middle East Journal*, *10*(4), 379-394.; Altunışık, M. B., & Tür, Ö. (2006). From distant neighbors to partners? Changing Syrian-Turkish relations. *Security Dialogue*, *37*(2), 229-248.; Magued, S. (2019). Turkey's economic rapprochement towards Syria and the territorial conflict over Hatay. *Mediterranean Politics*, *24*(1), 20-39.

⁵ Since the 1980s, Turkey and Syria had water conflict over the control of the water flow of the Euphrates. See details: Güner, S. (1997). The Turkish-Syrian war of attrition: The water dispute. *Studies in conflict & terrorism*, 20(1), 105-116.; Carkoglu, A., & Eder, M. (2001). Domestic concerns and the water conflict over the Euphrates-Tigris river basin. *Middle Eastern Studies*, 37(1), 41-71.; Scheumann, W. (2003). The euphrates issue in turkish-syrian relations. In *Security and Environment in the Mediterranean* (pp. 745-760). Springer, Berlin, Heidelberg.

loved by his people. Ankara urged al-Assad to enact democratic reform and endeavoured to play a mediating role between the Assad regime and the opposition groups. The main motivation behind Turkey's desire for a mediating role was its wish to "demonstrate its value as a regional interlocutor for the West, owing to the assumed leverage over [Assad] that would allow Ankara to steer him toward peaceful political change" (Hinnebusch 2015, 17). In this regard, Davutoglu, then Foreign Affairs Minister, and Hakan Fidan, then head of Turkey's National Intelligence Organisation (MIT), paid several visits to Damascus to convey Turkey's suggestions regarding reforms and persuade the Assad regime to accept Turkey's mediation with the opposition to find common ground in Syria. However, the Assad regime not only rejected this idea but perceived Turkey's suggestions to be an intervention in Syria's domestic issues, since the AKP elites demanded Assad's resignation (Barkey 2014, 104). Consequently, in August 2011, the AKP elites became increasingly critical of the Assad regime when it ultimately refused Turkey's call for reform and dialogue. In March 2012, Ankara even closed its embassy in Damascus and started to endorse the Syrian opposition in April 2012. The elites thus shifted their position from remaining neutral to gradually supporting the Syrian opposition. In doing so, they promoted Turkey as a defender of international law, human rights, democracy, core Islamic values and trans-religious principles, and as a moral actor in the Middle East standing with the oppressed (Güney Ateşoğlu 2013, 56). Following the downing of a Turkish jet by Syrian forces in June 2012, then Prime Minister Erdogan further declared that "every military element that approaches the Turkish border from Syria in a manner that constitutes a security risk or danger would be considered as a threat and would be treated as a military target"(The New York Times 2012). Therefore, Davutoglu frequently underlined that TFP towards the Syrian conflict was related to Turkish national and security interests, independent of external pressure or directives (Taşpınar 2012, 138–39). These remarks indicate that the AKP elites were heavily involved in the Syrian conflict and started to implement an interventionist foreign policy.

In this increasingly turbulent environment, the AKP elites aimed to present Turkey as a significant regional actor in the new era by utilising a "value-based discourse" (Demirtas-Bagdonas 2014, 147–48), but they initially perceived Iran as an obstacle to achieving this since each country was propping up different groups. The AKP elites further increased their unwavering support for the Syrian opposition by pioneering the foundation of the Friends of Syria Group, established in 2012, to exert pressure on the Syrian regime, and by allowing the nominal head of the FSA to base themselves in Turkey (C. Phillips 2017a, 39). Moreover, the

AKP government demanded that the US "set up no-fly zones or protected areas along the Turkish–Syrian border, just as it had done in northern Iraq during the 1990s" (Barkey 2014, 105). In doing so, the AKP elites sought to become a significant actor in the conflict and to gain leverage over Tehran.

While the conflict was intensifying, the AKP elites increased their critical discourse indirectly against Iran, as well as the Assad regime. For instance, in 2012, then Deputy Prime Minister Bulent Arinc accused Tehran of ignoring the violence in Homs, where almost 200 protesters had been killed (Ünver 2012, 106). In July 2013, Ahmet Davutoglu also stated that "all foreigners fighting in Syria [were] to immediately leave the country", implying Hezbollah, Tehran's Shia ally (Hurriyet Daily News 2013b). Behind their critique, the AKP elite's Sunnibased identity emerged as a foreign policy tool towards both the Syrian conflict and Tehran. They believed that the conflict would provide the MB with an opportunity to take over the Syrian regime, and hoped that an MB-affiliated government would replace the Assad regime in the post-conflict period (Okyay 2017, 834). Even before Turkish-Syrian ties deteriorated, the AKP had demanded the release of the MB's leaders and power-sharing between the Assad regime and the MB (Gupta 2016, 33). Accordingly, they deemed the MB to be a platform upon which Sunni solidarity could be established on Turkey's border, a belief that further complicated ties between Turkey and Iran. They also turned a blind eye to the movement of Sunni fighters and arms into Syria (Barkey 2014, 105). Based on this sectarian-based perception, therefore, the AKP elites opposed Iranian policies in the conflict.

Towards the end of 2013, however, Ankara revised its foreign policy towards Tehran due to the miscalculations made by the AKP at the beginning of the conflict. First, the AKP elites overestimated Turkey's ability to influence the Syrian conflict (Barkey 2014, 104). Propping up the Syrian opposition, the AKP elites envisaged that Assad would be deposed in six months, and thus rushed to support Sunni opposition groups by organising conferences in Istanbul and assisting with the formation of the FSA. Despite having lost some territory, the Assad regime still governed significant western parts of Syria and maintained power by means of Russian and Iranian support. Therefore, Turkey failed to realise that the toppling of the Assad regime would have a systemic effect, miscalculating the roles of Russia and Iran and the US in Syria (Expert 3, personal communication, July 21, 2020). At the same time, Syrian opposition groups were disorganised and had limited capability to influence the future of the Syrian conflict (C. Phillips 2016) while Turkey's relations with Saudi-led Sunni countries were

at stake due to the AKP elite's support for the MB-affiliated opposition groups. Turkey lost numerous Sunni state allies in the region, such as Saudi Arabia —except for Qatar, which also supported these groups. As a result, the AKP elites' quest for a Sunni-dominated post-Assad regime backfired; this was despite the fact that the AKP elites had joined the Sunni group of Saudi-led Gulf countries and Qatar, in an attempt to reap an economic dividend for Turkey by securing investment and funds from rich Sunni Gulf countries (Işiksal 2018, 24). This indicates that the elites' assertive sectarian policies and (mis) perception of the policies of other Sunni states in Syria culminated in economic and political losses for Ankara.

The AKP elites further misinterpreted US policies in Syria, assuming that Washington would seek regional transformation (Expert 3 personal communication, July 21, 2020). Notably, then Prime Minister Erdogan envisaged that the US was planning to attack the Assad regime after the presidential election of November 2012, since the Obama administration declared that using chemical weapons in Syria was a "red line". Even so, Washington remained hesitant about becoming mired in the Syrian conflict, a legacy of the Iraq war, and it continued to follow a more cautious policy towards the conflict than Ankara anticipated. As such, Washington merely intervened in the conflict when there was a threat to its interests, and focused more on ISIS than the Assad regime due to the rise of jihadism being perceived as a threat to the US and its allies in the region (C. Phillips 2017a, 41). In essence, Washington considered that there was no alternative Syrian opposition which would be a US ally in the post-Assad era, and deemed the opposition groups to be anti-US and anti-Israel (Ayman 2014, 22). Wary of further military entanglements in the Middle East, the Obama administration also hoped to pivot its stance from the Middle East towards Asia, due to the shift in international politics caused by the rise of China and the general economic and political importance of Asia. In short, the refusal of the Obama administration to enforce its own 'red line' following the chemical attack by Assad's forces in East Ghouta in April 2013 demonstrated that it would not intervene in the conflict (Expert 3, personal communication, July 21, 2020). Therefore, the AKP elites failed to persuade Washington to intervene militarily to topple the Assad regime due to their different perspectives on the future of the Syrian conflict.

In contrast to the US, the AKP elites focused on the overthrow of Assad rather than the rise of jihadists, and even supported jihadist groups, such as Jabhat al-Nusra, to remove the Assad regime. As a result, Turkey's image transformed from one of a "Muslim democracy" to an "Islamic bureaucracy" in international politics (Charountaki 2018, 274). While the number

of nationalist viewpoints increased among the AKP elites, they were also concerned about the US' unwavering support for the PYD in combatting ISIS. As a result, a lack of trust emerged between Ankara and Washington over the Syrian conflict, and notably the AKP elites periodically accused Washington of playing a passive role in the conflict. In this regard, in September 2013, Erdogan declared that "the world is bigger than five", implying the inefficiency of the UNSC in the conflict, including the US (Milliyet 2013). In May 2017, at a joint news conference with President Trump, Erdogan further asserted that "Turkey will never accept US alliance with Kurdish forces [PYD/YPG] fighting in Syria" (BBC News 2017b). Accordingly, US policies in Syria not only evolved diverged from the AKP elite's expectations, but undermined Ankara's policies in Syria.

Given their miscalculations towards the developments of the conflict, the trajectory of the Syrian conflict situation gradually forced Ankara to become aware of its limitations in shaping the new greater political environment. Therefore, the AKP government fuelled the need for alternative partners in Syria, except for the US and Saudi Arabia, which led to a more conciliatory foreign policy towards Tehran. Even though the UN organised the Geneva talks in 2017 to establish peace negotiations between the Syrian government and the opposition groups, Turkey and Iran, along with Russia, simultaneously adopted a triple negotiation system to find a diplomatic solution to the Syrian conflict and to secure Syria's national border from growing international violations. The three countries also conducted numerous mutual meetings and referred to themselves as guarantor countries fighting for peace in Syria (Nechepurenko 2017).

In conclusion, the AKP elites first perceived that the Syrian conflict was an opportunity to enhance Turkey's regional position, and thus followed overambitious policies in the conflict (Kuru 2015). However, their policies led to conflict with Turkey's western allies, including the US, and regional allies, including the Saudi Arabia. Increasingly, they believed that relying only on the US did not serve Turkey's national interests in the Syrian conflict. Accordingly, despite their disagreements, Turkey's ties with Iran in the Syrian conflict were not a choice but a necessity (Academic 4, personal communication, July 6, 2020). Therefore, the AKP elites put aside their sectarian policies towards Tehran based on cost-benefit calculations in the short-to-medium term. As noted in Chapter 5, with the rise of a nationalist and anti-western perspective among the AKP elites and its domestic considerations, Iran became a prominent

ally. The next section thus elucidates how domestic considerations motivated Ankara to ally with Tehran in the short-to-medium term as another intervening (unit-level) variable.

6.2.2 Domestic Considerations

As another intervening variable, domestic considerations played a significant role in shaping Ankara's short-term alliance choice with Tehran from the outset of the Syrian conflict to July 2018. At the beginning of the conflict, Ankara became concerned about the changing security environment derived from the instability on Turkey's border and its direct effects on Turkey's domestic politics. In particular, the rise of ISIS and the YPG/PYD posed a security burden to Turkey since both could revive Turkey's internal security issues. Accordingly, Turkey under the AKP government reformulated its strategies towards both the Syrian conflict and Iran due to the ever-shifting balance of power between state and non-state armed groups (Okyay 2017, 834).

Moreover, the AKP's power in domestic politics started to be questioned owing to financial costs, the refugee influx, and terrorist attacks stemming from its military interventionist Syrian policies. To consolidate its political survival, the AKP government sought to repair its external and internal position, even though it was able to restrain the domestic tensions and political consequences of policy reversals via its overwhelming control of the Turkish media (Barkey 2014, 112; C. Phillips 2017a, 45). Notably, President Erdogan deemed the Syrian conflict an internal security issue (Güney Ateşoğlu 2013, 54) which required that domestic considerations shape TFP. As such, the AKP-led Turkey extensively exploited TFP to maintain domestic stability, changing their policies from revisionist to maintaining the status quo. This suggests that the AKP elite's perceptions were not independent from domestic considerations, and internal security and economic considerations forced Turkey under the AKP government to implement pragmatic and interest-based policies towards Iran, despite their periodic clash of interests. In this regard, Schweller's balance of interest theory provides a useful perspective on why Ankara formed a short-term alliance with Tehran in consideration of cost-benefit calculations and profits. Without examining domestic considerations as an intervening variable, therefore, it can be difficult to understand how Turkey reshaped its foreign policy behaviour towards Iran from March 2011 to July 2018.

6.2.2.1 Security Consideration

Security factors became an overwhelming domestic consideration in shaping TFP towards Iran in the Syrian conflict. Following the fragmentation of Syria, numerous non-state armed groups emerged as influential actors which posed a security threat to other regional states, notably Syria's neighbours. According to Journalist 2 (personal communication, August 14, 2020), security concerns are mutual between Turkey and Iran. However, Iran's security perception was based on its own national security, while Turkey's security perception resulted from the rise of Syrian Kurds and the possible spill-over on its domestic Kurdish problem.

At the beginning of the Syrian conflict, Turkish policymakers arguably did not see the rise of PYD as a threat in the north of Syria, because they focused more on the overthrow of the Assad regime. Instead, Ankara made three demands of the PYD: no cooperation with the Assad regime; no de facto foundation based on ethnic or religious bases; and no engagement in activities that could endanger the security of the Turkish border (Gunter 2015, 108). In return, during a visit in July 2013, Salih Muslim, the former co-chairman of the PYD, declared that "the Syrian Kurds continue to see themselves as part of Syria and pose no threat to Turkey's territorial integrity". However, he also asserted that it was necessary to "establish a temporary-serving administration until the chaos in Syria is over" (Hurriyet Daily News 2013a).

Soon afterwards, in the face of Ankara's disapproval, the PYD established three cantons, thus reviving Turkey's concern regarding possible collaboration between the PKK and PYD (Elik 2018, 109). Notably, Erdogan upbraided the PYD and signalled a possible Turkish military operation in northern Syria as a "most natural right". Here, Davutoglu added, Turkey could intervene in northern Syria if the PYD became empowered (Demirtas-Bagdonas 2014, 149). These factors indicate that the empowerment of the PYD was subjugated to Turkey's internal PKK issue, and the AKP policymakers deemed that the rise of the PYD was a purely domestic matter for their political survival.

The aforementioned US-PYD alliance against ISIS further increased the AKP policymakers' concerns since the PYD would become a significant political actor on Turkey's southern border by means of US-led anti-ISIS operations (Işiksal 2017b, 98). Even though Turkey under the AKP government granted the use of Incirlik airbase, which already had a US Air Force complement, and collaborated with the US against ISIS in 2015, the US did not

withdraw its support for the PYD. Rather, Washington set up an air facility in Rmeilan in Kurdish-controlled northern Syria and sent special forces into Syrian Kurdish territory (Pierini 2016, 6). Thanks to US support, the YPG/PYD occupied Tal Abyad in June 2015, which increased Ankara's concerns since, "long-term control of Tal Abyad would further the YPG's goal of connecting the non-contiguous zones of the territory it holds across northern Syria" (Dalay 2017c, 79). Accordingly, the PYD became a more significant US ally than Turkey in combating ISIS, which sabotaged Turkey's ties with the US. In this context, Ankara demanded the termination of US support for the PYD and an "internationally-backed safe zone in between two PYD-controlled chunks of territory" (Okyay 2017, 836). Although the US encouraged the PYD to compose an alliance with its minor Syrian partners in order to meet Ankara's objections in November 2015 (S. Aydin and Ucar 2018), Ankara was at odds with the US over the empowerment of the PYD and sought an alternative way to neutralise the rise of this non-state armed group.

Given the collaboration with pro-Turkey Sunni groups, the AKP government launched numerous military attacks against the PYD, shifting the focus of its Syrian policy from ending the Assad regime to protecting Syria's peaceful territorial integrity (Ataman and Özdemir 2018, 21). In this regard, Ankara first conducted Operation Euphrates Shield from 24 August 2016 to 29 March 2017. The operation demonstrated how Ankara recognised that the PYD posed a tactical challenge to Turkey's domestic security, since the PYD's presence in Rojava could serve as a gateway for the PKK to expand its terrorist attacks along Turkey's border (C. Phillips 2017a, 43). Following this military operation, however, the international legitimacy of the PYD grew, thanks to further US support for the PYD. The US recognised the PYD internationally in its efforts in Syria and warned Turkey regarding its cross-border military activities in northern Syria (Gunes and Lowe 2015). The result was that the US support for the PYD led to a further deterioration in Turkish-US bilateral affairs, and the PKK was able to acquire safe havens adjacent to Turkish territory and close to its affiliated groups, the PYD/YPG.

The AKP's offensive attitude towards the PYD reignited tensions with the PKK, and thus worsened Turkey's own Kurdish problem. While the PYD drew international admiration for its six-months defence of Kobane from ISIS attacks in 2014–15, for instance, Ankara's unwillingness to aid the Kurdish militia drew heavy criticism from Kurds in Turkey against the AKP government and culminated in the reactivation of PKK insurgency and the termination of the Kurdish–Turkish peace process. Turkey also had to deal with regular terrorist attacks from

the PKK, including twin attacks on civilians in Ankara, killing 28 in February and 37 in March 2016 (C. Phillips 2017a, 43). It appears that Turkey under the AKP government was squeezed between protecting its internal power and clashing with the PKK/PYD.

In such an environment, the AKP government endeavoured to ensure its political survival. Notably, President Erdogan viewed the PKK attacks as an opportunity to deploy anti-Kurdish sentiments and Turkish nationalist discourse, to decrease the legitimate authority of the PKK-affiliated political party, the HDP and its leader, Selahattin Demirtas. This led to further offensive military operations against the PKK-affiliated PYD through Operation Euphrates Shield in Aleppo from August 2016 to March 2017, followed by Operation Olive Branch in Afrin from January to March 2018. All these factors indicate that Turkey's internal Kurdish issue was interwoven with the rise of Syrian armed Kurds and constrained the AKP government's ability to respond realistically to structural determinants. In particular, the rise of cross-border Turkish military offensives highlighted how the AKP government believed its political survival was now focused on preventing the rise of PYD in Syria. At the same time, Ankara was cognisant of the cost of unilateral military intervention in Syria, because Russia and Iran's increased support for the Assad regime would complicate the Turkish military intervention in Syria, and could internationalise Turkey's Kurdish issue (Ünver 2012, 106). Put differently, it was impossible for Ankara to achieve a permanent victory against both the Assad regime and the PYD without a political alliance.

Therefore, collaboration with Iran over the rise of the PYD was crucial for Turkey, even though at times Ankara had blamed Iran for supporting the PKK. According to Academic 4 (personal communication, July 6, 2020), Iran had used the PJAK and PYD against Turkey due to Turkish involvement in Syria, an act that arguably furthered the declaration of autonomy of Syrian Kurds in north Syria to counter-balance Turkey. Similarly, Academic 3 (personal communication, July 9, 2020) added that when the Assad regime started to lose influence in northern Syria, both the regime and Iran ignored the rise of the PYD in the same area. Even Iran utilised the rise of the PYD in Syria since Turkey was concerned about the rise of PYD's influence. However, in the long-term, the rise of the PYD also became a domestic concern for Iran because, as long as the Kurdish position was rising in the Middle East, Iran would have difficulty managing the Kurdish issue both internally and externally in the increasingly turbulent region (Laoutides 2016, 105). Expert 4 (personal communication, July 24, 2020)

control of Kurds would be beneficial to the territorial integrity of Syria. As such, Iran refrained from damaging its ties with Turkey completely.

Equally, the US presence in Syria and its support to the PYD would undermine Iran's regional expectations, as well as Turkey's. Despite the defeat of ISIS in 2017, the US continued its military presence in northern Syria, supporting the PYD/YPG, which raised suspicions and concerns in Turkey and Iran. Academic 3 (personal communication, July 9, 2020) added that Iran realised it would not win in Syria and sought to cooperate with Turkey by changing its policies. As such, the two states' shared desire to hamper the empowerment of the PYD increased the degree of cooperation between Tehran and Ankara, which further supports the notion that Turkey chose to ally with Tehran for cost-benefit reasons, profit-oriented and interest-driven policies.

As well as the rise of the PYD, the crisis in migration emerged as another purely domestic security matter which compelled the AKP government to establish ever-deepening interaction with Iran. According to a Global Appeal Report from the UN Refugee Agency, in 2018, Turkey hosted approximately 2.9 million Syrian refugees in special camps in provinces such as Gaziantep, Sanliurfa, and Kilis (adjacent to the Syrian border), and in other cities such as Istanbul and Ankara; Turkey was also becoming a transit route for refugees passing legally or illegally to European countries (The UN Refugee Agency 2019). This influx of Syrian refugees created "a difficult social situation in the south-eastern regions of the country, where sporadic episodes of violence have occurred", and thus the AKP government had to deal with the considerable economic and security burden this imposed (Alema 2017, 10; Parlar Dal 2016, 1403–4). In particular, the refugee flow from Syria resulted in illegal border crossings by gangs smuggling petrol and other goods and attacked border posts. Petty crime and the black market in the Turkish provinces adjacent to Syria increased due to the poor economic situation of refugees (Kirişci 2014, 36). The refugee flow also had a knock-on effect on Turkey's Kurdish problem in the south-east of the country, and threatened to become a back door for Kurdish terrorism (Aras and Yorulmazlar 2016a, 2262; Oktav 2013, 198). All these factors posed a political challenge for the AKP government and revived the general Turkish population's criticism of the AKP's military interventionist policies in Syria. These concerns only looked set to increase once Lebanon closed its borders to Syrian refugees in 2014, followed by Jordan in June, 2016, after a suicide car bomb near a crossing killed six Jordanians (Işiksal 2018, 27). Accordingly, the AKP-led Turkey demanded a safe zone to prevent the stream of Syrian

refugees, but this was rejected by the US. Turkey was forced to find another solution regarding the refugee crisis.

Consequently, security considerations pushed Turkey to liaise with Iran, as well as Russia. In doing so, Ankara pursued an alternative choice to determine the future of the conflict and remove itself from its geopolitical isolation, even though Ankara's ties with the US and its Sunni state partners (Qatar and Saudi Arabia) had resulted in an "ideological impediment" to enhance its collaboration with Russia, and Iran particularly (Aghaie Joobani and Mousavipour 2015, 151). The Astana and Sochi processes, controlled by Turkey, Iran, and Russia since January 2017, were designed to shape the future of the Syrian conflict. Despite the different long-term interests of the three states, for Turkey, the processes also highlighted the need for improved ties with Iran over both the rise of PYD/YPG and the impact of the refugee crisis in the post-Syrian conflict. With the Astana talks, for instance, almost 3.5 million refugees living in Turkey would be repatriated to Syria (Ataman and Özdemir 2018, 22). As revealed in Expert 1 (personal communication, August 17, 2020), these processes demonstrated that Turkey put aside a solely western-based foreign policy in favour of a more pragmatist and profit-oriented foreign policy towards Iran in the short-to-medium term. Therefore, Turkey's use of short-term interest-based policies towards Iran supports the balance of interest interpretation since pragmatic security collaboration with Tehran was more beneficial than competitiveness until July 2018.

6.2.2.2 Economic Considerations

Prior to the outbreak of the Syrian conflict, Turkey had been "Syria's largest individual source of imports, with sales of \$1.8 bn, and among their largest export destinations" (Werman 2019). With the outset of the conflict, however, the deterioration of Turkish-Syrian affairs hampered any economic relations between Ankara and Damascus. In such an increasingly turbulent environment, Turkey not only lost its economic partner but also became mired in an internal economic crisis. Notably, the refugee issue created by the Syrian conflict and the AKP's unwavering support for the Syrian Sunni opposition sparked economic unrest in Turkey.

The economic problems caused by the refugee crisis surpassed Ankara's financial capabilities as Turkey could not manage the cost of aiding the refugees and maintaining camps and facilities. Until 2016, for instance, Ankara disbursed approximately \$9 bn in humanitarian aid for almost 2.6 million Syrian refugees and there were 26 camps in different parts of Turkey,

amounting to 20 times more than the aid from international organisations (Parlar Dal 2016, 1403–4). Until the end of 2018, this expenditure was expected to increase to over \$37 bn in national resources going to Syrian refugees, as President Erdogan stated at an international gathering on migration (Sonmez 2019). It appears that the Syrian refugees became an additional financial burden for AKP-led Turkey, which was already confronted with economic difficulties such as the decline of the Turkish lira, rising inflation and foreign-source dependency (Tan 2018).

The refugee issue also played another significant economic role regarding the increase in the unemployment rate among Turkish locals because many Syrian refugees displaced locals by informally and illegally working for lower wages and longer hours. They worked mainly in the agriculture, construction and textile sectors, and labour-intensive industries, often as seasonal workers (Isiksal, Zhakanova Isiksal, and Apeji 2020; Sonmez 2019). Moreover, the cost of living and house prices were increasing daily due to the pressure caused by the rising numbers of refugees. For instance, property prices in Gaziantep, Sanliurfa, and Kilis doubled, and rents increased threefold in 2014 (İçduygu 2015, 133). All these factors revived hostility among Turkish citizens as they believed that they could not find work due to the Syrian refugees (The Economist 2019). Numerous social and economic tensions emerged and prompted demonstrations against the refugees, some of whom were evicted from their homes and businesses.

To resolve this issue, Ankara initially sought to liaise with the EU and negotiated several times the terms of financial aid to Turkey. In March 2016, Davutoglu met with the EU leaders to create a border management deal which would offer irregular migrants who had entered Greece after 20 March 2016 the chance to go back to Turkey; in return, Ankara would send legal Syrian refugees settling in Turkey to the EU. Turkey would receive ϵ 6 bn and visa liberalisation for its citizens from the EU by the end of June 2016. However, as Pierini (2016, 10) argues, "the deal quickly became embroiled in litigation about the concessions offered by the EU and the conditions to be met by Turkey". As such, Turkey's quest for collaboration with the EU regarding the issue of refugees stalled.

Further revenue loss for Turkey arising from the Syrian conflict was also felt in the tourism sector, notably in the southeast of the country. The impact on tourism was severe, and seriously crippled the domestic economy. For example, between 2015 and 2016, international

tourist arrivals dropped by 30.05 percent and tourism revenue fell \$9.4 bn (Bloomberg 2017; Ozgen and Reyhanogullari 2017). The isolation of Turkey in international and regional areas further culminated in economic stagnation with a \$4–5 bn trade deficit until 2016 (Zelyut 2016). At the same time as it was losing revenue, AKP-led Turkey increased military spending on operations in Syria. According to the Stockholm International Peace Research Institute (SIPRI), "Turkey's 2018 military spending increased to \$19 bn, a 24 percent rise from the previous year, which was the highest increase in military expenditure among the top 15 military spenders" (Ahval 2019). As such, the AKP government was held responsible for economic stagnation and became trapped in a quagmire of domestic and foreign issues.

To maintain political survival and increase electoral support, the AKP-led Turkey sought to improve the economy, and so it looked towards more cordial commercial ties with Iran. In this regard, the AKP government laid enormous stress on economic diplomacy and bilateral trade agreements with Iran. According to the Turkish Statistical Institute, during the era of the Syrian conflict, exports from Turkey to Iran were worth approximately \$10 bn in 2012 but decreased nearly \$4 bn between 2014–2015. In 2016, these exports slightly increased to almost \$5 bn, but shrank again to \$3.2 bn in 2017 and \$2.4 bn in 2018 (Turkish Statistical Institute 2018a). Imports from Iran to Turkey also fluctuated, being worth nearly \$12 bn in 2012, \$10 bn in 2014–2015, and \$5 bn in 2016. There was an increase to nearly \$7.5 bn in 2017 but this was followed by a further decrease to \$7 bn in 2018 (Turkish Statistical Institute 2018b). These figures demonstrate that Turkish-Iranian bilateral trade did not entirely collapse despite their mutual competition over the Syrian conflict, and their economic inter-dependency avoided the outbreak of an actual conflict between Turkey and Iran. However, the Turkish economy seems to have been more dependent on Iran given that imports, mainly in the energy sector exceeded exports between 2012–2018.

To further maintain its commercial ties with Iran, the AKP government set a goal of reaching \$30 bn in bilateral trade, and opposed any US sanctions against Iran (Erkus 2019). As such, despite the nuclear-related sanctions imposed by the EU and US against Iran, Turkey maintained economic ties with Iran and encouraged firms to continue doing business there (Karahan 2016). Ankara, therefore, prioritised its economic relationship with Iran over its diplomatic relations with the US and EU. Since 2017, the AKP government also regarded the trilateral summits between Russia, Iran, and Turkey —which aimed to end the Syrian conflict —as an opportunity to develop Turkey's economy, since enabling stability in Syria would

decrease Turkey's expenditure on Syrian refugees and defence. Therefore, commercial ties between Turkey and Iran increased in 2017 even though both countries were engaged in joint efforts in Syria, and Turkish relations with the US had cooled. Following the collaboration over the Syrian talks, as Centingulec (2018) notes, "[Turkish] trade volume rose for the first time since 2012, standing at \$10.7 bn at the end of 2017, representing a \$1.1 bn increase from 2016".

Ultimately, the domestic economic issues which arose in Turkey because of the Syrian conflict affected TFP towards Iran from the outbreak of the Syrian conflict to July 2018. Owing to these economic issues, the AKP-led Turkey pushed the need for short-term collaboration with Iran despite the discord over the Syrian conflict for its long-term interests. This demonstrates that economic considerations motivated Ankara to ally with Tehran, as Ankara found that an alliance with Tehran would bring more benefits than competition in the short-to-medium term, in line with Schweller's balance of interest theory.

6.3 Concluding Remarks

This chapter illustrates the relevance of NCR for explaining a regional power's foreign policy during a period of turbulent change in the regional system, in this case shaped by the Syrian conflict. Given that NCR examines structural changes in the regional and international system and the relative power of a state as systemic variables, the changing regional system in the Syrian conflict and Turkey's relative power position determined TFP towards Iran at this time. Alongside other Arab uprisings, Turkey had the self-confidence to play a significant role in shaping a new balance of power, and thus followed interventionist and assertive policies in Syria to expand its relative power position in the long term. As such, Turkey initially implemented a competitive foreign policy attitude towards Iran at a certain extent, even though the latter was pursuing its own interests in the Syrian conflict with widely differing aims. There were tensions between Ankara and Damascus, the Iranian ally, due to Ankara's support for the Syrian opposition and its harsh criticism of the Assad regime.

However, domestic factors played an intervening role in the short-to-medium term in shaping TFP towards Iran, as NCR highlights. Accordingly, the AKP elite's perception and domestic considerations played an influential role in TFP towards Iran in the Syrian conflict. Throughout, the AKP elites misread the regional system and made numerous mistakes, such as miscalculating the power of the Assad regime, and US policies in Syria. They notably followed sectarian identity-based foreign policy principles in Syria by propping up Sunni opposition groups in an attempt to fill the power vacuum in Syria; however, Turkey's financial and military support for the jihadist Sunni armed groups weakened the relationship between Turkey and other regional and international states, such as Saudi Arabia and the US, leading to regional and international isolation for Turkey. With its increasingly anti-Kurdish and ethno-nationalist policies, the AKP elites put aside their sectarian policies and chose a short-term pragmatic alliance with Tehran.

Moreover, the empowerment of the PYD and ISIS posed security challenges for Turkey. In consequence, the AKP government shifted its priorities from removing the Assad regime to preventing the rise of the PYD and ISIS; simultaneously, Turkey needed alternative trading partners due to the domestic economic challenges caused by the refugee influx and military expenditure. In such an environment, Ankara, though clearly in regional political competition with Tehran, pragmatically engaged with Iran by joining talks to resolve both security threats to its territorial integrity and worsening economic problems. It appears that AKP-led Turkey followed a status quo position because of domestic constraints, putting aside revisionist policies for a while. Since the onset of the Syrian conflict in March 2011, TFP towards Iran moved from an era of low intensity, controlled tension to a time of rapprochement, which indicates that the cool nature of TFP towards Iran in the first years of the Syrian conflict was in a stark contrast to the more recent efforts at rapprochement towards Iran. Despite its opposition to the Assad regime, Turkey sought collaboration with Tehran, the Syrian regime's ally, to safeguard its interests by shifting its focus from overthrowing the regime to ensuring its national security. In this regard, Schweller's balance of interest approach is a helpful position from which to understand Turkey's alliance choice with Tehran from March 2011 to July 2018 because Ankara followed an interest-based policy towards Iran and deemed that short-term bilateral collaboration with Tehran was more profitable than competition in the future of the conflict. Accordingly, Table 6.1 below shows TFP towards Iran in the Syrian conflict, enumerating the short-term behaviours.

Systemic Variables (Independent)	Intervening Variables (Unit-Level)	Dependent Variables (Foreign Policy Behaviour in the Short-to-Medium Term)
 Inter-state competition The Rise of Non-State Armed Groups Turkey's Relative Power Position 	 AKP Elite Perception Domestic Considerations (Security and Economic) 	CollaborationStatus-Quo BehaviourBalance of Interest

Table 6-1 TFP towards Iran in the Syrian Conflict from March 2011 to July 2018

Given the role of systemic and intervening variables in TFP towards Iran since the outset of the Syrian conflict (March 2011 to July 2018), the next chapter considers the Kurdistan Region of Iraq as the second case study used in this thesis to provide insight into how systemic and unit-level variables determined TFP towards Iran since the Arab Uprisings.

CHAPTER 7 TURKISH FOREIGN POLICY TOWARDS IRAN SINCE THE ARAB UPRISINGS: DECEMBER 2010 TO JULY 2018 — THE CASE OF THE KURDISTAN REGION OF IRAQ

The Kurdistan Region of Iraq (KRI) is a semi-autonomous area in Erbil in the north of Iraq and has officially been governed by the Kurdistan Regional Government (KRG) since 2005. Since the Kurdish civil war of the mid-1990s, two political parties have controlled the KRG in a power-sharing agreement: the Kurdish Democratic Party (KDP), under the Barzanis and centred on Erbil, and the Patriotic Union of Kurdistan (PUK), controlled by the Talabanis and centred on Sulaymaniyah (Handy 2018, 189). The different viewpoints of these parties in KRI politics have influenced the foreign policy of the KRG. While the KDP represents tribal, conservative Sunni groups, the PUK is seen as a representative of more progressive, left-wing, and secular Kurdish groups. Although the parties' internal competition has been reflected in KRI politics since the 1990s, there are other political parties in the KRI, such as the Goran Movement and the New Generation Movement (Thornton 2015). As such, domestic politics in the KRI are competitive, reflecting a range of political perspectives among Kurdish groups.

When the Arab Uprisings aggravated political instability and ethnic and sectarian divisions in the Middle East and North Africa, the central Iraqi government had difficulty consolidating its power, which had been already questionable. The rise of non-state armed groups and the internalisation of the Syrian conflict close to Iraq further worsened the existing power vacuum and state fragility in Iraq. Accordingly, this anarchic environment engulfed the politics of the KRI, and the KRG strove to utilise this downturn by following a more assertive and revisionist foreign policy than the central Iraqi government. Notably, the outbreak of the Syrian conflict in March 2011 and the US presence in Iraq bolstered the KRG's interest-driven regional aims (Cağaptay 2012, 8; Stansfield 2013, 278–79). For instance, Erbil hoped to expand its territory by seizing areas to which it had longstanding claims, following ISIS attacks against the KRI (Musiol 2019, 124). Erbil further declared its intention to hold an independence referendum in June 2017, and, as a result, Erbil's policies not only upended the regional power structure, but they have also led to an increase in its political importance.

As well as its political significance, the KRI is an economically important region. Since 2009, the KRG has increased its oil and gas production by establishing cooperation with international oil companies such as Exxon Mobil, Chevron, the English Gulf Keystone, Total of France, Gazprom of Russia, and the Turkish company, Genel Energy (Charountaki 2012, 193; Gunes 2019a, 22). This economic wealth has attracted both international and regional powers in the wake of the Arab Uprisings, and reinvigorated competitiveness between Turkey and Iran. While Turkey sought more collaboration with the KRG against a Baghdad-Tehran partnership, in order to gain political and economic leverage, Iran hoped to establish cordial links with Erbil and Baghdad to enhance its influence in Iraq (Soguk 2015, 966). According to Interviews B (personal communication, July 1, 2020) and F (personal communication, August 17, 2020), C (personal communication, August 6, 2020) and I (personal communication, July 21, 2020), internal competition in the KRI has shaped TFP towards Iran. What is interesting is that the secular Talabani Kurds cooperated with Iran, but the conservative Barzanis worked more closely with Turkey because of the competition between the latter and the PKK. Both countries thus cooperated with different Kurdish groups in the KRI, and Ankara's and Tehran's economic and political ties with Erbil played an influential role in their bilateral ties.

Given the background of the KRI, Chapter 7 contains an analysis of the role of the KRI in TFP towards Iran since the Arab Uprisings as the second case study. In so doing, the chapter covers how Turkey's relative power position and the analysis of the AKP elite's perceptions and domestic considerations shaped TFP towards Iran in the KRI after the Arab Uprisings. The chapter is split into three sections. The first explains the role of the systemic variables on TFP towards Iran since the Arab Uprisings and Turkey's relative power position in the KRI. The second section analyses the directly intervening variables for Turkey in terms of shaping TFP towards Iran as a causal chain, explaining the role of the AKP elite's perception and the balance of interest approach in TFP towards Iran in the short-to-medium term. The final section summarises the chapter. Overall, as a case, the KRI can explain the role of NCR in shaping TFP towards Iran from December 2010 to July 2018.

7.1 The Role of the System-Level on Turkish Foreign Policy towards Iran since the Arab Uprisings (December 2010 to July 2018): The Case of the KRI

This section sheds light on the role of immediate systemic changes in TFP towards Iran in the case of the KRI. In this context, the US role in Iraq and the empowerment of ISIS and the
Kurdish groups are the primary and direct systemic variables considered in this chapter, even though other systemic variables have little effect on TFP towards Iran in the KRI.

7.1.1 US policies in Iraq

The US-led Iraqi War in 2003 changed the balance of power in Iraq, laying the foundation for the official autonomy of Iraqi Kurds. While Turkey, a US ally, refused to support the removal of Saddam Hussein, the KRG emerged as a significant US ally. For providing this assistance, Iraqi Kurds were rewarded with constitutional recognition and independent foreign relations with other states (Bengio 2017, 16–17), thus increasing the significance of Iraqi Kurds in regional politics following the 2003 US-led Iraqi War.

For Iran, Iraq is vital to its regional interests, and thus Iran is unwilling to see an anti-Iranian government both in Erbil and Baghdad. For this reason, Iran started to support Iraqi Shia actors and play a more significant role in successive Baghdad governments in the wake of the 2003 war (Khoury 2016a, 14). Given its cordial ties with Baghdad, Iran also sought to forge closer ties with the KRG, but, notably due to its anti-US policies. Tehran thus put more effort into hampering US influence on the KRG. For example, Iran maintained close ties with members of the PUK, "many of whom are Persian-speaking and have lived in Iran", compared to the KDP, a US ally (Mustafa and Aziz 2016, 161–62). In this way, Iran aimed to leverage the politics of the KRI in its favour to achieve its national interests.

Following the Iraq war, Turkey remained at odds with the US regarding the status of the Iraqi Kurds. While the US deemed the Kurds a close ally in toppling the Saddam regime, Turkey was concerned about the proximity to its borders of a possible Kurdish state, envisaging that such a state would trigger rebellion among the Kurdish minority in Turkey. Irrespective of its Kurdish groups, Turkey has always concerned itself with the Turkmen minority in Iraq, which comprised approximately 16 per cent of the 2005 Iraqi population and shared ethnic affiliation with Turkey. Indeed, Ankara believed that if Kurds emerged as the rising minority, they would disregard the Turkmen in Iraqi politics (Müftüler-Bac 2005, 68, 72). On the other hand, since 2008, Turkey has no longer viewed the Iraqi Kurds as a security threat (Kayhan-Pusane 2020) and instead sought to improve relations with Erbil under the 'strategic depth doctrine' and 'zero-problems-with-neighbours' principle; Ankara also opened a consulate in Erbil in 2010. According to Journalist 2 (personal communication, August 14, 2020), the

presence of Turkmens in Iraq was more important for Turkey, which sought to prevent the empowerment of Shia groups. Therefore, Ankara perceived the rise of Shia groups in Iraq as more threatening to the rights of the Turkmen minority than the rise of Sunni Kurdish groups.

When the US decreased its troops from Iraq in 2011, the power gap that re-emerged in Iraq permitted the rise of sectarian and ethnic-based policies. Following the withdrawal, the Shia-dominated Maliki government began excluding Sunni Arabs and Kurds from Iraqi domestic politics (Bozarslan 2018, 18–19), and consequently, bilateral Erbil-Baghdad relations became strained and much more competitive (Park 2014, 5–6). This domestic rivalry in Iraq influenced Ankara's ties with Tehran since both countries sought to shape Iraqi policies to align with their interests. The decrease in the number of US troops in 2011, notably, solidified the Iranian influence in Iraq, shifting the balance of power in favour of Tehran (Khoury 2016, 14). Iran thus opted to increase its ties with the Shia-dominated Baghdad government despite its desire for closer ties to Erbil.

In contrast to the Tehran-Baghdad partnership, the Ankara-Baghdad relationship reached a nadir when Ankara blamed the Baghdad government under Nouri al-Maliki for following a sectarian policy in Iraq. As such, Turkey promoted its relations with the KRG to depose the Shi'a-dominated Maliki government while also competing with Iran (Park 2014, 6). Expert 1 (personal communication, August 17, 2020) revealed that Turkey not only opposed Iranian influence in Iraq, but it also perceived an Iran-oriented Iraq as a security threat to its long-term national interests. Although the US presence in Iraq played a balancing role between Ankara and Tehran, as an external state power, the drawdown of US troops intensified Ankara's competitive policies towards Tehran. Consequently, Ankara highlighted the significance of the KRI, just as Iran sought to enhance control over the Baghdad government. Thus, the KDP under the Barzanis became a close Turkish ally to balance Iranian influence, while Tehran deepened its ties with the PUK under the Talabanis (Cagaptay 2012, 9; Thornton 2015a, 870). More precisely, Ankara followed a competitive policy towards Iran based on costbenefit calculations, and collaboration with Erbil seemed more beneficial for Ankara to prevent the rise of Iranian influence on Iraq's politics in the long term.

With the empowerment of ISIS in Iraq, especially after 2011, there was a revival of US fears that jihadist groups could threaten the US and its allies in the Middle East. Therefore, Washington prioritised the fight with ISIS by endorsing the Peshmerga, the official

paramilitary forces of the KRG (Paasche and Gunter 2016, 11). In this way, the KRG reemerged as a valuable US partner in Iraq. However, the US continued to emphasise the importance of close cooperation between Baghdad and the KRG. For instance, the US recommended that Baghdad should decide which military supplies would reach the KRG (Strategic Comments 2015, 2). In doing so, Washington aimed to hamper Iranian influence over Baghdad; as such, Washington and Turkey had shared concerns about counterbalancing Iranian long-term influence (Lindenstrauss and Aksoy 2012, 55). Although the US emphasis on the political integrity of Iraq appeared to favour Turkish interests, the Turkish-US relationship remained strained due to the US' unwavering support for the PYD; the result was that Turkey began to exhibit disquiet about the presence of the US in Iraq. Given the cordial nature of Turkish-KRG affairs, Turkey perceived the Barzanis to be a more significant ally than the US due to US support to the PYD/YPG. In this regard, as a systemic force, the US role in Iraq determined TFP towards Iran. With a decreased US presence in Iraq, Turkey perceived Tehran as a competitor and sought to establish warming ties with Erbil to prevent the rise of Iranian influence in the long-term. However, Ankara has always been on the alert for US support for Kurdish groups, and so, as will be explained in the section on unit-level variables, the close ties between the US and Kurdish groups may have paved the way for a short-term alliance between Tehran and Ankara due to profit-oriented calculations and mutual interestdriven policies.

7.1.2 The Syrian Conflict

The Syrian conflict, which started in March 2011, upended the regional power structure in Iraq in the wake of the Arab Uprisings. In particular, the anarchic environment and lack of institutional legitimacy in Iraq which stemmed from the Syrian conflict provided significant opportunities for non-state armed groups, including the ISIS and the PYD/YPG, to achieve specific aims and influence the regional agenda. These groups sparked political unrest in Iraq and the KRI, as neighbours of Syria, and indeed the rise of Syrian Kurds impacted Baghdad-Erbil ties since each party took up different positions at the beginning of the Syrian conflict. While the KRG appreciated the Syrian Kurds to a certain extent, Iraqi the Maliki government, along with Tehran, sought to promote ties with the Assad regime. As such, the Iraqi and the Syrian Kurds clashed when the Baghdad government looked to "send the Iraqi army to the territory linking Kurdistan Region of Iraq [the KRI] with the Kurdish areas of Syria", to prevent the KRG from assisting Syrian Kurds (Neriah 2012, 14). However, the empathy of the Iraqi Kurds towards their Syrian brethren vanished when the KDP under the Barzanis realised that the PYD would diminish the regional influence of the KDP. The rise of ISIS further increased the difficulties when Iraqi Kurds started to fight with the jihadist group in their territory and tried to find external state support against its fight with the ISIS (Gunes 2019b, 35).

In such an environment, Turkey's strained affairs with both the Assad regime and Tehran led to closer ties with Erbil than ever before, since the latter was "the only corridor through which [Turkey could] reach out to the Arab world" (Mustafa and Aziz 2016, 162). Accordingly, Turkey desired no Iranian influence in the KRI since Iran had already increased its power in Syria. More precisely, the competition between Ankara and Tehran in the Syrian conflict spread across the KRI, and this tumultuous environment shaped TFP towards Iran as a systemic variable.

7.1.2.1 The rise of ISIS

By utilising the tumultuous environment, ISIS, as a non-state armed actor, sought to enhance its influence in both Syria and Iraq, and thus posed a threat to both Baghdad and Erbil. This jihadist group seized sophisticated and advanced weapons provided by the US to Iraqi armed forces and took control of oil wells in the KRI. Despite being forewarned by intelligence services of the KRG regarding possible attacks on Mosul, the KRG and Baghdad failed to defend their territories against ISIS attacks in August 2014 (D. L. Phillips 2014, 352). Worse still, they were incapable of preventing the humanitarian crisis which now unfolded from the ISIS assault on the Yazidi minorities in northern Nineveh. These formidable challenges and the weakness of the Peshmerga and Iraqi forces in fighting ISIS resulted in external intervention in Iraq. Notably, the KRG demanded US support in preserving its regional position and recapture control of its lost territories (Gunes 2019b, 35–36). In doing so, the KRG paved the way for a more substantial US presence and operations in the KRI.

With the US support to the fore, Iraqi Kurds enjoyed western military and financial support against the ISIS. At the beginning of 2015, for instance, the US built a military base in Erbil to provide logistical support. More than a year later, the US signed a memorandum of understanding on military coordination with the KRG. Given the close alliance between the US and the KRG, the Rudaw, a media group in the KRI, stated that "Since the beginning of the war with ISIS, there has been steady traffic of US high-level military and civilian delegations

to the Kurdistan Region and US-KRG relations entered a new phase in their strategic and special relationship" (Bengio 2017, 40–41).

This US-KRG entente also encouraged Iraqi Kurdish policymakers to seek their longterm regional aims. The KRG deemed ISIS's capture of Mosul as an opportunity to gain territory and push for independence because it assumed that a united Iraq would be impossible. In this regard, Phillips (2014, 352) notes that "By the end of June [2014], Iraqi Kurdistan [the KRI] included lands from the border with Syria in the west to Iran in the east". The status of disputed territories —including the city of Kirkuk and its neighbourhood, which had been left by the Iraqi government following an ISIS attack —drove a wedge between the KRG and the Baghdad government (Bengio 2017, 38; Gunes 2019b, 22). Tensions increased because Erbil held resource-rich areas, including Kirkuk, and had extended "its territory by more than 40 per cent" (Bengio 2017, 39; Dalay 2017d). On 25 September 2017, Erbil further attempted to gain independence by holding a referendum "in which 93.73% of the voters cast a vote in favour of independence" (Gunes 2019b, 22). Ultimately, Erbil utilised the anarchic environment stemming from ISIS attacks, with the US support and an independence referendum.

The KRG's enthusiasm backfired, however, since it failed to evaluate other external and regional state actors' perceptions. Even though Washington appeared to support Iraqi Kurdish autonomy, its primary concern was the territorial integrity of Iraq due to the rise of ISIS (Koolaee and Hafezian 2017, 82). As such, the US sought to mediate between Baghdad and the KRG to protect its interests, with the US embassy in Baghdad stating that "We support the peaceful reassertion of federal authority, consistent with the Iraqi constitution, in all disputed areas". Similarly, the US President Donald Trump asserted that "We don't like the fact that they are clashing, but we're not taking sides" (Gunter 2018, 29). Therefore, although the KRG aimed to have US support in its quest for independence, this was to no avail.

The rise of ISIS in Iraq also influenced the role of Ankara and Tehran in the KRI, with both regional middle powers deeming ISIS a threat. Tehran's concerns were due to the ISIS' Salafist ideology and its attacks against Baghdad. Following the fall of Mosul to the ISIS in June 2014, Tehran unwaveringly propped up both Baghdad and Erbil. When trying to liberate the two Kurdish districts of Jalawla and Sa'diah, the Peshmerga received direct Iranian support against ISIS (Mustafa and Aziz, 2016, 4–5). Notably, the Islamic Revolutionary Guard Corps (IRGC) cooperated with Iraqi Shiites by training and organising them (Nasr 2018, 112–13).

According to Mustafa and Aziz (2016, 4–5), "the threat to Erbil by ISIS allowed Iran to show its goodwill, as the primary aid donor in the region". By cooperating in defence of Erbil, therefore, political and economic rapprochement emerged between Tehran and Erbil. Erbil essentially viewed Tehran as an alternative trading partner when it was suffering economically due to the refugee crisis caused by ISIS attacks, substantial military costs, and increased oil prices (Bengio 2017, 39; Mustafa and Aziz 2016, 162).

In contrast to Tehran, Ankara initially ignored the rise of ISIS, mainly because of their mutual opposition to the Assad regime and the PYD/YPG. For this reason, Ankara at first ignored Erbil's request to fight against ISIS in August 2014 and Turkey's ignorance damaged Ankara-Erbil ties. Fuad Hussein, Barzani's chief of staff, stated that "the US, France and Europe came to our rescue but our neighbour [Turkey] did not do the same even though we asked for its help" (Thornton 2015, 873). According to Thornton (2015, 873), the main reason behind Turkey's inaction was Erdogan's assumption that "aiding Kurds against pious Sunni warriors in the shape of ISIS" would result in the loss of hard-line Islamist supporters. However, this reasoning had limited impact on Turkey's attitude towards ISIS, because the more prominent reason was that Turkey perceived the rise of the PYD as a more significant threat. Turkey therefore opted for a wait-and-see approach until the international community denounced Turkey's inaction, and ISIS directly threatened Turkey's national security.

When faced with security challenges, however, Ankara changed policy against ISIS in Iraq and joined the US-led international coalition in September 2014 despite its reluctance to help the PYD/YPG in the siege of Kobane in northern Syria. In collaboration with Erbil and Baghdad's efforts to purge ISIS from Iraq, in 2015, Ankara also decided to train Kurdish and Sunni Muslim forces and deployed a small military contingent to Bashiqa near Mosul in northern Iraq. Expert 2 (personal communication, August 19, 2020) argued that alongside the friendship between the Shi'a-dominated Maliki government and Iran, Turkey's concern was also increased by the establishment of the mostly Shia Hashd al-Sha'bi group sponsored by the Iraqi government to fight ISIS. Similarly, Academic 2 (personal communication, August 6, 2020) uncovered that Ankara blamed Tehran and Baghdad for following sectarian policies after the establishment of Hashd al-Sha'bi in 2014. Ankara thus viewed its cordiality with Erbil as a counterbalance against the ties between Tehran and Baghdad, and it opted for a unified Iraq with dominant groups (Sunnis, Shi'ites, Kurds and Turkmens), while Iran aspired to a weak Shia dominant Iraq (Lindenstrauss and Aksoy 2012, 54).

However, Ankara's military presence exasperated Baghdad, which was concerned about Turkey's support for Sunni tribes and politicians; Baghdad even claimed that Turkey's military base in Bashiqa impugned Iraqi sovereignty (Riva 2017). Tehran, meanwhile, seemed to share Baghdad's concerns about Turkey's military presence, and thus hoped to become a more significant actor and KRG supporter than Turkey against ISIS attacks, to hamper Turkish influence in the KRI (Sinkaya 2018). In this regard, Tehran was quicker to declare its willingness to supply military support to the Peshmerga than other state actors, including the US. In response, President Masoud Barzani stated that "We [the KRG] asked for weapons and Iran was the first country to provide us with weapons and ammunition" (Collard 2014). Turkey's competition with Iran, therefore, re-emerged in the KRI as the two regional powers attempted to forge influential and exclusive ties with Erbil.

7.1.2.2 The rise of the PYD

When the PYD declared autonomy over the region of Rojava in the north of Syria in 2012, the KRG sought to curb the rise of the PYD even though President Masoud Barzani initially endorsed the Syrian Kurds and called on the Assad regime to accept Kurds' rights. The KRG's support for Syrian Kurds was short-lived, however, as the KRG realised that empowering the PYD/YPG would void Erbil's regional influence. Indeed, the PYD's close ties with the PKK —the ideological rival of the KDP —and US support for the PYD evolved contrary to Erbil's expectations. With US help, the PYD started to increase its influence in the region, and the rise of this alternative Kurdish group prompted Erbil to be concerned that its own impact would falter. Given Erbil's fears, President Barzani remarked that "Any support to the PYD means support for the PKK. They are exactly one and the same thing", implying the growing US support to the PYD/YPG (Daily Sabah 2016a). Barzani even condemned the PYD's declaration of autonomy and refused its request to open diplomatic relations with the KRG (Wilgenburg 2014).

Having criticised the PYD, the KRG attempted to organise all those Kurdish political parties which were ideologically closer to the KDP and the PUK under the umbrella of the Kurdish National Council (KNC). In doing so, it sought to enhance its influence over Syrian Kurds and push the PYD to one side, due to its political differences with the KRG (Unver 2016). This attempt by Erbil thus made clear that the KRG started to perceive the PYD as a rival in the region. In return, the PYD opposed to joining the KNC, for other reasons. As a Syrian Kurdish group, the PYD hoped to establish "an ordering based on the idea of

autonomous assemblies" while "the KNC aimed at autonomy for the Kurds within a Syria in which the Ba'ath regime would be replaced" (Jongerden 2019, 65). As a result, the Erbil agreement on creating a Kurdish Supreme Committee and power-sharing Kurdish-controlled areas was never put into action due to these disagreements, even though both sides signed the deal in July 2012 (Soz 2016).

Another reason for the KRG's competitive actions against the PYD was the friendship between Ankara and Erbil. Turkey had encouraged the KRG to establish the KNC to restrict the rise of the PYD. With Ankara's support, in November 2013, the KRG blamed the PYD for "acting unilaterally in creating an autonomous region and [for] putting pressure on Kurdish opposition members who do not support the PYD" (Gunes and Lowe 2015, 10). By supporting the KRG, Ankara wanted Erbil to play a balancing role against the PYD (Bengio 2014, 276). In 2014, Ankara allowed Erbil to move Peshmerga forces through Turkey during the siege of Kobani, even though Turkey was reluctant to help the PYD due to its affiliation with the PKK. The reason for Ankara and Erbil's volte-face was to control the area lost by the PYD, as they hoped to fill the power vacuum in Kobani. Indeed, due to a deal signed between Erbil and the PYD in June 2012, Erbil sent Peshmerga forces to support the PYD and decide how power in Kobani would be shared (Tanir 2014). Therefore, the cordial ties between Erbil and Ankara shaped Erbil's relations with the PYD.

By contrast, the emergence of ISIS laid the foundation for a short-term rapprochement between the PYD and the KRG. In August 2014, the YPG and PKK conducted intensive operations to save Yazidis from ISIS in Iraq and defend Sinjar. In return, two months later, the Peshmerga forces assisted the YPG's defence of Kobani against ISIS and supplied military equipment (International Crisis Group 2018). In October 2014, this mutual assistance resulted in the Duhok Agreement on power-sharing between the Erbil-controlled KNC and the PYD. Nevertheless, the friendship between the KRG and Turkey always dominated the collaboration between the two Kurdish groups due to the KRG's economic and political dependence on Ankara (Gunes and Lowe 2015, 11–12). As such, the partnership between the KRG and the PYD was short lived due to the power struggle between the two Kurdish groups and the Ankara-Erbil friendship. As a result, the rise of the PYD determined Turkish policies in the KRI, culminating in warmer relations between the KRG and Turkey. Thanks to pragmatic and close ties with Erbil, Ankara had the opportunity to enhance its influence more than Iran. In conclusion, Erbil became a more significant ally of Ankara in an anarchic environment in which Turkey's ties with both Syria and Iran were cooling (Charountaki 2012, 195–96). Following the Syrian conflict, Erbil became a more important ally for Turkey since Ankara was becoming isolated by increased collaboration between Tehran and the Assad regime. From Turkey's perspective, if the Assad regime remained in power, Iran might increase its regional influence, resulting in a corresponding decrease in Turkey's. Given Ankara's concerns, Erbil served as a conduit for Ankara's aid to opposition Sunni groups in Syria (Koolaee and Hafezian 2017, 82; Lindenstrauss and Aksoy 2012, 54). Ankara, therefore, attempted to use the KDP to prevent both the rise of the PYD and Iranian influence in the long term. In other words, closer ties with Erbil meant the rise of Ankara's relative power position in the KRI for Turkey compared to Tehran. Having explained the role of systemic variables on TFP towards Iran, the next section highlights how domestic factors emerged as a causal link in shaping TFP towards Iran in the short-to-medium term.

7.2 The Role of the Unit-Level on Turkish Foreign Policy towards Iran since the Arab Uprisings: December 2010 to July 2018 —the Case of the KRI

Without elucidating intervening (unit-level) variables, it is difficult to explain TFP towards Iran in the KRI because domestic factors explain the character, type, and adjustment of Turkey's policy attitude towards Iran in the short-to-medium term. In this context, the perception of the AKP elite and profit-oriented domestic considerations shaped TFP towards Iran in the KRI.

7.2.1 AKP Elite Perception

Before the AKP came to office in 2002, the Kemalists (secular parties, the Turkish military, and extremist nationalists) deemed the Kurdish entity on Turkey's borders to be a security threat, because of the belief that the PKK and the KRG were similar ethnic entities (Tol 2014, 69). Although the Kemalists refrained from establishing ties with Iraqi Kurds, the AKP changed Turkey's attitude towards the Middle East and de-securitised Turkey's perception of Iraqi Kurds, especially after 2008. As the governing party, the AKP attempted to solve the domestic Kurdish issue through the Peace Process in 2013, with the help of Iraqi Kurds and, accordingly, the AKP elites preferred greater collaboration with them (Kayhan-Pusane 2016, 23).

The foreign policy principles actualised by Foreign Minister Ahmet Davutoglu, such as 'zero-problems-with-neighbours', created the opportunity for an Ankara-Erbil detente, and Turkey's perception of Iraqi Kurds became more progressive (Bengio 2014, 275). For example, the friendship between Erdogan and Barzani permitted growing levels of interdependence, indicating a shift from a security-oriented policy to a "political and civilian approach", even though the majority of the Turkish public maintained a sceptical view of Kurds (Dalay 2017b, 3, 5; Mustafa and Aziz 2016). Given the personal bonds between the two leaders, in March 2011, Erdogan became the first Turkish prime minister to visit the KRI, and even used the term "Kurdistan region"⁶ to refer to northern Iraq where the Iraqi Kurds lived, and said that "in Erbil, we have felt like being in our city. Now we feel Diyarbakir as your city" (Kayhan-Pusane 2020, 4). In so doing, Erdogan hoped to increase his Kurdish supporters and distinguished the PKK and the KDP (Kayhan-Pusane 2016, 23-24). Other AKP elites further mentioned their ideas regarding an independent Kurdish state. To illustrate, in 2014, the then AKP spokesperson, Huseyin Celik, noted that "in the past an independent Kurdish state was a reason for war [for Turkey], but no one has the right to say this now...If Iraq is divided and it is inevitable, they are our brothers..." (Kayhan-Pusane 2020, 4). The remarks of the AKP elites show that the AKP's policy-makers used Iraqi Kurds to shape Turkey's Iraq policy and were unconcerned about "the prospect of an independent Kurdistan in northern Iraq" (Tol 2014, 69) as long as the KRG serves the AKP's interest. More precisely, the AKP elites perceived the KRG as a means by which they could expand their own interests.

Accordingly, individual interests played a determining role in the goodwill between the AKP and the KDP. Notably, the similarity in their domestic politics culminated in cooperation because Erdogan and Barzani were both blamed for following authoritarian domestic policies, based on their interests and decisions. The secularists, the Gulenists and the liberals mostly criticised Erdogan's policies, and Barzani's legitimacy was denounced by Kurdish opposition parties, including the PUK and Gorran (Özpek 2018, 45, 53). Therefore, similar internal

⁶ Previous Turkish leaders had refrained from using the word 'Kurdistan', since they believed that this would aggravate the Kurdish issue in Turkey. For details *see*: Yavuz, M. Hakan (2001). Five stages of the construction of Kurdish nationalism in Turkey, *Nationalism and Ethnic Politics*, 7:3, 1–24; White, Paul J. (1999) Citizenship under the Ottomans and Kemalists: How the Kurds were excluded, *Citizenship Studies*, 3:1, 71–102

criticisms of the two leaders' legitimacy created greater personal closeness between Erdogan and Barzani.

The AKP elites also sought closer ties with the KRG due to Turkey's animus towards the PKK and the rise of the PKK-affiliated PYD (Park 2014, 8). During the Peace Process initiated in the first three months of 2013, the AKP elites deemed Barzani more "popular and an alternative Kurdish leader" than the PKK leaders due to their mutual distrust of "conservative Sunni Muslims with links to the Naqshbandi religious order, and a network of business ties among members of their inner circles" (Kayhan-Pusane 2016, 23; Strategic Comments 2017, 2). They thus perceived the KDP and Masoud Barzani as the only Kurdish representatives and refused to deal with other Kurdish groups. According to Özpek (2018, 55), the policies of the AKP elites "became more visible after the PKK-linked Democratic Union Party (PYD) gained control of some towns in northern Syria during the civil war", since they encouraged Barzani to lead the Kurdish formation in Syria. Therefore, ethnic differences between the AKP and the KDP played a limited role in shaping their bilateral affairs, and the interests of the former predominated in the pursuit of closer ties with the KRG. In other words, the AKP elites ignored the Kurdish ethnicity of the KDP for as long as they had mutual interests.

Given these individual interests, national interests became another reason for the friendship of the AKP elites with the KRG, because the former were undecided about the rise of Shia Iranian influence over Baghdad (Park 2014, 9). The Iraqi parliamentary elections held on 7 March 2010, notably, ushered in a period of disagreement between Ankara and Baghdad (Tocci 2013, 70). During this term, the AKP endorsed the Iraqi politician Ayad Allawi's non-sectarian movement in the elections and urged "the involvement of Sunni Arabs in Iraqi political structure" to impede the rise of sectarianism in Iraq (Bengio 2014, 271; Park 2014, 16). Since then, the AKP blamed the Shi'a-dominated Maliki government for following sectarian policies in Iraq and assumed that the Maliki government would fail to consolidate the internal balance of power (Bengio 2014, 271). Indeed, Maliki's increasing sectarian policies justified the AKP politicians' concerns. Following the decreased US presence in Iraq, Maliki attempted to arrest Tariq Al-Hashemi, the Sunni Vice President of Iraq from 2006–2012, who had close ties with Turkey (Park 2014, 16; Stephen Larrabee 2013, 142). In this regard, the AKP elites emphasised the significance of the KRG and opted for collaboration with it, in contrast to the Iran-aligned Maliki government, which denounced Ankara's protection of Al-

Hashemi and accused Ankara of intervening in Iraq's internal affairs (Gunter 2013, 446; Park 2012, 113–14).

Nevertheless, the AKP elites changed their stance towards Iraqi Kurds after the September 2017 independence referendum, and this shift coincided with the termination of the Peace Process and the rise of PKK-affiliated political party HDP in Turkish domestic politics in 2015. When the HDP gained seats in the Turkish Parliament, surpassing the steep 10% threshold in the June 2015, national election, the AKP lost its majority and required a coalition partner to form the government. This election failure surprised the AKP elites, and the party revised its domestic policies by pursuing nationalist viewpoints with a far-right political party, the Nationalist Movement Party (MHP). As a result, the AKP elites adopted re-securitisation policies towards the Iraqi Kurds following the referendum. They started to tread less carefully in their relationship with Iraqi Kurds and slammed the 2017 independence referendum. Notably, President Erdogan shifted his position towards the Iraqi Kurds, despite his warming ties with Barzani, saying that "We are against a Kurdish state even if it were established in Argentina" (Stansfield 2013, 277).

Bearing in mind that the rise of nationalism and re-securitisation in Turkish politics increased the AKP elites' aggressive policies towards Iraqi Kurds (Park 2019, 53), a short-term collaboration between Ankara and Tehran emerged when the KRG declared its desire to hold an independence referendum in 2017. This was because both were concerned about a potential spill-over effect from the referendum on their Kurdish minority, perceiving a "generation of internal instability in the KRG and the fact that weapons from the KRG might find their way into the hands of militant Kurds in either Turkey or Iran" (Thornton 2015). As a result, the AKP elites abandoned their de-securitisation policies towards the Iraqi Kurds following the 2017 referendum for a short period of time and formed a brief alliance with Tehran as this was more beneficial than competition due to their mutual concerns regarding Kurdish independence. This indicates that cost-benefit calculations forced Turkey to cooperate with Iran, and that these ties were profit-oriented and interest-driven. The change of perception of the AKP elites towards the Iraqi Kurds further was pertinent to Turkey's long-term security considerations, which will be examined in the following subsection.

7.2.2 Domestic Considerations

Economic and security considerations pushed Ankara to establish warming ties with the KRG. Based on Ankara's quest for close economic interests, as one report noted, "Turkey was the sole conduit for the Kurdistan Region's oil exports, which averaged around 600,000 barrels per day. The region also accounted for almost all of Turkey's exports to Iraq, which in 2016 totalled US\\$7.5 billion, making the region Turkey's third-largest export market" (Strategic Comments 2017, 2). In the competitive environment of the Arab Uprisings, Ankara perceived Erbil as "an alternative source of energy, a buffer against a hostile Baghdad and Iran" (Tol 2014, 70). Until 2014, Turkish companies, with almost 50,000 Turkish employees, represented 60 per cent of all the companies in the KRI. The volume of trade between Ankara and Erbil significantly improved, and Ankara-Erbil economic cordiality was dubbed "oil diplomacy" (Bengio 2014, 274; Mustafa and Aziz 2016). However, in the case of the KRI, security considerations rather than economic considerations were more dominant in Turkey's collaborative affairs with Iran because Ankara's security-based concerns forced Turkey to ally with Tehran in the short-tomedium term, despite their rivalry in the KRI in the longer term. This subsection thus emphasises Turkey's domestic security considerations regarding the KRI under the Schweller's balance of interest theory. As mentioned earlier, Turkey's security considerations were based on the presence of the PKK both in its territory and proximity. Thus, the Ankara-Erbil collaboration mainly depended on the KRG's status quo policy against the PKK (Gunes 2019b, 35), while the Ankara-Tehran alignment primarily resulted from the situation of Kurdish groups in the region. In other words, Turkey's quest for a short-term alliance with Iran was conditional upon Ankara's need for Tehran's support when Ankara believed that Erbil's support was insufficient to tackle Turkey's concern regarding the Kurdish issue.

With the termination of the Peace Process between the PKK and the AKP government and the rise of HDP in Turkish domestic politics in 2015, Ankara became less disposed to the empowerment of Kurdish groups, despite its warming ties with Erbil. Since then, Turkey perceived any Kurdish movement as a security threat and sought collaboration against the rise of Kurdish groups. With the aforementioned increase of HDP at home, the AKP elites began to follow a more nationalist approach, and opposed any independent Kurdish state in Turkey's proximity; they highlighted the territorial integrity of Iraq by claiming that an independent Kurdish state would aggravate regional anarchy (Gunes 2019b, 35; Sinkaya 2018, 857). In this regard, Ankara denounced the 2017 independence referendum of Iraqi Kurds, voicing the concern that any independent Kurdish state would trigger Kurds in Turkey to seek independence.

Despite Turkey's condemnation, in March 2017, the KRG flew the Kurdish flag on Kirkuk's Iraqi government buildings. In return, Ankara declared that "Kirkuk was historically a Turkmen city, regardless of whether some accept it or not", deeming the KRG as an occupant and warning Barzani about Turkey's future operations in the KRI. President Erdogan further noted that "the KRG's adventure was an act of treachery and will be over when we close the oil taps, all [their] revenues will vanish, and they will not be able to find food when our trucks stop going to northern Iraq" (Park 2019, 52). The reason for Ankara's threats and objections was its concern over the future of Kirkuk, as Ankara assumed the annexation of oil-rich Kirkuk would actualise an independent Kurdish state (Park 2014, 8). In this regard, Turkey sought to work with Iran as another opponent of a Kurdish state.

During this period , Ankara was also experiencing a cooling in its relationship with the US due to the latter's support for the PYD; Ankara assumed that the KRG referendum was a US project, as was the rise of the PYD in northern Syria (Park 2019, 54). Likewise, Iran viewed the referendum as a threat to its regime's security since a possible Kurdistan state supported by the US, Israel, and the Gulf Countries would endanger the Iranian regime. As such, Iran emerged as a viable partner for Turkey in opposing the Kurdish referendum, should the US criticise Turkish actions in the KRI. Erdogan notably opted to liaise with Iran and the Iraqi government after understanding that he could not prevent the referendum despite his friendship with Barzani (Dalay 2017b, 4–6). As a result, pragmatic and profit-oriented collaboration emerged between Ankara and Tehran towards the 2017 independence referendum of the Iraqi Kurds for a short-time period.

Before the referendum, Turkey conducted several military exercises in Silopi, a Kurdish majority settlement in Turkey's Sirnak Province, adjacent to Iraq and Syria. Ankara further worked with Tehran and Baghdad to pursue joint counter-measures against the KRG's quest for independence, and emphasised the territorial unity of Iraq (Dalay 2017b, 7; Gunter 2018, 29). After the referendum, Ankara threatened, along with Baghdad, to close its borders with the KRI and to reroute trade shipments through Iran (Sumer and Joseph 2018, 13). However, these Turkish measures were insufficient to prevent the referendum and had little effect on the result. Given its similar concerns and shared policies with Ankara, Tehran also feared that, if

the referendum had been successful, a new civil war would have emerged in the region, and Iran would have lost Iraq (Expert 1 personal communication, August 17, 2020). Tehran was also concerned that Ankara and Tel Aviv would gain an edge over Iran by collaborating with the Barzanis, even though Israeli support for the referendum was met with unease in Ankara. To hamper this probability, Tehran increased its collaboration with the PUK, as well as with Ankara, by following more active policies in the KRI than Turkey (Gunter 2018; Sinkaya 2018). In essence, Tehran and Ankara's policies towards the referendum demonstrated that Tehran had a more powerful position than Ankara due to its partnership with various actors in the KRI. In contrast to Tehran, Ankara had close ties only with the KDP under the Barzanis, and therefore required more external support than Tehran in the KRI.

The 2017 referendum further aggravated the domestic competition and political fragmentation in the KRI because the PUK, Gorran, and Kurdistan Islamist Group (Komal) assumed that Barzani would utilise the referendum to consolidate his power. These Kurdish parties further opposed Turkish attacks against the PYD/YPG in Afrin, Syria, and "visited Afrin in February, 2018 to express solidarity with the YPG's resistance" (Park 2019, 56). They thus opted to liaise with Tehran in contrast to Ankara. With these partnerships, Tehran increased support for Baghdad in regaining Kirkuk from the Iraqi Kurds on October 16, 2017. During this period, Ankara and opposition Iraqi Kurdish parties kept silent about Iranian operations in Kirkuk (Gunter 2018, 29–30). Journalist 1 (personal communication, July 1, 2020) noted that Turkey preferred Iranian control in Kirkuk rather than Kurdish groups. This shows that, except for the Barzanis, Turkey's limited relations with Kurdish groups in the KRI prompted a need for Tehran's support on the 2017 referendum. Ankara thought short-term cooperation with Tehran would be beneficial to Ankara's security interests in the KRI when Ankara was disappointed with the referendum of the Iraqi Kurds.

Moreover, trust issues emerged between Ankara-Erbil because the KDP failed to see the rise of nationalism in Turkey's domestic politics. The findings from Expert 3 (personal communication, July 21, 2020) further revealed that Turkey felt betrayed. This led to Turkish criticism of the referendum and more prudent and flexible policies towards Erbil, with the KRI losing importance for Turkey. However, regarding the 2017 referendum, it was noted in Expert 1 (personal communication, August 17, 2020) that the dependence of the KRG on Turkey increased, and that the tie between Turkey and the KRG was so close that the KRG was like a Turkish city. Although the KRG has more recently been off Turkey's external agenda, the 2017 referendum failure forced the KRG to seek closer ties with Turkey because of the latter's projection of military power in operations in northern Syria.

The rise of the PKK and PJAK close to the Qandil mountains resulted in further collaboration between Ankara and Tehran, since their attacks from the KRI posed a severe challenge to both countries. With the PKK/PJAK using the Qandil mountains as a refuge from Turkish and Iranian attacks (Unver 2016, 78), Ankara and Tehran assumed that Erbil was ignoring the PKK/PJAK because Erbil sometimes hesitated to restrain their operations from Qandil, despite the competition between Barzani and Ocalan. Ankara thus accused Erbil of not taking military measures against PKK militants in northern Iraq. In contrast to Ankara's aggressive policies towards the PKK, Barzani indeed believed that "Turkey's Kurdish problem can only be resolved by dialogue and by some degree of autonomy for Turkey's Kurdish provinces" (Park 2012, 117–19). In this regard, Park (2019, 51) notes that "Barzani was perhaps proving to be a less valuable strategic asset than Ankara had hoped, and Ankara a less accommodating neighbour than Erbil thought it had become". Barzani's periodic hesitation thus pushed Ankara to see collaboration with Tehran against the PKK-PJAK issue.

Academic 3 (personal communication, July 9, 2020) noted that Turkey's concerns regarding the PKK was greater than Iran's PJAK concern since the PKK was more effective in Turkey than the PJAK in Iran. Similarly, it was argued by Academic 4 (personal communication, July 6, 2020) that Iran thought it could control the Kurdish separatist movements because of the authoritarian nature of the regime. In comparison, therefore, Turkey was more concerned about the Kurds than Iran, and this was a concern which had arisen when Turkey established a nation state following the demise of the Ottoman Empire. With the foundation of the Turkish Republic, all ethnicities were regarded as unitary under the idea of Turkishness, but ethnicities in Iran historically lived together regardless of race and identity. Therefore, Turkey's Kurdish issue is different from Iran's; Turkey fears Kurdish separation and follows security-oriented policies, and Ankara's harsh response towards the referendum vindicated its real perception of the Kurds. In this regard, Ankara required an alliance against any Kurdish separatism more than Tehran.

7.3 Concluding Remarks

The KRI became a field both of cooperation and competition for Turkey and Iran, and there was no 'win-win policy' in Turkish-Iranian affairs there due to co-determination of systemic and unit-level factors. The case study in this chapter highlights the relevance of NCR in explaining TFP towards Iran from December 2010 to July 2018, by analysing particular systemic and unit-level variables in the KRI. NCR examines structural changes in regional and international systems and the relative power of a state as systemic variables. As such, the three factors of the US role in Iraq, the Syrian conflict, and Turkey's relative power position can be seen to have directly determined TFP towards Iran in the KRI. More precisely, these developments had determinant roles on TFP for Ankara's long-term interests. These three factors will now be examined in turn.

First, the US role in Iraq shaped Turkish behaviour in the country because Turkey hoped to fill the power vacuum left by the decreased US presence. Bearing in mind that Tehran increased its influence over the Shi-a dominated Maliki government in Baghdad and established cordial ties mostly with the PUK in the KRI, Ankara tried to establish closer ties with the KDP under the Barzanis to balance the Iranian impact throughout Iraq. As a result, Ankara's desire for friendly relations with the KRG enraged Baghdad and Tehran. Second, the Syrian conflict empowered non-state armed actors, including ISIS and the PYD/YPG, in Turkey's proximity, leading Ankara to reconsider its foreign policy behaviour regarding its security interests. Despite initially ignoring the rise of ISIS due to its aggressive policies against the Assad regime and the PYD, Ankara later opted to support Erbil in order to protect its own long-term interests in the KRI. In contrast, Tehran followed closer ties with Erbil in fighting ISIS, but Ankara had no desire to lose its long-term interests and gains in the KRI.

Moreover, the rise of the PYD held a potential risk for Ankara in terms of a loss in its regional power position, as becoming an influential actor of the PYD meant empowering the PKK and the states supporting these two parties. As such, Ankara pursued greater collaboration with Erbil to prevent the empowerment of PYD, as a mutual concern, through economic, political and security strategies. Indeed, Ankara's quest for collaboration with Erbil was based on power maximisation to improve and maintain its regional power position. Given these systemic variables, it appears that Ankara retained competitive policy behaviours towards

Tehran due to their similar power position in the international system and their conflict of interests in the longer-term.

Nevertheless, the regional balance of power evolved contrary to Turkey's expectations in the short-to-medium term. In this regard, Turkey put aside its long-term interests and policies towards Iran in favour of collaboration. NCR helps to explain Turkey's volte-face policy, since it can be argued that unit-level variables played a determinant role in the state's short-tomedium term foreign policy behaviour. As such, the main reasons behind Turkey's quest for cooperation with Iran were the twin perceptions of the AKP elite, and domestic security considerations at the unit level. These elites initially pursued a policy of de-secularisation towards Iraqi Kurds since they assumed that the latter could resolve Turkey's PKK issue and prevent the rise of the PYD in the region; however, their perception altered due to the rise of domestic nationalism. Accordingly, the AKP elites started to follow nationalist and anti-Kurdish policies in internal and external politics, reflecting a more ambivalent policy shift in Ankara's ties with Erbil. Following the Iraqi Kurdish independence referendum in 2017, significant contentious issues arose between Ankara and Erbil, as the former feared the referendum would threaten its domestic stability by having a spill-over effect among Kurds in Turkey.

In this period, Ankara also drove a wedge between itself and Washington due to the former's unwavering support for the PYD. As such, the rise of nationalist policies in Turkey further cooled its ties with the US, alongside other issues such as the US' reluctance to expel Fethullah Gulen after the 2016 coup attempt in Turkey. Ultimately, Turkish-US relations entered a period of uncertainty and disagreement, and Turkey faced significant issues with its long-term Western ally due to its domestic security concerns. Given the shift in the AKP elite's perception and Turkey's domestic security considerations —arising directly from the developments in the KRI —Ankara chose pragmatic and profit-oriented short-term cooperation with Tehran, as well as Baghdad. In this regard, Turkey's quest for cooperation with Tehran vindicated Schweller's balance of interest approach. In collaborating with Tehran, Ankara hoped to counterbalance threats, and, at the same time collaboration with Tehran provided more gains than the US in the KRI, since Tehran shared mutual concerns with Turkey. Therefore, a status quo-based foreign policy with Tehran was a better option for Turkish interests in the short-to-medium term than a revisionist one. In this regard, Schweller's balance with Tehran in

the case of the KRI. Table 7.1 summarises TFP towards Iran since the Arab Uprisings in the KRI.

Systemic Variables (Independent)	Intervening Variables (Unit-Level)	Dependent Variables (Foreign Policy Behaviour in the Short-to-Medium Term)
 US Policies Syrian Conflict The Rise of Non-State Armed Groups (ISIS and the PYD/YPG) Turkey's Relative Power Position 	 AKP Elite Perception Domestic Security Consideration 	 Collaboration Status Quo Behaviour Balance of Interest

Table 7-1 TFP towards Iran in the KRI: December 2010 to July 2018

This chapter summarises previous chapters, presents the main conclusions of the research, and then explores the key contributions, limitations, and recommendations for future research.

8.1 Executive Summary and Arguments

The objective of this thesis was to offer an original understanding of Turkish foreign policy (TFP) towards Iran amid the upheavals caused by the Arab uprisings, from December 2010 to July 2018, by using NCR. Turkey-Iran relations have always been volatile, involving both conflict and cooperation. With the start of the Arab uprisings, Turkey's relations with Iran reverted to a state of conflict and cooperation. Despite a lack of trust between Ankara and Tehran, Turkey cooperated with Iran in the short-to-medium term primarily due to domestic factors. As a result, the thesis explains which factors resulted in competition or cooperation in shaping TFP toward Iran from 2010 to 2018. Accordingly, Chapters 2–3 constituted the background. As a theoretical chapter, Chapter 2 explained how NCR provides a better conceptual explanation for TFP towards Iran from December 2010 to July 2018, because the theory employs both systemic and unit-level variables to analyse a state's foreign policy behaviour. Chapter 3 analysed the historic role of systemic and bilateral determinants in TFP towards Iran until the Arab Uprisings. Consequently, the thesis then asked:

1. To what extent did structural forces shape the formulation of TFP towards Iran since the Arab Uprisings, between December 2010 and July 2018?

Chapter 4 sheds light on the role of systemic factors and Turkey's relative power position towards Iran since the Arab Uprisings, from December 2010 to July 2018. With the outset of the revolts across the Arab world, the regional system experienced significant changes, including the role of Muslim Brotherhood (MB)-affiliated Islamic parties, the Sunni-Shia divide, the role of global powers in the Middle East, and the empowerment of non-state armed groups. As such, Turkey's relative power position and consequently, its foreign policy were influenced by these systemic changes, especially for its long-term policies. At the beginning of the Arab Uprisings, Turkey first followed a wait-and-see policy, a position shaped by its economic and political ties with previous authoritarian regimes under the zero-problems-withneighbours policy and economic interdependence. However, the increasingly anarchic environment pushed Turkey to reconsider its initial policy towards the uprisings, and it thus tried to fill the power vacuum as a regional middle power, following a power-maximisation policy. Bearing in mind Turkey's desire to be a regional model for political and social development, Turkey competed with Iran, as another influence-seeking regional power. While it cannot be determined with any degree of certainty, it seems that in the long-term Turkey will maintain its competitive policies towards Iran since the two countries are likely to seek to dominate the regional developments for their own interests.

2. How did unit-level (domestic) factors interact with structural imperatives to shape TFP towards Iran since the Arab Uprisings, between December 2010 and July 2018?

Chapter 5 defined the role of the unit-level variables on TFP towards Iran since the Arab Uprisings. The chapter points out that even though systemic shifts and Turkey's relative power position played an influential role in shaping TFP towards Iran, the AKP elite's perception and domestic considerations, as unit-level variables, forced Turkey to cooperate with Iran in the short-to-medium term. Bearing in mind that Schweller's balance of interest theory can explain the reason for the alliance choice between the two states, the chapter sheds light on how Turkey's interest-based and profit-oriented policies culminated in its cooperation with Iran in the short-to-medium term. The AKP elites first seemed to follow sectarian policies towards the uprisings due to its unwavering support for MB-affiliated parties and Sunni armed groups in the region. However, their pro-MB policies backfired and caused Turkey to become increasingly isolated in the region, notably from the Saudi-led Gulf countries and the US.

While following Sunni-based policies in the region, the AKP elite refrained from breaking its ties with Shia-dominated Iran for two reasons: Turkey's isolation pushed the AKP elite to establish closer relations with Iran; and domestic security and economic considerations prevented Ankara from maintaining competitive policies against Tehran. Following the Arab Uprisings, unit-level factors were more influential drivers of TFP towards Iran, since, instead of sectarian and identity policies, Ankara opted for interest-driven and profit-oriented policies. Ankara saw a short-term alliance with Tehran more beneficial than their competition. However, NCR highlights how unit-level variables play a significant role in a state's foreign policy in the short to medium term, while in the longer term a state's relative power position is the primary driver of its foreign policy. As such, it appears that the impact of the AKP elite's perception and domestic considerations is, in the short term at least, more influential in shaping TFP towards Iran.

As well as the main research questions, the thesis asked:

3. Since the Arab Uprisings, did Turkey cooperate with its western allies or turn away from them over its relations with Iran?

At the beginning of the Arab Uprisings, Turkey followed a western-based policy and cooperated with its western allies in Libya and Syria. Later, however, Ankara conflicted with these allies over certain developments, including the toppling of the MB-affiliated Morsi regime in Egypt, the future of the Assad regime, and the rise of the PYD/YPG in Syria. More precisely, Turkey's interests determined its relations with its western allies: when its interests clashed with those of its allies, Ankara forged a different path. In the Syrian conflict, Turkey initially sought to cooperate with its western allies to topple the Assad regime, but these allies changed the priority from overthrowing the Assad regime to the prevention of ISIS. Since Turkey maintained the original aim of toppling the Assad regime, it adopted different policies from its allies until the rise of ISIS began to threaten Turkey's domestic security. Moreover, the empowerment of the PYD through US support led to a lack of trust between Turkey and the US, with Turkey criticising US support for the PYD in fighting ISIS. Finally, following the coup attempt in 2016, the lack of trust increased between Turkey under the AKP government and its western allies, since Ankara noted how they remained silent about the coup attempt. This increasing distrust of western allies brought Ankara closer to Russia and Iran, who immediately voiced their support for the AKP government. This marked a decisive shift in Ankara's relations with its erstwhile NATO allies.

4. Did Turkey follow a value-based or interest-driven foreign policy after the Arab uprisings?

At the outset of the Arab Uprisings, Turkey claimed to support a value-based policy towards the Arab Uprisings and democratic change in the region. In this regard, it aimed to become a role model for Islamic values and democratic principles. Despite these aspirations, Turkey's actual policies reveal that it followed an interest-driven policy in the region. While claiming to endorse democratic change, for instance, Turkey remained quiet about the Shia-dominated uprising in Bahrain. In doing so, Turkey followed a pro-Sunni policy in Bahrain by cooperating with other Sunni states. Turkey further supported MB-affiliated parties throughout the region and opposed any movements against these groups. It also opposed a possible Kurdish state on its borders when the PYD declared autonomy in northern Syria. Even though Turkey pursued anti-Assad policies in Syria, the rise of the PYD emerged as a more significant issue for Turkey's interest in the region, because a stronger PYD could inflame the Kurdish problem in Turkey because of the close ties between the PYD and the PKK. As such, Turkey focused on its own interests after the Arab Uprisings. When these interests were at stake Turkey adopted more interventionist policies, as in Syria. Moreover, its support for MB-affiliated parties shows how Turkey followed interest-based policies, because the AKP elite assumed that the ideological affinity between the AKP and MB-affiliated parties would be in Turkey's favour in the region.

5. How did Turkey's relative power position and the analysis of the AKP elite's perceptions and domestic considerations shape TFP towards Iran in the Syrian conflict and the Kurdistan Region of Iraq after the Arab Uprisings?

Chapters 6–7 defined TFP towards Iran following the Arab Uprisings in the Syrian conflict and the Kurdistan Region of Iraq, respectively, as case studies of the thesis. Both chapters examine systemic factors, Turkey's relative power position, the AKP elite's perception, and Turkish domestic security and economic considerations. Focusing on the Syrian conflict, Chapter 6 explained that the internalisation of the conflict (inter-state competition and the rise of non-state armed groups), Turkey's relative power, and its foreign policy in the Syrian conflict directly emerged as independent variables in determining TFP towards Iran. As two regional middle powers, Turkey viewed Iran as a rival at the outset of the Arab Uprisings, and so it competed with Iran to increase its long-term influence throughout the region, by promoting itself as a role model. As such, Turkey's competition with Iran was maintained in Syria, where Iran played a dominant role in supporting the Assad regime, while Turkey endorsed Sunni groups.

However, Turkey's domestic politics (the AKP elite's perception and domestic considerations) prevented competitive policies against Iran. Although the AKP elite followed an identity-based policy and supported Sunni groups in the Syrian conflict, its misconception

of the post-conflict term pushed Turkey to establish closer ties with Iran in the short to medium term. The AKP elite believed that its western allies would continue to want to end the Assad regime quickly, but western countries, notably the US, began to prioritise fighting with ISIS over the fall of Assad. In this regard, Turkey conflicted with its western allies and began to follow more interventionist and over-ambitious policies than its western allies. Given its aim for the post-conflict term, Turkey endorsed Sunni opposition groups, including the Free Syrian Army and Al-Qaeda-affiliated Jabhat Al-Nusra. Although Turkey initially ignored the rise of ISIS due to its opposition to Assad, Turkey's sectarian policy backfired and was criticised by its western allies. The rise of the PYD further resulted in divergent policies being pursued between Turkey and its western allies. While the US supported the PYD, Turkey viewed the Kurdish group as a security threat due to its PKK affiliation, and criticised the US for supporting a terrorist organisation. As such, Turkey was isolated due to its over-ambitious and sectarian policies in the Syrian conflict, and Turkish policymakers sought an alternative partner in Turkey's policies in Syria.

As well as the AKP elite's perception, Turkey's domestic economic and security considerations determined TFP towards Iran in the Syrian conflict. Before the Arab Uprisings, Turkey had established close economic and trade ties with Syria under the zero-problems-withneighbours policy, but once the conflict began Turkey lost economic partners in various uprising countries, including Syria. Moreover, Turkey's interventionist and overambitious policies in Syria and the immense refugee flow from the latter increased economic costs for Turkey. At the same time as Turkey was dealing with problematic political and economic ties with other Sunni states and its western allies, Iran emerged as an alternative economic partner for Turkey. Alongside its economic considerations, security concerns arising from the rise of ISIS and the PYD pushed Turkey to cooperate with Iran. While ISIS conducted several terrorist attacks both in Turkey and in its proximity, the affiliation between the PKK and PYD posed a considerable security threat to Turkish domestic security.

Chapter 7 shed light on TFP towards Iran in the Kurdistan Region of Iraq (KRI), as the second case study in the thesis. Like the Syrian conflict, systemic and unit-level variables played determinant roles in TFP towards Iran since the Arab Uprisings, from December 2010 to July 2018. As systemic variables, US policies in Iraq, the developments of the Syrian conflict, and Turkey's relative power position and its foreign policy strategies directly shaped Turkey's ties with Iran and created competition between the two regional middle powers. As

such, Turkey deemed Iran a rival in the KRI, with both countries supporting different Kurdish groups. Turkey sought to establish closer ties with the KRG, notably with the KDP under the Barzanis, to prevent the rise of Iranian influence across Iraq. This was because Iran had already improved its relative power position following the decreased US presence in Iraq via closer ties with the Shia-dominated Maliki government in Iraq and the PUK, another significant Kurdish group in the KRI.

Equally, however, the AKP elite's perception and domestic short term security considerations played a direct influential role in Ankara's alliance choice with Iran in the KRI. Following the rise of the HDP in Turkish domestic politics in 2015 and the 2016 coup attempt, the AKP elite started to adopt nationalistic discourse regarding the Kurds, reflecting a change in their ties with the KRG even though they had previously followed de-secularisation policies with Erbil. Notably, the 2017 independence referendum raised the elite's concerns regarding a possible Kurdish state in Turkey's proximity. Moreover, as in the Syrian conflict, ISIS and the PYD remained a security threat to Turkey, and so Turkey needed an ally in Iran over the KRI, despite their mutual competition in Syria.

To sum up, this thesis offers an original understanding and interpretation of TFP towards Iran since the Arab Uprisings, by establishing a causal mechanism between independent and dependent variables. As such, the thesis uses the AKP elite's perception and domestic security and economic considerations as intervening variables to clarify the cause-effect relationship between systemic shifts and Turkey's relative power position (as independent variables) and TFP towards Iran (the dependent variable) in the short to medium term. In this regard, NCR provides a robust theoretical framework for the intervening variables to explain the causal link between independent and dependent variables. Schweller's balance of interest also helps explain Turkey's alliance choice with Iran, considering Turkey's economic and security considerations. Based on this approach, Turkey opted to follow status quo and profit-oriented policies with Iran in the short to medium term. These explanations were evaluated through the literature survey and expert interviews. Thus, process-tracing methodology facilitated the understanding of the causal link between systemic shifts and Turkey's relative power position, and TFP towards Iran in the short to medium term.

8.2 Contributions of the Research

The analysis of bilateral relations between Turkey and Iran, as two regionally significant middle powers, has increasingly gained currency in the literature of Middle Eastern studies. However, few studies have explained the Turkish perspective towards Iran via a holistic approach, and majority of literature regarding Turkish-Iranian affairs since the Arab Uprisings have focused on their rivalry in Syria. Notably, TFP towards Iran between 2011-2018 is still under-theorised as most studies either lack a sufficient time frame or focus on bilateral affairs rather than focusing solely on the Turkish side. The first contribution of this thesis is therefore to add new knowledge which helps fill this gap, through the analysis of TFP towards Iran via a theoretically informed account. Based on NCR, the most important contribution of the thesis conceptually is the causal relationship identified between the independent variables (systemic shifts arising from the Arab Uprisings and Turkey's relative power position) and the dependent variable (TFP towards Iran) in explaining Turkey' ties with Iran. As unit-level variables, the AKP elite's perception and the domestic considerations of Turkey provide a fruitful explanation for the causal link between the systemic shifts arising from the Arab Uprisings and Turkey's relative power position, and TFP towards Iran. Therefore, the thesis fills a gap in the literature by providing the co-determination of intervening and independent variables in TFP towards Iran through an analysis of different unit-level variables in these studies' explanations of TFP.

The second significant achievement of this study is that Turkey's desire for cooperation with Iran can be explained through NCR and Schweller's idea of a balance of interest. Due to Turkey's domestic economic and security considerations, Turkey sought to establish closer ties with Iran, proving that interest-driven and profit-oriented aims defined TFP towards Iran in the short-to-medium term. More precisely, Turkey was motivated by both opportunities for gain and its security and economic concerns in forming an alliance with Iran, and thus opted to follow status quo behaviour towards Iran despite their competitiveness in the wake of the Arab Uprisings. The alliance provided shelter for Turkey against its regional isolation and security threats in its proximity. Although several studies have touched upon the economic and security cooperation between Turkey and Iran, this thesis is the first study to examine Turkey's alliance of interest theory to define Turkey's domestic considerations is a more appropriate approach for explaining Turkey's alignment choices and rapprochement with Iran from the start of the

Arab Uprisings in December 2010 to July 2018. As a result, this study differs from other theorydriven studies on TFP toward Iran in that it makes use of a novel application of NCR that is more sensitive to Turkey's peculiarities toward Iran.

Moreover, the thesis is the first comprehensive study examining TFP towards Iran in the Kurdistan Region of Iraq within a theoretical framework, as well as providing different theoretical perspectives on Turkey's amity and enmity relations with Iran in Syria. The study clearly articulates why Turkey chose to align with Iran despite initial competition in these areas.

Finally, the thesis provides a comprehensive explanation of TFP towards Iran framed by the period both during and immediately after the Arab Uprisings. Despite the events in the Middle East changing very frequently, the thesis may be a good starting point for future studies seeking to expand this research further, by investigating the most recent developments in explaining certain TFP outcomes and Turkey's relations with Iran. Ultimately, this thesis provides a theoretical contribution and time-wise contribution to the literature on both TFP and, more broadly, the international relations of the Middle East.

8.3 Limitations of the Research

One of the main limitations or possible criticisms of this thesis is the period of December 2010 to July 2018. The frequent changes in Turkish politics and in the Middle East could have affected the research and the maintenance of up-to-date data. For this reason, the selected period was chosen to be able to analyse developments easily. After July 2018, Turkey's long-standing parliamentary system was completely transformed into a heavily centralised presidential one, and Turkish domestic politics entered a new phase, even though the new system was approved by voters in the referendum on April 16, 2017. Although new analysis of more recent events is required, this thesis provides an initial step in the analysis of the role of the presidential system on TFP.

Another limitation of this study concerns access to the AKP elite due to the turmoil being experienced in Turkey during the active research period, and the possible reluctance of politicians to expose their views and ideas. This may have resulted in the lack of interviews from political personalities, and specifically AKP politicians. Thus, the empirical aspect of the thesis would have been stronger if AKP interviewees had provided first-hand accounts of their beliefs and their own opinions on certain policy decisions. To mitigate this issue and increase the reliability and objectivity of the research, the thesis identified expert/specialist interviewees from different ideological standpoints and professions who support the AKP or have detailed knowledge of its members relevant across the period of the study.

The final limitation of this study relates to the communication process with the interviewees. Due to the very busy schedule of many interviewees, interview scheduling was quite difficult to manage; some interviewees had to cancel appointments, even though the interviews were online due to the Covid-19 pandemic. Greater value would have been added to the research if more interviews could have been conducted, thereby allowing new critical viewpoints to be accumulated and different ideas to be explored. Due to the social and political turmoil in Turkey, however, some of those contacted for interview seemed reluctant to speak on the sensitive issues under examination in this thesis.

8.4 Potential Further Research

Certain issues in this research require further investigation. Although the thesis focuses on the Turkish perspective, it would also be interesting to see if a similar NCR model could be applied to a study of bilateral relations from both Turkish and Iranian perspectives, to see if the same or similar intervening variables were at work. A deeper study integrating a theoretical framework on their bilateral affairs would doubtless be of benefit, not only for the understanding of Turkey and its foreign policy, but also for Iran and its foreign policy process towards Turkey.

As mentioned, the thesis was unable to investigate all the possible variables and factors that could provide a better understanding of TFP regarding the Middle East. Following the presidential system, the AKP elite's perception as an intervening variable revolved very much around Erdogan's personality and his ideology, and indeed President Erdogan has played an increasingly central role in shaping TFP. The role of the Nationalist Movement Party (MHP), a far-right political party, has also increased in Turkish domestic politics due to its alliance with the AKP since the presidential referendum. It is notable that the nationalistic perception of the MHP is more apparent in TFP in the Syrian conflict. Therefore, Erdogan's perceptions and the role of the MHP could be analysed as intervening variables under the NCR in future work. Last but not least, a deeper analysis could be undertaken regarding the role of the PKK

and/or its affiliations on TFP towards Iran, or their bilateral affairs. NCR was used to analyse the role of non-state actors in shaping foreign policy as an intervening variable. It could thus provide a theoretical framework for these topics in future work.

BIBLIOGRAPHY

- Abushouk, Ahmed Ibrahim. 2016. "The Arab Spring: A Fourth Wave of Democratization?"DigestofMiddleEastStudies25(1):52–69.https://onlinelibrary.wiley.com/doi/10.1111/dome.12080.
- Aghaie Joobani, Hossein, and Mostafa Mousavipour. 2015. "Russia, Turkey, and Iran: Moving Towards Strategic Synergy in the Middle East?" *Strategic Analysis* 39(2): 141–55. http://www.tandfonline.com/doi/abs/10.1080/09700161.2014.1000658.
- Ahval. 2019. "Turkey's Military Spending Increased by 24 Percent Report." Ahval. https://ahvalnews.com/turkish-military/turkeys-military-spending-increased-24-percentreport.
- Akbarzadeh, Shahram, and James Barry. 2017. "Iran and Turkey: Not Quite Enemies but Less than Friends." *Third World Quarterly* 38(4): 980–95. https://www.tandfonline.com/doi/full/10.1080/01436597.2016.1241139.
- Akgün, Mensur, Gökçe Perçinoğlu, and Sabiha Senyücel Gündoğar. 2012. *The Perception of Turkey in the Middle East.* https://d1wqtxts1xzle7.cloudfront.net/3249365/middle_east_report.pdf?responsecontentdisposition=inline%3B+filename%3DThe_Perception_of_Turkey_In_the_Middle_E.pdf &Expires=1606169578&Signature=aC9GTDQs51v-F4sC9ESfQyPpk7FKyvnxIHcFcVdIsD0IpACsJ3Y3XC6ow42E3O.
- Akkoyunlu, Karabekir. 2012. "Turkey's Iranian Conundrum: A Delicate Balancing Act." In *In Another Empire: A Decade of Turkey's Foreign Policy under the Justice and Development Party*, eds. Kerem Oktem, Ayse Kadioglu, and Mehmet Karli. Istanbul Bilgi University.
- Al-Ali, Zaid. 2019. The Social Justice Blind Spots in the New Arab Constitutions. https://tcf.org/content/report/social-justice-blind-spots-new-arab-

constitutions/?session=1.

- Al-Anani, Khalil. 2015. "Upended Path: The Rise and Fall of Egypt's Muslim Brotherhood."TheMiddleEastJournal69(4):527–43.http://www.ingentaconnect.com/content/10.3751/69.4.12.
- Al-Rasheed, Madawi. 2011. "Yes It Could Happen Here." *Foreign Policy*. https://foreignpolicy.com/2011/03/01/yes-it-could-happen-here-2/ (December 10, 2020).
- Alema, Francesco D. 2017. *The Evolution of Turkey* 's Syria Policy. http://www.radikal.com.tr/yorum/turkiye-merkez-ulke- (December 21, 2020).
- Aloudah, Haitham Saad. 2016. "Understanding the Sources of Turkish Foreign Policy Change towards the Middle East during the Justice and Development Party (AKP) Era: An Empirical Examination." University of Exeter. https://ethos.bl.uk/OrderDetails.do?uin=uk.bl.ethos.702294.
- Altunışık, Meliha Benli. 2009. "Turkey-EU Relations: Creating New Synergies in the Middle East." In *The European Neighborhood Policy and the Southern Mediterrenean*, eds. M Comelli, Atila Eralp, and C Ustun. METU Press, 141–56.
 - ——. 2011. Challenges to Turkey's Soft Power in the Middle East. https://www.files.ethz.ch/isn/142590/Challenges to Turkey's Soft Power in the Middle East_06.2011.pdf.
- 2014. "Turkey as an 'Emerging Donor' and the Arab Uprisings." *Mediterranean Politics* 19(3): 333–50.
 http://www.tandfonline.com/doi/abs/10.1080/13629395.2014.959761.
- Altunışık, Meliha Benli, and Lenore G. Martin. 2011. "Making Sense of Turkish Foreign Policy in the Middle East under AKP." *Turkish Studies* 12(4): 569–87. http://www.tandfonline.com/doi/abs/10.1080/14683849.2011.622513.
- Amour, Philipp O. 2017. "Israel, the Arab Spring, and the Unfolding Regional Order in the Middle East: A Strategic Assessment." *British Journal of Middle Eastern Studies* 44(3): 197

293-309. https://www.tandfonline.com/doi/full/10.1080/13530194.2016.1185696.

- Aras, Bülent. 1996. "Israil'in Yeni Stratejisinde Orta Asya ve Kafkasya'nin Yeri." Avrasya Dosyasi 2(4).
- . 1997. "The Place of the Palestinian-Israeli Peace Process in Turkish Foreign Policy." Journal of South Asian and Middle Eastern Studies 20(2): 24. https://search.proquest.com/docview/1311900869?pqorigsite=gscholar&fromopenview=true&imgSeq=1 (December 1, 2020).
- . 1999. "Israil-Iran-Turkiye Uclusunde Gelisen Iliskiler." Avrasya Dosyasi 5(1): 200–210.
- ———. 2001. "Turkish Foreign Policy towards Iran: Ideology and Foreign Policy in Flux." Journal of Third World Studies 18(1): 105–24.
- ——. 2009. "Turkey's Rise in the Greater Middle East: Peace-Building in the Periphery." Journal of Balkan and Near Eastern Studies 11(1): 29–41. http://www.tandfonline.com/doi/abs/10.1080/19448950902724380.
- ———. 2010. Arabulucu Değil, Oyun Kurucu Dış Politika. https://www.setav.org/arabulucudegil-oyun-kurucu-dis-politika/ (December 11, 2020).
- ——. 2014a. "Arap Bahari'nin Jeopolitigi." Ortadogu Analiz 6(64): 40–42.
- 2014b. "Post-Arab Spring Geopolitics, ISIS and Turkey." Ortadogu Analiz 6(65): 10–
 13.
- Aras, Bülent, and Sevgi Akarçeşme. 2012. "Turkey and the Arab Spring." International Journal: Canada's Journal of Global Policy Analysis 67(1): 39–51. http://journals.sagepub.com/doi/10.1177/002070201206700104.
- Aras, Bülent, and Aylin Gorener. 2010. "National Role Conceptions and Foreign Policy Orientation: The Ideational Bases of the Justice and Development Party's Foreign Policy Activism in the Middle East." *Journal of Balkan and Near Eastern Studies* 12(1): 73–92.

http://www.tandfonline.com/doi/abs/10.1080/19448950903507453.

- Aras, Bülent, and Rabia Karakaya Polat. 2008. "From Conflict to Cooperation: Desecuritization of Turkey's Relations with Syria and Iran." *Security Dialogue* 39(5): 495–515.
- Aras, Bülent, and Şule Toktaş. 2010. "Turkey's New Dynamics in Domestic and Foreign Policy." Journal of Balkan and Near Eastern Studies 12(1): 1–3. http://www.tandfonline.com/doi/abs/10.1080/19448950903507602.
- Aras, Bülent, and Emirhan Yorulmazlar. 2014. "Turkey and Iran after the Arab Spring: Finding
 a Middle Ground." *Middle East Policy* 21(4): 112–20. http://doi.wiley.com/10.1111/mepo.12100.
- 2016a. "State, Region and Order: Geopolitics of the Arab Spring." *Third World Quarterly* 37(12): 2259–73. https://www.tandfonline.com/doi/full/10.1080/01436597.2016.1205442.
- 2016b. "Turkish-Iranian Relations: A Long-Term Perspective." *Center for American Progress*: 1–8. https://cdn.americanprogress.org/wp-content/uploads/2016/07/07081708/TurkishEssay_Aras.pdf.
- Arik, Umut. 1992. "The New Independent States and Turkish Foreign Policy." *International Journal of Turkish Studies* 1(2).
- Arkan, Zeynep, and Müge Kınacıoğlu. 2016. "Enabling 'Ambitious Activism': Davutoğlu's Vision of a New Foreign Policy Identity for Turkey." *Turkish Studies* 17(3): 381–405. http://www.tandfonline.com/doi/full/10.1080/14683849.2016.1185943.
- Aşkar Karakir, İrem, and Nilüfer Karasulu. 2011. "Iran-Turkey Relations in the 2000s: Pragmatic Rapprochement." *Ege Akademik Bakis (Ege Academic Review)* 11(1): 111– 111. http://www.onlinedergi.com/MakaleDosyalari/51/PDF2011_1_10.pdf.
- Ataman, Muhittin, and Çağatay Özdemir. 2018. "Turkey's Syrian Policy: Shifting Priorities,
 Constant Objectives." *Türkiye Ortadoğu Çalışmaları Dergisi* 5(2).
 199

http://dergipark.gov.tr/doi/10.26513/tocd.466046.

- Ayata, Bilgin. 2015. "Turkish Foreign Policy in a Changing Arab World: Rise and Fall of a Regional Actor?" *Journal of European Integration* 37(1): 95–112.
- Aydin, Aydin. 2015. "Turkey-Iran Relations in the Light of Nuclear Power Debates and Arab Spring, Turkey's Role and The Future of Relations." *International Journal of Alanya Faculty of Business* 7(2): 123–30.
- Aydin, Mustafa. 2006. "Determinants of Turkish Foreign Policy: Changing Patterns and Conjunctures during the Cold War." *Middle Eastern Studies* 36(1): 103–39. http://www.tandfonline.com/doi/abs/10.1080/00263200008701300.
- Aydin, Mustafa, and Damla Aras. 2005. "Political Conditionality Of Economic Relations Between Paternalist States: Turkey's Interaction With Iran, Iraq, And Syria." *Arab Studies Quarterly* 27(1/2): 21–43.
- Aydin, Selcuk, and Muhammed Ali Ucar. 2018. "Divisions in Europe over Turkey's Operation in Syria." *TRT World*. https://www.trtworld.com/opinion/divisions-in-europe-overturkey-s-operation-in-syria-14822 (December 21, 2020).
- Aydin, Ufuk. 1994. "A Close Look To The Future Of Turkish-Iranian Relations." Naval Postgraduate School Monterey Ca. https://apps.dtic.mil/sti/citations/ADA288669 (November 30, 2020).
- Ayman, S. Gülden. 2012. "Regional Aspirations and Limits of Power-Turkish-Iranian Relations in the New Middle East." *Hellenic Studies* 20(1): 85–114.

2014. "Turkey and Iran: Between Friendly Competition and Fierce Rivalry." Arab Studies Quarterly 36(1): 6. http://www.jstor.org/stable/10.13169/arabstudquar.36.1.0006.

- Ayoob, Mohammed. 2011. "Beyond the Democratic Wave in the Arab World: The Middle East's Turko-Persian Future." *Insight Turkey* 13(2): 55–70.
 - ——. 2012. "The Arab Spring: Its Geostrategic Significance." *Middle East Policy* 19(3): 200

-. 2014. Will the Middle East Implode? John Wiley & Sons, Ltd.

- Aziz, T. M. 1993. "The Role of Muhammad Baqir Al-Sadr in Shici Political Activism in Iraq from 1958 to 1980." *International Journal of Middle East Studies* 25(2): 207–22.
- Azuolas, Bagdonas. 2012. "Russia's Interests in the Syrian Conflict: Power, Prestige, and Profit." *European Journal of Economic and Political Studies* 5(2): 55–77.
- Bagdonas, Ąžuolas. 2015. "Turkey as a Great Power? Back to Reality." *Turkish Studies* 16(3): 310–31. http://www.tandfonline.com/doi/full/10.1080/14683849.2015.1069712.
- Bahgat, Gawdat. 2015. "Geopolitics of Energy: Iran, Turkey, and Europe." *Mediterranean Quarterly* 26(3): 49–66. https://read.dukeupress.edu/mediterraneanquarterly/article/26/3/49-66/1930.
- Bal, Idris. 2000. *Turkey's Relations with the West and the Turkic Republics: The Rise and Fall of the 'Turkish Model'*. Ashgate Pub Ltd.

Balci, Ali. 2013. Turkiye Dis Politikasi. Nesil Basim Yayin Gida Ticaret ve Sanayi A.S.

- Bank, André, Thomas Richter, and Anna Sunik. 2014. "Durable, Yet Different: Monarchies in the Arab Spring." *Journal of Arabian Studies* 4(2): 163–79. https://www.tandfonline.com/action/journalInformation?journalCode=rjab20 (December 9, 2020).
- Barkey, Henri J. 2010. "Iran and Turkey." In *The Iran Primer: Power, Politics, and US Policy*, ed. Robin B Wright. US Institute of Peace Press, 163–67.
- 2014. "Turkey's Syria Predicament." *Survival* 56(6): 113–34.
 https://iiss.tandfonline.com/doi/abs/10.1080/00396338.2014.985440 (September 16, 2020).
- . 2016. "Erdogan's Foreign Policy Is in Ruins." Foreign Policy.

- Barkin, J. Samuel. 2003. "Realist Constructivism." *International Studies Review* 5(3): 325–42. http://isr.oxfordjournals.org/cgi/doi/10.1046/j.1079-1760.2003.00503002.x (November 23, 2020).
- Barkin, Samuel. 2009. "Realism, Prediction, and Foreign Policy." *Foreign Policy Analysis* 5(3): 233–46. http://fpa.oxfordjournals.org/cgi/doi/10.1111/j.1743-8594.2009.00091.x.
- Barnett, Michael. 2005. "Social Constructivism." In *The Globalization of World Politics: An Introduction to International Relations*, eds. John Baylis, Steve Smith, and Patricia Owens. Oxford University Press, 148–65.
- Barrans, Tiffany. 2016. Turkey-Iran Relations: Pragmatic Economics & the Ideological Ceiling to Strategic Relations.
- Başkan, Birol. 2018. "Islamism and Turkey's Foreign Policy during the Arab Spring." *Turkish Studies* 19(2): 264–88. https://www.tandfonline.com/doi/full/10.1080/14683849.2017.1405346.
- Başkan, Birol, and Özgür Pala. 2020a. "Making Sense of Turkey's Reaction to the Qatar Crisis." *The International Spectator* 55(2): 65–78. https://www.tandfonline.com/doi/full/10.1080/03932729.2020.1739846.
- ———. 2020b. "Making Sense of Turkey's Reaction to the Qatar Crisis." *The International Spectator* 55(2): 65–78.
- Baylouny, Anne Marie, and Creighton A. Mullins. 2018. "Cash Is King: Financial Sponsorship and Changing Priorities in the Syrian Civil War." *Studies in Conflict and Terrorism* 41(12): 990–1010. https://www.tandfonline.com/doi/abs/10.1080/1057610X.2017.1366621 (December 21, 2020).
- Bazzi, Mohamad. 2017. "The Growing U.S.-Iran Proxy Fight in Syria ." *The Atlantic*. https://www.theatlantic.com/international/archive/2017/06/iran-syria-trump-saudiarabia-escalation-isis/530844/ (December 10, 2020).
- BBC News. 2011. "Turkey and Iran 'Collaborating against Kurdish Rebels."" BBC News.
- . 2015. "Russia Joins War in Syria: Five Key Points." *BBC News*.
- ——. 2017a. "Syria Conflict: US Sends Arms to Kurdish Forces Fighting IS." BBC News.
- ———. 2017b. "Syria War: Turkey Will Never Accept US Alliance with Kurds Erdogan." BBC News. https://www.bbc.co.uk/news/world-middle-east-39944167 (May 17, 2021).
- Bengio, Ofra. 2004. The Turkish-Israeli Relationship. Springer.
- 2014. "Ankara, Erbil, Baghdad: Relations Fraught with Dilemmas." In *Conflict, Democratization, and the Kurds in the Middle East: Turkey, Iran, Iraq, and Syria*, eds. David Romano and Mehmet Gurses. Palgrave Macmillan, 267–82.
- ———. 2017. The Begin-Sadat Center for Strategic Studies (BESA) *The Kurds in a Volatile Middle East.*
- Bicchi, Federica. 2014. "Europe and the Arab Uprisings: The Irrelevant Power?" In *The New Middle East: Protest and Revolution in the Arab World*, ed. Fawaz A. Gerges. Cambridge University Press, 429–46.
- Bicer, Savas. 2016. "NATO and Turkey: What Is Required? Redefining NATO Partnership or Rewriting NATO's Charter." In *Turkey's Foreign Policy and Security Perspectives in the* 21st Century, ed. Sertif Demir. Brown Walker Press, 77–97.
- Bilefsky, Dan. 2007. "Sarkozy Blocks Key Part of EU Entry Talks on Turkey." *The New York Times*. https://www.nytimes.com/2007/06/25/world/europe/25iht-union.5.6325879.html (April 23, 2021).
- Bilgin, Hasret Dikici. 2008. "Foreign Policy Orientation of Turkey's Pro-Islamist Parties: A Comparative Study of the AKP and Refah." *Turkish Studies* 9(3): 407–21. http://www.tandfonline.com/doi/abs/10.1080/14683840802267363.

Bilgin, Pınar. 2007. "Making Turkey's Transformation Possible: Claiming 'Security- Speak'—

Not Desecuritization!" *Southeast European and Black Sea Studies* 7(4): 555–71. http://www.tandfonline.com/doi/abs/10.1080/14683850701726039.

———. 2009. "Securing Turkey through Western-Oriented Foreign Policy." New Perspectives on Turkey 40: 103–23.

Bleda, Tansug. 2000. Maskeli Balo. Dogan Yayincilik.

- Bloomberg. 2017. "Ulkemize Gelen Turist Sayisi Bir Onceki Yila Oranla Dususe Gecti." Bloomberg.
- Booysen, Hanlie. 2020. "Qatar's Calculated Gamble on the Syrian Muslim Brotherhood." In *The Regional Order in the Gulf Region and the Middle East*, Cham: Springer International Publishing, 195–215. http://link.springer.com/10.1007/978-3-030-45465-4_7.
- Boz, Hakan. 2013. "Israil, Turkiye'den Ozur Diledi: Iran Neden Rahatsiz Oldu?" *21. Yuzyil Turkiye Enstitusu*. https://21yyte.org/tr/merkezler/bolgesel-arastirma-merkezleri/guney-kafkasya-iran-pakistan-arastirmalari-merkezi/israil-turkiyeden-ozur-diledi-iran-neden-rahatsiz-oldu.
- Bozarslan, Hamit. 2018. "When the Present Sends Back to the Past: Reading the Kurdish Issue in the 2010s." *Middle East Critique* 27(1): 7–24. https://www.tandfonline.com/doi/abs/10.1080/19436149.2017.1411109 (November 5, 2020).
- Bozdaglioglu, Yucel. 2004. Turkish Foreign Policy and Turkish Identity: A Constructivist Approach. Routledge.
- Brawley, Mark R. 2009. "Neoclassical Realism and Strategic Calculations: Explaining Divergent British, French, and Soviet Strategies toward Germany between the World Wars (1919–1939)." In *Neoclassical Realism, the State, and Foreign Policy*, Cambridge University Press, 75–98. https://www.cambridge.org/core/books/neoclassical-realismthe-state-and-foreign-policy/neoclassical-realism-and-strategic-calculations-explainingdivergent-british-french-and-soviet-strategies-toward-germany-between-the-world-wars-

19191939/1F1BEB80E681E32 (November 21, 2020).

- Brown, Nathan J. 2011. Carnegie Endowment *Islamists: Politics Beckons Once Again*. https://carnegieendowment.org/2011/02/22/islamists-politics-beckons-once-again-pub-42697 (December 10, 2020).
- Buzan, Barry. 2003. "Regional Security Complex Theory in the Post-Cold War World." In *Theories of New Regionalism*, eds. Fredrick Söderbaum and Timothy M.Shaw. Palgrave Macmillan, 140–59.
- Buzan, Barry, Charles Jones, and Richard Little. 1993. *The Logic of Anarchy:Neorealism to Structural Realism*. New York: Columbia University Press.
- Buzan, Barry, and Ole Wæver. 2003. *Regions and Powers: The Structure of International Security*. Cambridge University Press.
- Byman, Daniel L. 2014. "Sectarianism Afflicts the New Middle East." Survival 56(1): 79–100. https://www.tandfonline.com/doi/abs/10.1080/00396338.2014.882157 (December 21, 2020).
- Byman, Daniel L., and Kenneth M. Pollack. 2011. "Regional Actors: The Changing Balance of Power in the Middle East." In *The Arab Awakening: America and the Transformation of the Middle East*, ed. Kenneth M. Pollack. Brooking Institution Press, 243–50.
- Cağaptay, Soner. 2012. "Arab Spring Heats Up Kurdish Issue The Washington Institute for Near East Policy." *The Washington Institute*: 8–10. https://www.washingtoninstitute.org/policy-analysis/view/arab-spring-heats-up-kurdishissue (November 4, 2020).

——. 2016. "Turkey's Rewarming Ties with Iran." *Washington Institute*. http://www.washingtoninstitute.org/policy-analysis/view/turkeys-rewarming-ties-with-iran.

-. 2020. Erdogan's Empire: Turkey and the Politics of the Middle East. IB Tauris.

- Çakmak, Cenap, and Murat Ustaoğlu. 2017. "Politics vs. Trade: A Realist View on Turkish– Israeli Economic Relations." *Israel Affairs* 23(2): 303–23.
- Calabrese, John. 1998. "Turkey and Iran: Limits of a Stable Relationship." *British Journal of Middle Eastern Studies* 25(1): 75–94. https://www.tandfonline.com/doi/full/10.1080/13530199808705655.
- Carpenter, Ted Galen. 2013. "Tangled Web: The Syrian Civil War and Its Implications." *Mediterranean Quarterly* 24(1): 1–11. https://read.dukeupress.edu/mediterraneanquarterly/article/24/1/1-11/1832.
- Carpes, Mariana. 2014. "From Breadcrumbs to Threads of Wool: Building a Neoclassical Realist Approach for the Study of Regional Powers Nuclear Choices." the University of Hamburg and GIGA German Institute of Global and Area Studies.
- Cavusoglu, Mevlut. 2017. "Turkish Foreign Policy in a Time of Perpetual Turmoil." *Insight Turkey* 19(1): 11–16.
- Celikkanat, Ersin. 2007. "11 Eylul Sonrasi ABD-Iran Iliskileri ve Siyasi Boyutta Turkiye'ye Etkileri." *Guvenlik Stratejileri Dergisi* 3(6): 123–52.
- Cetingulec, Mehmet. 2018. "Turkish-Iranian Trade Revived amid Growing Cooperation in Syria." *Al-Monitor*.

Cetingules, Mehmet. 2018. "Turkish-Iranian Trade Revived amid Growing Cooperation in Syria." *Al-Monitor*. https://www.almonitor.com/pulse/ru/contents/articles/originals/2018/03/turkey-iran-syria-traderevived-amid-rapport.html.

- Cetinsaya, Gokhan. 2002. "From the Tanzimat to the Islamic Revolution: Continuity and Change." *Turkish Review of Middle East Studies* 13: 113–34.
 - ——. 2003. *Turkish--Iranian Relations since the Revolution*. Ortadogu ve Balkan Incelemeleri Vakfi.

- Çevik, Senem B. 2019. "Reassessing Turkey's Soft Power: The Rules of Attraction." *Alternatives* 44(1): 50–71.
- Chan, Steve, Weixing Hu, and Kai He. 2019. "Discerning States' Revisionist and Status-Quo Orientations: Comparing China and the US." *European Journal of International Relations* 25(2): 613–40. http://journals.sagepub.com/doi/10.1177/1354066118804622 (November 22, 2020).
- Chang, I-Wei Jennifer. 2014. "Chinese Policies on the Arab Spring." In *The International Politics of the Arab Spring*, ed. Robert Mason. Springer, 177–91.
- Charountaki, Marianna. 2012. 17 Perceptions: Journal of International Affairs *Turkish Foreign Policy* and the *Kurdistan* Regional Government. https://dergipark.org.tr/en/pub/perception/624977 (November 4, 2020).

2018. Iran and Turkey: International and Regional Engagement in the Middle East.
 IB Tauris.

- Cheema, Sujata Ashwarya. 2015. "Turkey–Iran Economic Ties Post-2002." In *Perspectives on Turkey's Multi-Regional Role in the 21st Century*, ed. Mujib Alam. KW Publishers, 101–34.
- Ciplak, Bilal. 2014. "Democracy Promotion and Turkey."
- Cockburn, Patrick. 2015. The Rise of Islamic State: ISIS and the New Sunni Revolution. Verso Books.
- Coetzee, Eben, and Heidi Hudson. 2012. "Democratic Peace Theory and the Realist-Liberal Dichotomy: The Promise of Neoclassical Realism?" *Politikon* 39(2): 257–77. http://www.tandfonline.com/doi/abs/10.1080/02589346.2012.683942.
- Collard, Rebecca. 2014. "Iraq: Iran Arms Kurds in Fight Against ISIS ." *Time*. https://time.com/3196580/iran-kurds-isis-erbil-iraq/ (May 18, 2021).

Cook, Steven A, and Elizabeth Sherwood-Randall. 2006. Generating Momentum for a New 207

Era in US-Turkey Relations.

- Cooley, Alexander, Daniel Nexon, and Steven Ward. 2019. "Revising Order or Challenging the Balance of Military Power? An Alternative Typology of Revisionist and Status-Quo States." *Review of International Studies* 45(4): 689–708. https://doi.org/10.1017/S0260210519000019 (November 22, 2020).
- Cornell, Svante E. 2012. "What Drives Turkish Foreign Policy?" *Middle East Quarterly* 19(1): 13–24.
- Criss, Nur Bilge, and Serdar Guner. 1999. "Geopolitical Configurations." *Security Dialogue* 30(3): 365–76. http://journals.sagepub.com/doi/10.1177/0967010699030003015.
- Crosston, Matthew D. 2014. "Cold War and Ayatollah Residues: Syria as a Chessboard For." *Strategic Studies Quarterly* 8(4): 94–111. https://www.jstor.org/stable/26270818?seq=1#metadata_info_tab_contents (December 21, 2020).
- Çuhadar, Ç. Esra, Juliet Kaarbo, Barış Kesgin, and Binnur Özkeçeci-Taner. 2020. "Turkish
Leaders and Their Foreign Policy Decision-Making Style: A Comparative and Multi-
Method Perspective." *Turkish Studies*: 1–27.
https://www.tandfonline.com/doi/full/10.1080/14683849.2020.1724511.
- Daily Sabah. 2012. "Turkey's Army Is NATO's Second Largest." *Daily Sabah*. https://www.dailysabah.com/turkey/2012/04/16/turkeys-army-is-natos-second-largest (May 7, 2021).
- ———. 2016a. "Support for PYD Means Support for PKK, KRG President Barzani Says." Daily Sabah. https://www.dailysabah.com/war-on-terror/2016/03/23/support-for-pydmeans-support-for-pkk-krg-president-barzani-says (November 5, 2020).
 - ——. 2016b. "Turkey Will Not Be Europe's Firefighter." Daily Sabah.
- Dal, Emel Parlar. 2015. "A Normative Approach to Contemporary Turkish Foreign Policy: The Cosmopolitanism–Communitarianism Divide." *International Journal: Canada's Journal* 208

 of
 Global
 Policy
 Analysis
 70(3):
 421–33.

 http://journals.sagepub.com/doi/10.1177/0020702015584306.

 </td

- Dalacoura, Katerina. 2012. "The 2011 Uprisings in the Arab Middle East: Political Change and Geopolitical Implications." *International Affairs* 88(1): 63–79. https://academic.oup.com/ia/article-lookup/doi/10.1111/j.1468-2346.2012.01057.x.
- Dalay, Galip. 2017a. "IS There Really a Turkey-Iran Rapprochement?" *Al-Jazeera*. https://www.aljazeera.com/indepth/opinion/2017/09/turkey-iran-rapprochement-170907101356869.html.
 - ——. 2017b. "Report Evolution of Turkey Iraqi Kurdistan 's Relations." *Al Jazeera Center for Studies* (December). https://studies.aljazeera.net/en/reports/2017/12/evolution-turkey-iraqi-kurdistans-relations-171220092851950.html (November 5, 2020).
 - 2017c. "The Kurdish Fight against ISIS: Realizing the Virtual Kurdistan through Factionalized Politics in a Fragmented Homeland." In *Non-State Armed Actors in the Middle East: Geopolitics, Ideology, and Strategy*, Springer International Publishing, 77–95. https://link.springer.com/chapter/10.1007/978-3-319-55287-3_4 (December 21, 2020).
- ——. 2017d. "Why Iranian Kurds Have Been Resistant to the Regional Turmoil so Far | Middle East Eye." *Middle East Eye*. https://www.middleeasteye.net/big-story/whyiranian-kurds-have-been-resistant-regional-turmoil-so-far (May 28, 2020).
- Dalay, Galip, and Dov Friedman. 2013. "The AK Party and the Evolution of Turkish Political Islam's Foreign Policy." *Insight Turkey* 15(2): 123–39.
- Dallmayr, Fred. 2011. "Radical Changes in the Muslim World: Turkey, Iran, Egypt."Globalizations8(5):639–46.http://www.tandfonline.com/doi/abs/10.1080/14747731.2011.622872.
- Danforth, Nicholas. 2008. "Ideology and Pragmatism in Turkish Foreign Policy:From Ataturk to the AKP." *Turkish Policy Quarterly* 7(3): 83–95.

- Dannreuther, Roland. 2015. "Russia and the Arab Spring: Supporting the Counter-Revolution."JournalofEuropeanIntegration37(1):77–94.http://www.tandfonline.com/doi/abs/10.1080/07036337.2014.975990.
- Darwich, May. 2017. "Creating the Enemy, Constructing the Threat: The Diffusion of Repression against the Muslim Brotherhood in the Middle East." *Democratization* 24(7): 1289–1306. https://www.tandfonline.com/doi/full/10.1080/13510347.2017.1307824.
- Darwich, May, and Juliet Kaarbo. 2020. "IR in the Middle East: Foreign Policy Analysis in Theoretical Approaches." *International Relations* 34(2): 225–45.
- Davidson, Jason W. 2016. The Origins of Revisionist and Status-Quo States *The Origins of Revisionist and Status-Quo States*. Palgrave Macmillan.
- Davutoglu, Ahmet. 2013. "Zero Problems in a New Era." Foreign Policy.
- Davutoğlu, Ahmet. 2013. "The Three Major Earthquakes in the International System andTurkey."TheInternationalSpectator48(2):1–11.https://www.tandfonline.com/doi/full/10.1080/03932729.2013.796781.

Davutoğlu, Ahmet. 2001. Strategic Depth: Turkey's International Standing. Kure Yayinlari.

- ——. 2009. "Turkish Foreign Policy and the EU in 2010." *Turkish Policy Quarterly* 8(3): 11–17.
- Demir, Idris. 2017. "Turkey's Foreign Policy towards the Middle East under the Shadow of the Arab Spring." In *Turkey's Foreign Policy Towards the Middle East: Under the Shadow of the Arab Spring*, ed. Idris Demir. Cambridge Scholars Publishing, 1–6.
- Demirtas-Bagdonas, Özlem. 2014. "Reading Turkey's Foreign Policy on Syria: The AKP's Construction of a Great Power Identity and the Politics of Grandeur." *Turkish Studies* 15(1): 139–55. https://www.tandfonline.com/doi/abs/10.1080/14683849.2014.890412 (December 21, 2020).

Deringil, Selim. 1989. Turkish Foreign Policy During the Second World War: An "Active" 210

Neutrality. Cambridge University Press.

- Dias, Vanda Amaro. 2014. "A Critical Analysis of the EU's Response to the Arab Spring and Its Implications for EU Security." *Human Security Perspectives* 10(1): 26–61.
- Đidić, Ajdin, and Hasan Kösebalaban. 2019. "Turkey's Rapprochement with Russia: Assertive Bandwagoning." *The International Spectator* 54(3): 123–38.
- Dikkaya, Mehmet. 1999. "Orta Asya'da Yeni Buyuyk Oyun Turkiye, Rusyan ve Iran." *Avrasya Dosyasi* 5(3): 189–208.
- Dinc, Cengiz, and Mustafa Yetim. 2012. "Transformation of Turkish Foreign Policy toward the Middle East: From Non-Involvement to a Leading Role." *Alternatives: Turkish Journal of International Relations* 11(1): 67–84.
- Dinçer, Osman Bahadır, and Mehmet Hecan. 2020. "Democratisation in Ambiguous Environments: Positive Prospects for Democracy in the MENA Region after the Arab Spring." *Third World Quarterly* 41(12): 2087–2108. https://www.tandfonline.com/doi/full/10.1080/01436597.2020.1811084.
- Dismorr, Ann. 2008. Turkey Decoded. Saqi Books.
- Doyle, Michael W. 1986. "Liberalism and World Politics." *American Political Science Review* 80(04): 1151–69. http://www.journals.cambridge.org/abstract_S0003055400000228 (November 23, 2020).
- Dueck, Colin. 2009. "Neoclassical Realism and the National Interest: Presidents, Domestic Politics, and Major Military Interventions." In *Neoclassical Realism, the State, and Foreign Policy*, Cambridge University Press, 139–69. https://www.cambridge.org/core/books/neoclassical-realism-the-state-and-foreignpolicy/neoclassical-realism-and-the-national-interest-presidents-domestic-politics-andmajor-military-interventions/65990B9DABAE629A11E6192708493D68 (November 22, 2020).
- Dunne, Tim. 2008. "Liberalism." In *The Globalization of World Politics: An Introduction to* 211

International Relations, eds. John Baylis, Steve Smith, and Patricia Owens. Oxford University Press, 110–22.

- Duran, Burhanettin, and Nuh Yilmaz. 2013. "Islam, Models and the Middle East: The New Balance of Power Following the Arab Spring." Perceptions XVIII(4): 139-70.
- Dursun-Özkanca, Oya. 2019. "Turkey–West Relations: The Politics of Intra-Alliance Opposition." Turkey-West Relations: The Politics of Intra-alliance Opposition: 1–214. https://www.cambridge.org/core/books/turkeywestrelations/B7BA02FDC15C2178EB211D5BB51648AA (July 19, 2021).
- Ebrahimi, Mansoureh, Kamaruzaman Yusoff, and Mir Mohamadali Seyed Jalili. 2017. "Economic, Political, and Strategic Issues in Iran-Turkey Relations, 2002-2015." *Contemporary* Review ofthe Middle East 4(1): 67-83. http://journals.sagepub.com/doi/10.1177/2347798916681336.
- Eder, Thomas Stephan. 2014. China-Russia Relations in Central Asia: Energy Policy, Beijing's New Assertiveness and 21st Century Geopolitics. Springer.
- Edwards, Alex. 2013. "A Neoclassical Realist Analysis of American 'dual Containment' Policy in the Persian Gulf, 1991-2001." http://etheses.lse.ac.uk/704/ (November 21, 2020).
- Efegil, Ertan, and Leonard A. Stone. 2003. "Iran and Turkey in Central Asia: Opportunities for Rapproachment in the Post-Cold War Era." Journal of Third World Studies 20(1): 55–77.
- Ehteshami, Anoushiravan. 2014. "Middle East Middle Powers: Regional Role, International Impact." Uluslararası İlişkiler Dergisi 11(42): 29–49.
- Ehteshami, Anoushiravan, and Süleyman Elik. 2011. "Turkey's Growing Relations with Iran and Arab Middle East." Turkish **Studies** 12(4): 643-62. http://www.tandfonline.com/doi/abs/10.1080/14683849.2011.624322.
- Elbadawi, Ibrahim, and Samir Maksidi. 2017. Democratic Transitions in the Arab World. eds. Ibrahim Elbadawi and Samir Makdisi. Cambridge: Cambridge University Press.

http://ebooks.cambridge.org/ref/id/CBO9781316687000.

- Elhusseini, Fadi. 2015. "The Arab Spring and the Rise of Non-State Actors ." *Middle East Monitor*. https://www.middleeastmonitor.com/20150613-the-arab-spring-and-the-riseof-non-state-actors/ (December 10, 2020).
- Elik, Süleyman. 2012. Iran-Turkey Relations 1971-2011. New York: Routledge.
 - ——. 2018. "The Arab Spring and Turkish-Iranian Relations, 2011–2016." In *Turkey's Relations with the Middle East*, Cham: Springer International Publishing, 105–16. http://link.springer.com/10.1007/978-3-319-59897-0_7.
- Elman, Colin;, and Miriam Fendius Elman. 2003. *Progress in International Relations Theory : Appraising the Field*. MIT Press.
- Ennis, Crystal A., and Bessma Momani. 2013. "Shaping the Middle East in the Midst of the Arab Uprisings: Turkish and Saudi Foreign Policy Strategies." *Third World Quarterly* 34(6): 1127–44.
- Eralp, Atila. 1996. "Facing the Challenge: Post-Revolutionary Relations with Iran." In *Reluctant Neighbor*, United States Inst of Peace Press, 93–112.
- Erdogan, Birsen. 2016. Humanitarian Intervention and the Responsibility to Protect Humanitarian Intervention and Responsibility to Protect. Springer International Publishing.
- Ergun, Doruk. 2018. "External Actors and VNSAs: An Analysis of the United States, Russia, ISIS, and PYD/YPG." In Violent Non-State Actors and the Syrian Civil War, Cham: Springer International Publishing, 149–72. http://link.springer.com/10.1007/978-3-319-67528-2_8.
- Erkus, Sevil. 2019. "Sanctions on Iran Are Wrong': Turkey Warns US." Hurriyet Daily News.
- Ertosun, Erkan. 2012. "Turkish-Iranian Relations: The Legacy of the 1990s." Journal Of Business Economics and Political Science 1(2).

- Esfandiary, Dina, and Ariane M. Tabatabai. 2017. "A Comparative Study of U.S. and Iranian Counter-ISIS Strategies." *Studies in Conflict & Terrorism* 40(6): 455–69. https://www.tandfonline.com/doi/full/10.1080/1057610X.2016.1221265.
- Flanagan, Stephen J. 2013. "The Turkey–Russia–Iran Nexus: Eurasian Power Dynamics." The
Washington Quarterly 36(1): 163–78.
http://www.tandfonline.com/doi/abs/10.1080/0163660X.2013.751656.
- Foulon, Michiel. 2015. "Neoclassical Realism: Challengers and Bridging Identities." International Studies Review 17(4): 635–61. https://academic.oup.com/isr/articlelookup/doi/10.1111/misr.12255 (November 21, 2020).
- Frantzman, Seth J. 2017. "The New Middle East Crisis: Iran's Growing Relationships." *National Interests*.
- Fuller, Graham E. 2007. The New Turkish Republic: Turkey as a Pivotal State in the Muslim World. United States Institute of Peace Press.
- George, Alexander L., and Bennett Andrew. 2005. *Case Studies and Theory Development in the Social Sciences*. MIT Press.
- Gerges, Fawaz A. 2013. "The Islamist Moment: From Islamic State to Civil Islam?" *Political Science Quarterly* 128(3): 389–426. https://www.jstor.org/stable/23563255?seq=1#metadata_info_tab_contents (December 10, 2020).
- Gerring, John. 2008. "Case Selection for Case-Study Analysis: Qualitative and Quantitative Techniques." In *The Oxford Handbook of Political Methodology*, eds. Janet M. Box-Steffensmeier, Henry E. Brady, and David Collier. Oxford University Press, 645–84.
- Gökçe, Onur. 2016. "The Dynamics of Turkish-Israeli Relations." All Azimuth: A Journal of Foreign Policy and Peace 1(2): 55–55.
- Göksel, Oğuzhan. 2019. "Foreign Policy Making in the Age of Populism: The Uses of Anti-Westernism in Turkish Politics." *New Middle Eastern Studies* 9(1). 214

https://www108.lamp.le.ac.uk/ojs1/index.php/nmes/article/view/3108.

- Goldberg, Jeffrey. 2016. "The Obama Doctribe." *The Atlantic*. https://www.theatlantic.com/magazine/archive/2016/04/the-obama-doctrine/471525/ (March 31, 2021).
- Goldenberg, Ilan, and Julie Smith. 2017. "U.S.-Russia Competition in the Middle East Is Back ." Foreign Policy. https://foreignpolicy.com/2017/03/07/u-s-russia-competition-in-themiddle-east-is-back/ (December 9, 2020).
- Gregory Gause, F. 2014. BROOKINGS DOHA CENTER ANALYSIS PAPER Beyond Sectarianism: New Middle East Cold War. https://www.brookings.edu/research/beyondsectarianism-the-new-middle-east-cold-war/ (April 12, 2021).
- Grieco, Joseph M. 1990. *Cooperation Among Nations: Europe, America, and Non-Tariff Barriers* to Trade. Cornessl University Press. https://books.google.co.uk/books?hl=en&lr=&id=bcji1fUj_o0C&oi=fnd&pg=PR9&dq= Cooperation+among+nations:+Europe,+America,+and+nontariff+barriers+to+trade&ots=vPldsjWyUe&sig=xqQoJJvYULipA0eQa7wvzpW1V-M&redir_esc=y#v=onepage&q=Cooperation among nations%3A Eu (November 22, 2020).
- Gruen, George E. 1995. "Dynamic Progress in Turkish-Israeli Relations." *Israel Affairs* 1(4): 40–70.
- Grunstein, Judah. 2008. "Turkey's Foreign Policy Framework." *World Politics Review*. https://www.worldpoliticsreview.com/trend-lines/2342/turkeys-foreign-policy-framework (April 27, 2021).
- Gumus, Burak. 2013. "Adalet ve Kalkinma Partisin'nin Ortadogu Siyaseti." *Elektronik Siyaset Bilimi Arastirmalari Dergisi* 4(7): 75–99.
- Gundogan, Unal. 2003. "Islamist Iran and Turkey, 1979-1989: State Pragmatism and Ideological Influences." *Middle East Review of International Affairs* 7(1): 1–12.

- Güner, Serdar, and Dilan E Koç. 2018. "Leverages and Constraints for Turkish Foreign Policy in Syrian War: A Structural Balance Approach." *Uluslararasi Iliskiler* 15(59): 89–103. http://www.uidergisi.com.tr (March 19, 2021).
- Gunes, Cengiz. 2019a. The Kurds in a New Middle East *The Kurdish Resurgence in a Changing Middle East*. Springer International Publishing.
- ——. 2019b. The Kurds in a New Middle East: The Changing Geopolitics of a Regional Conflict *The Kurds in a New Middle East: The Changing Geopolitics of a Regional Conflict*. Palgrave Macmillan.
- Gunes, Cengiz, and Robert Lowe. 2015. "The Impact of the Syrian War on Kurdish Politics Across the Middle East." *Chatnam House* (July): 2–18.
- Güney Ateşoğlu, Nurşin. 2013. "A New Challenge for Turkey: Civil War in Syria." *Insight Turkey* 15(4). https://www.insightturkey.com/commentaries/a-new-challenge-for-turkey-civil-war-in-syria (September 16, 2020).
- Gunter, Michael M. 2013. "The Kurdish Spring." *Third World Quarterly* 34(3): 441–57. https://www.tandfonline.com/action/journalInformation?journalCode=ctwq20 (November 5, 2020).
 - ——. 2015. "Iraq, Syria, Isis and the Kurds: Geostrategic Concerns for the U.S. and Turkey." Middle East Policy 22(1): 102–11. http://doi.wiley.com/10.1111/mepo.12116.
- ——. 2018. "Kurdish Disunity in Historical Perspective." *Seton Hall Journal of Diplomacy and International Relations* 19.
- Gupta, Ranjit. 2016. "Understanding the War in Syria and the Roles of External Players: Way out of the Quagmire?" *Round Table* 105(1): 29–41. https://www.tandfonline.com/doi/abs/10.1080/00358533.2016.1128630 (December 21, 2020).

Gurkas, Berna. 2011. Turkiye-Iran Iliskileri.

- Gürzel, Aylin. 2014. "Turkey's Role as a Regional and Global Player and Its Power Capacity: Turkey's Engagement with Other Emerging States." *Revista de Sociologia e Política* 22(50): 95–105. http://www.scielo.br/scielo.php?script=sci_arttext&pid=S0104-44782014000200007&lng=en&tlng=en.
- Haas, Mark L. 2014. "Ideological Polarity and Balancing in Great Power Politics." SecurityStudies23(4):http://www.tandfonline.com/doi/abs/10.1080/09636412.2014.964991.
- Hachigian, Nina, and David Shorr. 2013. "The Responsibility Doctrine." *The Washington Quarterly* 36(1): 73–91. http://www.tandfonline.com/doi/abs/10.1080/0163660X.2013.751652.
- Hadfield-Amkhan, Amelia. 2010. British Foreign Policy, National Identity, and Neoclassical Realism. Rowman & Littlefield Publishers.
- Hale, William. 2012. Turkish Foreign Policy since 1774. 3rd ed. Taylor & Francis.
- 2015. "Reassessing Turkey's Relationships with Its Neighbours." In *Reconciling Cultural and Political Identities in a Globalized World*, ed. Michális S. Michael. London: Palgrave Macmillan UK, 185–209. http://link.springer.com/10.1007/978-1-137-49315-6_9.
- Han, Ahmet Kasim. 2016. "Paradise Lost: A Neoclassical Realist Analysis of Turkish Foreign Policy and the Case of Turkish-Syrian Relations." In *Turkey-Syria Relations: Between Enmity and Amity*, eds. Özlem Tür and Raymond Hinnebusch. Routledge, 55–69.
- Handy, Nathaniel. 2018. "Turkey's Relations with the Middle East." : 181–95. http://link.springer.com/10.1007/978-3-319-59897-0.
- Haşimi, Cemalettin. 2014. "Turkey's Humanitarian Diplomacy and Development Cooperation." *Insight Turkey* 16(1): 127–45.
- Haugom, Lars. 2019. "Turkish Foreign Policy under Erdogan: A Change in International Orientation?" *Comparative Strategy* 38(3): 206–23.

- Held, David, and Kristian Coates Ulrichsen. 2014. "The Arab Spring and the Changing Balance of Global Power ." *OpenDemocracy*. https://www.opendemocracy.net/en/north-africawest-asia/arab-spring-and-changing-balance-of-global-power/ (December 9, 2020).
- Hentov, Elliot. 2012. Asymmetry of Interest: Turkish-Iranian Relations since 1979. LAP Lambert Academic Publishing.
- ——. 2013. "The Ostensible 'Silent Victor'?: The Long-Term Impact of the Iran-Iraq War on Turkey." In *The Iran-Iraq War: New International Perspectives*, eds. Nigel Ashton and Bryan Gibson. Routledge, 125–45. https://www.taylorfrancis.com/ (November 30, 2020).
- Hinnebusch, Raymond. 2002. "Introduction: The Analytical Framework." In *The Foreign Policies of Middle East States*, eds. Raymond Hinnebusch and Anoushiravan Ehteshami. Lynne Rienner Publishers, 1–27.
 - ——. 2015. "Back to Enmity: Turkey-Syria Relations since the Syrian Uprising." *Journal* of German Orient Institute 56(1): 14–22.
- Hinnebusch, Raymond, and Anoushiravan Ehteshami. 1997. Syria and Iran: Middle Powers in a Penetrated Regional System. Routledge.
- Hokayem, Emile. 2014. "Iran, the Gulf States and the Syrian Civil War." *Adelphi Series* 54(447–448): 39–70. https://www.tandfonline.com/doi/full/10.1080/19445571.2014.995937.
- Hubbard, Ben, Anton Troianovski, Carlotta Gall, and Patrick Kingsley. 2019. "In Syria, Russia Is Pleased to Fill an American Void - The New York Times." *The New York Times*. https://www.nytimes.com/2019/10/15/world/middleeast/kurds-syria-turkey.html (December 10, 2020).
- Hudson, John. 2013. "U.S. Now Blamed for Undermining and Supporting Egypt's Coup." *Foreign Policy*. https://foreignpolicy.com/2013/07/10/u-s-now-blamed-for-underminingand-supporting-egypts-coup/ (December 10, 2020).

- Hudson, Valerie M. 2005. "Foreign Policy Analysis: Actor-Specific Theory and the Ground of International Relations." *Foreign Policy Analysis* 1(1): 1–30.
- Hurriyet Daily News. 2013a. "PYD Leader Arrives in Turkey for Two-Day Talks: Report." Hurriyet Daily News.
- ———. 2013b. "Turkish, Iranian Foreign Ministers Call for Cease-Fire in Syria during Ramadan ." *Hurriyet Daily News*. https://www.hurriyetdailynews.com/turkish-iranianforeign-ministers-call-for-cease-fire-in-syria-during-ramadan-50620 (May 17, 2021).
- ——. 2015. "Turkish Airstrikes Target ISIL in Syria." *Hurriyet Daily News*. https://www.hurriyetdailynews.com/turkish-airstrikes-target-isil-in-syria-85853.
- ——. 2018. "Istanbul Home to 538,000 Syrians, Highest Number in Turkey." *Hurriyet Daily News*. https://www.hurriyetdailynews.com/istanbul-home-to-538-000-syrians-highest-number-in-turkey-125948.
- İçduygu, Ahmet. 2015. "Syrian Refugees in Turkey: The Long Road Ahead | Migrationpolicy.Org." (September): 23. www.migrationpolicy.org/transatlantic. (December 21, 2020).
- Ihsan, Dagi. 2008. "Editor's Note." Insight Turkey 10(2): 3-4.
- Imset, Ismet G. 1992. *The PKK, A Report on Separatist Violence in Turkey*. Turkish Daily News Publications.
- Inat, Kemal. 2015. Ekonomik İşbirliği Örgütü'nün 30. Yılında Türkiye-İran Ekonomik İlişkileri . https://www.setav.org/ekonomik-isbirligi-orgutunun-30-yilinda-turkiye-iran-ekonomikiliskileri/ (December 1, 2020).
- International Crisis Group. 2016. "Turkey and Iran: Bitter Friends, Bosom Rivals." https://www.crisisgroup.org/middle-east-north-africa/gulf-and-arabianpeninsula/iran/b051-turkey-and-iran-bitter-friends-bosom-rivals.
 - —. 2018. Winning the Post-ISIS Battle for Iraq in Sinjar. 219

https://www.crisisgroup.org/middle-east-north-africa/gulf-and-arabianpeninsula/iraq/183-winning-post-isis-battle-iraq-sinjar (November 9, 2020).

- Ipek, Pinar. 2015. "Ideas and Change in Foreign Policy Instruments: Soft Power and the Case of the Turkish International Cooperation and Development Agency." *Foreign Policy Analysis* 11(2): 173–93. http://fpa.oxfordjournals.org/cgi/doi/10.1111/fpa.12031.
- Işiksal, Hüseyin. 2017a. "Conclusion: Turkey and the Middle East in an Age of Turbulence." In *Turkey's Relations with the Middle East: Political Encounters after the Arab Spring*, Springer International Publishing, 211–20. https://link.springer.com/chapter/10.1007/978-3-319-59897-0_14 (December 11, 2020).
 - ——. 2017b. "Political Chaos in Iraq, ISIS, and Turkish Foreign Policy: The High Cost of the Westphalian Delusion." In *Turkey's Relations with the Middle East: Political Encounters after the Arab Spring*, Springer International Publishing, 83–102. https://link.springer.com/chapter/10.1007/978-3-319-59897-0_6 (December 10, 2020).
- ——. 2018. "Turkish Foreign Policy, the Arab Spring, and the Syrian Crisis: One Step Forward, Two Steps Back." In *Turkey's Relations with the Middle East: Political Encounters after the Arab Spring*, Springer International Publishing, 13–30. https://link.springer.com/chapter/10.1007/978-3-319-59897-0_2 (September 16, 2020).
- Isiksal, Huseyin, Aliya Zhakanova Isiksal, and Yossi Apeji. 2020. "The Impact of Syrian Refugees on the Turkish Labor Market." *International Journal of Operations Management* 1(1): 27–34. https://researchleap.com/impact-syrian-refugees-turkish-labormarket/.
- Jenkins, Gareth H. 2012. Occasional Allies, Enduring Rivals: Turkey's Relations with Iran. www.silkroadstudies.org (November 30, 2020).
- Jervis, Robert. 1978. "Cooperation under the Security Dilemma." *Conflict After the Cold War: Arguments on Causes of War and Peace*: 425–41.

——. 2017. Perception and Misperception in International Politics. Princeton University

- Jones, Clive, and Yoel Guzansky. 2017. "Israel's Relations with the Gulf States: Toward the Emergence of a Tacit Security Regime?" *Contemporary Security Policy* 38(3): 398–419. https://www.tandfonline.com/doi/full/10.1080/13523260.2017.1292375.
- Jongerden, Joost. 2019. "Governing Kurdistan: Self-Administration in the Kurdistan Regional Government in Iraq and the Democratic Federation of Northern Syria." *Ethnopolitics* 18(1): 61–75. https://www.tandfonline.com/action/journalInformation?journalCode=reno20 (November 5, 2020).
- Juneau, Thomas. 2010. "Neoclassical Realist Strategic Analysis: A Statement." *Graduate* Student Conference of the European Consortium on Political Research 30(August).

2014. "US Power in the Middle East: Not Declining." *Middle East Policy* 21(2): 40–52.

- ——. 2015. Squandered Opportunity: Neoclassical Realism and Iran's Foreign Policy. Stanford University Press.
- 2018. "Iran's Costly Intervention in Syria: A Pyrrhic Victory." *Mediterranean Politics* 25(1): 26–44. https://www.tandfonline.com/doi/abs/10.1080/13629395.2018.1479362 (December 21, 2020).
- Kaarbo, Juliet. 2015. "A Foreign Policy Analysis Perspective on the Domestic Politics Turn in IR Theory." *International Studies Review* 17(2): 189–216. https://academic.oup.com/isr/article-lookup/doi/10.1111/misr.12213.
- Kamal, Arif. 2009. Dynamics of Peace and Stability in the Middle East Arena: Identifying the Contemporary Challenges and Options for Response.
- Kanat, Kilic Bugra. 2010. "Ak Party's Foreign Policy: Is Turkey Turning Away from the West?" *Insight Turkey* 12(1): 205–25.

- Kanat, Kiliç Buğra. 2014. "Theorizing the Transformation of Turkish Foreign Policy." *Insight Turkey* 16(1): 65–84.
- Kane, Sean. 2011. The Coming Turkish-Iranian Competition in Iraq.
- Kang, William, and Jaechun Kim. 2016. "Turco-Iranian Alignment: Balancing or Bandwagoning with the US?" *Journal of International and Area Studies* 23(1): 17–32.
- Kara, Mehtap, and Ahmet Sozen. 2016. "Change and Continuity in Turkish Foreign Policy: Evaluating Pre-AKP and AKP Periods' National Role Conceptions." Uluslararasi Iliskiler 13(52): 47–66.
- Karacasulu, Nilüfer. 2015. "Interpreting Turkey's Middle East Policy in the Last Decade*." *All Azimuth* 4(1): 27–38.
- Karacasulu, Nilüfer, and İrem Aşkar Karakir. 2014. "EU-Turkey Relations in the Context of the Middle East after the Arab Spring." *Insight Turkey* 16(4): 201–19.
- Karahan, Hatice. 2016. "The Post-Sanctions Iranian Economy and Turkey." The New Turkey.
- Karakoc, Abdullah. 2009. "Turkey's Relations with Iran and the United States: A Shift in Alignment?" Naval Postgraduate School.
- Karakoc, Julide. 2016. "Challenges to US Middle East Policy in the Post-Arab Uprisings Period." In New Actors and Issues in the Post-Arab Uprisings Period, eds. Julide Karakoc and Duygu Ersoy. Cambridge Scholars Publishing, 1–19.
- Kardaş, Şaban. 2010. "Turkey: Redrawing the Middle East Map Or Building Sandcastles?" *Middle East Policy* 17(1): 115–36. https://onlinelibrary.wiley.com/doi/10.1111/j.1475-4967.2010.00430.x.
- . 2021. "Revisionism and Resecuritization of Turkey's Middle East Policy: A Neoclassical Realist Explanation." *Journal of Balkan and Near Eastern Studies* 00(00): 1–12. https://doi.org/10.1080/19448953.2021.1888250.

- Kavak, Şeref. 2017. "The Arab Spring and the Emergence of a New Kurdish Polity in Syria." In *Turkey's Relations with the Middle East: Political Encounters after the Arab Spring*, Springer International Publishing, 197–210. https://link.springer.com/chapter/10.1007/978-3-319-59897-0_13 (September 16, 2020).
- Kaya, Karen. 2012. Turkey-Iran Relations after the Arab Spring. file:///Users/cangul/Downloads/2012-09-12 Turkey-Iran Relations After the Arab Spring (Kaya).pdf.
- Kayhan- Pusane, Özlem. 2016. "Turkey's Changing Relations with the Iraqi Kurdish Regional Government." *Middle East Review of International Affairs (Online)* 20(2): 20. https://www.researchgate.net/publication/334762451 (November 5, 2020).
- Kayhan-Pusane, Özlem. 2020. "The Role of Context in Desecuritization: Turkish Foreign Policy towards Northern Iraq (2008–2017)." *Turkish Studies* 21(3): 392–413. https://www.tandfonline.com/doi/abs/10.1080/14683849.2019.1675047 (November 5, 2020).
- Keohane, Robert. 1986. Neorealism and Its Critics. Columbia University Press. http://cup.columbia.edu/book/neorealism-and-its-critics/9780231063494 (November 21, 2020).
- Kerr, Siemon. 2017. "Oil Price More than Qatar Hurts Gulf Capital Inflows to Turkey." *Financial Times*.
- Kesgin, Barış. 2020. "Turkey's Erdoğan: Leadership Style and Foreign Policy Audiences." *Turkish Studies* 21(1): 56–82.
- Keskin, Arif. 2009. "Iran-Turkiye Iliskilerine Genel Bir Bakis." 21. Yuzyil Dergisi 12: 45-54.
- Keyman, E. Fuat. 2016. "Turkish Foreign Policy in the Post-Arab Spring Era: From Proactive to Buffer State." *Third World Quarterly* 37(12): 2274–87. https://www.tandfonline.com/doi/full/10.1080/01436597.2016.1199260.
- Keyman, E. Fuat, and Onur Sazak. 2015. "Turkey and Iran: The Two Modes of Engagement 223

in the Middle East." *Journal of Balkan and Near Eastern Studies* 17(3): 321–36. http://www.tandfonline.com/doi/full/10.1080/19448953.2015.1063803.

- Khatib, Lina. 2013. "Qatar's Foreign Policy: The Limits of Pragmatism." *International Affairs* 89(2): 417–31. https://academic.oup.com/ia/article-lookup/doi/10.1111/1468-2346.12025.
- Khoury. 2016a. "The Long View of the 2011 Arab Uprisings." *Bustan: The Middle East Book Review* 7(1): 1.
- Khoury, Nabeel A. 2013. "The Arab Cold War Revisited: The Regional Impact of the Arab Uprising." *Middle East Policy* 20(2): 73–87.
- ——. 2016b. "The Long View of the 2011 Arab Uprisings." *Bustan: The Middle East Book Review* 7(1): 1.
- Kiani, Davood. 2015. "Turkey's Middle Eastern Policy: A European Attitude." In Perspectives on Turkey's Multi-Regional Role in the 21st Century, ed. Mujib Alam. KW Publishers, 175–98.
- Kibaroglu, Mustafa. 2009. *Twists and Turns in Turkish-Iranian Relations*. https://mei.edu/publications/twists-and-turns-turkish-iranian-relations (December 1, 2020).
- Kibaroglu, Mustafa, and Yasemin Nun. 2008. "Iran." In *Turkey's Neighborhood*, ed. Mustafa Kibaroglu. Foreign Policy Institute, 143–67.
- Kirchner, Magdalena, and Şafak Baş. 2017. "Chaldiran 2.0: Conceptualizing Iranian-Turkish Relations after the Arab Spring." In *Contemporary Turkey at a Glance II: Turkey Transformed? Power, History, Culture*, Springer Fachmedien Wiesbaden, 163–76. https://link.springer.com/chapter/10.1007/978-3-658-16021-0_11 (November 24, 2020).
- Kirişci, Kemal. 2014. "Syrian Refugees and Turkey's Challenges: Going beyond Hospitality."
 Brookings. https://www.brookings.edu/research/syrian-refugees-and-turkeys-challengesgoing-beyond-hospitality/ (December 21, 2020).

- Kirişçi, Kemal. 2009. "The Transformation of Turkish Foreign Policy: The Rise of the Trading State." New Perspectives on Turkey 40: 29–56. https://www.cambridge.org/core/product/identifier/S0896634600005203/type/journal_ar ticle.
- Kitchen, Nicholas. 2010a. "Systemic Pressures and Domestic Ideas: A Neoclassical Realist Model of Grand Strategy Formation." *Review of International Studies* 36(01): 117--143.
 - 2010b. "Systemic Pressures and Domestic Ideas: A Neoclassical Realist Model of Grand Strategy Formation." *Review of International Studies* 36(1): 117–43.
 /core/journals/review-of-international-studies/article/systemic-pressures-and-domesticideas-a-neoclassical-realist-model-of-grand-strategyformation/E2DB12BFD2B50F898DCC4DA966177ED4 (September 16, 2020).
 - ——. 2012. LSE IDEAS After the Arab Spring: Power Shift in the Middle East?: The Contradictions of Hegemony: The United States and the Arab Spring.
- Koolaee, Elaheh, and Hossein Hafezian. 2017. 14 International Studies Journal (ISJ) International Studies Journal *The Kurdish Question and Iranian-Turkish Relations: 1991-*2015.
- Kosebalaban, Hasan. 2011. "Turkey and the New Middle East: Between Liberalism and Realism." *Perceptions* 16(3): 93–114. https://dergipark.org.tr/en/pub/perception/625030 (November 23, 2020).
- Kösebalaban, Hasan. 2011. Turkish Foreign Policy: Islam, Nationalism, and Globalization. Palgrave Macmillan.
- ———. 2020. "Transformation of Turkish Foreign Policy Toward Syria: The Return of Securitization." *Middle East Critique* 29(3): 335–44.
- Kramer, Heinz, and Ali Cimen. 2001. Avrupa ve Amerika Karsisinda Degisen Turkiye. Timas Yayinlari.
- Kumral, Mehmet Akif. 2020. Exploring Emotions in Turkey-Iran Relations *Exploring* 225

Emotions in Turkey-Iran Relations. Springer International Publishing.

- Küntay, Burak. 2014. "Relations Between Turkey and Iran After the Arab Spring: The Extent of a Pragmatic Rapprochement." *Abant İzzet Baysal Üniversitesi Sosyal Bilimler Enstitüsü Dergisi*.
- Kunu, Serkan, and Sertac Hopoglu. 2016. "Turkey and Iran: An Analysis Based on Mutual Trade and Defense Spending." *Journal of Kafkas University, Faculty of Economics and Administrative Sciences* 7(12): 109–26.
- Kuru, Ahmet T. 2015. "Turkey's Failed Policy toward the Arab Spring: Three Levels of Analysis." *Mediterranean Quarterly* 26(3): 94–116.
- Kutlay, Mustafa. 2011. "Economy as the 'Practical Hand' of 'New Turkish Foreign Policy': A Political Economy Explanation." *Insight Turkey* 13(1): 67–88.
- Laoutides, Costas. 2016. "How Foreign Is the Kurdish Issue in Iran's Foreign Policy?" In *Iran in the World*, Palgrave Macmillan US, 93–109. https://link.springer.com/chapter/10.1007/978-1-137-58577-6_6 (December 21, 2020).
- Larrabee, F Stephen, and Ian O Lesser. 2003. *Turkish Foreign Policy in an Age of Uncertainty*. RAND Corporation.
- Legrenzi, Matteo. 2015. The GCC and the International Relations of the Gulf: Diplomacy, Security and Economic Coordination in a Changing Middle East. IB Tauris.
- Legro, Jeffrey W., and Andrew Moravcsik. 1999. "Is Anybody Still a Realist?" *International Security* 24(2): 5–55. https://www.mitpressjournals.org/doix/abs/10.1162/016228899560130 (November 23, 2020).
- Lesch, David, and Mark Haas. 2012. *The Arab Spring: Change and Resistance in the Middle East*. Westview Press.
- Lesch, David, and Mark L. Haas. 2016. The Arab Spring: The Hope and Reality of the 226

Uprisings. Westview Press.

Lesser, Ian O. 2011. Turkey's Third Wave--and the Coming Quest for Strategic Reassurance.

Liel, Alon. 2001. Turkey in the Middle East: Oil, Islam, and Politics. Lynne Rienner.

- Lindemann, Björn Alexander. 2014. Cross-Strait Relations and International Organizations: Taiwan's Participation in IGOs in the Context of Its Relationship with China. Springer Science & Business Media.
- Lindenstrauss, Gallia, and Furkan Aksoy. 2012. "Turkey and Northern Iraq: Tightening Relations in a Volatile Environment." *Strategic Assessment* 15(3): 49–58.
- Lobell, Steven E. 2009. "Threat Assessment, the State, and Foreign Policy: A Neoclassical Realist Model." In *Neoclassical Realism, the State, and Foreign Policy*, eds. Steven E. Lobell, Norrin M. Ripsman, and Jeffrey W. Taliaferro. Cambridge: Cambridge University Press, 42–74. https://www.cambridge.org/core/product/identifier/CBO9780511811869A012/type/book _part.
- Lobell, Steven E., Norrin M. Ripsman, and Taliferro W. Jeffrey. 2009. *Neoclassical Realism, the State, and Foreign Policy Neoclassical.*
- Lobell, Steven E., Norrin M. Ripsman, and Jeffrey W. Taliaferro. 2009. *Neoclassical Realism, the State, and Foreign Policy*. Cambridge University Press.

——. 2016. Neoclassical Realist Theory of International Politics. Oxford University Press.

- Lynch, Marc. 2013. "The Middle East Power Vacuum." Foreign Policy. https://foreignpolicy.com/2013/10/25/the-middle-east-power-vacuum/ (December 9, 2020).
- ———. 2016a. In Uncharted Waters: Islamist Parties Beyond Egypt's Muslim Brotherhood. https://carnegieendowment.org/2016/12/16/in-uncharted-waters-islamist-parties-beyondegypt-s-muslim-brotherhood-pub-66483 (December 10, 2020).

——. 2016b. *The New Arab Wars: Uprisings and Anarchy in the Middle East*. PublicAffairs.

- MacGillivray, Iain William. 2019. "Complexity and Cooperation in Times of Conflict: Turkish-Iranian Relations and the Nuclear Issue." *British Journal of Middle Eastern Studies*: 1–25. https://www.tandfonline.com/doi/full/10.1080/13530194.2019.1596783.
- Majidyar, Ahmad. 2017. "Iran and Turkey Discuss Ways to Cooperate in Syria and Iraq." *Middle East Institute*.
- Malmvig, Helle. 2014. "Power, Identity and Securitization in Middle East: Regional Order after the Arab Uprisings." *Mediterranean Politics* 19(1): 145–48. http://www.tandfonline.com/doi/abs/10.1080/13629395.2013.856181.
- Mango, Andrew. 1994. Turkey: The Challenge of a New Role. Praeger.
- Martin, Lenore G. 2019. "Analysing a Tumultuous Relationship: Turkey and the US in the Middle East." Asian Journal of Middle Eastern and Islamic Studies 13(2): 262–77. https://www.tandfonline.com/doi/abs/10.1080/25765949.2019.1605571 (September 16, 2020).
- McCurdy, Daphne. 2008. "Turkish-Iranian Relations: When Opposites Attract." *Turkish Policy Quarterly* 7(2): 87–106.
- McLean, Wayne. 2015. "Understanding Divergence between Public Discourse and Turkish Foreign Policy Practice: A Neoclassical Realist Analysis." *Turkish Studies* 16(4): 449– 64.

Mearsheimer, John. 2001. The Tragedy of Great Power Politics. WW Norton & Company.

- Meibauer, Gustav et al. 2021. "Forum: Rethinking Neoclassical Realism at Theory's End." *International Studies Review* 23(1): 268–95.
- Menashri, David. 1998. "Iran and Central Asia-Radical Regime, Pragmatic Politics." In *Central Asia Meets the Middle East*, ed. David Menashri., 73–97.

- Mercan, Murat. 2010. "Turkish Foreign Policy and Iran." *Turkish policy quarterly* 8(4): 13–19.
- Metawe, Mohamed. 2013. "How and Why the West Reacted to the Arab Spring: An Arab Perspective." *Insight Turkey* 15(3): 141–55. https://www.insightturkey.com/articles/how-and-why-the-west-reacted-to-the-arab-spring-an-arab-perspective.
- Middle East Monitor. 2018. "Saudi-UAE-Kurdish Military Meeting in Northern Syria ." *Middle East Monitor*. https://www.middleeastmonitor.com/20180531-saudi-uae-kurdishmilitary-meeting-in-northern-syria/ (December 11, 2020).
- Milliyet. 2008. "'AK Parti Kapatılmasın' Kararı Çıktı." *Milliyet*. https://www.milliyet.com.tr/siyaset/ak-parti-kapatilmasin-karari-cikti-972729 (April 22, 2021).
 - 2013. "Dünya 'beş'ten Büyüktür ." Milliyet.
 https://www.milliyet.com.tr/siyaset/dunya-besten-buyuktur-1767594 (May 17, 2021).
- ———. 2015. "İran'la Dış Ticaret Hedefi 30 Milyar \$." *Milliyet*. http://www.milliyet.com.tr/iran-la-dis-ticaret-hedefi-30/ekonomi/detay/2040073/default.htm.
- Mohammed, Idrees. 2011. "Turkey and Iran Rivalry on Syria." *Alternatives: Turkish Journal of International Relations* 10(2–3): 87–99.
- Monier, Elizabeth. 2014. "The Arabness of Middle East Regionalism: The Arab Spring and Competition for Discursive Hegemony between Egypt, Iran and Turkey." *Contemporary Politics* 20(4): 421–34. http://www.tandfonline.com/doi/abs/10.1080/13569775.2014.968474.
- Moravcsik, Andrew. 1997. "Taking Preferences Seriously: A Liberal Theory of International Politics." *International Organization* 51(4): 513–53. https://www.cambridge.org/core/product/identifier/S0020818397440171/type/journal_ar ticle.

Morgenthau, Hans J. 1947. Scientific Man vs. Power Politics. University of Chicago Press.

- Müftüler-Bac, Meltem. 2005. "Turkey and the United States: The Impact of the War in Iraq." *International Journal* 61(1): 61–81.
- Müftüler-Baç, Meltem, and Yaprak Gürsoy. 2010. "Is There a Europeanization of Turkish Foreign Policy? An Addendum to the Literature on EU Candidates." *Turkish Studies* 11(3): 405–27. http://www.tandfonline.com/doi/abs/10.1080/14683849.2010.506734.
- Müftüler, Meltem, and Müberra Yüksel. 1997. "Turkey: A Middle Power in the New Order."
 In *Niche Diplomacy*, Palgrave Macmillan UK, 184–96. https://link.springer.com/chapter/10.1007/978-1-349-25902-1_10 (April 15, 2021).

Murinson, Alexander. 2012. Turkish Foreign Policy in the Twenty-First Century.

- Musiol, Marek. 2019. "The Secularization of Kurdish Self-Determination as a Challenge for the Sectarian Balance of Power in the Middle East." *Polish Political Science Yearbook* 48.
 https://heinonline.org/HOL/Page?handle=hein.journals/ppsy48&id=107&div=12&collec tion=journals (November 4, 2020).
- Mustafa, Sara-Salahaddin, and Sardar Aziz. 2016. "Turkey and the Iraqi Kurdistan Federal Region: Bonds of Friendship." In *Iraqi Kurdistan in Middle Eastern Politics*, ed. Alex Danilovich. Routledge, 149–71.
- Narizny, Kevin. 2017. 51 On Systemic Paradigms and Domestic Politics A Critique of the Newest Realism.
- Narli, Nilüfer. 1993. "Cooperation or Competition in the Islamic World: Turkish-Iranian Relations from the Islamic Revolution to the Gulf War and Beyond." *CEMOTI* 15(1): 265–93.
- Nasr, Vali. 2018. "Iran Among the Ruins." *Foreign Affairs* (March/April): 108–18. https://www.foreignaffairs.com/articles/middle-east/2018-02-13/iran-among-ruins (November 5, 2020).

Nechepurenko, Ivan. 2017. "Russia, Turkey and Iran Propose Conference on Postwar Syria's 230

Future." *New York Times*. https://www.nytimes.com/2017/11/22/world/europe/russia-turkey-iran-syria-war-peace-talks.html (December 21, 2020).

- Neriah, Jacques. 2012. *The Future of Kurdistan: Between Turkey, the Iraq War, and the Syrian Revolt.* Jerusalem Center for Public Affairs.
- Noi, Aylin Unver. 2012. "The Arab Spring, Its Effects On The Kurds, And The Approaches Of Turkey, Iran, Syria, And Iraq On The Kurdish Issue." *Middle East Review of International Affairs* 16(2): 15–29.
- ———. 2015. "Turkey and Iran: From Competition to Cooperation?" Huffington Post. https://www.huffingtonpost.com/aylin-unver-noi/turkey-and-iran-fromcomp_b_7127854.html.
- Norton, Augustus Richard. 2015. "The Geopolitics of the Sunni-Shia Rift." In *Regional Insecurity After the Arab Uprisings*, London: Palgrave Macmillan UK, 129–48. http://link.springer.com/10.1057/9781137503978_7.
- Oğuzlu, H. Tarik. 2007. "Soft Power in Turkish Foreign Policy." Australian Journal ofInternationalAffairs61(1):81–97.http://www.tandfonline.com/doi/abs/10.1080/10357710601142518.
- 2020. "Turkish Foreign Policy in a Changing World Order." *All Azimuth* 9(1): 127–39. https://dergipark.org.tr/tr/download/article-file/920715 (November 24, 2020).
- Oguzlu, Tarik. 2010. "Turkey and Europeanization of Foreign Policy?" *Political Science Quarterly* 125(4): 657–83. https://onlinelibrary.wiley.com/doi/10.1002/j.1538-165X.2010.tb00689.x.
- Oğuzlu, Tarık. 2011. "Arab Bahari ve Yansimalari." Ortadogu Analiz 3(36): 8–16.
- ——. 2012. The "Arab Spring" and the Rise of the 2.0 Version of Turkey's "Zero Problems with Neighbors" Policy . www.sam.gov.tr (December 11, 2020).

Oktav, Özden Zeynep. 2013. "The Syrian Uprising and the Iran-Turkey-Syria Quasi Alliance:

A View from Turkey." In *Turkey-Syria Relations: Between Enmity and Amity*, eds. Özlem Tür and Raymond A. Hinnebusch. Ashgate Publishing Limited, 193–203.

- ——. 2015. "The Syrian Uprising and Turkey-Syria-Iran Relations." In *Perspectives on Turkey's Multi-Regional Role in the 21st Century*, ed. Mujib Alam. KW Publishers, 81–100.
- ———. 2018. "Understanding Iran's Approach to Violent Non-State Actors: The ISIS and YPG Cases." In Violent Non-State Actors and the Syrian Civil War, Cham: Springer International Publishing, 193–210. http://link.springer.com/10.1007/978-3-319-67528-2_10.
- Okyay, Asli S. 2017. "Turkey's Post-2011 Approach to Its Syrian Border and Its Implications for Domestic Politics." *International Affairs* 93(4): 829–46. https://academic.oup.com/ia/article/93/4/829/3897494 (September 16, 2020).
- Olson, Robert W. 1998. The Kurdish Question and Turkish-Iranian Relations: From World War I to 1998. Mazda Pub.
- 2000. "Turkey-Iran Relations, 1997 to 2000: The Kurdish and Islamist Questions."
 Third World Quarterly 21(5): 871–90. http://www.tandfonline.com/doi/full/10.1080/713701072.
- ———. 2001. Turkey's Relations with Iran, Syria, Israel, and Russia, 1991-2000: The Kurdish and Islamist Questions. Mazda Pub.

——. 2004. Turkish-Iran Relations 1979-2004 Revolution, Ideology, War, Coups and Geopolitics. Mazda Pub.

- Onis, Ziya. 2011. "Multiple Faces of the 'New' Turkish Foreign Policy: Underlying Dynamics and a Critique." *Insight Turkey* 13(1): 47–65.
- Önis, Ziya. 2012. "Turkey and the Arab Spring: Between Ethics and Self-Interest." *Insight Turkey* 14(3): 45–63.

- Öniş, Ziya. 2012. "Turkey and the Arab Spring: Between Ethics and Self-Interest." SSRN Electronic Journal. http://www.ssrn.com/abstract=2017639.
- 2014. "Turkey and the Arab Revolutions: Boundaries of Regional Power Influence in a Turbulent Middle East." *Mediterranean Politics* 19(2): 203–19.
- Öniş, Ziya, and Mustafa Kutlay. 2017. "The Dynamics of Emerging Middle-Power Influence in Regional and Global Governance: The Paradoxical Case of Turkey." *Australian Journal of International Affairs* 71(2): 164–83. https://www.tandfonline.com/doi/full/10.1080/10357718.2016.1183586.
- ÖNİŞ, Ziya, and Mustafa Kutlay. 2013. "Rising Powers in a Changing Global Order: The Political Economy of Turkey in the Age of BRICS." *Third World Quarterly* 34(8): 1409–26. http://www.tandfonline.com/doi/abs/10.1080/01436597.2013.831541.
- Öniş, Ziya, and Şuhnaz Yilmaz. 2009. "Between Europeanization and Euro-Asianism: Foreign Policy Activism in Turkey during the AKP Era." *Turkish Studies* 10(1): 7–24. https://www.tandfonline.com/doi/abs/10.1080/14683840802648562 (September 16, 2020).

Oran, Baskin. 2001. Turk Dis Politikasi Cilt II, 1980-2001. Iletisim Yayinlari.

- Ospina, Mariano V., D. Gray, and Buies Creek. 2014. "Syria, Iran, and Hizballah: A Strategic Alliance." *Global Security Studies* 5(1): 27–36.
- Ovalı, Ali Şevket, and İlkim Özdikmenli. 2019. "Ideologies and the Western Question in Turkish Foreign Policy: A Neo-Classical Realist Perspective." All Azimuth: A Journal of Foreign Policy and Peace. http://dergipark.gov.tr/doi/10.20991/allazimuth.514465.
- Ozcan, Ali Nihat. 2000. "Iran'in Turkiye Algilamasi ve Politik Arac Olarak Teror." *Stratejik Analiz* 1: 49–53.
- Özcan, Mesut. 2016. *Harmonizing Foreign Policy*. Routledge. https://www.taylorfrancis.com/books/9781351931618.

- Özcan, Nihat Ali, and Özgür Özdamar. 2010. "Uneasy Neighbors: Turkish-Iranian Relations Since the 1979 Islamic Revolution." *Middle East Policy* 17(3): 101–17. http://doi.wiley.com/10.1111/j.1475-4967.2010.00454.x.
- Ozdamar, Ozgur; Halistoprak, B.Toygar; Sula, İ. Erkam. 2014. "From Good 'Neighbor to Model': Turkey's Changing Roles in the Middle East in the Aftermath of Die Arab Spring." *Uluslararasi Iliskiler* 11(42): 93–113.
- Ozdamar, Ozgur. 2015. Turkish-Iranian Relations after the Framework Agreement.
- 2016. "Domestic Sources Of Changing Turkish Foreign Policy Toward The Mena During The 2010s A Role Theoretic Approach." In *Domestic Role Contestation, Foreign Policy, and International Relations,* Routledge, 89–104. https://www.taylorfrancis.com/chapters/domestic-sources-changing-turkish-foreignpolicy-toward-mena-2010s-role-theoretic-approach-özgürözdamar/e/10.4324/9781315623580-11 (March 19, 2021).
- Özdamar, Özgür. 2016. "Domestic Sources of Changing Turkish Foreign Policy toward the MENA during the 2010s: A Role Theoretic Approach." In *Domestic Role Contestation, Foreign Policy, and International Relations,* Routledge, 89–104. https://www.taylorfrancis.com/ (November 23, 2020).
- Özdamar, Özgür, and Balkan Devlen. 2019. "Man vs. the System: Turkish Foreign Policy After the Arab Uprisings." In *Fear and Uncertainty in Europe*, Cham: Springer International Publishing, 177–95. http://link.springer.com/10.1007/978-3-319-91965-2_9.
- Ozdamar, Ozgur, and Zeynep Taydas. 2008. "Foreign Policy, Public Opinion and the Iraq War: The Turkish Case." In *International Studies Association Annual Convention*,.
- Özer, Yonca, and Fatmanur Kaçar. 2018. "The EU's Stance Toward VNSAs During the Syrian Crisis: YPG and ISIS Cases." In *Violent Non-State Actors and the Syrian Civil War*, Cham: Springer International Publishing, 173–92. http://link.springer.com/10.1007/978-3-319-67528-2_9.

- Ozgen, Nurettin, and Muslum Reyhanogullari. 2017. "Suriye (Ic) Savasinin Turkiye Turizmine Etkileri: Samandag (Hatay) Ornegi." *Yüzüncü Yıl Üniversitesi Sosyal Bilimler Enstitüsü Dergisi* 1(3): 1–19.
- Ozkan, Mehmet, and Hasan Korkut. 2013. "Turkish Foreign Policy towards the Arab Revolutions." *Epiphany* 6(1). http://epiphany.ius.edu.ba/index.php/epiphany/article/view/59.
- Özpek, Burak Bilgehan. 2018. "Paradigm Shift between Turkey and the Kurds: From 'Clash of the Titans' to 'Game of Thrones.'" *Middle East Critique* 27(1): 43–60. https://www.tandfonline.com/doi/abs/10.1080/19436149.2017.1415513 (November 5, 2020).
- Özpek, Burak Bilgehan, and Yelda Demirağ. 2014. "Turkish Foreign Policy after the 'Arab Spring': From Agenda-Setter State to Agenda-Entrepreneur State." *Israel Affairs* 20(3): 328–46.
- Ozturk, Osman Metin. 1999. "Turkiye-Israil Askeri Isbirligi Uzerine." Avrasya Dosyasi.
- Paasche, Till F., and Michael M. Gunter. 2016. "Revisiting Western Strategies against the Islamic State in Iraq and Syria." *Middle East Journal* 70(1): 9–29.
- Pahlavan, Tschanguiz H. 1996. "Turkish-Iranian Relations: An Iranian View." In *Reluctant Neighbor*, ed. Edward Ricardo Braithwaite. United States Inst of Peace Press, 71–92.
- Pamuk, Humeyra. 2015. "Turkey's Erdogan Says Can't Tolerate Iran Bid to Dominate Middle East." *Reuters*.
- Parasiliti, Andrew, Kathleen Reedy, and Becca Wasser. 2017. RAND Corporation PreventingStateCollapseinSyria.RANDCorporation.https://www.rand.org/pubs/perspectives/PE219.html (December 21, 2020).
- Park, Bill. 2012. "Turkey, the US and the KRG: Moving Parts and the Geopolitical Realities." *Insight Turkey* 14(3): 109–25.

2014. "Turkey-Kurdish Regional Government Relations after the U.S. Withdrawal from Iraq: Putting the Kurds on the Map?" In *Turkey: Issues and Relations with the U.S. and the Kurds of Iraq*, , 85–133. https://www.jstor.org/stable/resrep11855?seq=1#metadata_info_tab_contents (November 5, 2020).

- 2019. "Explaining Turkey's Reaction to the September 2017 Independence Referendum in the KRG: Final Divorce or Relationship Reset?" *Ethnopolitics* 18(1): 46–60. https://www.tandfonline.com/doi/abs/10.1080/17449057.2018.1525165 (November 5, 2020).
- Parlar Dal, Emel. 2016. "Impact of the Transnationalization of the Syrian Civil War on Turkey: Conflict Spillover Cases of ISIS and PYD-YPG/PKK." *Cambridge Review of International Affairs* 29(4): 1396–1420. https://www.tandfonline.com/doi/abs/10.1080/09557571.2016.1256948 (December 21, 2020).
- Parsi, Trita. 2007. *Treacherous Alliance: The Secret Dealings of Israel, Iran, and the United States*. Yale University Press.
- Patton, Marcie J. 2006. "The Economic Policies of Turkey's AKP Government: Rabbits from a Hat?" *The Middle East Journal* 60(3): 513–36. https://www.ingentaconnect.com/content/10.3751/60.3.15.
- Perra, Antonio. 2016. "From the Arab Spring to the Damascus Winter: The United States, Russia, and the New Cold War." *Contemporary Review of the Middle East* 3(4): 363–86. http://journals.sagepub.com/doi/10.1177/2347798916664578.
- Phillips, Christopher. 2016. *Battle for Syria: International Rivalry in the New Middle East.* Yale University Press.
- 2017a. "Eyes Bigger than Stomachs: Turkey, Saudi Arabia and Qatar in Syria." *Middle East Policy* 24(1): 36–47. https://onlinelibrary.wiley.com/doi/full/10.1111/mepo.12250 (December 21, 2020).

——. 2017b. "Eyes Bigger than Stomachs: Turkey, Saudi Arabia and Qatar in Syria." Middle East Policy 24(1): 36–47.

- Phillips, David L. 2014. "ISIS Crisis." American Foreign Policy Interests 36(6): 351–60. https://www.tandfonline.com/doi/abs/10.1080/10803920.2014.993254 (November 5, 2020).
- Pieper, Moritz. 2013. "Turkish Foreign Policy toward the Iranian Nuclear Programme: In Search of a New Middle East Order after the Arab Spring and the Syrian Civil War." *Alternatives - Turkish Journal of International Relations* 12(3): 81–92.
- Pierini, Marc. 2016. Carnegie Europe Papers In Search of an EU Role in the Syrian War. https://carnegieeurope.eu/2016/08/18/in-search-of-eu-role-in-syrian-war-pub-64352 (December 21, 2020).
- Polat, Rabia Karakaya. 2008. "The Kurdish Issue: Can the AK Party Escape Securitization?" *Insight Turkey* 10(3): 75–86. https://www.jstor.org/stable/26330792?seq=1#metadata_info_tab_contents (November 24, 2020).
- Pollack, Jonathan D. 2011. "China: Unease from Afar." In *The Arab Awakening: America and the Transformation of the Middle East*, ed. Kenneth M. Pollack. Brooking Institution Press, 298–304.
- Rabasa, Angel, and F. Stephen Larrabee. 2008. *The Rise of Political Islam in Turkey*. RAND Corporation.
- Rathbun, Brian. 2008. "A Rose by Any Other Name: Neoclassical Realism as the Logical and Necessary Extension of Structural Realism." *Security Studies* 17(2): 294–321. https://www.tandfonline.com/doi/abs/10.1080/09636410802098917 (November 21, 2020).
- Republic of Turkey Ministry of Foreign Affairs. 2017. "Article by H.E. Mr. Mevlüt Çavuşoğlu Titled 'Turkey-EU Relations: Investing in Our Common Future', Published in Bled

Strategic Times, 4 September 2017." Republic of Turkey Ministry of Foreign Affairs.

- Reuters. 2017. "Erdogan Says Turkey and Iran Discussing Joint Action against Kurdish Militants." *Reuters*.
- Rezaei, Farhad. 2019. "Iran and Turkey: Frenemies for Ever?" In Iran's Foreign Policy After the Nuclear Agreement, Cham: Springer International Publishing, 189–213. http://link.springer.com/10.1007/978-3-319-76789-5_8.
- Ripsman, Norrin M., Jeffrey W. Taliaferro, and Steven E. Lobell. 2016. *Neoclassical Realist Theory of International Politics*. Oxford University Press. https://books.google.co.uk/books?hl=en&lr=&id=SODeDAAAQBAJ&oi=fnd&pg=PP1 &dq=Neoclassical+realist+theory+of+international+politics&ots=FND5yf5_d1&sig=w p3bi-XKPD8qE40YH3GGgNbIMxk&redir_esc=y#v=onepage&q=Neoclassical realist theory of international politics& (November 21, 2020).
- Riva, Nadia. 2017. "Withdrawal of Turkish Troops from Iraq Soon to Come: Iraqi PM." *Kurdistan24.net*. https://www.kurdistan24.net/en/news/8d1c0a10-3277-42d4-9130e74ea039972f (November 5, 2020).
- Roberts, David B. 2014. "Qatar, the Ikhwan, and Transnational Relations in the Gulf, Vision of Gulf Security." In *Middle East Political Science (POMEPS)*, , 54–60.
- Robins, Philip. 2003. Turkish Foreign Policy since the Cold War: Suits and Uniforms. Hurst & Co. Publishers Ltd.
- 2007. "Turkish Foreign Policy since 2002: Between a'post-Islamist'government and a Kemalist State." *International Affairs* 83(2): 289–304. https://academic.oup.com/ia/article-lookup/doi/10.1111/j.1468-2346.2007.00619.x.
 - 2013. "Turkey's 'Double Gravity' Predicament: The Foreign Policy of a Newly Activist Power." *International Affairs* 89(2): 381–97. https://academic.oup.com/ia/article-lookup/doi/10.1111/1468-2346.12023.
 - . 2014. "The Foreign Policy of Turkey." In *The Foreign Policies of Middle East States*, 238
ed. Anoush Hinnebuch, Raymond and Ehteshami. Lynne Reiner, 315–36.

- Romano, David. 2015. "The "Arab Spring's" Effect on Kurdish Political Fortunes." *Insight Turkey* 17(3): 53–63.
- Rose, Gideon. 1998. "Neoclassical Realism and Theories of Foreign Policy on JSTOR." *World Politics* 51(1): 144–72. https://www.jstor.org/stable/25054068 (September 16, 2020).
- Roussos, Sotiris. 2017. "Greece and Regional Dynamics in the Eastern Mediterranean." In *Foreign Policy Under Austerity*, eds. Aristotle Tziampiris and N. Litsas Spyridon. Springer, 95–115.
- Roy, Olivier. 2012. "The Transformation of the Arab World." *Journal of Democracy* 23(3): 5– 18. http://muse.jhu.edu/content/crossref/journal_of_democracy/v023/23.3.roy.html.
- Ryan, Curtis R. 2009. Inter-Arab Alliances: Regime Security and Jordanian Foreign Policy. University Press of Florida.
- Sabah. 2009. "Yeni Osmanlılar Sözü Iyi Niyetli Değil'." Sabah. https://www.sabah.com.tr/siyaset/2009/12/04/yeni_osmanlilar_sozu_iyi_niyetli_degil (April 23, 2021).
- Sabio, Oso. 2015. Rojava: An Alternative to Imperialism, Nationalism, and Islamism in the Middle East. Lulu.com.
- Sadik, Giray. 2012. "Magic Blend or Dangerous Mix? Exploring the Role of Religion in Transforming Turkish Foreign Policy from a Theoretical Perspective." *Turkish Studies* 13(3): 293–317. http://www.tandfonline.com/doi/abs/10.1080/14683849.2012.717447 (March 19, 2021).
- Şahin Mencütek, Zeynep, N. Ela Gökalp Aras, and Bezen Balamir Coşkun. 2020. "Türkiye'nin Suriye Kaynaklı Kitlesel Göçe Yanıtı: Neoklasik Realist Bir Analiz." Uluslararası İlişkiler Dergisi. https://dergipark.org.tr/en/doi/10.33458/uidergisi.856928.

- Salem, Paul. 2011. *Turkey's Image in the Arab World*. https://www.files.ethz.ch/isn/129502/Paul_Salem_FINAL.pdf (December 11, 2020).
- Salloukh, Bassel F. 2013. "The Arab Uprisings and the Geopolitics of the Middle East." TheInternationalSpectator48(2):32–46.https://www.tandfonline.com/doi/full/10.1080/03932729.2013.787830.
- Salman, Mohammad, Moritz Pieper, and Gustaaf Geeraerts. 2015. "Hedging in the Middle East and China-U.S. Competition 1." Asian Politics & Policy 7(4): 575–96. http://doi.wiley.com/10.1111/aspp.12225.
- Sandal, Nukhet A. 2014. "Middle Powerhood as a Legitimation Strategy in the Developing World: The Cases of Brazil and Turkey." *International Politics* 51(6): 693–708. http://link.springer.com/10.1057/ip.2014.33.
- Santini, Ruth Hanau. 2011. "Europe: Muddling Through." In *The New Middle East: Protest* and Revolution in the Arab World, ed. Kenneth M. Pollack. Brooking Institution Press, 285–91.
- Sariaslan, Fatma. 2013. "2000'li Yıllarda Türkiye-İran Ekonomik İlişkileri." Akademik Ortadogu 7(2): 65–91.
- Sayari, Sabri. 1997. "Turkey and the Middle East in the 1990s." *Journal of Palestine Studies* 26(3): 44–55. https://online.ucpress.edu/jps/article/26/3/44/51893/Turkey-and-the-Middle-East-in-the-1990s.
- ———. 2000. "Turkish Foreign Policy in the Post-Cold War Era: The Challenges of Multi-Regionalism." *Journal of International Affairs* 54(1): 169–82.

Sayigh, Yezid, and Avi Shlaim. 1997. The Cold War and the Middle East. Clarendon Press.

Schmidt, Brian. 2008. "The Primacy of National Security." In Foreign Policy: Theories, Actors, Cases, eds. Steve Smith, Amelia Hadfield-Amkhan, and Tim Dunne. Oxford University Press, 188–201.

- Schweller, Randall L. 1994. "Bandwagoning for Profit: Bringing the Revisionist State Back In." *International Security* 19(1): 72.
- ——. 1998. *Deadly Imbalances: Tripolarity and Hitler's Strategy of World Conquest.* Columbia University Press.
- ———. 2003. "The Progressiveness of Neoclassical Realism." *Progress in international relations theory: appraising the field*: 311–47.
 - ——. 2004. "Unanswered Threats: A Neoclassical Realist Theory of Underbalancing on JSTOR." *International Security* 29(2): 159–201.
- ———. 2006. Unanswered Threats: Political Constraints on the Balance of Power. Princeton University Press.
- ——. 2015. "Rising Powers and Revisionism in Emerging International Orders." NIC Global Trends Meeting: 1–14. https://eng.globalaffairs.ru/articles/rising-powers-andrevisionism-in-emerging-international-orders/ (November 22, 2020).
- 2018. "Opposite but Compatible Nationalisms: A Neoclassical Realist Approach to the Future of US–China Relations." *The Chinese Journal of International Politics* 11(1): 23–48. https://academic.oup.com/cjip/article/11/1/23/4844056.
- Şen, Gülriz. 2019. "Dynamics of Estrangement and Realignment in Turkey–Iran Relations in the 2000s." In *Turkey's Pivot to Eurasia*, Routledge, 147–65. https://www.taylorfrancis.com/ (November 24, 2020).
- Shabbir, Muhammad. 2013. "Emerging Middle East: Interplay of the New Power Centers." ISSRA: 25–44.
- Shaheen, Kareem. 2015. "Isis 'controls 50% of Syria' after Seizing Historic City of Palmyra." *The Guardian*. https://www.theguardian.com/world/2015/may/21/isis-palmyra-syriaislamic-state (April 14, 2021).

Sharma, Suraj. 2016. "Saudi Arabia Backs Turkish Action against Kurds in Northern Syria ."

Middle East Eye. https://www.middleeasteye.net/news/saudi-arabia-backs-turkish-action-against-kurds-northern-syria (December 11, 2020).

- Shlaim, Avi. 2014. "Israel, Palestine, and the Arab Uprisings." In *The New Middle East: Protest and Revolution in the Arab World*, ed. Fawaz A. Gerges. Cambridge University Press, 380–401.
- Sinkaya, Bayram. 2004. "Conflict and Cooperation in Turkey-Iran Relations: 1989-2001." Middle East Technical |University.
- ——. 2005. "Turkey-Iran Relations in the 1990s and the Role of Ideology." *Perception Journal of International Affairs* 10(1): 1–16.
- ——. 2012. "Rationalization of Turkey-Iran Relations: Prospects and Limits." *Insight Turkey* 14(2): 137–56.
- 2018. "The Kurdish Question in Iran and Its Effects on Iran-Turkey Relations." *British Journal of Middle Eastern Studies* 45(5): 840–59. http://doi.org/10.1080/13530194.2017.1361315.

——. 2019. Turkey-Iran Relations after the JDP.

- Soguk, Nevzat. 2015. "With/Out a State, Kurds Rising: The Un/Stated Foreign Policy and the Rise of the Kurdish Regional Government in Iraq." *Globalizations* 12(6): 957–68. https://www.tandfonline.com/doi/abs/10.1080/14747731.2015.1100857 (November 4, 2020).
- Sönmez, Göktuğ. 2020. A Neoclassical Realist Approach to Turkey under JDP Rule. Cambridge Scholars Publishing.
- Sonmez, Mehmet. 2019. "Syrian Refugees in Turkey, Burden or Cheap Labor?" *Al-Monitor*. www.al-monitor.com/pulse/originals/2019/02/turkey-syria-syrian-refugees- burden-or-cheap-labor.html.

Sorhun, Engin. 2012. Is Arap Spring Turkey's Winter?

- Soz, Jiwan. 2016. "Isolation of the Kurds in Syria." *Atlantic Council*: Hamed, Safin, 'Barzani Asks PKK to Quit Iraqi Kurd. https://www.atlanticcouncil.org/blogs/syriasource/isolation-of-the-kurds-in-syria/ (November 5, 2020).
- Stansfield, Gareth. 2013. "The Unravelling of the Post-First World War State System? The Kurdistan Region of Iraq and the Transformation of the Middle East." *International Affairs* 89(2): 259–82. https://academic.oup.com/ia/article-lookup/doi/10.1111/1468-2346.12017 (November 4, 2020).
- Stein, Aaron. 2014. "III. The End of 'Zero Problems', 2010–13." *Whitehall Papers* 83(1): 34–58. http://www.tandfonline.com/doi/abs/10.1080/02681307.2014.989692.
- 2015a. "Turkey's Evolving Syria Strategy ." Foreign Affairs.
 https://www.foreignaffairs.com/articles/turkey/2015-02-09/turkeys-evolving-syria-strategy (December 21, 2020).
- ———. 2015b. *Turkey's New Foreign Policy: Davutoglu, the AKP and the Pursuit of Regional Order*. Routledge.
- Stein, Aaron, and Philipp C. Bleek. 2012. "Turkish-Iranian Relations: From 'Friends with Benefits' to 'It's Complicated."" *Insight Turkey* 14(4): 137–50. https://www.insightturkey.com/articles/turkish-iranian-relations-from-friends-withbenefits-to-its-complicated (November 24, 2020).
- Steinberg, Guido. 2017. AQAP, ISIS, and Domestic Security in Saudi Arabia. https://www.jstor.org/stable/pdf/resrep04754.7.pdf (April 14, 2021).
- Stephen Larrabee, F. 2013. "Turkey's New Kurdish Opening." Survival 55(5): 133–46. https://www.tandfonline.com/doi/abs/10.1080/00396338.2013.841816 (November 5, 2020).
- Strategic Comments. 2015. "Iraqi Kurdistan: The Essential Briefing." *Strategic Comments* 21(2): iii–vi. https://www.tandfonline.com/doi/abs/10.1080/13567888.2015.1035506

(November 5, 2020).

 2017. "The Repercussions of the Iraqi Kurdistan Independence Referendum."
 Strategic Comments 23(10): iii–v. https://www.tandfonline.com/doi/abs/10.1080/13567888.2017.1416782 (November 5, 2020).

- Sula, İsmail Erkam. 2018. "An Eclectic Methodological Approach in Analyzing Foreign Policy: Turkey's Foreign Policy Roles and Events Dataset (TFPRED)." All Azimuth: A Journal of Foreign Policy and Peace. http://dergipark.gov.tr/doi/10.20991/allazimuth.476890.
- Sumer, Fahrettin, and Jay Joseph. 2018. "The Paradox of the Iraqi Kurdish Referendum on Independence: Contradictions and Hopes for Economic Prosperity." *British Journal of Middle Eastern Studies* 45(4): 574–88. https://www.tandfonline.com/doi/full/10.1080/13530194.2018.1430533 (November 5, 2020).
- Süsler, Buğra. 2020. Turkey, the EU and the Middle East *Turkey, the EU and the Middle East*. Routlege.
- Tabatabai, Ariane, and Dina Esfandiary. 2017. "Cooperating with Iran to Combat ISIS in Iraq."TheWashingtonQuarterly40(3):129–46.https://www.tandfonline.com/doi/full/10.1080/0163660X.2017.1370338.
- Tahiroglu, Merve, and Behnam Ben Taleblu. 2015. "Turkey and Iran: The Best of Frenemies." *Turkish Policy Quarterly* 14(1): 123–34.
- Taliaferro, Jeffrey W. 2012. "Neoclassical Realism and the Study of Regional Order." In International Relations Theory and Regional Transformation, Cambridge University Press, 74–104. https://www.cambridge.org/core/books/international-relations-theoryand-regional-transformation/neoclassical-realism-and-the-study-of-regionalorder/71DE2E837F2C2FA38636872C163C0A2E (November 21, 2020).

- Taliaferro, Jeffrey W., Steven E. Lobell, and Norrin M. Ripsman. 2009. "Introduction: Neoclassical Realism, the State, and Foreign Policy." In *Neoclassical Realism, the State, and Foreign Policy*, Cambridge University Press, 1–41. https://www.cambridge.org/core/books/neoclassical-realism-the-state-and-foreignpolicy/introduction-neoclassical-realism-the-state-and-foreignpolicy/9505ED036A7B8B3FCCE5AB56C68C77DD (November 21, 2020).
- Taliaferro, Jeffrey W. 2006. "State Building for Future Wars: Neoclassical Realism and the Resource-Extractive State." *Security Studies* 15(3): 464–95.
- Al Tamamy, Saud Mousaed. 2014. "Saudi Arabia and the Arab Spring: Opportunities and Challenges of Security." In *Regional Powers in the Middle East: New Consetellations after the Arab Revolts*, ed. Henner Furtig. Palgrave Macmillan, 191–209.
- Tan, Weizhen. 2018. "Erdogan Wins More Power, and It Might Not Help Turkey's Economy." *CNBC*.
- Tang, Shiping. 2009. "Taking Stock of Neoclassical Realism." International Studies Review 11(4): 799–803.
- Tanir, Ilhan. 2014. "Kobani and Kurdish Nationalism in Turkey Carnegie Endowment for International Peace." https://carnegieendowment.org/sada/57152 (November 5, 2020).
- Taşpinar, Ömer. 2011. "The Three Strategic Visions of Turkey." US-Europe Analysis Series (50): 1–5. https://www.brookings.edu/research/the-three-strategic-visions-of-turkey/ (November 24, 2020).
- Taşpınar, Ömer. 2012. "Turkey's Strategic Vision and Syria." *The Washington Quarterly* 35(3): 127–40. http://www.tandfonline.com/doi/abs/10.1080/0163660X.2012.706519.
- Terzi, Özlem. 2016. The Influence of the European Union on Turkish Foreign Policy. Routledge.
- The Economist. 2019. "As Turkey's Economy Slows, Attitudes toward Syrian Refugees Harden." *The Economist*.

- The Guardian. 2014. "Israel's Prime Minister Backs Kurdish Independence ." *The Guardian*. https://www.theguardian.com/world/2014/jun/29/israel-prime-minister-kurdish-independence (April 14, 2021).
- ——. 2016. "Syria: Turkey and Saudi Arabia Consider Ground Campaign Following Border Strikes." *The Guardian*. https://www.theguardian.com/world/2016/feb/14/turkey-andsaudi-arabia-consider-ground-campaign-in-syria-following-border-strikes (December 11, 2020).
- The New York Times. 1998. "Istanbul Mayor, an Islamist, Is Given 10-Month Jail Term." *The New York Times*. https://www.nytimes.com/1998/04/22/world/istanbul-mayor-anislamist-is-given-10-month-jail-term.html (April 22, 2021).
- ———. 2012. "Turkey Warns Syrian Forces Not to Approach Border ." *The New York Times*. https://www.nytimes.com/2012/06/27/world/middleeast/turkey-seeks-nato-backing-insyria-dispute.html (June 15, 2021).
- The UN Refugee Agency. 2019. UNHCR Global Appeal 2018-2019 Full Report. https://www.unhcr.org/uk/publications/fundraising/5a0c05027/unhcr-global-appeal-2018-2019-full-report.html. (June 15, 2021).
- Thornton, Rod. 2015. "Problems with the Kurds as Proxies against Islamic State: Insights from the Siege of Kobane." *Small Wars and Insurgencies* 26(6): 865–85. https://www.tandfonline.com/doi/abs/10.1080/09592318.2015.1095844 (November 4, 2020).
- Tocci, Nathalie et al. 2011. "Turkey and the Arab Spring. Implications for Turkish Foreign Policy from a Transatlantic Perspective." *Mediterranean Paper Series*: 1–44.

 2013. "Turkey's Kurdish Gamble." *International Spectator* 48(3): 67–77. https://www.tandfonline.com/doi/abs/10.1080/03932729.2013.814997 (November 5, 2020).

Tol, Gönül. 2014. Untangling the Turkey-KRG Energy Partnership: Looking Beyond

Economic Drivers. http://dx.doi.org/10.1080/14683849.2011.563938. (November 5, 2020).

- ———. 2016. "Turkey Running Out of Options on Syria." Middle East Institute. https://www.mei.edu/publications/turkey-running-out-options-syria#_ftn3.
- Totten, Michael J, David Schenker, and Hussain Abdul-Hussain. 2012. "Arab Spring or Islamist Winter?: Three Views." World Affairs 174(5): 23–42. https://www.jstor.org/stable/23210473?seq=1#metadata_info_tab_contents (December 10, 2020).
- Trager, Eric. 2017. "The Muslim Brotherhood Is the Root of the Qatar Crisis." *The Atlantic*. https://www.theatlantic.com/international/archive/2017/07/muslim-brotherhoodqatar/532380/ (December 9, 2020).
- Trenin, Dmitri. 2016. "The Revival of the Russian Military: How Moscow Reloaded." *Foreign Affairs* 95(23): 23–29.
- Tugal, Cihan. 2016. Fall of the Turkish Model. Verso.
- Tür, Özlem. 2011. "Economic Relations with the Middle East Under the AKP—Trade, Business Community and Reintegration with Neighboring Zones." *Turkish Studies* 12(4): 589–602. http://www.tandfonline.com/doi/abs/10.1080/14683849.2011.622515.
- ——. 2015. "Engaging with the Middle East: The Rise and Fall of Turkish Leadership in the 2000s." In *Turkey's Public Diplomacy*, New York: Palgrave Macmillan US, 67–84. http://link.springer.com/10.1057/9781137466983_4.
- Tur, Ozlem, and Ahmet Kasim Han. 2011. "A Framework for Understanding the Changing Turkish Foreign Policy of the 2000s." In *Turkey in the 21st Century: Quest for a New Foreign Policy*, ed. Özden Zeynep Oktav. Ashgate Pub Ltd, 7–31.
- Turel, Yilmaz. 1999. Iran Islam Devrimi ve Turkiye Iran Iliskilerine Etkisi. Strateji.

Turhan, Ebru, and Emirhan Yorulmazlar. 2018. "Turkish Foreign Policy towards the Arab

Spring: Between Western Orientation and Regional Disorder." In *Turkey, the Arab Spring and Beyond*, eds. Bülent Aras and E. Fuat Keyman. Routledge, 97–112.

- Turkish Statistical Institute. 2018a. "Exports by Country and Year (Top 20 Country in
Exports)."TurkishStatisticalInstitute.http://www.turkstat.gov.tr/PreTablo.do?alt_id=1046.
- ——. 2018b. "Imports by Country and Year (Top 20 Country in Imports)." *Turkish Statistical Institute*.
- Turpin, Lee David. 2019. "UK-EU Military Cooperation and Brexit from a Neoclassical Realist Perspective: No Big Deal?" In *Peace, Security and Defence Cooperation in Post-Brexit Europe*, Cham: Springer International Publishing, 3–27. http://link.springer.com/10.1007/978-3-030-12418-2_1.
- Tuysuzoglu, Gokhan. 2014. "Degisen Bolgesel Denklemler Isiginda Turkiye-Israil Iliskileri'nde Isbirligini Tetikleyen Unsurlar." *Journal of the Human and Social Science Researches* 3(3): 588–609.
- Ünal, Serhan, and Eyüp Ersoy. 2014. "Political Economy of Turkish-Iranian Relations: Three Asymmetries." *Ortadoğu Etütleri* 5(2): 141–64.
- Unver, H. Akin. 2016. "Schrodinger's Kurds: Transnational Kurdish Politics in the Age of Shifting Borders." *Journal of International Affairs* 69(2): 65–100. https://www.jstor.org/stable/10.2307/26494339%0A.

———. 2018. "Computational International Relations What Can Programming, Coding and Internet Research Do for the Discipline?" *arXiv* (February 2018).

- Ünver, H. Akin. 2012. "How Turkey's Islamists Fell out of Love With Iran." *Middle East Policy* 19(4): 103–9. http://doi.wiley.com/10.1111/j.1475-4967.2012.00563.x.
- 2016. "Schrödinger'S Kurds: Transnational Kurdish Geopolitics in the Age of Shifting Borders." *Journal of International Affairs* 69(2): 65–98. https://jia.sipa.columbia.edu/schrodingers-kurds-transnational-kurdish-geopolitics-age-

shifting-borders (November 5, 2020).

- Uslu, Nasuh. 2000. "The Factors Affecting Turkey's Relations with the United States in the Post-Cold War Period." *The Turkish Yearbook* 31(2): 183–225.
- Uygur, Hakki. 2012. *Iran ve Arab Bahari*. http://file.setav.org/Files/Pdf/20121122110322_seta-iran_ve_arap_bahari.pdf.
- Uzun, Ozum. 2013. "The 'Arab Spring' and Its Effect on Turkish-Iranian Relations." *Ortadogu Etutleri* 4(2).
- Vasquez, John A. 1997. "The Realist Paradigm and Degenerative versus Progressive Research Programs: An Appraisal of Neotraditional Research on Waltz's Balancing Proposition." *American Political Science Review* 91(4): 899–912. https://www.cambridge.org/core/journals/american-political-sciencereview/article/realist-paradigm-and-degenerative-versus-progressive-research-programsan-appraisal-of-neotraditional-research-on-waltzs-balancingproposition/B98FE902AFEEA66660F438F1D624C5 (November 23, 2020).
- Walt, Stephen M. 2009. "Alliances in a Unipolar World ." World Politics 61(1): 86–120.
- Waltz, Kenneth N. 1986. "Reflections on Theory of International Politics: A Response To My Critics." In *Neorealism and Its Critics*, ed. Robert Keohane. Columbia University Press.
- ——. 2001. *Man, the State and the War*. Columbia University Press.
- 2002. "The Continuity of International Politics." In Worlds in Collision: Terror with the /Future of Global Order, eds. Ken Booth and Timothy Dunne. Palgrave, 348–54. https://uq.rl.talis.com/items/BE810937-E99D-ED18-00D3-0638EC77824A.html (November 22, 2020).
- . 2010. *Theory of International Politics*. Waveland Press.
- Wang, Bo. 2011. "Turkey-Iran Reconciliatory Relations: Internal and External Factors." Journal of Middle Eastern and Islamic Studies (in Asia) 5(1): 1–18.

https://www.tandfonline.com/doi/full/10.1080/19370679.2011.12023171.

- Wastnidge, Edward. 2019a. "Imperial Grandeur and Selective Memory: Re-Assessing Neo-Ottomanism in Turkish Foreign and Domestic Politics." *Middle East Critique* 28(1): 7–28. https://www.tandfonline.com/doi/full/10.1080/19436149.2018.1549232.
- . 2019b. "Transnational Identity Claims, Roles and Strategic Foreign Policy Narratives in the Middle East." *Global Discourse* 9(4): 605–25. https://www.ingentaconnect.com/content/10.1332/204378919X15718897799871.
- Wendt, Alexander. 1992. "Anarchy Is What States Make of It: The Social Construction of Power Politics." *International Organization* 46(2): 391–425. https://www.cambridge.org/core/product/identifier/S0020818300027764/type/journal_ar ticle.
- Werman, Alexander. 2019. "Turkey: Impact of the Syrian War's next Stage." Global Risk Insights. https://globalriskinsights.com/2019/03/turkey-syrian-war/.
- Wilgenburg, Wladimir van. 2014. "Iranian Kurdish Parties Prefer Dialogue with Government." *Al-Monitor*. http://www.al-monitor.com/pulse/originals/2014/01/iranian-kurdish-partieswaning-support-exiles-pkk-turkey.html.
- Williams, Paul A. 2019. "The Rise and Fall of Turkey in the Arab Spring." In *Routledge Handbook of International Relations in the Middle East*, Taylor and Francis, 376–90. https://www.taylorfrancis.com/https://www.taylorfrancis.com/chapters/edit/10.4324/978 1315229591-27/rise-fall-turkey-arab-spring-paul-williams (May 9, 2021).
- Winrow, Gareth M. 1997. "Turkey and the Newly Independent States of Central Asia and the Transcaucasus." *MERIA Journal* 2(5): 215–16.
- Wivel, Anders. 2005. "Explaining Why State X Made a Certain Move Last Tuesday: The Promise and Limitations of Realist Foreign Policy Analysis." *Journal of International Relations and Development* 8(4): 355–80. http://link.springer.com/10.1057/palgrave.jird.1800064.

- Wohlforth, William Curti. 1993. The Elusive Balance: Power and Perceptions During the Cold War. Cornell University Press.
- Yackley, Jean Ayla, and Bozorgmehr Sharafedin. 2016. "Rivals Turkey and Iran Seek to 'Manage Differences."" *Reuters*.
- Yalvaç, Faruk. 2014. "Approaches to Turkish Foreign Policy: A Critical Realist Analysis." *Turkish Studies* 15(1): 117–38.
- Yavuz, M. Hakan. 2009. Secularism and Muslim Democracy in Turkey Secularism and Muslim Democracy in Turkey. Cambridge: Cambridge University Press.
- Yeni Şafak. 2018. "Saudi Funds US-Backed PKK Terror Force in Northern Syria." Yeni Şafak. https://www.yenisafak.com/en/world/saudi-arabia-in-talks-with-ypg-pkk-to-form-newforce-3412467 (December 11, 2020).
- Yeşiltaş, Murat. 2013. "The Transformation of the Geopolitical Vision in Turkish Foreign Policy." *Turkish Studies* 14(4): 661–87.
- ——. 2014. "The New Era in Turkish Foreign Policy: Critiques and Challenges." *Insight Turkey* 16(3): 25–36.
- Yesilyurt, Nuri. 2017. "Explaining Miscalculation and Maladaptation in Turkish Foreign Policy towards the Middle East during the Arab Uprisings: A Neoclassical Realist Perspective." All Azimuth: A Journal of Foreign Policy and Peace 6(2): 65–83.
- Yeşilyurt, Nuri. 2017. "Explaining Miscalculation and Maladaptation in Turkish Foreign Policy towards the Middle East during the Arab Uprisings: A Neoclassical Realist Perspective." All Azimuth 6(2): 65–83.
- Yesilyurt, Nuri, and Atay Akdevelioglu. 2009. "Turkey's Middle East Policy under the JDP Rule." *The Turkish Yearbook of International Relations* 40: 39–70.
- Yetim, Mustafa, and Ridvan Kalayci. 2011. "Turkiye-Iran Iliskileri: Sifir Sorun Mu, Nukleer Sorun Mu?" *Akademik Ortadogu* 5(2): 85–108.

- Yilmaz, Suhnaz. 2017. "Middle Powers and Regional Powers." Oxford Bibliographies in International Relations. https://www.oxfordbibliographies.com/view/document/obo-9780199743292/obo-9780199743292-0222.xml (April 15, 2021).
- Yorulmazlar, Emirhan, and Ebru Turhan. 2015. "Turkish Foreign Policy towards the Arab Spring: Between Western Orientation and Regional Disorder." *Journal of Balkan and Near Eastern Studies* 17(3): 337–52. http://www.tandfonline.com/doi/full/10.1080/19448953.2015.1063805.
- Yücesoy, Vahid. 2020. "The Recent Rapprochement between Iran and Turkey: Is It Durable or Is It a Relationship of Convenience?" *Turkish Studies* 21(2): 274–96. https://www.tandfonline.com/doi/full/10.1080/14683849.2019.1657773.
- Zakaria, Fareed. 1999. From Wealth to Power: The Unusual Origins of America's World Role. Princeton University Press.
- . 2011. The Post-American World. WW Norton & Company.
- Zaman, Amberin. 2014. "Turkey Ignored Direct Warnings of ISIS Attack on Mosul." Al-Monitor. https://www.al-monitor.com/pulse/originals/2014/06/mosul-turkish-consulateisis-ankara-syria-iraq-kidnappings.html (December 21, 2020).
- Zelyut, Evren Devrim. 2016. "Suriye'de Savas Devam Ederse Turkiye Ekonomik Krize Girer." *Aydinlik*.

APPENDIX



ETHICAL APPROVAL-SGIA

Dear Cangul,

The following project has received ethical approval:

Project Title: Turkish foreign policy towards Iran since the Arab Uprisings: A Neoclassical Realist Account.

Start Date: *30 June 2020*. End Date: *31 August 2020*. Reference: *SGIA-2020-04-07T22:36:07-jndx45* Date of ethical approval: *10 June 2020*.

You should print this email and keep it for your records.

Please note the following restriction:

- The approval relates only to proposed research you have submitted for approval and you are required to follow all procedures outlined in the application.
- If you wish to change the proposal, for example, to widen the scope of inquiry or substantially expand the use of data collected, then you MUST reapply to the Ethics & Risk Committee for further approval.
- Should you end up traveling for research you must submit individual insurance applications through the university's VIATOR system (https://apps.dur.ac.uk/travel.forms/). Those applications should include the risk assessment form attached to this ethics application.

We wish you all the best with your research.

Please be aware that if you make any significant changes to the design, duration or delivery of your project, you should contact your department ethics representative for advice, as further consideration and approval may then be required.

If you have any queries regarding this approval or need anything further, please contact sgia.ethics@durham.ac.uk

If you have any queries relating to the ethical review process, please contact your supervisor (where applicable) or departmental ethics representative in the first instance. If you have any queries relating to the online system, please contact <u>research.policy@durham.ac.uk</u>.



Consent Form

Project title: Turkish Foreign Policy towards Iran since the Arab Uprisings: A Neoclassical Realist Account Researcher(s): Cangul Altundas-Akcay Department: School of Government and International Affairs Contact details: cangul.altundas-akcay@durham.ac.uk Supervisor name: Prof Clive A. Jones Supervisor contact details: c.a.jones@durham.ac.uk

This form is to confirm that you understand what the purposes of the project, what is involved and that you are happy to take part. Please initial each box to indicate your agreement:

I confirm that I have read and understand the information sheet dated [dd/mm/yy] and the privacy notice for the above project.	
I have had sufficient time to consider the information and ask any questions I might have, and I am satisfied with the answers I have been given.	
I understand who will access to personal data, how the data will be stored and what will happen to the data at the end of the project.	
I agree to take part in the above project.	
I consent to being audio recorded and understands how recordings will be used in research outputs.	
I understand that my participation is voluntary and that I am free to withdraw at any time without giving a reason.	
I understand that if I feel uncomfortable in any way during the interview, I have the right to decline to answer any question, from start to end of the interview.	

I understand that there are two possible risks (facing heavy scrutiny by the AKP government and ethical considerations arising from Skype) regarding the participation in this study.	
I understand that I will be anonymized, so the researcher will not identify me by name in any report using information obtained from this study, and that my confidentiality as a participant in this study will remain secure.	

Participant's Signature	Date
(NAME IN BLOCK LETTERS)	
Researcher's Signature	_ Date
(NAME IN BLOCK LETTERS)	

Debriefing Sheet



Project title: Turkish Foreign Policy towards Iran since the Arab Uprisings: A Neoclassical Realist Account

Thank you for taking part in this study. This research project examines the influence of Turkey's relative power position, the perceptions of the AKP elite and domestic factors (economic and security considerations) in shaping Turkish foreign policy towards Iran since the Arab Uprisings.

In this regard, the thesis seeks to trace the causal relationship between the independent (Turkey's relative power position from the Arab Uprisings) variable, the intervening (the AKP elite perception and domestic considerations of Turkey) variable and the dependent variable (Turkish foreign policy outcomes towards Iran since the Arab Uprisings).

As such, you were asked to reflect upon Turkey's relative power in regional politics, AKP elite perceptions of Ankara's relative power as well as the domestic factors that have influenced and shaped Turkish Foreign policy towards Iran since the Arab Uprisings. You were also asked to reflect upon Turkey's policies towards the Syrian civil war and Iraqi Kurdistan and to consider how these policies were shaped by Ankara's perceptions of Iran's regional policies.

With your agreement, I made an audio recording of the interviews. This was solely for the purposes of research and to help me recall with accuracy what you and other people told me. The recordings will be destroyed on completion of the research. Any information you supplied me will be held in strict confidence, viewed only by the researcher (Cangul Altundas Akcay) and then anonymised.

The records of this study will be kept secure and private. All files containing any information provided by you are password protected. In any research report that may be published, no

information will be included that will make it possible to identify you individually. There will be no way to connect your name to your responses at any time during or after the study.

If you would like further information about the study or would like to know about what my findings are when all the data have been collected and analyzed then please contact me on 'cangul.altundas-akcay@durham.ac.uk'. I cannot however provide you with your individual results.

If taking part in this study has raised any specific concerns including withdrawing your data then I would suggest you contact me as soon as possible.

Kind regards, Cangul Altundas-Akcay



Project title: Turkish Foreign Policy towards Iran since the Arab Uprisings: A Neoclassical Realist Account

Researcher(s): Cangul Altundas Akcay

Department: School of Government and International Affairs

Contact details: cangul.altundas-akcay@durham.ac.uk

Supervisors name: Prof. Clive A. Jones

Supervisors contact details: <u>c.a.jones@durham.ac.uk</u>

You are invited to take part in a study that I am conducting as part of my PhD at Durham University. This study has received ethical approval from School of Government and International Affairs ethics committee of Durham University.

Before you decide whether to agree to take part it is important for you to understand the purpose of the research and what is involved as a participant. Please read the following information carefully. Please get in touch if there is anything that is not clear or if you would like more information.

Note: The rights and responsibilities of anyone taking part in Durham University research are set out in our 'Participants Charter':

https://www.dur.ac.uk/research.innovation/governance/ethics/considerations/people/charter/

What is the purpose of the study?

The aim of this study is to identify how academics/journalists/scholars think about the influence of Turkey's relative power position in determining its relations with Iran since the Arab uprising. In order to do this, the project looks to understand the role of the following: the elite perceptions of the ruling AKP towards Iran as well as the role of domestic factors (economic and security considerations) in shaping ties between Ankara and Tehran.

Why have I been invited to take part?

You have been invited because of your expertise in Turkish and/or Iranian affairs of TFP or/towards Iran. Your insights and thoughts can help inform and deepen the analysis of the Turkish-Iranian relations.

Do I have to take part?

Your participation is voluntary, and you do not have to agree to take part. If you do agree to take part, you can withdraw at any time, without giving a reason.

What will happen to me if I take part?

If you agree to take part in the study, you will be asked to reflect upon Turkey's relative power position in international affairs, the role of elite decision-makers in the AKP, as well as wider economic and security considerations, that have shaped Ankara's approach towards Tehran since the Arab uprisings. This will also cover Turkey's and Iran's policies towards the Syrian conflict and Iraqi Kurdistan.

The interview will be online through Skype in English where possible (to reduce the necessity to translate and to reduce translation errors). Yet, if you are uncomfortable speaking English, the interview can be conducted in Turkish. This interview will be audio-recorded using an encrypted recorder. You will be informed when recording begins, is paused or stopped.

The interview will last approximately one hour and will be informed by semi-structured questions. I may ask other questions as part of a snow-balling process, but you have right at any time to decline to respond any questions you do not wish to answer.

Are there any potential risks involved?

There is a potential risk. You may face heavy scrutiny due to the AKP government's growing assaults on journalists, intellectuals and academics. To mitigate this, your data will be anonymized. In this way, I will not identify you by name in any report using information obtained from this study. Your TOTAL ANONYMITY will remain secure.

With a Skype video interview though, there are additional ethical considerations to take into account: the issues created by online interaction is mediated through the use of technology (which is owned by third parties); the verification of participant's identity; and issues raised by the interview environment and the nature of recording the interview. In this regard, it is important, from an ethical standpoint, to remind you that your discussions online may be accessed and stored by governments agencies or corporations. In order to mitigate this issue, I will create a specific Skype account. At the termination of the study, the Skype account will be

closed and subsequently all the participants' online details and data will be removed. To protect your privacy, you will be advised to select appropriate locations from which you will be interviewed.

Will my data be kept confidential?

All information obtained during the study will be kept confidential. To mitigate risk and maintain subject privacy/confidentiality/anonymity, the data will be viewed only by me. You will have the option to withdraw from/end the interview at any time or to retract specific statements. You will also remain anonymous. Following the interview, the data then will be transcribed by me to protect the data as well as your anonymity. I will transcribe the recording from the encrypted voice recorder and delete it from the recorder. To ensure anonymity, a list that identifies you with the recordings will be kept separately and made unidentifiable by name, with a random number code attributed to you and your recording. I will then remove identifying information from the transcripts and imported them into Nvivo, a software program that facilitates qualitative data analysis and enables users to assign a code to specific lines or segments of the text.

I will also delete/destroy data containing personal details that would lead to the identification of you (e.g. your email addresses, telephone numbers, physical addresses, etc but *not* consent forms) as soon as possible. I will retain contact details only until you have participated in the study, or until you have informed me that you do not wish to participate. If you indicate that you would like to receive a copy of my notes or the research findings at the end of the study, then I will retain your contact details until this information has been sent out. I will also adhere to the University of Durham Data Protection Principles

(https://www.dur.ac.uk/data.protection/dp_principles/).

What will happen to the results of the project?

The results will be used for the purpose of the PhD's degree dissertation in Government and International Affairs, Durham University. However, they might additionally be disseminated at conferences or published in peer-reviewed publications. In this case, your anonymity will still be ensured in any papers.

All research data and records needed to validate the research findings will be stored for [10] years after the end of the project.

Durham University is committed to sharing the results of its world-class research for public benefit. As part of this commitment the University has established an online repository for all Durham University Higher Degree theses which provides access to the full text of freely available theses. The study in which you are invited to participate will be written up as a thesis. On successful submission of the thesis, it will be deposited both in print and online in the University archives, to facilitate its use in future research. The thesis will be published open access after the expiration of restricted access of the thesis.

Who do I contact if I have any questions or concerns about this study?

If you have any further questions or concerns about this study, please speak to the researcher (Cangul Altundas-Akcay). If you remain unhappy or wish to make a formal complaint, please submit a complaint via the University's <u>Complaints Process</u>.

Thank you for reading this information and considering taking part in this study. Signature the participant:

Date:

Signature of the researcher:

Date:

Privacy Notice



PART 1 – GENERIC PRIVACY NOTICE

Durham University has a responsibility under data protection legislation to provide individuals with information about how we process their personal data. We do this in a number of ways, one of which is the publication of privacy notices. Organisations variously call them a privacy statement, a fair processing notice or a privacy policy.

To ensure that we process your personal data fairly and lawfully we are required to inform you:

- Why we collect your data
- How it will be used
- Who it will be shared with

We will also explain what rights you have to control how we use your information and how to inform us about your wishes. Durham University will make the Privacy Notice available via the website and at the point we request personal data.

Our privacy notices comprise two parts – a generic part (ie common to all of our privacy notices) and a part tailored to the specific processing activity being undertaken.

Data Controller

The Data Controller is Durham University. If you would like more information about how the University uses your personal data, please see the University's <u>Information Governance</u> <u>webpages</u> or contact Information Governance Unit:

Telephone: (0191 33) 46246 or 46103

E-mail: information.governance@durham.ac.uk

Information Governance Unit also coordinate response to individuals asserting their rights under the legislation. Please contact the Unit in the first instance.

Data Protection Officer

The Data Protection Officer is responsible for advising the University on compliance with Data Protection legislation and monitoring its performance against it. If you have any concerns regarding the way in which the University is processing your personal data, please contact the Data Protection Officer:

Jennifer Sewel University Secretary Telephone: (0191 33) 46144 E-mail: <u>university.secretary@durham.ac.uk</u>

Your rights in relation to your personal data

Privacy notices and/or consent

You have the right to be provided with information about how and why we process your personal data. Where you have the choice to determine how your personal data will be used, we will ask you for consent. Where you do not have a choice (for example, where we have a legal obligation to process the personal data), we will provide you with a privacy notice. A privacy notice is a verbal or written statement that explains how we use personal data.

Whenever you give your consent for the processing of your personal data, you receive the right to withdraw that consent at any time. Where withdrawal of consent will have an impact on the services, we are able to provide, this will be explained to you, so that you can determine whether it is the right decision for you.

Accessing your personal data

You have the right to be told whether we are processing your personal data and, if so, to be given a copy of it. This is known as the right of subject access. You can find out more about this right on the University's <u>Subject Access Requests webpage</u>.

Right to rectification

If you believe that personal data, we hold about you is inaccurate, please contact us and we will investigate. You can also request that we complete any incomplete data.

Once we have determined what we are going to do, we will contact you to let you know.

Right to erasure

You can ask us to erase your personal data in any of the following circumstances:

- We no longer need the personal data for the purpose it was originally collected
- You withdraw your consent and there is no other legal basis for the processing
- You object to the processing and there are no overriding legitimate grounds for the processing
- The personal data have been unlawfully processed
- The personal data have to be erased for compliance with a legal obligation
- The personal data have been collected in relation to the offer of information society services (information society services are online services such as banking or social media sites).

Once we have determined whether we will erase the personal data, we will contact you to let you know.

Right to restriction of processing

You can ask us to restrict the processing of your personal data in the following circumstances:

- You believe that the data is inaccurate, and you want us to restrict processing until we determine whether it is indeed inaccurate
- The processing is unlawful, and you want us to restrict processing rather than erase it
- We no longer need the data for the purpose we originally collected it but you need it in order to establish, exercise or defend a legal claim and
- You have objected to the processing and you want us to restrict processing until we determine whether our legitimate interests in processing the data override your objection.

Once we have determined how we propose to restrict processing of the data, we will contact you to discuss and, where possible, agree this with you.

Retention

The University keeps personal data for as long as it is needed for the purpose for which it was originally collected. Most of these time periods are set out in the <u>University Records Retention</u> <u>Schedule</u>.

Making a complaint

If you are unsatisfied with the way in which we process your personal data, we ask that you let us know so that we can try and put things right. If we are not able to resolve issues to your satisfaction, you can refer the matter to the Information Commissioner's Office (ICO). The ICO can be contacted at: Information Commissioner's Office Wycliffe House Water Lane Wilmslow Cheshire SK9 5AF

Telephone: 0303 123 1113

Website: Information Commissioner's Office

PART 2 – TAILORED PRIVACY NOTICE

This section of the Privacy Notice provides you with the privacy information that you need to know before you provide personal data to the University for the particular purpose(s) stated below.

Project Title: Turkish Foreign Policy towards Iran since the Arab Uprisings: A Neoclassical Realist Account

Type(s) of personal data collected and held by the researcher and method of collection:

Personal data will be collected through interview. This will include 'special category data' including the views of academics and journalists, researchers who have sufficient information regarding Turkish foreign policy and/or Iranian foreign policy. The interviews will be conducted via 'Skype', and audio recordings will be used to collect the data.

Lawful Basis

The legal basis for processing the participant's personal data on this occasion is Article 6(1a) consent of the data subject. In addition to the legal basis for processing his/her personal data, For 'special category data' (including: personal data revealing racial or ethnic origin, political opinions, religious or philosophical beliefs, or trade union membership, and the processing of genetic data, biometric data for the purpose of uniquely identifying a natural person, data concerning health or data concerning a natural person s sex life or sexual orientation), the basis for processing the participant sensitive personal data on this occasion is Article 9(2j) processing

is necessary for archiving purposes in the public interest, scientific or historical research purposes or statistical purposes.

How personal data is stored:

All personal data will be held securely and strictly confidential to the researcher. The participant will have the option to withdraw from (end) the interview at any time or to retract specific statements. The participant will also totally anonymous, Thus, he/she will be allocated an anonymous number for data collection which will not be connected to his/her name or identity. Signed consent forms will be stored separately to project data. All personal data in electronic form will be stored on a password protected computer. Data will not be available to anyone except for the researcher. The conversation will be recorded and stored on an encrypted device until it has been transcribed by the researcher. No-one else will have access to the recording, and it will be erased once the transcript has been completed.

How personal data is processed:

The researcher is collecting the data to identify how academics/journalists/scholars think about the influence of Turkey's relative power position, the perceptions of the AKP elite, as well as the influence of domestic factors, notably economic and internal security considerations in shaping Turkish foreign policy towards Iran since the Arab Uprisings. As such, information will be entered into a database for analysis. The interviews will be complementary to the analytical research and used to support or challenge the insights derived from secondary data. The data will be completely anonymized and the original records, including any information which can identify the participant personally, will be destroyed. The recorded conversation will be transcribed by the researcher, and personal information will be coded and anonymized. The original recording will then be erased.

Withdrawal of data

The participant can request withdrawal of his/her data at any time. In this case, please contact the researcher.

Who the researcher shares personal data with:

The relevant data will NOT be shared with anybody, so ONLY the researcher will conduct the interview with the voice recorder and then transcribe the data. After completing the interview via Skype, the researcher will delete the contact information and chat history. After transcribing

the data, the researcher will also delete the voice record from the recorder. The identifiable data will not be transferred outside the UK/EU.

Please be aware that if you disclose information which indicates the potential for serious and immediate harm to yourself or others, the research team may be obliged to breach confidentiality and report this to relevant authorities. This includes disclosure of child protection offences such as the physical or sexual abuse of minors, the physical abuse of vulnerable adults, money laundering, or other crimes covered by prevention of terrorism legislation. Where you disclose behavior (by yourself or others) that is potentially illegal but does not present serious and immediate danger to others, the researcher will, where appropriate, signpost you to relevant services, but the information you provide will be kept confidential (unless you explicitly request otherwise).

How long personal data is held by the researcher?

The researcher will hold personal data till the completion of transcribing the data, after which it will be anonymized if necessary. Yet a signed consent form may be hold till the completion of PhD for audit purposes.

How to object to the processing of your personal data for this project:

If you have any concerns regarding of your personal data, or you wish to withdraw your data from the project, contact Cangul Altundas-Akcay, <u>cangul.altundas-akcay@durham.ac.uk</u>.

Further information:

Cangul Altundas-Akcay. Email: <u>cangul.altundas-akcay@durham.ac.uk</u> Tel: +447470033477



INTERVIEW QUESTIONS

- 1. Can you reflect upon what factors have influenced Turkey's relative power position in the light of the Arab Uprisings?
- 2. To what extent has Turkey has been turning away from its traditional Western allies?
- 3. Do you think that there is any evidence of ideological incentives in shaping TFP?
- **4.** Can you reflect upon the consistency in the AKP's domestic and regional policies towards the Kurds?
- **5.** Do you think that Turkey has followed an identity or interest-based foreign policy towards Iran since the Arab Uprisings?
- **6.** How has personnel changes across the AKP elites over 10 years impacted TFP towards Iran since the Arab Uprisings?
- **7.** How important were domestic political considerations in determining the policy choices of AKP elites' towards Iran?
- **8.** In what ways, if any, did Turkey's internal security dilemmas (PKK, coup attempt, ISIS) shape Turkey's relations with Iran?
- **9.** In what ways, if any, did economic considerations impact upon Turkey's relations with Iran?
- 10. Can you explain the impact of the Iraqi Kurds on Turkish-Iranian bilateral ties?
- 11. Can you explain the impact of the Syrian conflict on Turkish-Iranian bilateral ties?



LIST OF INTERVIEWEES

Name	Type of Interview	Date
Academic 1	Skype	04.08.2020
Academic 1	Зкуре	04.08.2020
Journalist 1	Skype	01.07.2020
Academic 2	Skype	06.08.2020
Academic 3	Skype	09.07.2020
Journalist 2	Skype	14.08.2020
Expert 1	Skype	17.08.2020
Expert 2	Skype	19.08.2020
Academic 4	Skype	06.07.2020
Expert 3	Skype	21.07.2020
Academic 5	Skype	22.07.2020
Expert 4	Skype	24.07. 2020
Academic 6	Skype	25.08.2020